

Government Gazette Staatskoerant

Vol. 450

Pretoria, 6

December 2002 Desember No. 24134







AIDS HELPLINE: 0800-0123-22 Prevention is the cure

CONTENTS

No.

Page Gazette No. No.

GOVERNMENT NOTICE

Minerals and Energy, Department of

Government Notice

GOVERNMENT NOTICE

DEPARTMENT OF MINERALS AND ENERGY

No. 1520

6 December 2002

DRAFT REGULATIONS UNDER THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002 (Act No. 28 of 2002).

The Minister of Minerals and Energy intends promulgating draft regulations in terms of section 107(1) of the Mineral and Petroleum Resources Development Act, 2002 (Act No. 28 of 2002), as set out in the Schedule hereto.

Interested parties are invited to submit written comments on the draft regulations on or before 6 February 2003. Comments must be submitted to:

The Director-General,
Department of Minerals and Energy,
Private Bag X59,
PRETORIA
0001

The draft regulations will also be available on the Department's website, www.dme.gov.za

REPUBLIC OF SOUTH AFRICA

DRAFT MINERAL AND PETROLEUM RESOURCES DEVELOPMENT REGULATIONS

IN TERMS OF THE

MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002, (ACT NO. 28 OF 2002)

SCHEDULE

ARRANGEMENT

CHAPTER 1 DEFINITIONS AND INTERPRETATION

Short title and commencement Statutory authorization

1. Definitions and interpretation

CHAPTER 2 MINERAL, SOCIAL AND ENVIRONMENTAL REGULATION

Part I: Mineral regulation

- 2. Manner of lodging applications and plan to be submitted
- 3. Consultation with interested and affected parties
- 4. Transferability and encumbrance of permits and rights
- 5. Application for assistance to historically disadvantaged persons
- 6. Application for a reconnaissance permission
- 7. Application for a prospecting right
- 8. General terms and conditions of a prospecting right
- 9. Prospecting work programme
- 10. Progress report in respect of prospecting
- 11. Application for the renewal of a prospecting right
- 12. Application for a mining right
- 13. Mining work programme
- 14. General terms and conditions for mining right
- 15. Application for the renewal of a mining right
- 16. Information, data and statistical returns in respect of mining or processing of minerals
- 17. Annual report in respect of mining
- 18. Application for a mining permit
- 19. Application for the renewal of mining permit
- 20. Application for a retention permit
- 21. General terms and conditions of retention permit
- 22. Application for the renewal of a retention permit
- Purpose and objectives of the Regional Mining Development and Environmental Committees
- Establishment and functions of the Regional Mining Development and Environmental Committees
- 25. Process of handling objections by the Committee

Part II: Social and Labour Plan

- 26. Submission of a Social and Labour Plan and its applicability
- 27. Objects of Social and Labour Plan
- 28. Contents of Social and Labour Plan
- 29. Social and Labour Plan forum

Part III: Environmental regulation

- 30. Environmental reports to be compiled for the application of a mining right
- 31. Scoping Report
- 32. Environmental Impact Assessment Report
- 33. Investigation of cumulative impacts
- 34. Contents of Environmental Management Programme
- 35. Independent review
- Environmental report to be compiled for the application of a reconnaissance permission, prospecting right or mining permit
- 37. Environmental Management Plan
- 38. Method(s) for financial provision

- 39. Standard forms for financial provision
- 40. Quantum of financial provision
- 41. Monitoring and performance assessment of environmental management programme or plan
- 42. General requirements for mine closure
- 43. Application for a closure certificate
- 44. Application to transfer environmental liabilities to a qualified person
- 45. Qualifications of person regarding transfer of liabilities and responsibilities
- 46. Environmental Risk Report
- 47. Closure objectives
- 48. Contents of Closure plan

Part IV: Pollution Control and Waste Management Regulation

- 49. Principles of pollution control and waste management
- 50. Air quality management and control
- 51. Fire prevention
- 52. Noise control
- 53. Blasting, vibration and shock
- 54. Water management pollution control
- 55. Disposal of waste material
- 56. Soil pollution and erosion control
- 57. Sanitation of surface
- 58. Management of residue stockpiles and deposits

CHAPTER 3 PETROLEUM EXPLORATION AND PRODUCTION

Part I: Petroleum Regulation

- 59. Application for a reconnaissance permit
- 60. General terms and conditions for a reconnaissance permit
- 61. Reconnaissance work programme
- 62. Progress report in respect of reconnaissance activities
- 63. Supply of data in respect of reconnaissance
- 64. Application for a technical co-operation permit
- 65. General terms and conditions for a technical co-operation permit
- 66. Technical co-operation work programme
- 67. Progress report in respect of technical co-operation activities
- 68. Application for an exploration right
- 69. General terms and conditions for an exploration right
- 70. Exploration work programme
- 71. Progress reports in respect of exploration
- 72. Supply of data in respect of exploration
- 73. Application for the renewal of an exploration right
- 74. Application for a production right
- 75. General terms and conditions for a production right
- 76. Production work programme
- 77. Application for the renewal of a production right
- 78. Information, data and statistical returns in respect of production of petroleum

Part II: Social and Labour Plan

- 79. Submission of a Social and Labour Plan and its applicability
- 80. Objectives of Social and Labour Plan
- 81. Contents of Social and Labour Plan
- 82. Social and Labour Plan forum

Part III: Environmental regulation for petroleum

- 83. Environmental reports to be compiled for the application of a production right
- 84. Scoping Report
- 85. Environmental Impact Assessment Report
- 86. Investigation of cumulative impacts
- 87. Contents of Environmental Management Programme

- 88. Independent review
- Environmental report to be compiled for the application of a reconnaissance permit or exploration right
- 90. Environmental Management Plan
- 91. Method(s) for financial provision
- 92. Standard forms for financial provision
- 93. Quantum of financial provision
- 94. Monitoring and performance assessment of environmental management programme or plan
- 95. General requirements for closure of reconnaissance, exploration and production operations
- 96. Application for a closure certificate
- 97. Application to transfer environmental liabilities to a qualified person
- 98.Qualifications of person regarding transfer of liabilities and responsibilities
- 99. Environmental Risk Report
- 100. Closure objectives
- 101. Contents of Closure plan

Part IV: Pollution Control and Waste Management

- 102. Principles of pollution control and waste management
- 103. Air quality control
- 104. Fire prevention
- 105. Noise control
- 106. Blasting, vibration and shock
- 107. Water management pollution control
- 108. Disposal of waste material
- 109. Soil pollution and erosion control
- 110. Sanitation of surface
- 111. Management of residue stockpiles and deposits

CHAPTER 4 APPEALS

- 112. Appeal against administrative action by Regional Manager, the Petroleum Agency or an
- 113. Appeal against administrative action by Director-General or Petroleum Agency

CHAPTER 5 GENERAL AND MISCELLANEOUS REGULATIONS

- 114. Refusal of prospecting right, mining right, mining permit, retention permit, reconnaissance permit, technical co-operation permit, exploration right, production right
- 115. Permission to remove bulk samples of minerals during prospecting or exploration
- 116. Notice for the suspension or cancellation of a prospecting right, mining right, mining permit or retention permit, reconnaissance permit, technical co-operation permit, exploration right, production right
- 117. Withdrawal of the suspension notice
- 118. Notice to prohibit or restrict the granting of a prospecting right, mining right, reconnaissance permit, exploration right, production right
- Notice of profitability and curtailment of mining or production operations affecting employment
- Notification to Regional Manager / Petroleum Agency if holder is prevented from commencing with operations
- 121. Notice to land owner or lawful occupier of land if holder is prevented from commencing with operations
- 122. Notice of expropriation of property
- 123. Certificate to authorize person to enter prospecting area, mining area, retention area, reconnaissance area, exploration area, production area
- 124. Notice by authorized person if holder contravenes provisions of Act
- 125. Balance sheet: Liquidator's report

CHAPTER 6 TRANSITIONAL ARRANGEMENTS

126. Lodgement for the conversion of an old order prospecting right in terms of the Minerals Act, 1991, (Act No. 50 of 1991) to one in terms of the Mineral and Petroleum Resources Development Act, 2002, (Act No. 28 of 2002).

127. Lodgement for the conversion of an old order mining right in terms of the Minerals Act, 1991, (Act No. 50 of 1991) to one in terms of the Mineral and Petroleum Resources Development Act, 2002, (Act No. 28 of 2002).

ANNEXURE 1

APPLICATION FORMS FOR PERMISSIONS, RIGHTS AND PERMITS TO BE COMPLETED IN TERMS OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002

- Form A: Application form for a reconnaissance permission in terms of section 13 of the Act, 2002
- Form B: Application form for a prospecting right in terms of section 16 of the Act, 2002
- Form C: Application form for the renewal of a prospecting right in terms of section 18(1) of the Act, 2002)
- Form D: Application form for a mining right in terms of section 22(1) of the Act, 2002
- Form E: Application form for the renewal of a mining right in terms of section 24(1) of the Act, 2002
- Form F: Application form for a mining permit in terms of section 27(2) or the renewal thereof in terms of section 27(8) of the Act, 2002
- Form G: Application form for a retention permit in terms of section 31(1) or the renewal thereof in terms of section 34(1) of the Act, 2002
- Form H: Application for the renewal of retention permit in terms of section 34(1) of the Act, 2002
- Form I: Lodgment form for the conversion of an old order prospecting right in terms of the Minerals Act, 1991, (Act No 50 of 1991), to one in terms of the Mineral and Petroleum Resources Development Act, 2002
- Form J: Lodgment form for the conversion of an old order mining right in terms of the Minerals Act, 1991, (Act No 50 of 1991), to one in terms of the Mineral and Petroleum Resources Development Act, 2002
- Form K: Application form for a reconnaissance permit in terms of section 74 of the Act, 2002
- Form L: Application form for a technical co-operation permit in terms of section 76 of the Act, 2002
- Form M: Application form for an exploration right in terms of section 79 and the renewal thereof in terms of section 81 of the Act, 2002
- Form N: Application form for a production right in terms of section 83 and the renewal thereof in terms of section 85 of the Act, 2002

MISCELLANEOUS PROCEDURES

- Form O: Notice of consultation in terms of section 10 of the Act
- Form P: Transfer and encumbrance of permits and rights in terms of section 11 of the Act
- Form Q: Application form for assistance to historically disadvantaged persons in terms of section 12 of the Act
- Form R: Refusal by Minister to grant permits/rights
- Form S: Permission to remove and disposal of minerals and petroleum in terms of section 20 of the Act
- Form T: Terms and conditions: Mineral beneficiation in terms of section 26 of the Act
- Form U: Application form in terms of section 43(2) of the Act to transfer environmental liabilities to qualified person
- Form V: Notice for suspension / cancellation in terms of section 47 of the Act
- Form W: Withdrawal of suspension Notice in terms of section 47(5) of the Act
- Form X: Notice for prohibition or restriction of granting of right/permit in terms of section 49 of Act
- Form Y: Notice in terms of section 49 of the Act: Profitability and curtailment of mining or production operations affecting employment
- Form Z: Notice by holder in terms of section 54(1) of the Act where the holder of any right or permit in terms of the Act cannot commence with operation
- Form AA Notice to land owner in terms of section 54(2) of the Act where the holder of any right or permit in terms of the Act cannot commence with operation
- Form BB Expropriation Notice in terms of section 55 of the Act
- Form CC Notice in terms of section 88(1) of the Act regarding contravention of provisions of Act
- Form DD Form in terms of section 94 of the Act

Form EE Application for a Closure certificate in terms of section 43(3) of the Act Form FF Closure certificate in terms of section 43 of the Act

PERMISSIONS, RIGHTS AND PERMITS IN TERMS OF THE ACT

Form GG Prospecting Right
Form HH Mining Right
Form II Mining Permit
Form JJ Reconnaissance Permit
Form KK Technical Co-operation Permit
Form LL Exploration Right
Form MM Operation Right

STANDARD FORMS FOR FINANCIAL PROVISION FOR THE REMEDIATION OF ENVIRONMENTAL DAMAGE

Form NN Standard form for approved contributions to a dedicated Trust Fund Form OO Standard form for a written guarantee from a bank or financial institution Form PP Standard form for financial deposits placed with the Department

ANNEXURE 2

MAXIMUM QUANTITIES OF ORES AND MINERALS REMOVABLE

CHAPTER 1

DEFINITIONS AND INTERPRETATION

SHORT TITLE AND COMMENCEMENT

These Regulations shall be called the Mineral and Petroleum Resources Development Regulations, 2002, and shall commence on the date to be determined by the Minister by notice in the *Government Gazette*.

STATUTORY AUTHORISATION

Section 107(1) of the Act authorises these Regulations.

DEFINITIONS AND INTERPRETATION

These Regulations shall be read in conjunction with the Act. In the event of conflict between the provisions of the Act and the Regulations, the provisions of the Act shall apply.

- In these Regulations, unless the context indicates otherwise
 - (a) "applicant" means any natural or juristic person who applies for a reconnaissance permission, prospecting right, mining right, mining permit, retention permit, exploration right, production right, reconnaissance permit or technical co-operation permit or renewal thereof in terms of the Act.
 - (b) "environmental impact assessment" means a process applied to assess the environmental consequences of applications for a mining or production right in terms of the Act and include the compilation of a scoping and environmental impact assessment report concerning such applications.
 - (c) "environmental management programme performance assessment" means a systematic, periodic, objective and documented evaluation of compliance with the environmental management programme approved in terms of the Act and the continued appropriateness and adequacy of the approved environmental management programme.
 - (d) "environmental management programme monitoring" means the use of quantitative and qualitative data gathering techniques in order to ascertain whether the requirements of an approved environmental management programme are being complied with and to supply supporting information for the environmental management programme performance assessment.
 - (e) "closure certificate" means a certificate issued in terms of section 43(1) of the Act.
 - (f) "form" means a document or standard form attached to the Regulations and which forms part of the Annexure hereto.
 - (g) "Social and Labour Plan forum" means a forum established in terms of the Regulations.
 - (h) "hazard" means a source of or exposure to danger.
 - (i) "installation" means installation as defined in the Mine Health and Safety Act, 1996, (Act No. 29 of 1996), as amended.
 - (j) "integrated environmental management" means a philosophy that prescribes an approach for ensuring that environmental considerations are fully integrated into all stages of the reconnaissance, prospecting, exploration, mining and production process in order to achieve a desirable balance between conservation and development.
 - (k) "interested and affected party" means a natural or juristic person or an association of persons with a direct interest in the proposed or existing operation or who will be directly affected by the proposed or existing operation.

- (1) "latent environmental impacts" means environmental impacts that may occur due to natural events or disasters.
- "optimal prospecting" means prospecting conducted by the holder of a prospecting (m) right in accordance with the prospecting work programme.
- "optimal mining" means mining conducted by the holder of a mining right or mining (n) permit in accordance with the mining work programme.
- "optimal" means in the case of petroleum operations, that the work is being carried out (0) in accordance with good internationally accepted oil-field practice subject to approval by the Petroleum Agency.
- "post-closure management" means the ongoing management of residual (p) environmental impacts for a specified period as determined after closure in terms of section 43 (1) of the Act has been obtained.
- "produced water" means water produced with petroleum from the sub-surface in the (q) course of production operations, and separated from the petroleum with the intention of discharging it into the environment.
- "regional mining development and environmental committee" means the committee (r) established in terms of section 64 of the Act and as contemplated in Regulations.
- "regional office" means an office of the Department situated within each region as (s) contemplated in section 8 of the Act. For the purposes of Chapter 3, this is the office of the Petroleum Agency.
- "residual environmental impacts" means the environmental impacts remaining after (t) mitigation.
- "risk assessment" means a process of gathering data and making assumptions to (u) estimate short and long term harmful effects on human health and the environment from exposure to hazards associated with the use of a particular product or technology; or establishing the probability of an event occurring, the factors that could bring about that event, likely exposure levels and the acceptability of the impact (4) resulting from exposure.
 - "risk" means the probability that injury or harm to persons and environmental (v) degradation will occur.
- "risk management" means using the information from the risk assessment to make . (w) and implement decisions about risk based on the balance between cost and benefits, for a range of options that mitigate or eliminate risk .
 - "qualified person" means a person who -(x)

The party Hally and the

- is qualified by virtue of his or her knowledge, training, skills and experience to organise the work and its performance; and
- is familiar with the provisions of the Act and other related legislation which apply to the work to be performed; and
- has been trained to recognise any potential or actual problem in the performance of the work.

harring visit in the about the terms.

resident i tradiç çarmının işkile, berilik barılını (ö. 1801)

market and real resides of the comment of the first section of

CHAPTER 2

MINERAL, SOCIAL AND ENVIRONMENTAL REGULATION

PART 1: MINERAL REGULATION

Manner of lodging applications and plan to be submitted

- Any application for a reconnaissance permission, prospecting right, mining permit, retention permit or reconnaissance permit, lodged in terms of the Act, must be completed on the appropriate Form in Annexure 1 of the Regulations.
 - (2) An application contemplated in sub-regulation (1) must be submitted to the regional office where the property or land is situated, by
 - (a) hand delivery;
 - (b) registered post;
 - (c) fax copy; or
 - (d) electronic format,
 - (3) An application contemplated in sub-regulation (1) must be accompanied by a plan that must contain –
 - (a) the co-ordinates of the area applied for;
 - (b) the north point;
 - (c) the scale to which the plan has been drawn:
 - (d) the name, number and location of the area or land covered by the application; and
 - (e) in relation to farm boundaries and surveyed points-
 - (i) the size and shape of the proposed area;
 - the boundaries of the land comprising the subject of the application concerned;
 - (iii) the layout of the proposed prospecting or mining operations;
 - (iv) surface structures and servitudes; and
 - (v) the topography of the area.

Consultation with interested and affected parties

- Any comments lodged by an interested and affected party or parties in terms of section 10(1)(b) of the Act, must be in writing and addressed to the relevant Regional Manager.
 - (2) Any objections lodged by an interested and affected party or parties against the application for a right or permit in terms of the Act, must set out clearly and concisely the facts upon which it is based and must be addressed to the relevant Regional Manager in writing.

Transferability and encumbrance of permits and rights

- (1) An application for the transferability and encumbrance of a prospecting right, mining right, technical co-operation permit, exploration right and production right must be in the form of Form P in Annexure 1.
 - (2) The cessionary, transferee, lessee, sub-lessee, assignee or the person to whom the right has been alienated or disposed of, must apply and sign as co-signatory to the application for the consent of the Minister in which he or she must undertake to comply with the provisions of section 11(2) of the Act as set out in the undertaking in Form P, Annexure 1.

Application for assistance to historically disadvantaged persons

 An application for assistance to historically disadvantaged person in terms of section 12 of the Act must be made to the Minister in the form of Form Q in Annexure 1.

Application for a reconnaissance permission

- An application for a reconnaissance permission in terms of section 13 of the Act must be completed on Form A in Annexure 1 and must contain the following-
 - (a) full particulars of the applicant;

- (b) a description of the area, land or properties to which the application relates;
- (c) detailed documentary proof of the applicant's financial ability or access thereto, which may include
 - bank statement;
 - (ii) budget approved for proposed operation;
 - (iii) loan obtained for proposed operation; or
 - (iv) resolution by company or board for the proposed operation.
- (d) proof of technical ability to conduct the said reconnaissance operation;
 (i) in-house expertise and the number of professionals used on the
 - i) in-house expertise and the number of professionals used on the project; and/or
 - contractors, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act; and/or
 - (iii) external expertise or consultants, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act.
- full particulars of a reconnaissance work programme and the duration thereof; and
- (f) full particulars of the estimated expenditure regarding the reconnaissance operations.

Application for a prospecting right

- 7. (1) An application for a prospecting right in terms of section 16 of the Act must be completed on Form B in Annexure 1 and must contain-
 - (a) full particulars of the applicant;
 - a plan as described in Regulation 2(3) indicating the co-ordinates and boundaries of the area applied for;
 - (c) an indication of the mineral for which the right is required;
 - (d) the period for which the right is required;
 - (e) a detailed prospecting work programme as contemplated in Regulation 9 with estimated expenditure during prospecting operations per year;
 - detailed documentary proof of the applicant's technical ability or access thereto to enable the applicant to carry out the prospecting operations and to mitigate and rehabilitate relevant environmental impacts;
 - (g) a description of how the applicant's technical ability or access thereto will be provided by making use of –
 - in-house expertise and the number of professionals used on the project; and/or
 - contractors, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act; and/or
 - (iii) external expertise or consultants, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act.
 - (h) detailed documentary proof of the applicant's financial ability or access thereto, which may include-
 - (i) bank statement;
 - (ii) budget approved for proposed operation;
 - (iii) loan obtained for proposed operation; or
 - (iv) resolution by company or board for the proposed operation.
 - a list of existing rights and permits held by the applicant to be compiled in a table format which indicate the region, the location with regard to the farm name and the existing right or permit number;
 - any other information, data or documentation which the Minister may request in connection with the information under paragraphs (a) to (i).
 - (2) After receiving the information required, the Regional Manager must submit the application forthwith to the Minister for consideration.

General terms and conditions of a prospecting right

- 8. (1) The terms and conditions of the prospecting right may be drafted in the form of Form GG, Annexure 1, or as the Minister may determine from time to time, and may contain particulars regarding-
 - (a) the parties to the agreement;
 - (b) the prospecting area applied for;

- (c) the mineral or associated minerals to be prospected for;
- (d) the period for which prospecting right is required;
- (e) prospecting fees payable;
- (f) the prospecting work programme;
- (g) indemnity;
- (h) joint and several liability;
- (i) waiver or lenience:
- (j) arbitration;
- (k) environmental protection;
- (I) cost;
- (m) severability; and
- (n) domicilium citandi et executandi of the parties; and
- any additional terms and conditions that the Minister may determine to be applicable from time to time.

(2) The holder of a prospecting right must -

- (a) exercise any right granted to him or her or it in terms of the provisions of this Act reasonably and in such manner that the rights and interests of the owner of any land to which the prospecting right relates are not adversely affected, except to the extent to which such owner is compensated;
- in the employment of employees, adopt and apply the principles outlined in the South African national affirmative action programmes, where possible;
- investigate, communicate and assess the potential environmental impacts of the proposed prospecting operation in accordance with the provisions of the Act;
- (d) prevent, minimise, mitigate and manage the environmental impacts in accordance with the approved environmental management plan:
- (e) monitor and assess the continued appropriateness, adequacy and environmental performance of the approved environmental management plan as determined;
- make appropriate and adequate financial provision for rehabilitation and remediation of environmental damage and revise such provision as required in terms of the Act;
- (g) where the holder of a prospecting right is a corporate body, or association of persons give proper notice in writing to the Minister within thirty (30) days of any change relating to the following-
 - (i) the change of name;
 - (ii) the registered address and principal place of business;
 - (iii) the change as to the directors of the company, or member of a closed corporation;
 - (iv) the share capital of the company;
 - (v) the beneficial owner or owners of the shares issued by such company in contravention to the provisions of the Competition Act;
- (h) in the case of a natural person, provide written notice to the Minister of any change of the address of such person within thirty (30) days.

Prospecting work programme

- The prospecting work programme as contemplated in section 18(3)(b) of the Act, shall contain the following -
 - (a) the mineral to be prospected for;
 - a description of the geology substantiated by geological maps;
 - (c) a description of how the mineral reserve and mineral distribution of the prospecting area will be determined through -
 - (i) prospecting work to be performed;
 - geochemical survey to be carried out;
 - (iii) geophysical survey to be undertaken;
 - (d) a description of the prospecting method or methods to be implemented which may include the following-
 - any excavations, trenching, pitting and drilling to be carried out;
 - (ii) any bulk sampling and testing to be carried out; and
 - (iii) any other prospecting methods to be applied.
 - (e) all planned prospecting activities must be conducted in phases and within specific timeframes;

 (f) technical data detailing the prospecting method or methods to be implemented and the time required for each phase of the proposed prospecting operation;

(g) details with documentary proof of the following-

- the applicant's technical ability or access thereto to conduct the intended operation; and
- the applicant's financial ability or access thereto to conduct the intended operation.
- (h) a cost estimate of the expenditure to be incurred for each phase of the prospecting operations where the expenditure must be broken down into -

(i) direct prospecting costs;

- (ii) labour costs; and
- (iii) costs pertaining to the rehabilitation and management of environmental impacts; and
- (iv) any other direct cost.

Progress report in respect of prospecting

- 10. (1) Within six months from the granting of a prospecting right, and every six months thereafter for the duration of the term of the prospecting right, the holder of a prospecting right must submit progress reports as contemplated in section 21(1(b) of the Act, to the Regional Manager detailing progress achieved in connection with the prospecting work programme.
 - (2) The progress report must contain the following-

(a) prospecting work conducted during the reporting period;

- (b) a surface plan of the prospecting area, the location, extent and depth of all boreholes, trenches or excavations completed;
- (c) the lithology, mineral content and mineral distribution identified in such boreholes, trenches or excavations;
- (d) any geological and or pre-feasibility study reports completed on the mineral or minerals obtained from the excavation, trench or borehole;
- actual expenditure incurred in terms of the prospecting area and the basis on which it was calculated;
- (g) details with regard to the execution and compliance with the approved environmental management plan;
- (h) a summary of performance regarding compliance to the Mine Health and Safety Act, 1996, (Act No. 29 of 1996);
- (i) prospecting work to be conducted during the next reporting period; and
- (i) any other information gathered by the holder of a prospecting right or requested by the Minister from time to time.
- (3) In respect of other prospecting methods employed excluding any form of excavation, the following information is required in the progress report-

(a) full details of any geophysical surveys conducted, including -

- flight plans or surface plans showing all flight lines or traverse lines;
 and
- (ii) any data in writing or digital format gathered during surveys.
- (b) full details of any geochemical surveys conducted, including-

surface plans showing all sample points;

(ii) details of all analysis carried out on these samples; and

(iii) full analytical results of all samples.

- (c) full results of any other investigations or tests performed, including-
 - positions of any other samples and bulk samples collected, including where applicable-

(aa) petrographic descriptions of thin sections;

- (bb) full information and analytical results of any age determinations;
- (cc) full results of any physical properties determined;
- (dd) full results of any chemical analysis performed;
- (ee) full results of any sedimentological investigations; and
- (ff) full results of any other tests or procedures performed.
- (4) True copies of any geological maps or plans produced during the prospecting operation must also be included in the progress report.

(5)The holder of a prospecting right must supply the Minister with a list of borehole core information generated during any drilling programme and must obtain permission from the Minister to destroy the core.

Application for the renewal of a prospecting right

11. An application for the renewal of a prospecting right in terms of section 18(1) of the (1) Act must be in the form of Form C in Annexure 1, and must contain-

full particulars of the applicant;

(b) a plan as described in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for;

the mineral for which the renewal of the prospecting right is required; (c)

(d) the period for which a renewal of the prospecting right is needed subject to the maximum period as stated in section 18(4) of the Act;

(e) reasons as to why a renewal is required;

- a report reflecting the results and interpretation of completed prospecting (f) activities and the costs thereof;
- details with regard to the execution and compliance with the approved (i) environmental management plan also reflecting the extent of environmental rehabilitation conducted and completed thus far and the costs thereof;
- (k) a summary of performance regarding compliance to the Mine Health and Safety Act, 1996 (Act No 29 of 1996); and
- a detailed prospecting work programme for the renewal period. (j)
- (2) After receiving the information required, the Regional Manager must submit the application forthwith for the renewal of the prospecting right to the Minister for consideration.

Application for a mining right

- An application for a mining right in terms of section 22(1) of the Act must be 12. (1)completed on Form D in Annexure 1 and must contain
 - full particulars of the applicant; (a)

(b) the name of the proposed mine;

- (c) a plan as described in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for:
- (d) the mineral for which the right is required;
- the period for which the right is required; (e)
- a mining work programme; (f)

a social and labour plan; (g)

detailed documentary proof of the applicant's technical ability or access thereto to enable the applicant to carry out the mining activities and to mitigate and rehabilitate relevant environmental impacts;

(i) a description of how the applicant's technical ability or access thereto will be provided by making use of -

- in-house expertise and the number of professionals used on the project; and/or
- contractors, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act; and/or
- external expertise or consultants, where a signed agreement must be provided to act on behalf of the applicant and take responsibility to comply to the requirements of the Act.
- detailed documentary proof of the applicant's financial ability or access (i) thereto, which may include
 - bank statement:
 - budget approved for proposed operation;
 - (iii) loan obtained for proposed operation; or

(iv) resolution by company or board for the proposed operation.

- (k) a list of existing rights and permits held by the applicant to be compiled in a table format which indicate the region, location with regard to farm name and the existing right or permit number; and
- any other information, data or documentation which the Minister may request in connection with the information required under paragraphs (a) to (k).

(2) After receiving the information required, the Regional Manager must submit the application forthwith for the mining right to the Minister for consideration.

Mining work programme

13. (1) An application for a mining right in terms of section 22(1)(b) of the Act, must include a mining work programme, which shall become an obligation in terms of the right granted. Such mining work programme must contain-

(a) full particulars of the applicant;

 a plan, co-ordinated according to generally accepted standards, showing the area under application;

(c) a feasibility study which must include the following-

- details of the identified mineral deposit concerned with regard to the type of mineral to be mined, its locality, extent, depth, geological structure, mineral content and mineral distribution;
- details of the market for, and the market's requirements in respect of, the mineral concerned;
- (iii) details and costing of the mining technique, mining technology, and production rates applicable to the intended mining operation;
- (iv) details and costing of the technological process applicable to the extraction of the mineral for preparation of the mineral to suit market requirements;
- details and costing of the technical skills and expertise and associated labour implications, required to conduct the intended mining operation;
- (vi) details with regard to the applicable timeframes and scheduling of the various implementation phases of the intended mining operation, and a technically justified estimate of the period required for the mining of the mineral deposit concerned;
- (vii) details and costing of the regulatory requirements which will be applicable to the mining operation;
- (viii) details regarding other relevant costing, capital expenditure requirements, and expected revenue applicable to the mining
- (ix) a detailed cash flow forecast and valuation, excluding financing, of the mining operation, which forecast must also clearly indicate how the applicable regulatory costs will be accommodated therein;
- details regarding the applicant's resources or intended mechanisms to finance the mining operation, and detail regarding the impact of such financing arrangements on the cash flow forecast; and
- (xi) an assessment of the estimated costs to mitigate the potential environmental impact.

General terms and conditions of a mining right

- 14. (1) The terms and conditions of the mining right in terms of section 25(2)(c) of the Act may be drafted in the form of Form HH in Annexure 1 or as the Minister may determine from time to time, and must contain particulars regarding
 - (a) parties to the agreement;

(b) the mining area applied for;

- (c) the mineral or associated minerals to be mined;
- (d) the period for which the mining right is required;
- (e) consideration (royalties or minimum royalties payable);
- (f) restrictions and obligations applicable on the holder;
- (g) indemnity;
- (h) transferability;
- (i) protection of boreholes, shafts, etc.;
- protection of property and persons;
- (k) records and returns;
- (I) joint and several liability;
- (m) waiver of lenience;
- (n) arbitration;
- (o) environmental protection;
- (p) cost;
- (q) severability;
- (r) domicilium citandi et executandi of the parties; and

- (s) any additional terms and conditions which the Minister may request to be applicable to an application.
- (2) The holder of a mining right must -
 - exercise any right granted to him or her or it, reasonably and with due regard to such rights and interests of the holder of the right in the land or area to which the mining right relates;
 - engage in joint-ventures or any other form of co-operation with the Black Economic Empowerment entities to develop technical skills and infrastructure;
 - investigate, communicate and assess the potential environmental impacts of the proposed mining operation in accordance with the provisions of the Act;
 - (d) prevent, minimise, mitigate and manage the environmental impacts in accordance with the approved environmental management programme;
 - (e) monitor and assess the continued appropriateness, adequacy and environmental performance of the approved environmental management programme as determined;
 - (f) make appropriate and adequate financial provision for rehabilitation and remediation of environmental damage and revise such provision as required in terms of the Act;
 - (g) where the holder of a mining right is a corporate body or association of persons to give proper notice in writing to the Minister within 30 days of any change relating to the following-
 - (i) the change of name;
 - (ii) the registered address and principal place of business;
 - the change as to the directors of the company, or member of a closed corporation;
 - (iv) the share capital of the company; and
 - the beneficial owner or owners of the shares issued by such company in contravention to the provisions of the Competition Act.
 - (h) as far as possible make use of local contractors, services, equipment purchased or hired; and
 - in the case of a natural person, provide written notice to the Minister of any change of the address of such person within 30 days.

Application for the renewal of a mining right

- An application for the renewal of a mining right in terms of section 24(1) of the Act must be completed on Form E in Annexure 1 and must contain -
 - (a) full particulars of the applicant;
 - a plan as contemplated in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for;
 - (c) the mineral for which the renewal of the mining right is required;
 - (d) the period for which a renewal is required;
 - (e) a detailed revised mining work programme for the renewal period;
 - (b) a revised social plan for the renewal period;
 - (g) details with regard to the execution and compliance with the approved environmental management programme also reflecting the extent of environmental rehabilitation conducted and completed thus far and the costs thereof;
 - (h) details with regard to financial provision for environmental rehabilitation, documentary proof of the applicant's financial ability or access thereto, the approved methodology and the amount to be provided for the renewal of the mining right;
 - a summary of performance regarding compliance to the Mine Health and Safety Act, 1996, (Act No 29 of 1996);
 - any other information or documentation the Minister may request under paragraphs (a) to (i).
 - (2) After receiving the information required, the Regional Manager must submit the application forthwith for the renewal of the mining right to the Minister for consideration.

Information, data and statistical returns in respect of mining or processing of minerals

- 16. (1) The holder of a mining right must for the duration of such right and on the last day of every month, submit statistical returns to the Minister, regarding information and data in respect of mining or processing of minerals which include the following-
 - accurate returns of the work accomplished, production statistics for the reporting year and production forecast for the ensuing year, the persons employed, total mining costs and the results obtained;
 - (b) accurate returns for the preceding month of the production, purchases, sales and exports transacted by the holder of the mining right; and
 - (2) The returns must reach the Minister not later than the 15th of the month in respect of which it is made and must be forwarded as set out in forms to be provided by the Department for that purpose.
 - (3) The holder of a mining right shall furnish the Minister annually and in such forms as provided by the Department for this purpose by him or her with returns regarding matters pertaining to employees, salaries and wages.
 - (4) Such holder shall also provide the Minister at his or her request with such other returns and data as may be reasonably required. Each such return and other data shall be furnished on or before a date as requested by the Minister and specified on the forms as provided or otherwise notified in writing.

Annual report in respect of mining

- 17. The prescribed annual report as provided for in section 25(2)(h) of the Act must contain -
 - (1) full information in respect of the mine's business for that year, reflecting -

(a) production figures;

- (b) all financial information in respect of costs, capital expenditure and revenue.
- (2) annual progress regarding the execution of the social and labour plan; and
- (3) annual progress in terms of the socio-economic empowerment as provided for in section 100 of the Act.

Application for a mining permit

18. (1) An application for a mining permit in terms of section 27(2) of the Act must be completed on Form F in Annexure 1 and must contain-

(a) full particulars of the applicant;

- a plan as contemplated in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for, and not exceeding 1,5 ha per application;
- (c) an indication of the mineral, associated minerals or group of minerals for which the permit is required;

(d) the period for which the permit is required, not exceeding two (2) years;

(e) documentary evidence to prove the applicant's technical ability and financial resources or access thereto that are readily available or how it will be provided for, to enable the applicant to carry out the mining activities and to mitigate and rehabilitate relevant environmental impacts;

 details of any mineral or group of minerals and the quantity thereof, which the applicant intends to remove and dispose of during mining operations;

- (g) any other information, data or documentation which the Minister may request under paragraphs (a) to (f).
- (2) After receiving the information required, the Regional Manager must submit the application forthwith for the mining permit to the Minister for consideration.

Application for the renewal of a mining permit

- 19. (1) An application for the renewal of a mining permit in terms of section 27(8) of the Act must be completed on Form F in Annexure 1 and must contain-
 - (a) full particulars of the applicant;

- (b) a plan as contemplated in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for;
- (c) an indication of the mineral for which the permit is required;
- (d) the period for which the permit is required, not exceeding one (1) year;
- (e) documentary evidence to prove the applicant's technical ability and financial resources or access thereto that are readily available or how it will be provided for, to enable the applicant to carry out the mining activities and to mitigate and rehabilitate relevant environmental impacts; and
- (f) any other information, data or documentation that the Minister may request in connection with the information required under paragraphs (a) to (e).
- (2) After receiving the information required, the Regional Manager must submit the application forthwith for the renewal of the mining permit to the Minister for consideration.

Application for a retention permit

- 20. (1) An application for a retention permit in terms of section 31(1) of the Act must be completed on Form G in Annexure 1 and must contain -
 - (a) full particulars of the applicant;
 - (b) the period for which the retention permit is required:
 - particulars of the prospecting right in terms of which prospecting operations was conducted on the area concerned;
 - (d) a plan as contemplated in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for;
 - a detailed geological description of the area of land to which the application relates;
 - a description of the mineral or group of minerals for which the retention permit is required;
 - (g) documentary evidence to prove the existence of the mineral reserves occurring within the area under application;
 - (h) detail with regard to the execution of and compliance with the approved environmental management plan;
 - particulars, substantiated by documentary proof, of the technical ability and financial resources of, or available to the applicant;
 - the reasons why mining operations cannot be carried out on the area concerned;
 - (k) an indication of the circumstances in which and the earliest date on which, such mineral or group of minerals can in the opinion of the applicant, be mined or won on a profitable basis; and
 - any other information, data or documentary evidence that the Minister may request.
 - (2) After receiving the information required, the Regional Manager must submit the application forthwith for the retention permit to the Minister for consideration.

General terms and conditions of retention permit

- 21. (1) Every retention permit is, unless expressly waived, varied or suspended in writing by the Minister, subject to the conditions that the holder shall in relation to the retention area -
 - carry out progressive programmes of work for the purposes of testing prevailing market conditions and metallurgical testing;
 - (b) not carry out any programme involving substantial disturbance of the prospecting area without the written approval of the Minister and comply with such directions as the Minister consider appropriate for the protection of the environment or other aspect of or in relation to the land in carrying out that programme;
 - (c) not sell any mineral obtained from the retention area, except with the prior approval, in writing, of the Minister, nor remove from the retention area any mineral other than that which is required for testing purposes;
 - (d) not erect any permanent building or facility except with the approval in writing of the Minister; and
 - (2) Any other conditions, not inconsistent with this Regulation or the specific provision of the Act imposing conditions to which the retention permit is subject, or as the Minister may determine and make an endorsement on the retention permit.

Application for the renewal of a retention permit

- 22. An application for renewal of a retention permit in terms of section 34(1) of the Act must be completed on Form H in Annexure 1, and must-
 - (a) reflect updated information of market circumstances prevailing; and
 - (b) any other information or data which the Minister may request.

Purpose and objectives of the Regional Mining Development and Environmental Committees

- (1) The purpose and objectives of the Regional Mining Development and Environmental Committees include the following-
 - to advise the Minister and Regional Manager with regard to the implementation and enforcement of the Act;
 - (b) to promote co-operative Governance; and
 - (c) to ensure adequate participation by interested and affected persons.

Establishment and functions of the Regional Mining Development and Environmental Committees

- 24. (1) A Regional Mining Development and Environmental Committee contemplated in section 64(1) of the Act, must be established for each region referred to in section 7 of the Act within 30 days after the promulgation of the Act.
 - (2) The constitution and working procedure of the Committee must be established by the Minerals and Mining Development Board.
 - (3) The Committee shall consists of such persons who by virtue of their knowledge and understanding of economic development and mining environmental management, can contribute to informed decision-making pertaining to sound environmental management and optimal utilization of mineral resources.
 - (4) The Committee must consist of not more than 14 members appointed by the Minerals and Mining Development Board and approved by the Minister and shall include
 - the Regional Manager designated by the Minerals and Mining Development Board, to be the Chairperson;
 - a representative of the Mine Health and Safety Inspectorate of the Department; and
 - (c) representatives of relevant organs of State;
 - (d) organised labour, organised business, any relevant non-governmental organisation and any relevant community based organisation.
 - (5) A further two persons may be co-opted in cases where their specific qualifications or knowledge will contribute to informed decision-making.
 - (6) Any person co-opted by the Committee, shall be deemed to be a member of the Committee for that meeting and be entitled to vote on the relevant question before the Committee.
 - (7) The Committee must meet to discuss, evaluate and make recommendations to the Minister only if objections against the application for a prospecting right, mining right or mining permit and/or the approval of a draft environmental management programme or plan, as the case my be, have been lodged by interested or affected persons.

Process of handling objections by the Committee

- 25. (1) Any objections received regarding any application in terms of this Act must be in writing and submitted to the Chairperson of the Committee.
 - (2) The Committee must consider the objections received within 30 days whereafter the Chairperson must submit the findings and recommendations in writing to the Board.
 - (3) The Board must advise the Minister within 14 days of the date on which the recommendations of the Committee have been received.

(4) Where disagreement regarding the recommendations to be made by the Committee cannot be resolved within the Committee, the Chairperson of the Committee must refer the matter to the Minerals and Mining Development Board who will make recommendations to the Minister.

PART II: SOCIAL AND LABOUR PLAN

Submission of a Social and Labour Plan and its applicability

- 26. (1) Any application for a mining right in terms of this Act shall be accompanied by a social and labour plan. The Minister may refer the said social and labour plan back to the applicant with proposals and the revised social and labour plan must then be lodged.
 - (2) The social and labour plan shall apply –

(a) to all applications for a mining right;

 to all applications for the conversion of an old order right in terms of the Minerals Act, 1991, (Act No 50 of 1991) to a mining right in terms of the Act;

(c) throughout the life of a mine referred to in sub-regulation (a) or (b);

(d) in the event of possible large-scale retrenchments at a mine referred to in sub-regulation (a) and (b) and when such retrenchments may impact on the sector or regional or national interest.

Objects of Social and Labour Plan

27. (1) The objects of the social and labour plan are to-

- (a) consider the development of the mining operation in the context of generally recognized standards of sustainable development by integrating social, economic and environmental factors in planning the mining operations throughout the life of the mine;
- endeavour to avoid job losses and a decline in employment by promoting alternative solutions to threatening job security and potential measures to prevent a decline in employment;
- endeavour to ameliorate the social and economic impact on individuals, regions and the economy in cases where job losses remain unavoidable after exploring alternative solutions and measures;
- endeavour to avoid the unnecessary establishment of settlements which cannot be sustained after the closure of the mine;

(e) put a social and labour plan strategy in place;

(f) put a human resource development strategy in place;

(g) include monitoring, contingency and reporting mechanisms; and

(h) endeavour to reach its conclusion once all the obligations in terms of the social and labour plan have been met.

Contents of Social and Labour Plan

- 28. (1) The social and labour plan proposal must include the following, but is not limited thereto-
 - a preamble, describing the background to the mine, in terms of its location, commodity to be mined, scale of operations and size of workforce;
 - (b) a description of the local economic background of the area in which the mine will be or is situated, the areas from which labour will mainly be sourced, the key economic activities of the area, and the socio-economic impact of the mine's operations in the area;
 - a summary of the objectives of the social and labour plan and how it will be achieved;
 - (d) the social and labour plan strategy of the mine;

(e) the human resource development strategy of the mine;

- (f) the implementation process, which must include the establishment of a future forum, and the action plans, timeframes and other relevant information during the construction, operational, decommissioning, closure and postclosure phase of the mine with respect to the social and labour plan objectives;
- (g) monitoring, contingency mechanisms and reporting mechanisms;
- (h) any additional information which may be requested by the Minister; and
- an undertaking by the applicant to adhere to the requirements and conditions set out in the social and labour plan.

Social and Labour Plan forum

- 29. (1) A social and labour plan forum comprising of the holder of a mining right, the employees and the relevant local authority, each of which shall be represented by a maximum of five (5) persons, will be established to ensure that the objectives of the social and labour plan is achieved.
 - (2) The purpose of the forum is-

(a) to implement the social and labour plan; and

 to monitor the compliance and the appropriateness of the social and labour plan.

PART III: ENVIRONMENTAL REGULATION

Environmental reports to be compiled for the application of a mining right

- 30. (1) An applicant who has applied for a mining right in terms of section 22 of the Act must conduct an environmental impact assessment and submit a draft environmental management programme to the Regional Manager within 180 days of the date of notification by the Regional Manager.
 - (2) An environmental impact assessment as contemplated in section 39(1) of the Act, includes the compilation of-

(a) a scoping report; and

(b) an environmental impact assessment report.

Scoping Report

- 31. (1) A scoping report must identify the impacts and reasonable alternatives to be addressed in the environmental impact assessment report. The result of the scoping process must determine the nature and extent of investigations to be undertaken in the environmental impact assessment report.
 - (2) A scoping report must -
 - (a) describe the methodology, including the stakeholder engagement process, of the scoping process;
 - (b) describe the existing status of the environment prior to mining;
 - (c) identify and describe the anticipated environmental, social and cultural impacts, including the cumulative effects of the proposed operation;
 - identify and describe reasonable alternatives to the project and alternative means of carrying out the project;
 - (e) propose the most appropriate approach and procedure to plan and develop the proposed mining project;
 - describe the views and concerns of relevant authorities and interested and affected persons and the participation process followed; and
 - (g) describe the nature and extent of further investigations required in the environmental impact assessment report.
 - (3) If the scoping report indicates that the proposed mining operation will result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must refuse the application.
 - (4) If the scoping report indicates that the proposed mining operation will not result in unacceptable pollution, ecological degradation or damage to the environment, the Regional Manager must request the applicant to submit -
 - an environmental impact assessment report based on the outcome of the scoping report; and
 - (b) a draft environmental management programme based on the environmental impact assessment report.

Environmental Impact Assessment Report

32. (1) An environmental impact assessment report must determine the nature, extent,

duration, probability and significance of the environmental, social and cultural impacts of the proposed mining operation, the reasonable alternatives and the potential mitigation measures for each impact during each of the life of the proposed mining operation.

- (2) An environmental impact assessment report must as a minimum ensure the following-
 - investigation of the environment likely to be significantly affected by the proposed mining operation and alternatives thereto;
 - investigation of the potential impact, including cumulative impacts, of the proposed operation and its alternatives on the environment, socio-economic conditions and cultural heritage;
 - an assessment of the nature, extent, duration, probability and significance of the potential environmental, social and cultural impacts of the proposed operation, including the cumulative impacts;
 - a comparative assessment of the feasible alternatives and their potential environmental, social and cultural impacts;
 - (c) investigate and describe mitigatory measures for each significant impact of the proposed activity and the preferred alternative to keep adverse impacts to a minimum.
 - (d) description of the stakeholder engagement process followed during the course of the assessment and an indication of how issues raised have been addressed;
 - reporting on knowledge gaps, the adequacy of predictive methods and underlying assumptions, and uncertainties encountered in compiling the required information; and
 - description of the arrangements for the monitoring and management of impacts, and the assessment of the effectiveness of such arrangements after their implementation; and
 - (g) inclusion of technical and supporting information as appendices.

Investigation of cumulative impacts

- 33. (1) In the environmental impact assessment report, the applicant must determine whether specific resources and ecological components are cumulatively affected by the proposed mining operation by considering-
 - (a) whether the proposed operation is one of several projects or activities in the same geographic area;
 - whether other projects or activities in the area have similar impacts on the specific resource and ecological components;
 - (c) whether these impacts have been historically significant for this resources and ecological components; and
 - (d) whether other investigations in the area have identified a cumulative impact concern.
 - (2) To avoid extending data and analytical requirements beyond those relevant for decision-making, the selection of geographic boundaries and time period should be, whenever possible, be based on the natural boundaries of the resources of concern and the period of time that the proposed project's impacts will persist.

Contents of Environmental Management Programme

- 34. (1) An environmental management programme must describe the environmental objectives and specific goals for the proposed operation, the appropriate management options for the ongoing management and monitoring of the impacts of the activity, until closure.
 - (2) A draft environmental management programme must provide -
 - (a) a description of the environmental objectives and specific goals for-
 - (i) mine closure;
 - the management of identified environmental impacts emanating from the proposed operation;
 - (iii) the socio-economic conditions; and
 - (iv) the historical and cultural aspects (if applicable);
 - (b) an outline of the implementation programme which must include -

- a description of the chosen appropriate technical and management options for each environmental impact, socio-economic condition and historical and cultural aspects of the preferred alternative under each phase of the mining operation;
- (ii) a layout of action plans to achieve the objectives and specific goals also indicating a time schedule of actions undertaken within the action plans to implement mitigatory measures and options for prevention, management and remediation;
- (iii) maintenance and emergency procedures or further remediation;

(iv) contingency measures;

- requirements for monitoring and environmental management programme performance assessment;
- (vi) financial provision which must include-

(aa) the determination of the quantum;

- (bb) details of the method of provision and revision thereof;
- (vii) an environmental awareness plan in accordance with guidelines published by the Department from time to time;

(viii) supporting information and appendices; and

- (ix) undertaking by the applicant to comply with the provisions of the Act
- (2) If the environmental impact assessment report and the draft environmental management programme indicate that the proposed mining operation will result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must refuse the application.
- (3) If the environmental impact assessment report and draft environmental management programme indicate that the proposed mining operation will not result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must approve the application.

Independent review

- 35. (1) The Minister may appoint a person to undertake an independent review, if a reasonable assessment indicates that the environmental impact assessment report and the draft environmental management programme or environmental management plan, as the case may be, cannot be executed satisfactorily by the applicant. Such appointment and execution shall be for the cost of the applicant.
 - (2) The Minister must notify the applicant and provide reasons for the need for such independent review.
 - (3) The person who will undertake such an independent review must provide proof that he or she has no vested interest or involvement in the proposed prospecting or mining operation.

Environmental report to be compiled for the application of a reconnaissance permission, prospecting right or mining permit

36. (1) An applicant who has applied for a reconnaissance permission, prospecting right or mining permit in terms of the Act must submit an environmental management plan to the Regional Manager within 180 days of the date of notification by the Regional Manager.

Environmental Management Plan

- 37. (1) An environmental management plan must address the potential impacts of the proposed operation, address feasible mitigatory and dedicated management and monitoring options for the proposed operation.
 - (2) The environmental management plan, shall be in the format provided by the Department and must contain-
 - a description of the environment likely to be affected by the proposed prospecting or mining operation;
 - (b) an assessment of the potential impacts of the proposed prospecting or mining operation on the environment, socio-economic conditions and cultural heritage;

- a summary of the significance of the potential impacts, the preferred alternative methods, proposed mitigation and management measures to minimise adverse impacts and optimize benefits, as well as the option of not implementing the activity;
- (d) financial provision which must include-
 - (i) the determination of the quantum;
 - ii) details of the method(s) of provision;
- (e) requirements for monitoring and performance assessment;
- (f) closure and environmental objectives;
- (g) a record of the public participation undertaken; and
- (h) an undertaking by the applicant regarding the implementation of the environmental management plan.

Method(s) for financial provision

- 38. Financial provision for rehabilitation and remediation of environmental damage must be made by one or more of the following methods-
 - approved contributions to a dedicated trust fund as provided for in section 10(1)(cH) of the Income Tax Act, 1962, (Act No. 58 of 1962) and registered with the South African Revenue Services; and / or
 - (b) a written guarantee from a bank or approved financial institution guaranteeing the amount specifically for rehabilitation and the remediation of environmental damage and the availability of funds on demand of the Department of Minerals and Energy if the proposed prospecting or mining operation should fail or become incapacitated; and/or
 - financial deposits to be placed within a central fund with Government or an institution to be approved by the Minister; and/or
 - any other methodology of financial provision to be approved by the Minister from time to time.

Standard forms for financial provision

- 39. The standard form for the-
 - (a) trust fund must be in the format set out in Form NN, Annexure 1;
 - (b) written guarantee must be in the format set out in Form OO, Annexure 1; and
 - (c) financial deposits must be in the format set out in Form PP, Annexure 1.

Quantum of financial provision

- 40. (1) The Minister shall request from the holder of a reconnaissance permission, prospecting right, mining right or mining permit, as the case may be, to submit financial statements from a financial institution as proof that the holder has the financial means to execute the requirements of the environmental management programme or plan.
 - (2) The quantum of the financial provision to be approved by the Minister must be based on the requirements of the environmental management programme or plan and shall include a detailed itemization of all actual costs required for-
 - (a) the rehabilitation of the surface of the area;
 - (b) the prevention and management of pollution from the atmosphere;
 - (c) the prevention and management of pollution of water and the soil;
 - (d) decommissioning and final closure of the operation; and
 - (e) post closure management of residual and latent impacts.
 - (3) The holder of a reconnaissance permission, prospecting right, mining right or mining permit must annually update and review the quantum of the financial provision -
 - (a) in consultation with a qualified person;
 - (b) as required in terms of the approved environmental management programme or environmental management plan; or
 - (c) as requested by the Minister.
 - (4) Any inadequacies with regard to the financial provision must be rectified by the holder of a reconnaissance permission, prospecting right, mining right or mining permit as-
 - determined in an amendment of the environmental management programme or plan as the case may be;
 - (b) within the timeframe provided; or
 - (c) as determined by the Minister.

Monitoring and performance assessments of environmental management programme or plan

- 41. (1) As part of the general terms and conditions for a prospecting right, mining right or mining permit and in order to ensure compliance with the approved environmental management programme or plan and to assess the continued appropriateness and adequacy of the environmental management programme or plan, the holder of such right must-
 - (a) conduct monitoring on a continuous basis;
 - (b) conduct performance assessments of the environmental management programme or plan as required; and
 - (c) compile and submit a performance assessment report to the Regional Manager to demonstrate adherence to sub-regulation (b).
 - (2) The frequency of performance assessment reporting shall be-
 - (a) in accordance with the period specified in the approved environmental management programme or plan, or, if not so specified;
 - (b) as agreed to in writing by the Minister; or
 - (c) biennially (every two years).
 - (3) The performance assessment report, shall be in the format provided in guidelines that will from time to time be published by the Department and contain-
 - (a) information regarding the period that applies to the performance assessment;
 - (b) the scope of the assessment;
 - (c) the procedure used for the assessment;
 - (d) the interpreted information gained from monitoring the approved environmental management programme or plan;
 - (e) the evaluation criteria used during the assessment;
 - (f) the results of the assessment; and
 - (g) recommendations on how and when deficiencies that are identified and/or aspects of non-compliance will be rectified.
 - (4) The holder of a prospecting right, mining right or mining permit may appoint an independent qualified person(s) to conduct the performance assessment and compile the performance assessment report provided that no such appointment shall relieve the holder of the responsibilities in terms of these regulations.
 - (5) After submission of the performance assessment report, the report shall be made available by the Regional Manager to any person on request.
 - (6) If upon consideration by the Regional Manager, the performance assessment executed by the holder is not satisfactory or the report submitted by the holder is found to be unacceptable, the holder must-
 - repeat the whole or relevant parts of the performance assessment and revise and resubmit the report; and/or
 - (b) submit relevant supporting information; and/or
 - (c) appoint an independent qualified person(s) to conduct the whole or part of the performance assessment and to compile the report.
 - (7) If a reasonable assessment indicates that the performance assessment cannot be executed satisfactorily by the holder or a qualified person(s) appointed by the holder, the Minister may appoint an independent performance assessment person(s). Such appointment and execution shall be for the cost of the holder.
 - (8) When the holder of a prospecting right, mining right or mining permit intends closing such operation, a final performance assessment shall be conducted and a report submitted to the Regional Manager to ensure that -
 - the requirements of the relevant legislation have been complied with;
 - (b) the closure objectives as described in the environmental management programme or plan have been met; and
 - (c) all residual environmental impacts resulting from the holder's operations have been identified and the risks of latent impacts which may occur have been identified, quantified and arrangements for the management thereof have been assessed.

(9) The final performance assessment and the submission of the required report shall either precede or accompany the application for a closure certificate in terms of the Act.

General requirements for mine closure

- 42. (1) The holder of a prospecting right, mining right, retention permit and mining permit must ensure that
 - the closure of a prospecting or mining operation incorporates a process which must start at the commencement of the operation and continue throughout the life of the operation and within the framework of sustainable development;
 - risks to elements of the environment must be quantified and managed proactively, which includes the gathering of relevant information throughout the life of a prospecting or mining operation;
 - (c) the safety and health of humans and animals are safeguarded from hazards resulting from prospecting or mining operations;
 - environmental damage or residual impacts are minimised to such an extent that it is acceptable to all involved interested and affected persons;
 - the land is rehabilitated to, as far as is practicable, its natural state, or to a
 predetermined and agreed standard or land use which conforms with the
 concept of sustainable development;
 - the optimal exploitation and utilization of mineral resources are not adversely affected; and
 - (g) prospecting or mining operations are closed efficiently and cost effectively.

Application for a closure certificate

- 43. (1) An application for a closure certificate of a prospecting right, mining right, retention permit or mining permit in terms of section 43(4) of the Act must be completed on the Form EE as set out in Annexure 1.
 - (2) An application for a closure certificate of a prospecting right, mining right, retention permit or mining permit must be accompanied with a closure plan.

Application to transfer environmental liabilities to a qualified person

- 44. (1) The holder of a prospecting right, mining right or mining permit may transfer liabilities and responsibilities as identified in the environmental management programme, or environmental management plan or any prescribed closure plan to a qualified person as contemplated in Regulation 45.
 - (2) An application to transfer environmental liabilities to a qualified person in terms of section 43(2) of the Act, must be completed on Form U as set out in Annexure 1 and be lodged to the Minister for consideration.
 - (3) When considering the transfer environmental liabilities and responsibilities in terms of section 43(2) of the Act, the Minister must consult with any State department which administers any law relating to matters affecting the environment.
 - (4) No transfer of environmental liabilities and responsibilities to a qualified person may be made unless the Chief Inspector of Mines and the Department of Water Affairs and Forestry have confirmed in writing that the person to whom the liabilities and responsibilities is transferred to, have the necessary qualifications pertaining to health and safety and management of potential pollution of water resources.

Qualifications of person regarding transfer of environmental liabilities and responsibilities

45. (1) For the purposes of transferring environmental liabilities and responsibilities as may be identified in the environmental management programme or plan and any prescribed closure plan, the person to whom such transfer is made must –

 have the expertise, resources and organisational abilities to integrate risk assessment, risk management and risk financing to ascertain the cost of environmental management:

 (b) have the expertise, financial and other resources to meet its obligations to carry out actions necessary to fulfil the environmental obligations as set out in

- the environmental management programme, environmental management plan or any prescribed closure plan concerned;
- (c) have appropriate experience in environmental management and mine health and safety matters;
- (d) have direct access to insurance products and alternative risk financing services appropriate to financing of exposure to risks;
- (e) have the ability to manage trusts set up in terms of section 10(1)(cH) of the Income Tax Act, 1962 (Act No 58 of 1962); and
- (f) have expertise and experience or proven access thereto to interpret and manage the findings of an environmental risk assessment.

Environmental Risk Report

- 46. (1) An application for a closure certificate in terms of section 43(3) of the Act must be accompanied by an environmental risk report which must include.
 - (a) the undertaking of a screening level environmental risk assessment where-
 - (i) all possible environmental risks are identified, including those which appear to be insignificant;
 - (ii) the process is based on the input from existing date;
 - (iii) the issues that are considered are qualitatively ranked as –
 (aa) a potential significant risk; and/or
 - (bb) a uncertain risk; and/or
 - (cc) an insignificant risk.
 - (b) the undertaking of a second level risk assessment on issues classified as potential significant risks where-
 - (i) appropriate sampling, data collection and monitoring be carried out;
 - (ii) more realistic assumptions and actual measurements be made; and
 - (iii) a more quantitative risk assessment is undertaken, again classifying issues as posing a potential significant risk or insignificant risk.
 - (c) assessing whether issues classified as posing potential significant risks are acceptable without further mitigation;
 - issues classified as uncertain risks be re-evaluated and re-classified as either posing potential significant risks or insignificant risks;
 - documenting the status of insignificant risks and agree with interested and affected persons;
 - identifying alternative risk prevention or management strategies for potential significant risks which have been identified, quantified and qualified in the second level risk assessment;
 - (g) agreeing on management measures to be implemented for the potential significant risks which must include-
 - (i) a description of the management measures to be applied;
 - (ii) a predicted long-term result of the applied management measures;
 - (iii) the residual and latent impact after successful implementation of the management measures;
 - (iv) time frames and schedule for the implementation of the management measures:
 - (v) responsibilities for implementation and long-term maintenance of the management measures;
 - (vi) financial provision for long-term maintenance; and
 - (vii) monitoring programmes to be implemented.

Closure objectives

- Closure objectives form part of the draft environmental management programme or environmental management plan, as the case may be, and must –
 - identify the key objectives for mine closure to guide the project design, development and management of environmental impacts;
 - (b) provide broad future land use objective(s) for the site; and
 - (c) provide proposed closure costs.

Contents of Closure plan

- 48. (1) A closure plan in terms of section 43(3)(d) of the Act, forms part of the environmental management programme or environmental management plan, as the case may be, and must include
 - (a) a description of the closure objectives and how these relate to the prospecting or mine operation and its environmental and social setting;

- a summary of the regulatory requirements and conditions for closure negotiated and documented in the environmental management programme or plan;
- (d) a summary of the results of the environmental risk report and details of identified residual and latent impacts;

(e) a summary of the results of progressive rehabilitation undertaken;

 a description of the methods to decommission each prospecting or mining component and the mitigation or management strategy proposed to avoid, minimize and manage residual or latent impacts;

(g) details of any long-term management and maintenance expected;

- (h) details of financial provision for monitoring, maintenance and post closure management, if required;
- a plan or sketch at an appropriate scale describing the final land use proposal and arrangements for the site;

(j) a record of interested and affected persons consulted; and

(k) technical appendices, if any.

PART IV: POLLUTION CONTROL AND WASTE MANAGEMENT REGULATION

Principles of pollution control and waste management

49. (1) In accordance with the national norms and standards for pollution control and waste management, the holder of any right or permit in terms of the Act must -

a) avoid the generation and production of pollution, waste and mine residue at

source; or

 (b) where the production thereof cannot altogether be avoided, be minimized and re-used or recycled; and

(c) where possible and otherwise disposed of in a responsible and prescribed

Air quality management and control

- 50. (1) The holder of a right or permit in terms of the Act, must comply with national air quality control or management norms and standards, where applicable.
 - (2) The assessment of impacts relating to air quality control and management, where appropriate, must form part of the environmental impact assessment report and draft environmental management programme or the environmental management plan, as the case may be.

Fire prevention

- 51. (1) All coal debris and bituminous rock shall be deposited at a site as identified and approved in the environmental management programme or plan.
 - (2) The holder of a mining right pertaining to the mining of coal shall be responsible for all damages resulting from the combustion of the coal residue stockpiles or deposits.

Noise control

- 52. (1) The holder of a right or permit in terms of the Act must comply with national norms and standards regarding noise control where applicable.
 - (2) The assessment of impacts relating to noise pollution control and management, where appropriate, must form part of the environmental impact assessment report and draft environmental management programme or the environmental management plan as the case may be.
 - (3) When the equivalent noise exposure, as defined in the South African Bureau of Standards Code of Practice for the Measurement and Assessment of Occupational Noise for Hearing Conservation Purposes, SABS 083 as amended, [in any at or in any mine or works where persons may travel or work, exceeds 85 dB (A)] exceeds the background plus seven (7) dB (A) decibels beyond the mine boundary or at the residential community or where non employees may travel, the holder shall take the necessary steps to reduce the noise to below this level.

- (4) Where compliance with sub-regulation (3) is not possible, the holder shall implement a hearing conservation programme, which complies with the recommendations given in the said SABS Code 083.
- (5) All mechanical equipment must be in a good working order and vehicles must adhere to the relevant noise requirements of the Road Traffic Act, 1997, (Act No. 93 of 1997).
- (6) All vehicles in operation must be in good working order and equipped with a silencer on their exhaust system.
- (7) Where appropriate, lubricants must be applied to ensure that surfaces which interact during mechanical movement do not generate undesirable noise levels.
- (8) Safety measures which generate noise such as reverse gear alarms on large vehicles must be appropriately calibrated or adjusted.
- (9) Appropriate non-metallic washers or isolation must be used with any joining apparatus to join screens such as corrugated iron to other structures and to each other. Such screens must be maintained in a fixed position.
- (10) Appropriate measures must be installed and or employed at the crushing operations or plant to act as screen and to reflect or reduce the noise level.
- (11) Noise levels must be recorded on a continuous basis at identified receptor sites which will include the residences of immediate neighbours and the borders of the mining area.

Blasting, vibration and shock

- 53. (1) In accordance with the provisions of the Mine Health and Safety Act, 1996, (Act No. 29 of 1996) blasting operations at prospecting or mining operations must be so designed and executed that the minimum shock, noise and vibration are generated.
 - (2) An assessment of impacts relating to blasting, vibration and shock control and management, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan, as the case may be.

Water management and pollution control

- 54. (1) The Regulations promulgated in the Government Notice No. 704 of 4 June 1999, in terms of the National Water Act, 1998, (Act No. 36 of 1998) shall apply to the water management and pollution control at all proposed or existing prospecting or mining operations.
 - (2) An assessment of impacts relating to water management and pollution control at proposed prospecting or mining operations, where appropriate, must form part of the environmental impact assessment report and draft environmental management programme or the environmental management plan, as the case may be.

Disposal of waste material

- 55. (1) The holder of any right or permit in terms of the Act, must comply with national norms and standards regarding the disposal of waste material, where applicable.
 - (2) The assessment of impacts relating to the disposal of waste material, where appropriate, must form part of the environmental impact assessment report and draft environmental management programme or the environmental management plan.
 - (3) Waste material from reduction works, beneficiation plants, coal preparation plants, screening and washing installations and generating stations at a mine shall be disposed of in accordance with the approved environmental management programme or environmental management plan and on an approved demarcated site.
 - (4) No dumping or impounding of rubble, litter, garbage, rubbish or discards of any description, whether solid or liquid, shall take place elsewhere than at the site or sites

- demarcated for the purpose in accordance with the approved environmental management programme or environmental management plan.
- (5) No sand dump or slimes dam shall be established on the bank of any stream, river, dam, pan or lake without written permission of the Minister and upon such conditions as he/she may prescribe or if and as approved in the environmental management programme or environmental management plan.
- (6) Coal debris shall not be allowed to accumulate on any ground where there exist, or where there are likely to occur, surface fissures or cavities as a result of underground operations.

Soil pollution and erosion control

- 56. (1) The holder of a right or permit in terms of the Act, must comply with national norms and standards regarding the management and control disposal of waste material, where applicable.
 - (2) The assessment of impacts relating to soil pollution and erosion control, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan.
 - (3) Except for the areas concerned in the approved areas for the depositioning of waste, all adverse chemical contamination of soils must be minimised to accepted levels.
 - (4) The spillage of hazardous chemicals onto soils or its escape or migration into surrounding soils from the approved depositioning area, must be prevented as far as possible.
 - (5) Oils, grease and hydraulic fluids must be disposed of in a waste disposal recepticle for collection or treatment or at a licensed facility or in accordance with a method approved in the environmental management programme or environmental management plan.
 - (6) Oils, grease and hydraulic fluid spills which occur must be cleaned up by removing all contaminated soil and disposing thereof in a waste disposal receptacle or at a licensed facility.
 - (7) The acidification, salination and mineralisation of soils through seepage of polluted water and no irrigation of land with any water must take place with the permission of the Minister of Water Affairs and Forestry or as approved in the environmental management programme or environmental management plan.
 - (8) The chemical and physical properties of top soil to be used for the purposes of rehabilitation must not be changed by introducing foreign material, gravel, rock, rubble or mine residue.

Sanitation of surface

- Toilet facilities shall be located in such a manner that no water or other pollution is caused.
 - (2) No person shall pollute the workings with faeces or urine or misuse the facilities provided or inappropriately foul the surrounding environment with faeces or urine. Acceptable hygienic and aesthetic practices must be adhered to.

Management of residue stockpiles and deposits

- 58. (1) The assessment of impacts relating to the management of residue stockpiles and deposits, where appropriate, must form part of the environmental impact assessment report and draft environmental management programme or the environmental management plan.
 - (2) Residue characterisation
 - (a) Mine residue must be characterised to identify any potentially significant health or safety hazard and environmental impact that may be associated

with the residue when stockpiled or deposited at the site(s) under consideration.

- (b) Residue stockpiles and/or deposits must be characterised in terms of its
 - physical characteristics, which may include -
 - (aa) the size distribution of the principal constituents;
 - (bb) the permeability of the compacted material;
 - (cc) void ratios of the compacted material:
 - (dd) the consolidation or settling characteristics of the material under its own weight and that of any overburden;
 - (ee) the strength of compacted material;
 - (ff) the specific gravity of the solid constituents; and
 - (gg) the water content of the material at the time of deposition, after compaction, and at other phases in the life of the deposit.
 - (ii) chemical characteristics, which may include -
 - (aa) the toxicity:
 - (bb) the propensity to oxidize and /or decompose:
 - (cc) the propensity to undergo spontaneous combustion;
 - (dd) the pH and chemical composition of the water separated from the solids:
 - (ee) stability and reactivity and the rate thereof; and
 - (ff) neutralising potential.
 - (iii) mineral content, which include the specific gravity of the residue particles and its impact on particle segregation and consolidation;
- (3) Classification of residue stockpiles and deposits
 - All residue stockpiles and deposits must be classified into one or a combination of the following categories –
 - the safety classification to differentiate between residue stockpiles and deposits of high, medium and low hazard on the basis of their potential to cause harm to life or property; and
 - the environmental classification to differentiate between residue stockpiles and deposits with -
 - (aa) a potentially significant impact on the environment due to its spatial extent, duration and intensity of potential impacts; or
 - (bb) no potentially significant impact on the environment.
 - (b) All mine residue stockpiles and deposits must be classified by a suitably qualified person(s).
 - (c) The classification of residue stockpiles and deposits shall determine the -
 - (i) level of investigation and assessment required;
 - requirements for design, construction, operation, decommissioning, closure and post closure maintenance; and
 - (iii) qualifications and expertise required of persons undertaking the investigations, assessments, design, construction thereof.
 - (d) The safety classification of residue stockpiles and deposits shall be based on the following criteria —

	Number of workers in zone of influence		Depth to underground mine workings	Classification
0	< 10	0 – R2 m	> 200m	Low hazard
1 – 10	11 – 100	R 2 m - R20 m	50 m - 200 m	Medium hazard
> 10	> 100	> R20 m	< 50 m	High hazard

- (e) A risk analysis must be carried out and documented on all high hazard residue stockpiles and deposits.
- (f) The environmental classification of residue stockpiles and deposits must be undertaken on the basis of –
 - (i) the characteristics of the residue;
 - (ii) the location and dimensions of the deposit (height, surface area);

- (iii) the importance and vulnerability of the environmental components that are at risk; and
- (iv) the spatial extent, duration and intensity of potential impacts.
- (g) The assessment of impacts and analyses of risks shall form part of the environmental impact assessment and environmental management programme or plan, as the case may be.
- (4) Site selection and investigation:
 - (a) The process of investigation and selection of a site must entail -
 - the identification of a sufficient number of possible candidate sites to ensure adequate consideration of alternative sites;
 - (ii) qualitative evaluation and ranking of all alternative sites;
 - (iii) qualitative investigation of the top ranking sites to review the ranking done in (ii);
 - (iv) a feasibility study to be carried out on the highest ranking site(s), involving -
 - (aa) a prelimenary safety classification;
 - (bb) an environmental classification;
 - (cc) geotechnical investigations; and
 - (dd) groundwater investigations.
 - (b) The geotechnical investigations may include-
 - the characterization of the soil profile over the entire area to be covered by the residue facility and associated infrastructure to define the spatial extent and depth of the different soil horizons;
 - the characterization of the relevant engineering properties of foundations soils and the assessment of strength and drainage characteristics.
 - (c) The groundwater investigations may include-
 - (i) the potential rate of seepage from the residue facility;
 - (ii) the quality of such seepage;
 - (iii) the geohydrological properties of the strata within the zone that could potentially be affected by the quality of seepage;
 - (iv) the vulnerability and existing potential use of the groundwater resource within the zone that could potentially be affected by the residue facility.
 - (d) From these investigations, a preferred site must be identified.
 - (e) Further investigation on the preferred site, shall include -
 - (i) land use;
 - (ii) topography and surface drainage;
 - (iii) infrastructure and man-made features;
 - (iv) climate;
 - (v) flora and fauna;
 - (vi) soils;
 - (vii) ground water morphology, flow, quality and usage; and
 - (viii) surface water.
 - (f) The investigations, laboratory test work, interpretation of data and recommendations for the identification and selection of the most appropriate and suitable site for the disposal of all residue that have the potential to generate leachate that could have a significant impact on the environment and groundwater must be carried out by a suitably qualified person.
- (5) Design of residue stockpile and deposit
 - (a) The design of the residue stockpile and deposit shall be undertaken by a suitably qualified person.
 - (b) An assessment of the typical soil profile on the site is required for residue stockpiles and deposits which -
 - (i) have a low hazard potential; and
 - (ii) have no significant impact on the environment.

- (c) The design of the residue stockpile and deposit must take into account all phases of the life cycle of the stockpile and deposit, from construction through to closure and must include —
 - (i) the characteristics of the mine residue;
 - (ii) the characteristics of the site and the receiving environment;
 - (iii) the general layout of the stockpile or deposit, whether it is a natural valley, ring dyke, impoundment or a combination thereof and its 3dimensional geometry at appropriate intervals throughout the planned incremental growth of the stockpile or deposit;
 - (iv) the type of deposition method used; and
 - (v) the rate of rise of the stockpile or deposit.
- (d) Other design considerations, as appropriate to the particular type of stockpile and deposit must be incorporated —
 - (i) the control of storm water on and around the residue stockpile or deposit by making provision for the maximum precipitation to be expected over a period of 24 hours with a frequency of once in a 100 years, in accordance with the regulations made under section 8 of the National Water Act, 1998;
 - the provision, throughout the system, of a freeboard of at least 0.5 m above the expected maximum water level, in accordance with regulations made under the National Water Act, 1998, to prevent overtopping;
 - (iii) keeping the pool away from the walls; where there are valid technical reasons for deviating from this, adequate motivation must be provided and the design must be reviewed by a qualified person as required in terms of sections 9(6) or 9(7) of the Mine Health and Safety Act, 1996;
 - (iv) the control of decanting of excess water under normal and storm conditions;
 - (aa) the retension of polluted water in terms of polluted water in terms of GN R991(9), where measures may be required to prevent water from the residue deposit from leaving the residue management system unless it meets prescribed requirements;
 - (bb) the design of the penstock, outfall pipe, under-drainage system and return water dams;
 - (cc) the height of the phreatic surface, slope angles and method of construction of the outer walls and their effects on shear stability;
 - (dd) the erosion of slopes by wind and water, and its control by
 - (ee) vegetation, berms or catchment paddocks; and
 - (ee) the potential for pollution.
- (e) A design report and operating manual shall be drawn up for all residue stockpiles and deposits which -
 - (i) have a medium to high hazard; and
 - (ii) have a potentially significant impact on the environment.
- (f) Relevant information must be included in the draft environmental management programme or environmental management plan.
- (6) Construction and operation of residue deposits:
 - (a) The holder of any right or permit in terms of the Act, must ensure that-
 - the residue deposits, including any surrounding catchment paddocks, is constructed and operated in accordance with the approved environmental management programme or environmental management plan;
 - (ii) the design of the residue deposit is followed implicitly throughout the construction thereof, and that any deviations from the design be approved by the Regional Manager and the environmental manage programme and environmental management plan be amended accordingly;
 - (iii) as part of the monitoring system, measurements of all residues transported to the site and of all surplus water removed from the site are recorded:

- (iv) the provision for appropriate security measures be implemented to limit unauthorised access to the site and intrusion into the residue deposit;
- (v) specific action be taken in respect of any sign of pollution;
- (vi) adequate measures be implemented to control dust pollution and erosion of the slopes; and
- (vii) details of rehabilitation of the residue deposit be provided in the draft environmental management programme or environmental management plan.
- (b) A system of routine maintenance and repair in respect of the residue deposit must be implemented to ensure the ongoing control of pollution, the integrity of rehabilitation and health and safety maters at the site.
- (7) Monitoring of residue stockpiles and deposits:
 - (a) A monitoring system for residue stockpiles and deposits with respect to potentially significant impacts as identified in the environmental assessment must be included in the environmental management programme or environmental management plan.
 - (b) In the design of a monitoring system for a residue stockpile or deposit, consideration must be given to –
 - baseline and background conditions with regard to air, surface and groundwater quality;
 - (ii) the air, surface and groundwater quality objectives;
 - (iii) residue characteristics;
 - (iv) the degree and nature of residue containment;
 - the receiving environment and secifically the climatic, local geological, hydrogeological and geochemical conditions;
 - (vi) potential migration pathways;
 - (vii) potential impacts of leachate;
 - (viii) the location of monitoring points and the prescribed monitoring protocols; and
 - (ix) the reporting frequency and procedures.
- (8) Decommissioning, closure and after care:
 - (a) The decommissioning, closure and post closure management of residue deposits must be addressed in the closure plan, which must contain the following -
 - the environmental classification, including assumptions on which the classification were based;
 - (ii) the closure objectives, final land use or capability;
 - (iii) conceptual description and details for closure and post closure management;
 - (iv) cost estimates and financial provision for closure and post-closure management; and
 - residual impacts, monitoring and requirements to obtain mine closure in terms of the Act.

CHAPTER 3

PETROLEUM EXPLORATION AND PRODUCTION

Application for a reconnaissance permit

- 59. (1) An application for a reconnaissance permit in terms of section 74 of the Act must be completed on Form K in Annexure 1, and lodged at the office of the Petroleum Agency and must contain-
 - (a) full particulars of the applicant;
 - (b) a plan as described in Regulation 2(3), indicating the co-ordinates and boundaries of the area applied for, referenced to the WGS84, Hartebeesthoek 94 datum;
 - (c) a clear statement of the technical motivation;

- the period for which the permit is required which in terms of section 75(4)(b) of the Act may not exceed one year;
- (e) a detailed work programme;
- (f) detailed documentary proof that the applicant has access to financial resources and has the technical ability to conduct the proposed reconnaissance operations optimally in accordance with the reconnaissance work programme;
- (g) a list of existing rights and permits held by the applicant to be compiled in a table format which indicate the location and the number of the existing rights or permits;
- (h) acknowledgement that the Petroleum Agency is entitled to require further information before determining whether or not to issue the reconnaissance permit applied for; and
- any other information, data or documentation that the Petroleum Agency may require in connection with information required under paragraphs (a) to (g).
- (2) After receiving the information required, the Petroleum Agency must submit the application forthwith to the Minister for consideration.

General terms and conditions for a reconnaissance permit

- 60. (1) The terms and conditions of the reconnaissance permit to be concluded between the applicant and the Minister in terms of section 75(1) of the Act, must be substantially in the form of Form JJ in Annexure 1, or as the Minister may determine from time to time, and must contain particulars regarding -
 - (a) parties to the agreement;
 - (b) description of the permit area;
 - (c) duration of the reconnaissance permit;
 - (d) fees payable:
 - (e) reports and samples to be furnished
 - (f) work programme;
 - (g) indemnity and insurance:
 - (h) waiver of lenience;
 - (i) arbitration;
 - (j) environmental protection;
 - (k) costs:
 - (I) severability;
 - (m) domicilium citandi et executandi of the parties;
 - (n) notices and representatives; and
 - (o) any additional terms and conditions that the Minister may determine from time to time.

Reconnaissance work programme

- The reconnaissance work programme shall contain the following-
 - the period required, subject in terms of section 75(4)(b) of the Act to a maximum of one year;
 - (b) an outline of the geological, geochemical, geophysical and other work to be performed;
 - technical data detailing the reconnaissance method or methods to be implemented for the proposed reconnaissance programme;
 - (d) an estimate of the expenditure to be incurred, which must include costs pertaining to the rehabilitation and management of environmental impacts;
 - (e) the programme for the marketing and sale of any data and conditions relating thereto which shall be limited to a maximum period of 10 years.

Progress reports in respect of reconnaissance activities

62. The holder of a reconnaissance permit must submit timeous, accurate summary progress reports to the Petroleum Agency on a monthly basis.

Supply of data in respect of reconnaissance

63. The holder of a reconnaissance permit must supply to the Petroleum Agency, in a format and medium as agreed by the Petroleum Agency, digital and, where appropriate, paper copies of all data, reports and interpretations generated, as soon as possible after completion of operations or projects.

Application for a technical co-operation permit

64. An application for a technical co-operation permit in terms of section 76 of the Act must be completed on Form L as set out in Annexure 1 of the Regulations and must contain-

(a) full particulars of the applicant;

(b) a plan indicating the co-ordinates and boundaries of the area applied for, referenced to the WGS84, Hartebeesthoek 94 datum;

(c) a clear statement of the technical motivation;

(d) the period for which the permit is required which in terms of section 77(4)(b) of the Act may not exceed one year;

(e) a detailed work programme;

(f) documentary evidence to prove that the applicant has access to financial resources and has the technical ability to conduct the proposed work programme;

(g) acknowledgement that the Petroleum Agency is entitled to require further information before determining whether or not to issue the permit applied for; and

(h) any other information, data or documentation that the Petroleum Agency may require in connection with information required under paragraphs (a) to (f).

General terms and conditions for a technical co-operation permit

65. The terms and conditions of the technical co-operation permit to be concluded between the applicant and the Petroleum Agency in terms of section 77(1) of the Act, may be as set out in Form KK, Annexure 1 of the Regulations or as the Minister may determine from time to time and must contain the following information-

(a) parties to the agreement;

- (b) description of the permit area;
- (c) duration of the technical co-operation permit;

(d) fees payable;

(e) reports and samples to be furnished

(f) work programme;

- (g) indemnity and insurance;
- (h) waiver of lenience;
- (i) arbitration;
- (j) cost;
- (k) option to acquire an exploration right;

(I) severability;

(m) domicilia citandi et executandi;

(n) a signature clause and clause for witnesses to sign; and

(o) any additional terms and conditions that the Minister may determine from time to time.

Technical co-operation work programme

66. The technical co-operation work programme shall contain the following-

 the period required, subject in terms of section 77(4)(b) of the Act to a maximum of one year;

(b) an outline of the geological, geochemical, geophysical and other work to be performed;

(c) technical data detailing the research and analysis methods to be implemented for the proposed work programme; and

(d) an estimate of the expenditure to be incurred.

Progress reports in respect of technical co-operation activities

67. The holder of a technical co-operation permit must supply to the Petroleum Agency, in a format and medium agreed by the Petroleum Agency, digital and, where appropriate, paper copies of all data generated, as soon as possible after completion of operations or projects.

Application for an exploration right

68. An application for an exploration right in terms of section 79 of the Act must be completed on Form M as set out in Annexure 1 of the Regulations or as the Minister may determine from time to time and must contain-

(a) full particulars of the applicant;

(b) a plan indicating the co-ordinates and boundaries of the area applied for, referenced to the WGS84, Hartebeesthoek 94 datum;

(c) a clear statement of the technical motivation;

- (d) the period for which the right is initially required in terms of section 80(5) of the Act, as well as any possible renewals;
- (e) a detailed exploration work programme;
 -) documentary evidence to prove that:
 - the applicant has access to financial resources and has the technical ability to conduct the proposed exploration operation optimally in accordance with the exploration work programme, such to be supported by a report from an independent assessor approved by the Petroleum Agency;
 - the applicant has the ability to comply with the provisions of the Mine Health and Safety Act, 1996 (Act No. 29 of 1996);
 - (iii) the granting of the right will further the objects referred in sections 2(d) and (f) of the Act:
- (g) acknowledgement that the Petroleum Agency and the Minister are entitled to require further information before determining whether or not the right applied for should be granted; and
- (h) any other information, data or documentation that the Petroleum Agency or the Minister may require in connection with information required under paragraphs (a) to (f).

General terms and conditions for an exploration right

- 69. (1) The terms and conditions of the exploration right to be concluded between the applicant and the Minister in terms of section 80(5) of the Act, must be as set out in Form LL, Annexure 1 of the Regulations or as the Minister may determine from time to time and must contain the following information-
 - (a) parties to the agreement;
 - (b) area applied for;
 - (c) duration of the exploration right;
 - (d) fees payable;
 - (e) work programme;
 - (f) indemnity and insurance;
 - (g) waiver of lenience;
 - (h) arbitration;
 - (i) environmental protection;
 - (j) cost;
 - (k) option to acquire a production right;
 - (I) severability;
 - (m) domicilia citandi et executandi;
 - (n) a signature clause and clause for witnesses to sign;
 - (o) any additional terms and conditions that the Minister may determine from time to time.
 - (2) In addition to any term and condition contained in the exploration right, the following general terms and conditions also apply to such right where the holder of an exploration right shall:
 - exercise any right granted in terms of the provisions of this Act reasonably and in such manner that the rights and interests of the owner of any land to which exploration right relates are not adversely affected, except to the extent to which such owner is compensated;
 - in the employment of employees, adopt and apply the principles outlined in the South African national affirmative action programmes, where possible;
 - (c) where possible, engage in joint-ventures or any other form of co-operation with Black Economic Empowerment entities to develop technical skills and infrastructure;
 - investigate, communicate and assess the potential environmental impacts of the proposed exploration operation in accordance with the provisions of the Act;
 - (e) prevent or manage to minimise or mitigate the environmental impacts in accordance with the approved environmental management programme;
 - (f) monitor and assess the continued appropriateness, adequacy and environmental performance of the approved environmental management programme as determined:
 - (g) make appropriate and adequate financial provision for rehabilitation and remediation of environmental damage and revise such provision as required in terms of the Act;
 - (h) where the holder of an exploration right is a corporate body, partnership or association of persons, give proper notice in writing to the Minister within thirty (30) days of any change relating to the following-
 - (i) name of right holder;

- (ii) registered address and principal place of business;
- (iii) directors of the company, or members of a close corporation;

(iv) share capital of the company;

- (v) beneficial owner or owners of the shares issued by such company in contravention of the provisions of the Competition Act, 1998 (Act No. 89 of 1998); and
- in the case of a natural person, provide written notice to the Minister of any change of the address of such person within thirty (30) days.

Exploration work programme

- 70. (1) The exploration work programme shall contain the following -
 - (a) the period for which the right is required, as well as any possible renewals;
 - (b) an outline of the geological, geochemical, geophysical, exploration drilling and other work to be performed;
 - (c) technical data detailing the exploration method or methods to be implemented and the time required for each stage of the proposed exploration operation;
 - (d) an estimate of the expenditure to be incurred for each stage of the exploration operation where the expenditure must be broken down into:

(i) exploration costs; and

- (ii) costs pertaining to the rehabilitation and management of environmental impacts.
- (2) In the case where an old order prospecting right must be converted into a new exploration right, the following information must, in addition to that required in item 4 of schedule II to the Act and to that required and as applied to the Regulations in terms of Chapter 3, also be included-
 - (a) a brief summary of the geology of the exploration target(s) with a compilation of the data and conclusions generated under the old order prospecting right;
 - a description of the contractual work commitment and exploration work which has been carried out and conclusions reached, which must include:

(i) past geological results as are outlined for the progress report;

- details of past expenditure broken down into exploration costs, manpower costs and cost pertaining to the rehabilitation and management of environmental impacts;
- data, previously compiled but not yet provided to the Petroleum Agency; and
- (iv) a statement reflecting rehabilitation work completed and the rehabilitation work uncompleted.

Progress reports in respect of exploration

- 71. (1) The holder of an exploration right, must submit timeous accurate summary progress reports to the Petroleum Agency on a monthly, quarterly and annual basis.
 - (2) Monthly progress reports must be submitted within 7 days of month end.
 - (3) Quarterly progress reports must be submitted within 21 days of quarter end and must include-
 - (i) the numbers of local and expatriate persons employed;

(ii) work done and money expended on operations;

- (iii) the site and depth of every well drilled or being drilled and the formations penetrated and particulars regarding any occurrence of petroleum and / or any other mineral of potential value encountered; and
- (iv) a statement reflecting rehabilitation work completed and the rehabilitation work uncompleted.
- (4) Annual progress reports must be submitted within 60 days of calendar year end and must include:
 - (i) a full report of the exploration operations carried out during the year;
 - (ii) a detailed statement of exploration expenditure incurred during the year;

(iii) a description of operations planned for the following year;

- (iv) a budget for exploration operations planned for the following year; and
- a statement reflecting rehabilitation work completed and rehabilitation work uncompleted.

- (5) Information in the progress reports will be kept confidential for the period contemplated in section 88(2) of the Act.
- (6) The Minister or the Petroleum Agency may reject any progress report or exploration work programme which does not comply with the requirements of the Act. Such reports or programmes must be re-submitted to the Minister within one (1) month.

Supply of data in respect of exploration

72. The holder of an exploration right must supply to the Petroleum Agency in a format and medium agreed by the Petroleum Agency, samples, digital and where appropriate, paper copies of all data generated, as soon as possible.

Application for the renewal of an exploration right

73. (1) An application for the renewal of an exploration right in terms of section 81 of the Act must be completed on Form M as set out in Annexure 1 of the Regulations and must contain:

(a) full particulars of the applicant;

- (b) a plan indicating the co-ordinates and boundaries of the renewal area applied for;
- the period for which the renewal is required, together with any possible further renewal periods;

(d) reasons why a renewal is required;

- a detailed report reflecting the exploration results, the interpretation thereof and the exploration expenditure incurred;
- (f) a report reflecting the extent of compliance with the requirements of the approved environmental management programme, the rehabilitation to be completed and the estimated cost thereof:

(g) a detailed exploration work programme for the renewal period; and

 (h) any other information or documentation that the Petroleum Agency or the Minister may require in connection with information required under paragraph (a) to (g).

Application for a production right

74. (1) An application for a production right in terms of section 83 of the Act must be completed on Form N as set out in Annexure 1 of the Regulations and must contain-

(a) full particulars of the applicant;

- a plan indicating the co-ordinates and boundaries of the area applied for referenced to the WGS84, Hartebeesthoek 94 datum;
- (c) a clear statement of the technical motivation;

(d) the period for which the right is required;

(e) a development and production work programme;

(f) a social and labour plan;

(g) documentary evidence to prove that:

- the applicant has access to financial resources and has the technical ability to conduct the proposed production operation optimally, such to be supported by a report from an independent assessor(s) approved by the Petroleum Agency;
- the applicant has the ability to comply with the relevant provisions of the Mine Health and Safety Act, 1996 (Act No. 29 of 1996);
- (iii) the granting of such right will further the object referred to in section 2(f) of the Act;
- details of the petroleum types and the quantity thereof, which the applicant intends to remove and dispose of during production operations;
- acknowledgement that the Petroleum Agency or the Minister is entitled to require further information before deciding whether or not the right applied for should be granted; and
- any other information, data or documentation that the Petroleum Agency or the Minister may require in connection with information required under paragraphs (a) to (h).

General terms and conditions for a production right

- 75. (1) The terms and conditions of the production right to be concluded between the applicant and the Minister in terms of section 84(1) of the Act must be as set out in Form MM, Annexure 1 of the Regulations or as the Minister may determine from time to time and must contain the following information-
 - (a) parties to the agreement;

(b) production area applied for;

(c) the petroleum type to be produced;

- (d) period for which the production right is required;
- (e) considerations, royalties or minimum royalties payable;
- (f) restrictions and obligations applicable on the holder;

(g) indemnity and insurance;

- (h) protection of boreholes and associated facilities;
- (i) protection of property and persons;
- (j) records and returns;
- (k) waiver of lenience;
- arbitration;
- (m) environmental protection;
- (n) cost;
- (o) severability;
- (p) domicilia citandi et executandi; and
- (q) a signature clause and clause for witnesses to sign;
- (2) In addition to any term and condition contained in the production right, the following general terms and conditions also apply to such right. The holder of a production right shall-
 - exercise any right granted in terms of the provisions of this Act reasonably and in such manner that the rights and interests of the owner of any land to which the production right relates are not adversely affected, except to the extent to which such owner is compensated;
 - in the employment of employees, adopt and apply the principles outlined in the South African national affirmative action programme;
 - carry out training programmes in order to encourage and promote the development of local communities where the production operation is carried out;
 - (d) where possible, engage in joint-ventures or any other form of co-operation with Black Economic Empowerment entities to develop technical skills and infrastructure;
 - investigate, communicate and assess the potential environmental impacts of the proposed production operation in accordance with the provisions of the Act;
 - (f) prevent, minimise, mitigate and manage the environmental impacts in accordance with the approved environmental management programme;
 - (g) monitor and assess the continued appropriateness, adequacy and environmental performance of the approved environmental management programme as determined;
 - (h) make appropriate and adequate financial provision for rehabilitation and remediation of environmental damage and revise such provision as required in terms of the Act;
 - where the holder of a production right is a corporate body, partnership or association of persons, give proper notice in writing to the Minister within 30 days of any change relating to the following:
 - (i) name of right holder;
 - (ii) registered address and principal place of business;
 - (iii) the directors of the company, or member of a closed corporation;
 - (iv) share capital of the company; and
 - beneficial owner or owners of the shares issued by such company in contravention of the provisions of the Competition Act, 1998 (Act No. 89 of 1998);
 - as far as possible make use of local contractors, services, equipment purchased or hired; and
 - (k) in the case of a natural person, provide written notice to the Minister of any change of the address of such person within 30 days.
- (3) The State shall have the option, exercisable within 60 (SIXTY) days of the date of acceptance of an application for a Production Right, by written notice to the applicant, to elect to retain and transfer to a wholly owned Affiliate of the State up to a maximum of 1/10th (ONE TENTH) undivided interest in the Production Right.

Production work programme

- 76. (1) The production work programme shall contain the following information-
 - type of petroleum to be produced and the period required for production up to a maximum of thirty years (30) for the initial period;

- (b) a comprehensive feasibility study including details of the measured petroleum resource and a plan of the area under application;
- (c) technical data detailing the production method to be used;
- (d) details with respect to the envisaged production rate, processing, and marketing arrangements;
- (e) a detailed forecast of capital requirements, sales revenues, operating costs and other costs presented in a cash flow format for the period required for production;
- (f) an assessment of the reserves carried out by an independent expert(s) approved by the Petroleum Agency or the Minister; and
- (g) an assessment of the development plan carried out by an independent expert(s) approved by the Petroleum Agency or Minister;
- (2) In the case where an old order mining right or OP26 right must be converted into a new production right, the following information must, in addition to that required in item 5 or item 7, as applicable, of Schedule II to the Act and to that required in terms of Regulations as applied to Chapter 3, be included-
 - a brief history of the production area and operations as well as an explanation of activities undertaken;
 - a schedule of reserves and resources classified in terms of the classification rules of the Society of Petroleum Engineers as from the date of conversion;
 - (c) annualised figures for past expenditure broken down into development costs, production costs and costs pertaining to the rehabilitation and management of environmental impacts;
 - (d) copies of production reports reflecting relevant data;
 - (e) data previously compiled but not yet provided to the Petroleum Agency; and
 - a statement reflecting rehabilitation work completed and the rehabilitation work uncompleted.

Application for the renewal of a production right

- 77. (1) An application for the renewal of a production right in terms of section 85(1) of the Act must be completed on Form N as set out in Annexure 1 of the Regulations and must contain-
 - (a) full particulars of the applicant;
 - a plan indicating the co-ordinates and boundaries of the renewal area applied for referenced to the WGS84, Hartebeesthoek 94 datum;
 - (c) the type of petroleum for which the renewal of the production right is required;
 - (d) the period for which the renewal is required;
 - (e) reasons why a renewal is required;
 - a detailed report reflecting the production results, the interpretation thereof and the production expenditure incurred;
 - (g) a report reflecting the extent of compliance with the requirements of the approved environmental management programme, the rehabilitation to be completed and the estimated cost thereof;
 - (h) a detailed production work programme for the renewal period;
 - (i) a revised social plan for the renewal period; and
 - any other information or documentation the Petroleum Agency or the Minister may require in connection with information required under paragraphs (a) to (i).

Information, data and statistical returns in respect of production of petroleum

- 78. (1) The holder of a production right must submit statistical returns to the Petroleum Agency monthly, regarding information and data in respect of production of petroleum which include the following-
 - (a) description of all work, improvements and rehabilitation undertaken;
 - (b) accurate returns of all production; and
 - accurate monthly returns of the sales and exports of petroleum transacted by the holder of the production right; and
 - (d) a rolling 12-month projection of petroleum production.
 - (2) The returns must reach the Petroleum Agency not later than 21 days after month end and must be written in such form as may be required by the Petroleum Agency or the Minister.

- (3) The holder of a production right shall furnish the Petroleum Agency annually and in such form as may be required in terms of the production right be prescribed by him or her with returns regarding matters pertaining to employees, salaries and wages.
- (4) Such holder shall also provide the Minister or the Petroleum Agency at its request with such other returns and data as may be reasonably required. Each such return and other data shall be furnished on or before a date to be determined by the Petroleum Agency and specified on the prescribed forms or otherwise notified in writing.

PART II: SOCIAL AND LABOUR PLAN

79. Submission of a Social and Labour Plan and its applicability

- (1) Any application for a production right in terms of this Act shall be accompanied by a social and labour plan. The Minister may refer the said social and labour plan back to the applicant with proposals and the revised social and labour plan must then be lodged.
- (2) The social and labour plan shall apply -

(a) to all applications for a production right;

(b) to all applications for the conversion of an old order right in terms of the Minerals Act, 1991, (Act No. 50 of 1991) to a production right;

(c) throughout the life of such operation referred to in (a) or (b);

(d) in the event of possible large-scale retrenchments at operations referred to in (a) and
 (b) and when such retrenchments may impact on the sector or regional or national interest.

Objectives of Social and Labour Plan

80. (1) The objects of a social and labour plan are to-

- (a) consider the development of the production operation in the context of generally recognized standards of sustainable development by integrating social, economic and environmental factors in planning the operations throughout the life of the operation;
- endeavour to avoid job losses and a decline in employment by promoting alternative solutions to threatening job security and potential measures to prevent a decline in employment;
- (c) endeavour to ameliorate the social and economic impact on individuals, regions and the economy in cases where job losses remain unavoidable after exploring alternative solutions and measures;
- endeavour to avoid the unnecessary establishment of settlements which cannot be sustained after the closure of the operation;
- (e) put a social and labour plan strategy in place;

(f) put a human resource development strategy in place;

(g) include monitoring, contingency and reporting mechanisms; and

(h) endeavour to reach its conclusion once all the obligations in terms of the social and labour plan have been met.

Contents of Social and Labour Plan

- 81. (1) The social and labour plan proposal must be submitted in the following format, but is not limited thereto-
 - (a) a preamble, describing the background to the production operation, in terms of its location, commodity to be mined, scale of operations and size of workforce;
 - (b) a description of the local economic background of the area in which the operation will be or is situated, the areas from which labour will mainly be sourced, the key economic activities of the area, and the socio-economic impact of the operation in the area.
 - (c) a summary of the objectives of the social and labour plan and how it will be achieved;
 - (d) the social and labour plan strategy of the production operation;

(e) the human resource development strategy of the production operation;

the implementation process, which must include the establishment of a forum, and the action plans, timeframes and other relevant information during the construction, operational, decommissioning, closure and post-closure phase of the production operation with respect to the social and labour plan objectives;

- (g) monitoring, contingency mechanisms and reporting mechanisms:
- (h) any additional information which may be requested by the Minister; and
- an undertaking by the applicant to adhere to the requirements and conditions set out in the social and labour plan.

Social and Labour Plan forum

- 82. (1) A social and labour plan future forum comprising the holder of a production right, the employees and the relevant local authority, each of which shall be represented by a maximum of five (5) persons, will be established to ensure that the objectives of the social and labour plan is achieved.
 - (2) The purpose of the forum is-
 - (a) to implement the social and labour plan; and
 - (b) to monitor the compliance and the appropriateness of the social and labour plan.

PART III: ENVIRONMENTAL REGULATION

Environmental reports to be compiled for the application of a production right

- 83. (1) An applicant who has applied for a production right in terms of section 83 of the Act must conduct an environmental impact assessment and submit a draft environmental management programme to the Petroleum Agency within 180 days of the date of notification by the Agency.
 - (2) An environmental impact assessment as contemplated in section 39(1) of the Act, includes the compilation of-
 - (d) a scoping report; and
 - (e) an environmental impact assessment report.

Scoping Report

- A scoping report must identify the impacts and reasonable alternatives to be addressed in the environmental impact assessment report. The result of the scoping process must determine the nature and extent of investigations to be undertaken in the environmental impact assessment report.
 - (2) A scoping report must
 - describe the methodology, including the stakeholder engagement process, of the scoping process;
 - (b) describe the existing status of the environment prior to mining;
 - identify and describe the anticipated environmental, social and cultural impacts, including the cumulative effects of the proposed operation;
 - identify and describe reasonable alternatives to the project and alternative means of carrying out the project;
 - (e) propose the most appropriate approach and procedure to plan and develop the proposed mining project;
 - describe the views and concerns of relevant authorities and interested and affected persons and the participation process followed; and
 - (g) describe the nature and extent of further investigations required in the environmental impact assessment report.
 - (4) If the scoping report indicates that the proposed mining operation will result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must refuse the application.
 - (4) If the scoping report indicates that the proposed mining operation will not result in unacceptable pollution, ecological degradation or damage to the environment, the Petroleum Agency must request the applicant to submit -
 - an environmental impact assessment report based on the outcome of the scoping report; and
 - a draft environmental management programme based on the environmental impact assessment report.

Environmental Impact Assessment Report

- 85. (1) An environmental impact assessment report must determine the nature, extent, duration, probability and significance of the environmental, social and cultural impacts of the proposed mining operation, the reasonable alternatives and the potential mitigation measures for each impact during each of the life of the proposed operation.
 - (2) An environmental impact assessment report must as a minimum ensure the following-

 (a) investigation of the environment likely to be significantly affected by the proposed production operation and alternatives thereto;

- investigation of the potential impact, including cumulative impacts, of the proposed operation and its alternatives on the environment, socio-economic conditions and cultural heritage;
- (c) an assessment of the nature, extent, duration, probability and significance of the potential environmental, social and cultural impacts of the proposed operation, including the cumulative impacts;

 (d) a comparative assessment of the feasible alternatives and their potential environmental, social and cultural impacts;

- investigate and describe mitigatory measures for each significant impact of the proposed activity and the preferred alternative to keep adverse impacts to a minimum;
- (f) description of the stakeholder engagement process followed during the course of the assessment and an indication of how issues raised have been addressed:
- reporting on knowledge gaps, the adequacy of predictive methods and underlying assumptions, and uncertainties encountered in compiling the required information; and
- (h) description of the arrangements for the monitoring and management of impacts, and the assessment of the effectiveness of such arrangements after their implementation; and
- (i) inclusion of technical and supporting information as appendices.

Investigation of cumulative impacts

86. (1) In the environmental impact assessment report, the applicant must determine whether specific resources and ecological components are cumulatively affected by the proposed mining operation by considering-

 (a) whether the proposed operation is one of several projects or activities in the same geographic area;

whether other projects or activities in the area have similar impacts on the specific resource and ecological components;

(c) whether these impacts have been historically significant for this resources and ecological components; and

 (d) whether other investigations in the area have identified a cumulative impact concern.

(2) To avoid extending data and analytical requirements beyond those relevant for decision-making, the selection of geographic boundaries and time period should be, whenever possible, be based on the natural boundaries of the resources of concern and the period of time that the proposed project's impacts will persist.

Contents of Environmental Management Programme

- 87. (1) An environmental management programme must describe the environmental objectives and specific goals for the proposed operation, the appropriate management options for the ongoing management and monitoring of the impacts of the activity, until closure.
 - (2) A draft environmental management programme must provide
 - a description of the environmental objectives and specific goals for-

(i) closure of the production operation;

(ii) the management of identified environmental impacts emanating from the proposed operation;

(iii) the socio-economic conditions; and

(iv) the historical and cultural aspects (if applicable);

(b) an outline of the implementation programme which must include -

- a description of the chosen appropriate technical and management options for each environmental impact, socio-economic condition and historical and cultural aspects of the preferred alternative under each phase of the mining operation;
- (ii) a layout of action plans to achieve the objectives and specific goals also indicating a time schedule of actions undertaken within the action plans to implement mitigatory measures and options for prevention, management and remediation;
- (iii) maintenance and emergency procedures or further remediation;

(iv) contingency measures;

 requirements for monitoring and environmental management programme performance assessment;

(vi) financial provision which must include-(aa) the determination of the quantum;

(bb) details of the method of provision and revision thereof;

(vii) an environmental awareness plan in accordance with guidelines published by the Department from time to time;

(viii) supporting information and appendices; and

- (ix) undertaking by the applicant to comply with the provisions of the Act
- (2) If the environmental impact assessment report and the draft environmental management programme indicate that the proposed production operation will result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must refuse the application.
- (3) If the environmental impact assessment report and draft environmental management programme indicate that the proposed production operation will not result in unacceptable pollution, ecological degradation or damage to the environment, the Minister must approve the application.

Independent review

- The Minister may appoint a person to undertake an independent review, if a reasonable assessment indicates that the environmental impact assessment report and the draft environmental management programme or environmental management plan, as the case may be, cannot be executed satisfactorily by the applicant. Such appointment and execution shall be for the cost of the applicant.
 - (2) The Minister must notify the applicant and provide reasons for the need for such independent review.
 - (3) The person who will undertake such an independent review must provide proof that he or she has no vested interest or involvement in the proposed prospecting or mining operation.

Environmental report to be compiled for the application of a reconnaissance permit or exploration right

An applicant who has applied for a reconnaissance permit or an exploration right in terms of the Act must submit an environmental management plan to the Petroleum Agency within 180 days of the date of notification by the Agency.

Environmental Management Plan

- 90. (1) An environmental management plan must address the potential impacts of the proposed operation, address feasible mitigatory and dedicated management and monitoring options for the proposed operation.
 - (2) The environmental management plan, shall be in the format provided by the Petroleum Agency and must contain-
 - (a) a description of the environment likely to be affected by the proposed operation:
 - (b) an assessment of the potential impacts of the proposed operation on the environment, socio-economic conditions and cultural heritage;
 - a summary of the significance of the potential impacts, the preferred alternative methods, proposed mitigation and management measures to

minimise adverse impacts and optimize benefits, as well as the option of not implementing the activity:

(d) financial provision which must include-

(i) the determination of the quantum;(ii) details of the method(s) of provision;

(e) requirements for monitoring and performance assessment;

(f) closure and environmental objectives;

- (g) a record of the public participation undertaken; and
- (h) an undertaking by the applicant regarding the implementation of the environmental management plan.

Method(s) for financial provision

- 91. (1) Financial provision for rehabilitation and remediation of environmental damage must be made by one or more of the following methods-
 - (a) approved contributions to a dedicated trust fund as provided for in section 10(1)(cH) of the Income Tax Act, 1962, (Act No. 58 of 1962) and registered with the South African Revenue Services; and / or
 - (b) a written guarantee of a bank or approved financial institution guaranteeing the amount specifically for rehabilitation and the remediation of environmental damage and the availability of funds on demand of the Department of Minerals and Energy if the proposed prospecting or mining operation should fail or become incapacitated; and/or

(c) financial deposits to be placed within a central fund with Government or an institution to be approved by the Minister; and/or

(d) any other methodology of financial provision to be approved by the Minister from time to time.

(2) The holder of a reconnaissance permit, exploration right or production right over any offshore area shall obtain and maintain for and in relation to Operations, insurances to indemnify the Agency, the State as set out in the permit and or right.

Standard forms for financial provision

- 92. The standard form for the-
 - (a) trust fund must be in the format set out in Form NN, Annexure 1;
 - (b) written guarantee must be in the format set out in Form OO, Annexure 1; and
 - (c) financial deposits must be in the format set out in Form PP, Annexure 1.

Quantum of financial provision

- 93. (1) The Minister shall request from the holder of a reconnaissance permit, an exploration or production right, as the case may be, to submit financial statements from a financial institution as proof that the holder has the financial means to execute the requirements of the environmental management programme or plan.
 - (2) The quantum of the financial provision to be approved by the Minister must be based on the requirements of the environmental management programme or plan and shall include a detailed itemization of all actual costs required for-
 - (a) the rehabilitation of the surface of the area;
 - (b) the prevention and management of pollution from the atmosphere;
 - (c) the prevention and management of pollution of water and the soil;
 - (d) decommissioning and final closure of the operation;
 - (e) post closure management of residual and latent impacts; and
 - (f) the prevention of leakage of water or petroleum between subsurface formations and/or surface; and
 - (g) the removal of any facilities and wellheads, and any other equipment on the seabed or surface of the Area as appropriate which may constitute an obstruction or environmental hazard.
 - (3) The holder of a reconnaissance permit, an exploration or production right must annually update and review the quantum of the financial provision -
 - (a) in consultation with a qualified person;
 - (b) as required in terms of the approved environmental management programme or environmental management plan; or
 - (c) as requested by the Minister.

- (4) Any inadequacies with regard to the financial provision must be rectified by the holder of a reconnaissance permit, an exploration or production right as-
 - (d) determined in an amendment of the environmental management programme or plan as the case may be;
 - (e) within the timeframe provided; or
 - (f) as determined by the Minister.

Monitoring and performance assessments of Environmental Management Programme or Plan

- 94. (1) As part of the general terms and conditions for a reconnaissance permit, an exploration or production right and in order to ensure compliance with the approved environmental management programme or plan and to assess the continued appropriateness and adequacy of the environmental management programme or plan, the holder of such permit or right must-
 - (a) conduct monitoring on a continuous basis;
 - (b) conduct performance assessments of the environmental management programme or plan as required; and
 - (c) compile and submit a performance assessment report to the Petroleum Agency to demonstrate adherence to sub-regulation (b).
 - (2) The frequency of performance assessment reporting shall be-
 - (a) in accordance with the period specified in the approved environmental management programme or plan, or, if not so specified;
 - (b) as agreed to in writing by the Minister; or
 - (c) biennially (every two years).
 - (3) The performance assessment report, shall be in the format provided in guidelines that will from time to time be published by the Department and contain-
 - (a) information regarding the period that applies to the performance assessment;
 - (c) the scope of the assessment;
 - (c) the procedure used for the assessment;
 - (d) the interpreted information gained from monitoring of the approved environmental management programme or plan;
 - (e) the evaluation criteria used during the assessment;
 - (f) the results of the assessment; and
 - (g) recommendations on how and when deficiencies that are identified and/or aspects of non-compliance will be rectified.
 - (4) The holder of a reconnaissance permit, an exploration or production right may appoint independent qualified person(s) to conduct the performance assessment and compile the performance assessment report provided that no such appointment shall relieve the holder of the responsibilities in terms of these regulations.
 - (5) After submission of the performance assessment report, the report shall be made available by the Petroleum Agency to any person on request.
 - (6) If upon consideration by the Petroleum Agency, the performance assessment executed by the holder is not satisfactory or the report submitted by the holder is found to be unacceptable, the holder must-
 - repeat the whole or relevant parts of the performance assessment and revise and resubmit the report; and/or
 - (b) submit relevant supporting information; and/or
 - (c) appoint an independent qualified person(s) to conduct the whole or part of the performance assessment and to compile the report.
 - (7) If a reasonable assessment indicates that the performance assessment cannot be executed satisfactorily by the holder or person(s) appointed by the holder, the Minister may appoint an independent performance assessment person(s). Such appointment and execution shall be for the cost of the holder.
 - (8) When the holder of a reconnaissance permit, an exploration or production right intends closing such operation, a final performance assessment shall be conducted and a report submitted to the Petroleum Agency to ensure that -
 - (a) the requirements of the relevant legislation have been complied with;
 - (b) the closure objectives as described in the environmental management programme or plan have been met; and

- (c) all residual environmental impacts resulting from the holder's operations have been identified and the risks of latent impacts which may occur have been identified, quantified and arrangements for the management thereof have been assessed.
- (9) The final performance assessment and the submission of the required report shall either precede or accompany the application for a closure certificate in terms of the Act.

General requirements for closure of reconnaissance, exploration or production operations

- 95. (1) The holder of a reconnaissance permit, an exploration or production right must ensure that –
 - (a) the closure of such operation incorporates a process which must start at the commencement of the operation and continue throughout the life of the operation and within the framework of sustainable development;
 - (b) risks to elements of the environment must be quantified and managed proactively, which includes the gathering of relevant information throughout the life of such operation:
 - the safety and health of humans and animals are safeguarded from hazards resulting from such operations;
 - environmental damage or residual impacts are minimised to such an extent that it is acceptable to all involved interested and affected persons;
 - (e) the land is rehabilitated to,as far as is practicable, its natural state, or to a predetermined and agreed standard or land use which conforms with the concept of sustainable development;
 - the optimal exploitation and utilization of mineral resources are not adversely affected; and
 - exploration and production operations are closed efficiently and cost effectively.

Application for a closure certificate

- 96. (1) An application for a closure certificate of a reconnaissance permit, exploration right and production right in terms of section 43(4) of the Act must be completed on the Form EE as set out in Annexure 1.
 - (2) An application for a closure certificate of such permit or right must be accompanied with a closure plan.

Application to transfer environmental liabilities to a qualified person

- 97. (1) The holder of a reconnaissance permit, exploration right and production right may transfer liabilities and responsibilities as identified in the environmental management programme, or environmental management plan or any prescribed closure plan to a qualified person.
 - (2) An application to transfer environmental liabilities and responsibilities to a qualified person in terms of section 43(2) of the Act, must be completed on Form U as set out in Annexure 1 of the Regulations and be lodged to the Minister for consideration.
 - (3) When considering the transfer environmental liabilities and responsibilities in terms of section 43(2) of the Act, the Minister must consult with any State department which administers any law relating to matters affecting the environment.
 - (4) No transfer of environmental liabilities to a qualified person may be made unless the Chief Inspector of Mines and the Department of Water Affairs and Forestry have confirmed in writing that the person to whom the liabilities and responsibilities is transferred to, have the necessary qualifications pertaining to health and safety and management of potential pollution of water resources.

Qualifications of person regarding transfer of environmental liabilities and responsibilities

98. (1) For the purposes of transferring environmental liabilities and responsibilities as may be identified in the environmental management programme or plan and any prescribed closure plan, the person to whom such transfer is made must —

- have the expertise, resources and organisational abilities to integrate risk assessment, risk management and risk financing to ascertain the cost of environmental management;
- (b) have the expertise, financial and other resources to meet its obligations to carry out actions necessary to fulfil the environmental obligations as set out in the environmental management programme, environmental management plan or any prescribed closure plan concerned;
- (c) have appropriate experience in environmental management specific to exploration and production operations, health and safety and management of water pollution:
- (d) have direct access to insurance products and alternative risk financing services appropriate to financing of exposure to risks;
- (e) have the ability to manage trusts set up in terms of section 10(1)(cH) of the Income Tax Act. 1962 (Act No 58 of 1962); and
- (f) have expertise and experience or access thereto to interpret and manage the findings of an environmental risk assessment.

Environmental Risk Report

- 99. (1) An application for a closure certificate in terms of section 43(3) of the Act must be accompanied by an environmental risk report which must include-
 - (a) the undertaking of a screening level environmental risk assessment where-
 - all possible environmental risks are identified, including those which appear to be insignificant;
 - (ii) the process is based on the input from existing date;
 - (iii) the issues that are considered are qualitatively ranked as -
 - (aa) a potential significant risk; and/or
 - (bb) a uncertain risk; and/or
 - (cc) an insignificant risk.
 - (b) the undertaking of a second level risk assessment on issues classified as potential significant risks where-
 - (i) appropriate sampling, data collection and monitoring be carried out;
 - (ii) more realistic assumptions and actual measurements be made; and
 - (iii) a more quantitative risk assessment is undertaken, again classifying issues as posing a potential significant risk or insignificant risk.
 - assessing whether issues classified as posing potential significant risks are acceptable without further mitigation;
 - issues classified as uncertain risks be re-evaluated and re-classified as either posing potential significant risks or insignificant risks;
 - documenting the status of insignificant risks and agree with interested and affected persons;
 - identifying alternative risk prevention or management strategies for potential significant risks which have been identified, quantified and qualified in the second level risk assessment;
 - (g) agreeing on management measures to be implemented for the potential significant risks which must include-
 - (i) a description of the management measures to be applied;
 - (ii) a predicted long-term result of the applied management measures;
 - (iii) the residual and latent impact after successful implementation of the management measures;
 - (iv) time frames and schedule for the implementation of the management
 - responsibilities for implementation and long-term maintenance of the management measures;
 - (vi) financial provision for long-term maintenance; and
 - (v) monitoring programmes to be implemented.

Closure objectives

- 100. (1) Closure objectives form part of the draft environmental management programme or environmental management plan, as the case may be, and must
 - identify the key objectives for closure of the operation to guide the project design, development and management of environmental impacts;
 - (b) provide broad future land use objective(s) for the site; and
 - (c) provide proposed closure costs.

Contents of Closure plan

- 101. (1) A closure plan forms part of the environmental management programme or environmental management plan, as the case may be, and must include –
 - a description of the closure objectives and how these relate to the reconnaissance, exploration or production operation and its environmental and social setting;
 - a summary of the regulatory requirements and conditions for closure negotiated and documented in the environmental management programme or plan;
 - a summary of the results of the environmental risk assessment and details of identified residual and latent impacts:

(d) a summary of the results of progressive rehabilitation undertaken;

 (e) a description of the methods to decommission each exploration or production component and the mitigation or management strategy proposed to avoid, minimize and manage residual or latent impacts;

(f) details of any long-term management and maintenance expected;

- (g) details of financial provision for monitoring, maintenance and post closure management, if required;
- a plan or sketch at an appropriate scale describing the final land use proposal and arrangements for the site;
- record of interested and affected persons consulted; and

(j) technical appendices, if any.

PART IV : POLLUTION CONTROL AND WASTE MANAGEMENT REGULATION

Principles of pollution control and waste management

- 102. (1) In accordance with the national norms and standards for pollution control and waste management, the holder of a reconnaissance permit, exploration right or production right must -
 - (a) avoid the generation and production of pollution, waste and mine residue at source; or
 - (b) where the production thereof cannot altogether be avoided, be minimized and re-used or recycled; and
 - (c) where possible and otherwise disposed of in a responsible and prescribed manner.

Air quality control

- 103. (1) The holder of any right or permit in terms of the Act, must comply with national air quality control or management norms and standards where applicable.
 - (2) The assessment of impacts relating to air quality control and management, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan, as the case may be.
 - (3) The holder of an exploration right or production right shall not without the prior written approval of the Petroleum Agency flare any petroleum.

Fire prevention

104. (1) The holder of a reconnaissance permit, an exploration right or production right shall conduct its operations in accordance with good international oil-field practice so as to minimise the risks of fire, explosion and/or uncontrolled escape of petroleum.

Noise control

- 105. (1) The holder of a right or permit in terms of the Act must comply with national norms and standards regarding noise control where applicable.
 - (2) The assessment of impacts relating to noise pollution control and management, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan as the case may be.

- (3) When the equivalent noise exposure, as defined in the South African Bureau of Standards Code of Practice for the Measurement and Assessment of Occupational Noise for Hearing Conservation Purposes, SABS 083 as amended, [in any at or in any operation or works where persons may travel or work, exceeds 85 dB (A)] exceeds the background plus seven (7) dB (A) decibels beyond the operation boundary or at the residential community or where non employees may travel, the holder shall take the necessary steps to reduce the noise to below this level.
- (4) Where compliance with sub-regulation (1) is not possible, the holder shall implement a hearing conservation programme, which complies with the recommendations given in the said SABS Code 083.
- (5) All mechanical equipment must be in a good working order and vehicles must adhere to the relevant noise requirements in terms of the Road Traffic Act, 1997, (Act No. 93 of 1997).
- (6) All vehicles in operation must be in good working order and equipped with a silencer on their exhaust system.
- (7) Where appropriate, lubricants must be applied to ensure that surfaces which interact during mechanical movement do not generate undesirable noise levels.
- (8) Safety measures which generate noise such as reverse gear alarms on large vehicles must be appropriately calibrated or adjusted.
- (9) Appropriate non-metallic washers or isolation must be used with any joining apparatus to join screens such as corrugated iron to other structures and to each other. Such screens must be maintained in a fixed position.
- (10) Appropriate measures must be installed and or employed at the crushing operations or plant to act as screen and to reflect or reduce the noise level.
- (11) Noise levels must be recorded on a regular basis at identified receptor sites which will include the residences of immediate neighbours and the borders of the mining area.

Blasting, vibration and shock

- 106. (1) In accordance with the provisions of the Mine Health and Safety Act, 1996, (Act No. 29 of 1996) blasting operations at reconnaissance, exploration and production operations must be so designed and executed that the minimum shock, noise and vibration are generated.
 - (2) An assessment of impacts relating to blasting, vibration and shock control and management, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan, as the case may be.

Water management and pollution control

- 107. (1) The Regulations promulgated in the Government Notice No. 704 of 4 June 1999, in terms of the National Water Act, 1998, (Act No. 36 of 1998) shall apply to the water management and pollution control at all proposed or existing reconnaissance, exploration or production operations.
 - (2) An assessment of impacts relating to water management and pollution control at proposed reconnaissance, exploration or production operations, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan, as the case may be.
 - (3) The holder of an offshore reconnaissance permit, exploration or a production right must ensure that produced water from an offshore installation-
 - is not diluted with sea-water or any other water prior to measurement of the hydrocarbon content.
 - (b) does not exceed 50 ppm by volume total hydrocarbons (i.e. both aliphatic and aromatic components) on a monthly average.

- (c) does not exceed 100 ppm by volume as a daily maximum.
- (4)The environmental management programme for the installation must include a process of continuous improvement in reducing the oil content of produced water.
- (5) For installations within 25 kilometres of shore or where the risk of pollution of sensitive " " Not a production environments is greater, the Minister may require that the permissible concentration of hydrocarbons in produced water be further reduced.
 - (6)The holder of an exploration right or a production right must ensure that in the event of the above limits being exceeded, a report is promptly submitted to Petroleum Agency, Department of Environmental Affairs and Tourism, and South African Maritime Safety Association, providing details and stating what remedial action is being taken.

Disposal of waste material

4 -4 - 2d 1 - 35

- (1) The holder of any right or permit in terms of the Act, must comply with national norms 108. and standards regarding the disposal of waste material, where applicable.
 - (2) The assessment of impacts relating to the disposal of waste material, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan.
 - Waste material from exploration and production operations shall be disposed of in (2)accordance with the approved environmental management programme or environmental management plan and on an approved demarcated site.
 - (3)No dumping or impounding of rubble, litter, garbage, rubbish or discards of any description, whether solid or liquid, shall take place elsewhere than at the site or sites demarcated for the purpose in accordance with the approved environmental management programme or environmental management plan.
 - (4)No sand dump or slimes dam shall be established on the bank of any stream, river, dam, pan or lake without written permission of the Minister and upon such conditions as he/she may prescribe or if and as approved in the environmental management programme or environmental management plan.
 - (5)No oil-based mud shall be used in exploration or production operations or disposed of without the prior written approval of the Petroleum Agency. Such approval shall only be granted after a full evaluation of the environmental assessment and handling methods are proposed.

Soil pollution and erosion control

4.

- 109. (1) The holder of a right or permit in terms of the Act, must comply with national norms and standards regarding the management and control of soil pollution and erosion, where applicable.
 - . (2) The assessment of impacts relating to soil pollution and erosion control, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan.
 - Except for the areas concerned in the approved areas for the depositioning of waste. all adverse chemical contamination of soils must be minimised to levels, as approved in the environmental management programme or the environmental management plan.
 - (4) The spillage of hazardous chemicals onto soils or its escape or migration into surrounding soils from the approved depositioning area, must be prevented.
 - (5) Oils, grease and hydraulic fluids must be disposed of in a waste disposal recepticle for collection or treatment or at a licensed facility or in accordance with a method approved in the environmental management programme or environmental management plan.

55

- (6) Oils, grease and hydraulic fluid spills which occur must be cleaned up by removing all contaminated soil and disposing thereof in a waste disposal receptacle or at a licensed facility.
- (7) The acidification, salination and mineralisation of soils through seepage of polluted water or irrigation of land must not take place without the permission of the Minister of Water Affairs and Forestry or as approved in the environmental management programme or environmental management plan.
- (8) The chemical and physical properties of top soil to be used for the purposes of rehabilitation, must not be changed by introducing foreign material, gravel, rock, rubble or mine residue.

Sanitation of surface

- Toilet facilities shall be located in such a manner that no water or other pollution is caused.
 - (2) No person shall pollute the workings with faeces or urine or misuse the facilities provided or inappropriately foul the surrounding environment with faeces or urine. Acceptable hygienic and aesthetic practices must be adhered to.

Management of residue stockpiles and deposits

- 111. (1) The assessment of impacts relating to the management of residue stockpiles and deposits, where appropriate, must form part of the environmental impact assessment report and environmental management programme or the environmental management plan.
 - (2) Residue characterisation
 - (a) Exploration or production residue must be characterised to identify any potentially significant health, safety and environmental impact that may be associated with the residue when stockpiled or deposited at the site(s) under consideration.
 - (b) Residue stockpiles and/or deposits must be characterised in terms of its
 - i) physical characteristics, which may include -
 - (aa) the size distribution of the principal constituents;
 - (bb) the permeability of the compacted material;
 - (cc) void ratios of the compacted material;
 - (dd) the consolidation or settling characteristics of the material under its own weight and that of any overburden;
 - (ee) the strength of compacted material;
 - (ff) the specific gravity of the solid constituents; and
 - (gg) the water content of the material at the time of deposition, after compaction, and at other phases in the life of the deposit.
 - (ii) chemical characteristics, which may include -
 - (aa) the toxicity;
 - (bb) the propensity to oxidize and /or decompose;
 - (cc) the propensity to undergo spontaneous combustion;
 - (dd) the pH and chemical composition of the water separated from the solids:
 - (ee) stability and reactivity and the rate thereof; and
 - (ff) neutralising potential.
 - mineral content, which include the specific gravity of the residue particles and its impact on particle segregation and consolidation;
 - (3) Classification of residue stockpiles and deposits
 - All residue stockpiles and deposits must be classified into one or a combination of the following categories –
 - the safety classification to differentiate between residue stockpiles and deposits of high, medium and low hazard on the basis of their potential to cause harm to life or property; and
 - the environmental classification to differentiate between residue stockpiles and deposits with -
 - (aa) a potentially significant impact on the environment due to its spatial extent, duration and intensity of potential impacts; or
 - (bb) no potentially significant impact on the environment.

- (b) All exploration and production residue stockpiles and deposits must be classified by a suitably qualified person(s).
- (c) The classification of residue stockpiles and deposits shall determine the -
 - (i) level of investigation and assessment required;
 - (ii) requirements for design, construction, operation, decommissioning, closure and post closure maintenance; and
 - qualifications and expertise required of persons undertaking the investigations, assessments, design, construction thereof.
- (d) The safety classification of residue stockpiles and deposits shall be based on the following criteria –

Number of residents in zone of influence	Number of workers in zone of influence		Depth to underground mine workings	Classification	
0 < 10		0 – R2 m	> 200m	Low hazard	
1 – 10	11 – 100	R 2 m - R20 m	50 m - 200 m	Medium hazard	
> 10	> 100	> R20 m	< 50 m	High hazard	

- (e) A risk analysis must be carried out and documented on all high hazard residue stockpiles and deposits.
- (f) The environmental classification of residue stockpiles and deposits must be undertaken on the basis of —
 - (i) the characteristics of the residue;
 - the location and dimensions of the deposit (height, surface area);
 - (iii) the importance and vulnerability of the environmental components that are at risk; and
 - iv) the spatial extent, duration and intensity of potential impacts.
- (g) An assessment of the environmental impacts shall be done on all environmental components which are significantly affected.
- (h) The assessment of impacts and analyses of risks shall form part of the environmental impact assessment and environmental management programme or plan.

(4) Site selection and investigation:

- (a) The process of investigation and selection of a site must entail -
 - the identification of a sufficient number of possible candidate sites to ensure adequate consideration of alternative sites;
 - (ii) qualitative evaluation and ranking of all alternative sites;
 - (iii) qualitative investigation of the top ranking sites to review the ranking done in (ii);
 - (iv) a feasibility study to be carried out on the highest ranking site(s), involving -
 - (aa) a prelimenary safety classification;
 - (bb) an environmental classification;
 - (cc) geotechnical investigations; and
 - (dd) groundwater investigations.
- (b) The geotechnical investigations may include-
 - the characterization of the soil profile over the entire area to be covered by the residue facility and associated infrastructure to define the spatial extent and depth of the different soil horizons;
 - the characterization of the relevant engineering properties of foundations soils and the assessment of strength and drainage characteristics.
- (c) The groundwater investigations may include-
 - (i) the potential rate of seepage from the residue facility;
 - (ii) the quality of such seepage;
 - (iii) the geohydrological properties of the strata within the zone that could potentially be affected by the quality of seepage;
 - (iv) the vulnerability and existing potential use of the groundwater resource within the zone that could potentially be affected by the residue facility.
- (d) From these investigations, a preferred site must be identified.
- (e) Further investigation on the preferred site, shall include
 - (i) land use:
 - (ii) topography and surface drainage;
 - (iii) infrastructure and man-made features;

- (iv) climate;
- (v) flora and fauna;
- (vi) soils:
- (vii) ground water morphology, flow, quality and usage; and
- (viii) surface water.
- (f) The investigations, laboratory test work, interpretation of data and recommendations for the identification and selection of the most appropriate and suitable site for the disposal of all residue that have the potential to generate leachate that could have a significant impact on the environment and groundwater must be carried out by a suitably qualified person.

(5) Design of residue stockpile and deposit

- (a) The design of the residue stockpile and deposit shall be undertaken by a suitably qualified person.
- (b) An assessment of the typical soil profile on the site is required for residue stockpiles and deposits which -

(i) have a low hazard potential; and

(ii) have no significant impact on the environment.

- (c) The design of the residue stockpile and deposit must take into account all phases of the life cycle of the stockpile and deposit, from construction through to closure and must include —
 - (i) the characteristics of the mine residue;

(ii) the characteristics of the site and the receiving environment:

- (iii) the general layout of the stockpile or deposit, whether it is a natural valley, ring dyke, impoundment or a combination thereof and its 3dimensional geometry at appropriate intervals throughout the planned incremental growth of the stockpile or deposit;
- (iv) the type of deposition method used; and

(v) the rate of rise of the stockpile or deposit.

- (d) Other design considerations, as appropriate to the particular type of stockpile and deposit must be incorporated –
 - the control of storm water on and around the residue stockpile or deposit by making provision for the maximum precipitation to be expected over a period of 24 hours with a frequency of once in a 100 years, in accordance with the regulations made under section 8 of the National Water Act, 1998;
 - the provision, throughout the system, of a freeboard of at least 0.5 m above the expected maximum water level, in accordance with regulations made under the National Water Act, 1998, to prevent overtopping;
 - (iii) keeping the pool away from the walls; where there are valid technical reasons for deviating from this, adequate motivation must be provided and the design must be reviewed by a qualified person as required in terms of sections 9(6) or 9(7) of the Mine Health and Safety Act, 1996;
 - the control of decanting of excess water under normal and storm conditions;
 - (aa) the retension of polluted water in terms of polluted water in terms of GN R991(9), where measures may be required to prevent water from the residue deposit from leaving the residue management system unless it meets prescribed requirements;
 - (bb) the design of the penstock, outfall pipe, under-drainage system and return water dams;
 - (cc) the height of the phreatic surface, slope angles and method of construction of the outer walls and their effects on shear stability;
 - (dd) the erosion of slopes by wind and water, and its control by

(ee) vegetation, berms or catchment paddocks; and

(ee) the potential for pollution.

 (e) A design report and operating manual shall be drawn up for all residue stockpiles and deposits which —

(i) have a medium to high hazard; and

- (ii) have a potentially significant impact on the environment.
- (f) Relevant information must be included in the draft environmental management programme or environmental management plan.

(6) Construction and operation of residue deposits:

(a) The holder of any right or permit in terms of the Act, must ensure that-

 the residue deposits, including any surrounding catchment paddocks, is constructed and operated in accordance with the approved environmental management programme or environmental management plan;

 the design of the residue deposit is followed implicitly throughout the construction thereof, and that any deviations from the design be approved by the Regional Manager and the environmental manage programme and environmental management plan be amended accordingly;

(iii) as part of the monitoring system, measurements of all residues transported to the site and of all surplus water removed from the site

are recorded;
(iv) the provision for appropriate security measures be implemented to limit unauthorised access to the site and intrusion into the residue deposit;

(v) specific action be taken in respect of any sign of pollution;

 (vi) adequate measures be implemented to control dust pollution and erosion of the slopes; and

 (vii) details of rehabilitation of the residue deposit be provided in the draft environmental management programme or environmental management plan.

(b) A system of routine maintenance and repair in respect of the residue deposit must be implemented to ensure the ongoing control of pollution, the integrity of rehabilitation and health and safety maters at the site.

(7) Monitoring of residue stockpiles and deposits:

- (a) A monitoring system for residue stockpiles and deposits with respect to potentially significant impacts as identified in the environmental assessment must be included in the environmental management programme or environmental management plan.
- (b) In the design of a monitoring system for a residue stockpile or deposit, consideration must be given to –
 - baseline and background conditions with regard to air, surface and groundwater quality;
 - (ii) the air, surface and groundwater quality objectives;

(iii) residue characteristics;

(iv) the degree and nature of residue containment;

- the receiving environment and specifically the climatic, local geological, hydrogeological and geochemical conditions;
- (vi) potential migration pathways;

(vii) potential impacts of leachate;

- (viii) the location of monitoring points and the prescribed monitoring protocols; and
- (ix) the reporting frequency and procedures.

(8) Decommissioning, closure and after care:

- (a) The decommissioning, closure and post closure management of residue deposits must be addressed in the closure plan, which must contain the following -
 - the environmental classification, including assumptions on which the classification were based;
 - (ii) the closure objectives, final land use or capability;
 - (iii) conceptual description and details for closure and post closure management;
 - (iv) cost estimates and financial provision for closure and post-closure management; and
 - residual impacts, monitoring and requirements to obtain closure in terms of the Act.

CHAPTER 4

APPEALS

Appeal against administrative action by Regional Manager, the Petroleum Agency or an officer

- 112. (1) Any person, hereinafter referred to as the appellant, who appeals under section 96(1)(a) of the Act against an administrative action of a Regional Manager, the Petroleum Agency or an officer, hereinafter referred to as the Respondent, must within 30 days after he or she has become aware of such administrative action, lodge a written notice of appeal with the Director-General.
 - (2) The notice of appeal shall state clearly-
 - (a) the actions appealed against; and
 - (b) the grounds on which the appellant rely.
 - (3) When lodging the notice of appeal, the appellant shall at the same time deposit with the Director-General the fee prescribed.
 - (4) The Director-General may in his or her discretion and on such terms and conditions as he or she may decide, condone the late noting of an appeal.
 - (5) After receipt of the notice of appeal, the Director-General must dispatch copies thereof to —
 - (a) the respondent; and
 - (b) any other person, hereinafter referred to as the affected party, whose rights may, in the opinion of the Director-General, be affected by the outcome of the appeal

and request the persons contemplated in paragraphs (a) and (b) to respond as provided for in sub-regulations (6) and (7).

- (6) The respondent must, within 21 days from receipt of the notice of appeal, submit to the Director-General written reasons for the administrative action appealed against.
- (7) The affected party must within 21 days from receipt of the notice of appeal, submit to the Director-General a replying submission indicating -
 - (a) the extent and nature of his or her rights;
 - (b) how the outcome of the appeal may affect his or her rights; and
 - (c) any other information pertaining to the grounds as set out in the notice of the appeal.
- (8) The Director-General must dispatch the documents referred to in sub-regulations (6) and (7) to the appellant by registered post and request him or her to reply in writing within 21 days from receipt thereof.
- (9) The Director-General must, within 30 days from the date contemplated in subregulation (8), either -
 - (a) confirm;
 - (b) set aside;
 - (c) amend the administrative action of the respondent; or
 - substitute any other administrative action for the administrative action of the respondent.

Appeal against administrative action by Director-General or Petroleum Agency

- Any person, hereinafter referred to as the appellant, who appeals under section 96(1)(a) of the Act against an administrative action of a Director-General or the designated agency, hereinafter referred to as the Respondent, must within 30 days after he or she has become aware of such administrative action, lodge a written notice of appeal with the Minister.
 - (2) The notice of appeal shall state clearly-
 - (a) the actions appealed against; and
 - (b) the grounds on which the appellant rely.

- (3) When lodging the notice of appeal, the appellant shall at the same time deposit with the Director-General the sum of R200.
- (4) The Minister may in his or her discretion and on such terms and conditions as he or she may decide, condone the late noting of an appeal.
- (5) After receipt of the notice of appeal, the Minister must dispatch copies thereof to -

(a) the respondent; and

(b) any other person, hereinafter referred to as the affected party, whose rights may, in the opinion of the Minister, may be affected by the outcome of the appeal

and request the persons contemplated in paragraphs (a) and (b) to respond as provided for in sub-regulations (6) and (7).

- (6) The respondent must, within 21 days from receipt of the notice of appeal, submit to the Minister written reasons for the administrative action appealed against.
- (7) The affected party must within 21 days from receipt of the notice of appeal, submit to the Minister a replying submission indicating -

(a) the extent and nature of his or her rights;

(b) how the outcome of the appeal may affect his or her rights; and

- any other information pertaining to the grounds as set out in the notice of the appeal.
- (8) The Minister must dispatch the documents referred to in sub-regulations (6) and (7) to the appellant by registered post and request him or her to reply in writing within 21 days from receipt thereof.
- (9) The Minister must, within 30 days from the date contemplated in sub-regulation (8), either -
 - (a) confirm:
 - (b) set aside;
 - (c) amend the administrative action of the respondent; or
 - substitute any other administrative action for the administrative action of the respondent.

CHAPTER 5

GENERAL AND MISCELLANEOUS REGULATIONS

Refusal of prospecting right, mining right, mining permit, retention permit, reconnaissance permit, technical co-operation permit, exploration right, production right

114. Refusal of a prospecting right, mining right, mining permit or retention permit, reconnaissance permit, technical co-operation permit, exploration right, production right, must be completed on Form R, Annexure 1.

Permission to remove bulk samples of minerals during prospecting or exploration

115. Permission to dispose of or remove bulk samples of minerals or petroleum during prospecting or exploration must be completed on Form S, Annexure 1.

Notice for the suspension or cancellation of a prospecting right, mining right, mining permit or retention permit, reconnaissance permit, technical co-operation permit, exploration right or production right

116. Notice for the suspension or cancellation of a prospecting right, mining right, mining permit or retention permit, reconnaissance permit, technical co-operation permit, exploration right, production right ,must be completed on Form V, Annexure 1.

Withdrawal of the suspension notice

117. Notice of the withdrawal of the suspension notice must be completed as per Form W, Annexure 1.

Notice to prohibit or restrict the granting of a prospecting right, mining right, reconnaissance permit, exploration right or production right

118. Notice to prohibit or restrict the granting of a prospecting right or mining right as contemplated in section 49 of the Act, must be completed on Form X, Annexure 1.

Notice of profitability and curtailment of mining or production operations affecting employment

119. Notice of profitability and curtailment of mining operations affecting employment must be completed as per Form Y, Annexure 1.

Notification to Regional Manager / Petroleum Agency if holder is prevented from commencing with operations

120. The holder of a prospecting right, mining right or mining permit, reconnaissance permit, technical co-operation permit, exploration right, production right, must notify the Regional Manager or Petroleum Agency if he or she is prevented from commencing with operations as per Form Z, Annexure 1.

Notice to land owner or lawful occupier of land if holder is prevented from commencing with operations

121. Notice given by the Regional Manager or Petroleum Agency to the land owner or lawful occupier of land, if holder is prevented from commencing with operations, must be completed as per Form AA, Annexure 1.

Notice of expropriation of property

122. Notice of the expropriation of property for the purpose of prospecting, mining, reconnaissance, exploration or production, must be completed on Form BB, Annexure 1.

Certificate to authorise person to enter prospecting area, mining area, retention, reconnaissance, exploration or production area

123. The certificate as per Form DD, Annexure 1, to authorise person to enter prospecting area, mining area, retention, reconnaissance, exploration or production area.

Notice by authorised person if holder contravenes provisions of Act

Notice given to the holder of a prospecting right, mining right, mining permit, reconnaissance permit, exploration right or production right, by an authorised person as contemplated in section 93(1) of the Act, must be completed on Form CC, Annexure 1.

Balance Sheet: Liquidator's report

125. The holder of a permit or right shall also file with the Director-General immediately on publication a copy of each annual report, including the balance sheet and profit and loss account, issued by the directors of such concern, and also a copy of the liquidator's report on liquidation.

CHAPTER 6

TRANSITIONAL ARRANGEMENTS

Lodgement for the conversion of an old order prospecting right in terms of the Minerals Act, 1991, (Act No. 50 of 1991) to one in terms of the Mineral and Petroleum Resources Development Act, 2002, (Act No. 28 of 2002)

An application for the conversion of an old order prospecting right in terms of section 6(1) of Schedule II of the Act, must be completed on Form I as set out in Annexure 1.

Lodgement for the conversion of an old order mining right in terms of the Minerals Act, 1991, (Act No. 50 of 1991) to one in terms of the Mineral and Petroleum Resources Development Act, 2002 (Act No. 28 of 2002)

127. The application for the conversion of an old order mining right in terms of section 7(1) of Schedule II of the Act must be in the form of Form J in Annexure 1.

			Al	NEXURE '	an a		25/6-7-2015	
PPLICATION TERMS O	N FORMS F F THE MINE	OR PERM	MISSION PETRO	IS, RIGHTS	AND PER	MITS TO DEVELO	BE COMPL PMENT AC	ETED T, 2002
	N FOR A RE							
NISTER		*			9 25			© _{\$1}
mission in ter	ms of section 13	3 of the Act o	over the fo	the u	ndersigned	hereby ap	oly for a rec	connaissa
1.	12		7701 (110 10	noming proper	ty or propertit		9	- 2
2.	10.0					*		''
3.								The same of the sa
	- 104		1	10 10	7.			<u></u>
6.00-00								 .
4.	<u> </u>	¥8		17				
0. 00				e e	, ,			×
4	ON MUST BE A	ICCOMPAN	IED BY:		, <u>r</u>			¥
4.	ON MUST BE A	1CCOMPAN	IIED BY:					
4	Access to	financial res	ources.			-		2
4	Access to Technical Reconnais undertake	financial res ability to con ssance work n.	ources. nduct the s programr	192	ince to which	the recon	naissance ope	

APPLICANT

SIGNED AT

FORM B:

APPLICATION	ON FOR A	PROSPECTING	RIGHT IN	TERMS OF	SECTION	16 OF THE	MINERAL
AND PETRO	LEUM RE	SOURCES DEVE	LOPMENT	ACT. (ACT	NO 28 OF	2002)	

	e Minister
1.	Full name of applicant
	Identity number (in the case of a person other than a natural person the registration number, where applicable, mus be furnished):
3.	Postal Address
4.	(a) Registered name of land (farm and subdivision) on which you propose to prospect:
	(b) Magisterial District
	Mineral; associated minerals or group of minerals for which you propose to prospect:
6.	Period for which the right is required:
TH	IE APPLICATION MUST BE ACCOMPANIED BY:
	Details of the Area Applied for: i. Provide a plan, drawn up in such a form and to such a scale as may be required by the Regional Manager, and signed and dated by the applicant showing: - the area applied for. The co-ordinates of the corner points are to be tabulated on the map in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown. ii. The plan should show in addition: - the name, number and situation on the farm or farms on which the area applied for is situated, - surface structures and servitudes, - the topography of the immediate vicinity, - the size of the area applied for in square kilometres. - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations. Provide details of any other matter relevant to the application.
В.	A Detailed Prospecting Work Programme
C.	Substantiate Financial and Technical Competence Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the prospecting activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.
D.	Existing rights and past compliance with provisions of the Act Provide details of existing rights and permits held by the applicant in the region where the application(s) is lodged. Report the extent of compliance with the requirements of the approved Environmental Management Plan for any prospecting or mining right that the applicant holds or has held in South Africa and has not contravened any provisions of the Act.
G.	Acknowledge that the Regional Manager is entitled to request further information before considering the prospecting right.
н.	Prescribed Fee An amount of R, being the prescribed non-refundable application fee, must accompany this application.
SIG	GNATURE OF APPLICANT DATE:

FORM C:
APPLICATION FOR THE RENEWAL OF A PROSPECTING RIGHT IN TERMS OF SECTION 18(1) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

1000	and the second second	- 17.	51 40	200 80	163	
1.	Full name of applicant	`				
2.	Identify number (in the must be furnished):	e case of a	person other than a	a natural person the n	egistration number,	where applica
3.	Postal Address			55 E		
			T	elephone Number:		••••••
4.(a)	Registered name o	of land (farm	and subdivision) on	which you propose to	continue to prospec	t:
(b)	Magisterial District	·····		••••••		
			V 1	#1		
5.	Mineral for which you p	propose to pr	ospect:			
3.	Period for which renew	al is required	1		18 g 13	
		ar io roquiroc	*		***************************************	*****************
	*					
7.		ls required			Anger	
7.	Reasons why renewal	5500		F 2 2		
7.		5500				
7.	Reasons why renewal	5500			, , , , , , , , , , , , , , , , , , ,	
7. THE	Reasons why renewal	BE ACCOM	PANIED BY:		X	
7.	APPLICATION MUST	BE ACCOMI	PANIED BY:		X	
7. THE	Reasons why renewal	BE ACCOMI	PANIED BY:		X	
7. THE A.	APPLICATION MUST A detailed report regar expenditures incurred.	BE ACCOMI	PANIED BY:	ation of completed pr	ospecting activities	and prospec
7. THE	A detailed report regar expenditures incurred. A report reflecting the	BE ACCOMP arding the re	PANIED BY: esults and interpret	ation of completed pr	ospecting activities	and prospec
7. THE A.	APPLICATION MUST A detailed report regar expenditures incurred.	BE ACCOMP arding the re	PANIED BY: esults and interpret	ation of completed pr	ospecting activities	and prospec
7. THE A.	A detailed report regard expenditures incurred. A report reflecting the environmental manage	BE ACCOMI arding the re extent of the ement plan as	PANIED BY: esults and interpret environmental rehals well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an	ospecting activities	and prospec
7. A. 3.	A detailed report regard expenditures incurred. A report reflecting the environmental manage. A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regar expenditures incurred. A report reflecting the environmental manage A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regard expenditures incurred. A report reflecting the environmental manage. A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regar expenditures incurred. A report reflecting the environmental manage A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regar expenditures incurred. A report reflecting the environmental manage A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regar expenditures incurred. A report reflecting the environmental manage A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.
7. THE	A detailed report regar expenditures incurred. A report reflecting the environmental manage A detailed prospecting	BE ACCOME arding the re extent of the ement plan as work program	PANIED BY: esults and interpret environmental reha s well as rehabilitation	ation of completed pr bilitation conducted an on to be completed an I period.	ospecting activities d completed in term d the estimated cos	and prospectors of the approximate thereof.

FORM D:

APPLICATION FOR A MINING RIGHT IN TERMS OF SECTION 22(1) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

ine	Minister
1.	Full name of applicant
2.	Identity number (in the case of a person other than a natural person the registration number, where applicable, must be furnished):
3.	Postal Address
	Telephone Number:
4. 	(a) Registered name of land (farm and subdivision) on which you propose to mine
	(b) Magisterial District
5.	Mineral; associated minerals or group of minerals for which you propose to mine:
6.	Period for which the right is required:

THE APPLICATION MUST BE ACCOMPANIED BY:

A. Details of the Area Applied for :

- ii. Provide a plan, drawn up in such a form and to such a scale as may be required by the Regional Manager, and signed and dated by the applicant showing:
 - the area applied for. The co-ordinates of the corner points are to be tabulated on the map in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.
- iii. The plan should show in addition:
 - the name, number and situation on the farm or farms on which the area applied for is situated.
 - surface structures and servitudes,
 - the topography of the immediate vicinity,
 - the size of the area applied for in square kilometres.
 - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.
- iii. Provide details of any other matter relevant to the application.

B. A Detailed Mining Work Programme

C. Substantiate Financial and Technical Competence

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the mining activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

D. Existing rights and past compliance with provisions of the Act

Provide details of existing rights held by the applicant in the region where the applications is lodged. Report the extent of compliance with the requirements of the approved Environmental Management Programme or Plan for any prospecting or mining right that the applicant holds or has held in South Africa and has not contravened any provision of the Act.

G. Acknowledge that the Regional Manager is entitled to request further information before considering the prospecting right.

H. Social and Labour Plan

A social and labour plan containing details of the socio-economic impact of the proposed mining project, as well as the measures to remedy any negative socio-economic impacts during the life of the mine and after closure:

I. Prescribed For An amount of application	70.74	1	being the pres	cribed nor	-refundable	application	n fee, must	accompany this
арризанон	10 - Feb. 1	802	* 2 H		26		20 ¹⁰	
En En		56	0909				83	
SIGNATURE OF	APPLICANT		Š.	w I	DATE:			••••

FORM E:

APPLICATION FOR THE RENEWAL OF A MINING RIGHT IN TERMS OF SECTION 24(1) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

In	e Minister		30	12	00.00		3 D
••••					ev p	0.8	7
			0		25 AM		
1.	Full name of applicant						2004
					.,		8:17 54
2.	Identify number (in the case of a person other to applicable, must be furnished):	han a natura	l person	the regi	stration n	umber,	wher
3.	Postal Address	i i		E p	Ÿ.		2 6.
17.77							
		Telephone N	umber: .				52.23
1	(a) Registered name of land (farm and subdivision) o	n which you n	ronoeo t	o continu	e to mine:		390
4.	(a) Registered flame of land (lann and subdivision) o						
							1
							100
	(b) Magisterial District						
				••••••			
5.	Mineral for which you propose to mine:				-100	j eg	
	,,						
_	Anna Property of the second property of the s						
6.	Period for which renewal is required			12 1 40			- 1
						•••••	
7.	Reasons why renewal is required				0 6	49	92
-	IS ARRUST OF ASSOCIATION OF				and the second		
TH	IE APPLICATION MUST BE ACCOMPANIED BY:			-			
A.	A report reflecting the extent of the environmental approved environmental management programme estimated costs thereof.	rehabilitation of as well as	conducte rehabilit	ed and co ation to	mpleted i be comp	n terms leted an	of the
В.	A detailed revised mining work programme for the re	enewal period				6	
C.	An amount of R, being the prescribe application.	ed non-refund	able app	lication fe	e, must a	ccompar	ny this
	707 II						
			320				*
	· · · · · · · · · · · · · · · · · · ·		- G - B				
			*			98	

SIC	GNATURE OF APPLICANT DATE:	2	42	61		*	

FORM F:

APPLICATION FOR A MINING PERMIT IN TERMS OF SECTION 27(2) AND THE RENEWAL THEREOF IN TERMS OF SECTION 27(8) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Th	ne Minister	- 1					
	Full name of applicant			P.		ű.	
2.	ldentity number (in the case of a papplicable, must be furnished):	person other	than a natural	person the	registration	numb	
3.	Postal	11:	y this way	egitera en e		8 8	Address
		Telephone N	Number:	·····	·····	 	
4. he	(a) Registered name of land (farm ctares:	and the same of th					araan Baaraa
	(b) Magisterial District						••••••
5.	Mineral; associated minerals or group	of minerals w	hich you propos	e to mine:			
6.	Period for which the right is required: .	·)	v .		4 i,	(not m	
92	two years), or Period for which the renewal of the rig than one year)	ht is required:				(not	more
Th	IE APPLICATION MUST BE ACCOMPA	ANIED BY:	The second second	5	-		
	the area applied for. in latitude / longitude should be shown. iv. The plan should show in the name, number a situated, surface structures at the topography of the the size of the area a major towns, roads, communities and fac iii. Provide details of any oth	addition: and situation of and servitudes, e immediate v applied for in s rail lines, cons bilities that counter matter rele	on the farm or factionity, square metres of servation areas, and be affected by the to the application to th	rms on which r kilometres, restricted an	datum. The	north	point for is
В,	Substantiate Financial and Technical Provide details with documentary resources that are readily available the mining activities and to mitigat	evidence to pr e or how they	ove the applications will be provided	for, to enabl	e the applic	cant to	carry out
c.	Existing rights and past compliance Provide details of existing rights he Report the extent of compliance Programme or Plan for any prosp Africa and has not contravened an	eld by the app with the requ pecting or min	licant in the regi irements of the ing right that th	approved E	nvironmen	tal Mar	agement
Э.	Acknowledge that the Regional Mana the mining permit.	iger is entitle	d to request fu	rther inform	ation befor	re cons	idering
i.	Prescribed Fee An amount of R, bein application.	g the prescrib	ed non-refundal	ole applicatio	n fee, must	accom	pany this
	¥ y	1		٠			-
SIG	NATURE OF APPLICANT	3	DATE:		***************************************		

FORM G:

APPLICATION FOR A RETENTION PERMIT IN TERMS OF SECTION 31(1) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002 (ACT NO. 28 OF 2002)

The	Minister	36	(4)	a a			- 4 - 4 -	11
	***********							. 14
1.		e of applicant						((a)
2.		ımber (in case of a pe			on the registration	number)		ż
3.		idressTelephone i					. 3	-4
4.	(a)	Registered name o				retain:	- 4	8
	-						12	
						73		
	(b)	Magisterial District				. 4	5	
5.	Allotted r	number of prospecting	right:			2.75	¥2	al.
6.	Period fo	r which the retention p	period is required	d t		0 a		** (0)
7.	Th	e application must be	accompanied b	у:	, i		* * * * * * * * * * * * * * * * * * *	
	(a) (b)	Full particulars of the Full particulars of preport. Proof that a minera	rospecting opera I reserve exist wi	itions conduct	ed on the relevan		f.	
	(d) ,	Proof that the minin conditions.	ig of the relevant	mineral would	d be uneconomic	al due to pr	evailing n	narket
	(e)	Proof that the application	cant has complie	ed with the rele	evant provisions of	f the Act.		
	(f)	If the area applied f provide the co-ordin	or is less than th nates for the area	ie originai area a to be retaine	a for which the prod.	ospecting ri	gnt was g	granted
8.	An amou	nt of R,	being the applic	ation fee mus	t accompany this	application	• = ==	
			9	05 18				
	+38	140	e e	2903		80	. 8	
	900	3						
SIG	NATURE O	F APPLICANT	e e e e e e e e e e e e e e e e e e e	DATE		# #[

FORM H:

APPLICATION FOR RENEWAL OF A RETENTION PERMIT IN TERMS OF SECTION 34(1) OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002 (ACT NO. 28 OF 2002)

I IIO N	viii notei			8 8	
	a 9		n	©	4
1.	Full name of applicant				1 10 2
2.	Identity number (in case of a pe	rson other than	a natural person th	e registration numb	per)
3.	Postal addressTelephone !	Number			ti k
4.	(a) Registered name or retention permit:		•••••	you wish to retain	the
5.	Allotted number of the retention				
6.	Period for which the retention p	period is required	d b		g a
7.	The application must be	accompanied b	y:		
*****	the relevant mi (b) The reasons fo	neral cannot be r the renewal be	e circumstances wheeconomically mine ing sought, mplied with the rele	d	
8.	An amount of R,	being the applic	ation fee must acc	ompany this applic	ation.
	De la serie de				# ²⁵
	a to the second				÷ 10
SIGN	ATURE OF APPLICANT		DATE		

FORM I:

FOR THE CONVERSION LODGEMENT FORM OF PROSPECTING RIGHT IN TERMS OF THE MINERALS ACT, (ACT NO 50 OF 1991), TO ONE IN TERMS OF THE MINERAL AND PETROLEUM RESOURCES **DEVELOPMENT ACT, (ACT NO 28 OF 2002)**

The	e Minister					, x = 0		
••••					61 - 81 - ₁₈	32	_ 2	
1.	Full name of applicant				**********			
••••								
	A 10 M 420 T	8 S	35 34 32	57	253	- 12 - 1	W B	
2.	Identity number (in the case of a applicable, must be furnished):	* V		_		W	(d) Tr	3 of 10
••••			*****************		***********			-
3.	Postal Address							
	(a) Registered name of land (fa				i			
5. 	Mineral; associated minerals or g	group of rue to pros	ninerals bein	g held u	nder the	old orde	r prospectii	ng right
6.	Number of old order prospecting				952			
7.	Period for which the right is requi	ired:	•••••					
ТН	E APPLICATION MUST BE ACC	OMPANII	ED BY:					

A. Details of the Area Applied for :

Provide a plan, drawn up in such a form and to such a scale as may be required by the Regional Manager, and signed and dated by the applicant showing:

the area applied for. The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.

The plan should show in addition:

- the name, number and situation on the farm or farms on which the area applied for is situated,
- surface structures and servitudes,
- the topography of the immediate vicinity,

the size of the area applied for in square kilometres.

major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.

Provide details of any other matter relevant to the application.

- An affidavit verifying that the holder is conducting or has conducted prospecting on the land to which the conversion relates immediately before this Act came into effect and setting out the periods during which such prospecting had been conducted;
- Information whether or not the prospecting right being held in terms of the old order prospecting right is encumbered by any mortgage bond or other right registered at the Deeds Office or Mining Titles Office and the registered number of such mortgage bond or other right;
- A statement setting out the terms and conditions which apply to the present old order prospecting right;
- A certified copy of the title deed in respect of the land to which the present prospecting right
- The original old order prospecting right or a certified copy thereof.

B. A Detailed Prospecting Work Programme

C. Substantiate Financial and Technical Competence

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the prospecting activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

D. Existing rights and past compliance with provisions of the Act

Provide details of existing rights held by the applicant in the region where the application(s) is lodged.

Report the extent of compliance with the requirements of the approved Environmental Management Programme for any prospecting or mining right that the applicant holds or has held in South Africa and has not contravened any provision of the Act.

G. Acknowledge that the Regional Manager is entitled to request further information before considering the conversion of the prospecting right.

accomp	any this	s applic	ation.		1							
		48	55			6	25					
9 22 30 1944				43	4			8	50.0	585		#2
		27		4.	•	20	12			37	* *	
2	100											- 2

FORM J:

LODGEMENT FORM FOR THE CONVERSION OF A OLD ORDER MINING RIGHT IN TERMS OF THE MINERALS ACT, (ACT NO 50 OF 1991), TO ONE IN TERMS OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Th	e Minister
1.	Full name of applicant
2.	Identity number (in the case of a person other than a natural person the registration number, where applicable, must be furnished):
3.	Postal Address Telephone Number:
4. 	(a) Registered name of land (farm and subdivision) on which you wish to continue mining:
	(b) Magisterial District
5.	Mineral; associated minerals or group of minerals being held under the old order mining right or any other mining right from previous legislation and for which you wish to continue to mine:
6.	Period for which the right is required:

THE APPLICATION MUST BE ACCOMPANIED BY:

A. Details of the Area Applied for :

- Provide a plan, drawn up in such a form and to such a scale as may be required by the Regional Manager, and signed and dated by the applicant showing :
 - the area applied for. The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.
- 4. The plan should show in addition:
 - the name, number and situation on the farm or farms on which the area applied for is situated.
 - surface structures and servitudes,
 - the topography of the immediate vicinity,
 - the size of the area applied for in square kilometres.
 - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.
- 3. Provide details of any other matter relevant to the application.
- 4. An affidavit verifying that the holder is conducting or has conducted mining on the land to which the_conversion relates immediately before this Act came into operation and setting out the periods during which such mining had been conducted;
- 5. Information whether or not the mining right being held in terms of the old order mining right is encumbered by any mortgage bond or other right registered at the Deeds Office or Mining Titles Office and the registered number of such mortgage bond or other right:
- A statement setting out the terms and conditions which apply to the present mining right;
- 7. A certified copy of the title deed in respect of the land to which the present mining right relates;
- 8. The original lod order mining right or a certified copy thereof.

B. A Detailed Mining Work Programme

C. Substantiate Financial and Technical Competence

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the mining activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

D. Existing rights and past compliance with provisions of the Act

Tilgir.

721.1

Provide details of existing rights held by the apllicant in the region where the application(s) is lodged.

Report the extent of compliance with the requirements of the approved Environmental Management Programme for any prospecting or mining right that the applicant holds or has held in South Africa and has not contravened any provision of the Act.

G. Acknowledge that the Regional Manager is entitled to request further information before considering the conversion of the prospecting right.

	An amount of R. accompany this		, being	the pres	cribed no	n-refund	able ap	plication	fee, must
ä	n state end	ti da la			385 10		si [#] 0		# E
191	gw Try,	20	14. C 25	71 -	D 0		100		a # #
	to explor	and the second second		te de la	i ja	, F - 1 x	0		

FORM K:

APPLICATION FOR A RECONNAISSANCE PERMIT IN TERMS OF SECTION 74 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Note that:

- a Reconnaissance Permit for petroleum is valid for a period not exceeding one year.
- it does not represent an exclusive right and is not transferable or renewable.
- an environmental management programme must be approved before seismic operations may commence.

Applications must be delivered to:

The Chief Executive, Petroleum Agency SA, PO Box 1174,

Parow, 7499,

Tel: 021 938 3500 Fax: 021 938 3520

Email: plu@petroleumagencysa.com

Street address:

Petroleum House, 151 Frans Conradie ave., Parow, 7500.

<u>A.</u> D	etails of Applicant	and the figure was the state of the
1) :	Full Name of Applicant	***************************************
2)	Registration Number of Company / Identify Number of I	A PROBLEM CONTROL OF THE CONTROL OF
3) :	Postal Address	
••••••	Telephone No	Fax No
4)	Black Empowerment Shareholding / Partners (%):	
	rea Applied for & Summary of Planned Operations Hydrocarbon Targeted in Reconnaissance Programme	(oil & / or gas or CBM) :
2) :	Period Applied for	
3) Reco	Magisterial District, Farm Name / sub-division or Offsho nnaissance Area lies	re Licence Block in which
4)	Summary of Planned Reconnaissance Operations :	1 N
Signa	ature of Applicant D	ate

THE APPLICATION MUST BE ACCOMPANIED BY:

C. Details of the Area Applied for :

- iv. Provide a plan, drawn up in such a form and to such a scale as may be required by the Petroleum Agency, and signed and dated by the applicant showing :
 - the area applied for. The corner points should be labelled clockwise A,, B, C etc.
 The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.
- v. The plan should show in addition:
 - the name, number and situation on the farm or farms on which the area applied for is situated.
 - surface structures and servitudes,
 - the topography of the immediate vicinity.
 - the size of the area applied for in square kilometres.
 - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.
 - for offshore applications, the Petroleum Agency numerical licence block(s) affected should be shown.
- iii. Provide details of any other matter relevant to the application.

D. Details of Planned Reconnaissance Operations :

- Specify the minimum proposed operations and expenditure for the block applied for ;
- ii. Provide a map showing the areas within which 2D and 3D seismic surveys are

planned;

iii

- Detail the proposed time framework for the proposed reconnaissance operations;
- vi. Provide a statement of the anticipated effect which the proposed reconnaissance operations may have on the environment and other users of the area.
- vii. Provide a budget for the proposed work programme.
- v. Note that provision must be made to provide copy of all resulting data, reports, information and samples to the Petroleum Agency SA.

E. Substantiate Financial and Technical Competence

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the reconnaissance activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

F. Past Compliance with Provisions of the Act

Provide a report by an independent third party of the extent of compliance with the requirements of the approved Environmental Management Programme for any petroleum right that the applicant holds or has held in South Africa.

G. Acknowledge that the Petroleum Agency is entitled to request further information before deciding whether or not to award the reconnaissance permit.

H. Prescribed Fee	T 47	章 强			69
An amount of R	, being the pres	scribed non-r	efundable appli	cation fee, mu	st be paid int
the account of the Petroleum	Agency. Proof of	avment mus	t be attached to	the application	n.

FORM L:

APPLICATION FOR A TECHNICAL CO-OPERATION PERMIT IN TERMS OF SECTION 76 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Note that a Technical Co-operation Pe	ermit:
---------------------------------------	--------

- is valid for a period not exceeding one year.
- is not an exclusive right and is not transferable or renewable.
- does not permit any exploration operations.

Applications must be delivered to :

The Chief Executive, Petroleum Agency SA, PO Box 1174,

Parow, 7499.

Tel: 021 938 3500 Fax: 021 938 3520

Email: plu@petroleumagencysa.com

Street address :

Petroleum House, 151 Frans Conradie ave., Parow, 7500.

<u>A.</u>	Details	of A	\ppl	icant

-		State Co. 10 March	
1)	Full Name of		
	Applicant:		
220			para half
2)	Registered Number of Company /	Identify Number of In	idividual ;
2)	Postal Address	FF 138	*
3)		F 100	
	••••••••••••••••••••••••••••••••••	•••••	
W 59	Telephone No	(8 / E)	Fay No.
	Telephone No		ax no
4)	Black Empowerment Shareholding	g / Partners (%)	
٠,			
•••••			
B. AI	rea Applied for & Summary of Planner Hydrocarbon Targeted in Technica		v/oil/gae/CBM\·
''	nydrocarbon rargeted in reclinica	ai co-operation study	y (OII / gas / CBM)
2)	Period Applied for :	# # # #	
-,	, o., o., o., o., o., o., o., o., o., o.		the second second second
		* -	
3) will F	Magisterial District, Farm Name / s ocus :	§ (9	8 U B 8
	•••••••••••••••••		
			•••••
41	Common Warls Coope of Tooksia	-10	
4)	Summary Work Scope of Technica		
			,

		*	v

THE APPLICATION MUST BE ACCOMPANIED BY:

C. Details of the Area Applied for :

- v. Provide a plan, drawn up in such a form and to such a scale as may be required by the Petroleum Agency, and signed and dated by the applicant showing :
 - the area applied for. The corner points should be labelled clockwise A,, B, C etc. The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.
- viii. The plan should show in addition:
 - the name, number and situation on the farm or farms on which the area applied for is situated.
 - surface structures and servitudes.
 - the topography of the immediate vicinity,
 - the size of the area applied for in square kilometres.
 - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by exploration operations.
 - for offshore applications, the Petroleum Agency numerical licence blocks affected should be shown.
- iii. Provide details of any other matter relevant to the application.

D. Details of Planned Technical Co-operation Study:

- i. Specify the minimum proposed work scope and expenditure for the area or areas applied for;
 - Detail the proposed time framework for the proposed Technical Co-operation Study;
 - Provide a budget for the proposed Technical Co-operation Study.

E. Substantiate Financial and Technical Competence:

Provide information to substantiate that the applicant has the technical and financial competence to carry out the planned Technical Co-operation study in a responsible manner.

F. Past Compliance with Provisions of the Act, Regulations, Agreements

Provide a report by an independent third party of the extent of compliance with the requirements of the approved Environmental Management Programme for the preceding Exploration Right period(s).

G. Acknowledge that the Petroleum Agency is entitled to request further information before deciding whether or not to award the reconnaissance permit.

H. Prescribed Fee

An amount of R...... being the prescribed non-refundable application fee, must be paid into the account of the Petroleum Agency. Proof of payment must be attached to the application.

FORM M:

APPLICATION FOR AN EXPLORATION RIGHT IN TERMS OF SECTION 79 AND THE RENEWAL THEREOF IN TERMS OF SECTION 81 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Note th	hat an Exploration Right : - is valid for a period not exceeding three	vears.		
	- is an exclusive right which may be rene	wed and transferred ar	nd encumbered	Lo
NB:	- an environmental management progra	mme must be approve	d before explor	ration operation
may co	ommence.	0. N° 1944 (MACCE - 1554 - 1692) (MACANANA (MACANANA 11 0 1 € 14 € 14 5. 14 6.00 (MACANANA (MAC		
	*		ng _{ti} fi	
Applic	ations must be delivered to:	x.	1000	
	The Chief Executive,	Street address:		
	Petroleum Agency SA,	Petroleum House,	73,	
	PO Box 1174,	151 Frans Conrad	ie avenue.	
	Parow, 7499.	Parow, 7500.		
	Tel: 021 938 3500			4 25 23
	Fax: 021 938 3520	80		
	Email: plu@petroleumagencysa.com		50 18	
11.7298			522 W E	*
A. Det	tails of Applicant		4.0	18
				The second second
1)	Full Name of Applicant			
2)	Registration Number of Company / Identi	ify Number of Individua	1 :	
3)	Postal Address :			

	12			g ear
	Telephone No	F	ax No	,
4)	Black Empowerment Shareholding / Part	ners (%):		***************************************
••••••	••••••••••••••••••••••••••••••••••		•••••	
D Ara	oce Applied for P. Cummons of Dispused O			
D. Ale	eas Applied for & Summary of Planned O	perauons	80 00 to 10	
1)	Hydrocarbon Targeted in Exploration Pro	gramme (oil / gas / CB	M) -	70 -1
	Tydrodalbon Targetod III Exploration 1 To	gramme (on r gas r ob	(*))	
2)	Period Applied for :			
-/	. onour pphou for	**************************	****************	******************
3)	Province / Magisterial District / Farm Nam	ne or Offshore Licence	Block included	in Exploration
Area:	- 10 M			ne menterentant

4)	Summary of Planned Exploration Operation	ons :		* OV 10 00430
5 8				
	1000			
442		7.707M 60.000 H		
		39		
Signati	ure of Applicant	ï	Date	
	The state of the s			
¥.				*

C. Details of the Area Applied for :

THE APPLICATION SHOULD BE ACCOMPANIED BY:

- vi. Provide a plan, drawn up in such a form and to such a scale as may be required by the Petroleum Agency, and signed and dated by the applicant showing:
 - the area applied for. The corner points should be labelled clockwise A,, B, C etc. The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.
- ix. The plan should show in addition:
 - the name, number and situation on the farm or farms on which the area applied for is situated.
 - surface structures and servitudes,
 - the topography of the immediate vicinity,
 - the size of the area applied for in square kilometres.
 - major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.
 - for offshore applications, the Petroleum Agency numerical licence blocks affected should be shown.
- iii. Provide details of any other matter relevant to the application.

D. Details of Planned Exploration Operations:

- i. Specify the minimum proposed operations and expenditure for the block or blocks applied for;
- ii. For seismic surveys, provide a map showing the areas within which 2D and 3D surveys are planned;
 - Detail the proposed time frame for the proposed exploration operations;
 - xi. Provide a budget for the proposed work programme.
 - v. Provide a statement of the anticipated effect which the proposed Exploration Operations may have on the environment and other users of the area.

E. Substantiate Financial and Technical Competence :

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the exploration activities and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

F. Past Compliance with Provisions of the Act

Provide a report by an independent third party of the extent of compliance with the requirements of the approved Environmental Management Programme for the preceding Exploration Right period.

G. Acknowledge that the Petroleum Agency is entitled to request further information before deciding whether or not to award the reconnaissance permit.

H. Prescribed Fee

An amount of R,	being the prescribed	non-refundable ap	plication fee, must be	paid
into the account of the Petroleum A	gency. Proof of payr	ment must be attache	ed to the application.	

FORM N:

APPLICATION FOR A PRODUCTION RIGHT IN TERMS OF SECTION 83 AND THE RENEWAL THEREOF IN TERMS OF SECTION 85 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002)

Note	that a	Production	Right .
INOTE	ulaic	i i ioduction	munit.

- is valid for a period not exceeding three years.
- is renewable.
- is an exclusive right which may be renewed and transferred and encumbered.

NB : an environmental management programme must be approved before production operations may commence.

Applications must be delivered to:

The Chief Executive, Petroleum Agency SA, PO Box 1174,

PO Box 1174, Parow, 7499,

Tel: 021 938 3500 Fax: 021 938 3520

Email: plu@petroleumagencysa.com

Street address: Petroleum House,

151 Frans Conradie ave.,

Parow, 7500.

<u>A.</u> D	etails of Applicant
1)	Full Name of Applicant :
2)	Registration Number of Company / Identify Number of Individual :
3)	Postal Address :
	Telephone NoFax No
4)	Black Empowerment Shareholding / Partners (%):
B. A	reas Applied for & Summary of Planned Operations
1)	Hydrocarbon targeted in Production Operation (oil / gas or CBM):
2) will ta	Province / Magisterial District / Farm Name / Offshore Licence Block in which Operation ake place :
ii.	
3)	Summary of Planned Production Operations :
Siana	ture of Applicant

THE APPLICATION MUST BE ACCOMPANIED BY:

C. Details of the Area Applied for :

Provide a plan, drawn up in such a form and to such a scale as may be required by the Petroleum Agency, and signed and dated by the applicant showing:

the area applied for. The corner points should be labelled clockwise A,, B, C etc.
 The co-ordinates of the corner points are to be tabulated on the plan in latitude / longitude referenced to the WGS84 Hartebeest94 datum. The north point should be shown.

ii The plan should show in addition :

- the name, number and situation on the farm or farms on which the area applied for is situated,
- surface structures and servitudes,
- the topography of the immediate vicinity,
- the size of the area applied for in square kilometres.
- major towns, roads, rail lines, conservation areas, restricted areas, and all features, communities and facilities that could be affected by proposed operations.
- for offshore applications, the Petroleum Agency numerical licence blocks affected should be shown.
- iii. Provide details of any other matter relevant to the application.

D. Details of Planned Production Operations :

- Provide a comprehensive geological report with details of the petroleum resources and reserves as determined by the geological study and feasibility study in respect of the area applied for;
- Provide an independent assessment of the reserves to be exploited under the production right;
- iii. Provide a comprehensive technical and feasibility report with details of the production method, capacity of production and scale of operations, processing and treatment of petroleum, infrastructure requirements as well as marketing arrangements for the sale of the petroleum products;
- iv. Provide a detailed forecast of capital requirements, sale revenues, operating costs, environmental management and remedial costs, environmental management programme financial guarantees, mine safety and health costs and other costs, presented in a cash flow format in respect of the period required for production operations;
- v. Provide an independent evaluation of the proposed development programme.
- vi Note that provision must be made to provide copy of all resulting data, reports, information and samples to the Petroleum Agency SA.

E. Substantiate Financial and Technical Competence:

Provide details with documentary evidence to prove the applicant's technical ability and financial resources that are readily available or how they will be provided for, to enable the applicant to carry out the production activities effectively and to mitigate and rehabilitate relevant environmental impacts satisfactorily.

F. Social and Labour Plan:

Provide full details of the proposed social and labour plan.

G. Past Compliance with Provisions of the Act

Provide a report by an independent third party of the extent of compliance with the requirements of the approved Environmental Management Programme for the preceding Exploration Right period(s).

H. Acknowledge that the Petroleum Agency is entitled to request further information before deciding whether or not to award the reconnaissance permit.

I. Prescribed Fee

An amount of R....., being the prescribed non-refundable application fee, must be paid into the account of the Petroleum Agency. Proof of payment must be attached to the application.

MISCELLANEOUS PROCEDURES

FORM O:

NOTICE OF CONSULTATION IN TERMS OF SECTION 10 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, (ACT NO 28 OF 2002):

A PROSPECTING RIGHT / MINING RIGHT / MINING PERMIT /
RECONNAISSANCE / TECHNICAL CO-OPERATION PERMIT / EXPLORATION
RIGHT / PRODUCTION RIGHT

De pleased to take							
	* H ***	3 3 4 7 7 7				es or or	
has applied for a in terms of			Pe	mit /		Riç	ght
		Si .	-			- 26	題
sectionblocks:	of the Minera	and Petroleu	m Resources D	evelopment /	Act over the fo	ollowing propert	ties /
91	% E				100		
1					-	***************************************	
0							
2		•••••••	******************				•••••
3	•						
4	N.	10					
4	• • • • • • • • • • • • • • • • • • • •		******************			•••••	•••••
(Annex a schedule	of properties	applied for if pe	ecessary				
,	- proportion	app	30000a.y.			3	
forward your object after the expiry of ti	ions in writing nis notice.	to the relevan	t Regional Man	ager within a	period of 14	(fourteen) days	
	# N	F32	£0				
¥5 16				S\$		× 3 5	
The second secon		(51)			er ett		
Dated at			on	the		day	of
	200		34 St		81		
38				**		9	
This notice will expi	re on :		day of		••••••	200 at	t
151100.		- F				38	
		· · · · · · · · · · · · · · · · · · ·					
	E	g to gen	\$\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\				
		V 19		192		19	
Designated officer	······································		.t.				
besignated officer	9 9 * 10 950	16 SE 18	9.21.8			\$7	
					50 (0)		

FORM P:

TRANSFER OR ENCUMBERANCE OF

PERMITS AND RIGHTS:
(in terms of Section 11 of the Mineral and Petroleum Resources Development Act, 2002)

West Control of the C		The second secon		THE STATE OF
# #	e en managina e	April 18		
l,				e
l,	*****		*****************	
of	2°		12 <u>12</u>	(company nama)
01	***************************************			(company name)
the undersigned applicant	Si di Cara di	. I		E
the undersigned applicant	and noider of a.	329	1 81	
proposting right mining ri	aht taahniaal aa anaratis.	normit / ovoloratio	n right / product	tion right
prospecting right, mining ri	grit, tecrifical co-operation	i permit / exploration	n ngni / produci	lion right
140	, riereby apply for the	R G		
1 Consists			10	
1. Cession		_ K		
2. Transfer		a service W	920 8 6	60 W
3. Letting				70 27 36 78
Sub-letting	99			4
Assignment		37	9 N S	
6. Alienation				
		a n . 9		
of the above-men	tioned right.			
The receiver of the right as	co-signatory bereaf unde	rtakes to comply wit	th the provision	s of section 11/2\(a)
(h) (a) (d) and analism 44	(4) -f the Ast	nakes to comply wit	iii ule provision	a or accitor i i (2)(a)
(b), (c), (d) and section 11(4) of the Act.	<u>†0.</u>		1
		2		
3	e _ 1	* 5		0
-	***			75 Table 1
Dated at	,on	theday o	f	200
	10 M	# # # # #	8 8 8	pri s
14	7 IT	30 G	*** ***	23
			4 4	
Signature of Applicant	en e	Signatur	e of the Recei	ver of the Right
500 61 00000 5t			TA BE	3. #
Details of Receiver of Rig	sht		, in	- W
			B 5:	
Name of representative (Pr	rint) ·			
Name of representative (F)				
Name of same				## EX
Name of company	i			
(i)	報 費用	8/2	¥8	
	1			
Address			1.00	141
	natio es		. d. 74	
	nation ex			
	natio es			

FORM Q:

APPLICATION FORM FOR ASSISTANCE TO HISTORICALLY DISADVANTAGED PERSONS IN TERMS OF SECTION 12 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002, (ACT NO. 28 OF 2002)

(The legible persons entitled to apply for assistance in terms of section 13 are those previously been disadvantaged by unfair discrimination. Assistance if offered will be subject to the Minister's terms and conditions).

(A) Full particulars of the applicant:	e e e e e e e e e e e e e e e e e e e
Name and Surname (registration number in case of a juristic person):	TOTAL STATE OF THE
	the second of th
Citizenship/primary listing of a company (if applicable) :	
Identification number:	
Directors/Members name(s) and surname(s):	
1	the afficiency factors in the
3	e i kristinsk sije in de stort
5	
7	
Postal Address (where the main business is carried in a case of a juris	tic person)

Telephone numbers :	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
E- mail address (if any)	
Fax numbers:	Difference of the sti
(B) Insolvent status of the applicant	i de la compania del compania del compania de la compania del compania de la compania de la compania del compania de la compania de la compania de la compania del compania
Five years prior to this application being lodged for assistance in ter estate been placed under sequestration or liquidation by an order of co	
YES/NO	
A 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ere en en
If yes, the following documents must be attached to this application:	
1 Copy of the court order (i.e. Provisional/Final.)	n 1981 y 198

- Rehabilitation court order (in the case of sequestration); or

Any other information that should be disclosed to enable the Minister to determine the solvency of the applicant.

Five years prior to this application being lodged on behalf of a juristic person, did any director/member in his/her capacity as a director/member been involved in a company or close corporation which was placed under judicial management or liquidation by an order of court.

VEQ/N	10	
I EO/IN	\mathbf{U}	

IF yes, the following documents must be lodged together with the application:

- A copy of the court order whether provisional or final (placing the entity under judicial management or liquidation)
- 2. A certified schedule 2 form CM27 copy (companies act, 1973 as amended);or
- 3. A certified schedule 4 CK1 copy (close corporations act, 1984 as amended)

Any other information that will enable the Minister to determine the individual director's/member's specific roles.

(C) Criminal Offences

Two years prior to the lodging of this application, did the applicant(s) or director(s)/ member(s) of a juristic person been convicted for a criminal offence with out an option of a fine.

YES/NO
If yes, briefly state the surrounding circumstances which lead to the commission of the criminal offence.
(If the space provided is insufficient, sworn statements may be used and attached to this application).

(D) Pre - knowledge and technical skills in the Mining and Minerals Industry

This part of the application must be addressed in a separate document under the following headings (discussion per heading is limited to the maximum of half a page):

- Involvement in the mining industry either as employer/employee
- The importance of the Environmental Management Program (EMP)
- The reasons, why an entrepreneurs/company social plan is important to the Development of the local
 economy were mining activities are carried out.

(E) Business case

Under this heading proved, a summarized vision of your business plan and attach your business plan to this application (The nature of the business plan will depend on the type i.e. small, medium or large mining operation the applicant is or intents to carry out).

(F) Financial Disclosure

If the applicant is already operating mining business, the following documentation as certified by the board of directors/members/trustees must be lodged together with the application:

- A recent balance sheet reflecting the profit and loose of the entity.
- The auditor's or accountant's report which must verify the correctness and status of the balance sheet and the audit and accounting books of the entity respectively.
- Pledges/bonds/loan agreements entered into with other financial sources.

Any other information that should be disclosed in order to enable the Minister to determine the financial need or status of the applicant.

If the applicant intents to enter the mining business, the following documentation as certified by a commission of oath must be lodged together with the application (if any):

- · Loan agreement entered into with other financial sources.
- · Short or long term investment in a form of shares, unit trust etc.

· Personal financial contributions.

Any other information that should be disclosed in order to enable the Minister to determine the financial need or status of the applicant.

FORM R:

REFUSAL BY THE MINISTER TO GRANT PERMITS/RIGHTS:

PROSPECTING RIGHT / MINING RIGHT / MINING PERMIT RECONNAISSANCE / TECHNICAL CO-OPERATION PERMIT / EXPLORATION RIGHT / PRODUCTION RIGHT

(in terms of Sections 13, 17, 23, 33, 75, 79,84 of the Mineral and Petroleum Resources Development Act, 2002)

			- 12	771	960000000000000000000000000000000000000	- Designation of the last	Marie appropria	Water-Transfer Transfer	5=10=0 (W)
Арр	lication Number:							5	
Nam	ne of Applicant:								
App	lication for (permit / right	type) :	***************************************						
Mine	eral type:								88
Ove	r farms / offshore petrole	um licence l	block :						
	an and a	80							-
•••••	, i		de 1, 73						
		-				······································			
(app	end list of farm names if ne	ecessary)		10 E	¥2.				
l, the	undersigned,ister of Minerals and Ene	rgy)							
herei	by refuse the permit / right	on the follow	ina arounds				0		
1.		*							
2.		· · · · · · · · · · · · · · · · · · ·						: 	
3.				, A					41
4.		••••••				•••••••			
5.		•••••••				•••••		•••••	(a)
6.			•••••						7.
7.			•••••						
8.	*,*,								10.
9.								•	
10.									
	ti .			-				it.	
Dated	d at	···	on th	ne		day of		200	
		11		r.					
Minis	ster of Minerals and Energ	MV.				7.05			

FORM S:

PERMISSION TO REMOVE AND DISPOSE OF MINERALS AND PETROLEUM PERMIT IN TERMS OF SECTION 20 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002, (ACT NO. 28 OF 2002)

The Minister of Minerals and Energy hereby grant permission for the removal and disposal of the undermentioned mineral(s) as follows:

(5)	Full particulars of the prospecting right holder:
(6)	Name and surname (in case of a juristic person, registration number)
(7)	Identification number of the holder
(8)	The physical address where the holder carries out business
(9)	Class (e.g. A, B, C, D or E)
(10)) Mineral/Ore
(11)	
	T

(MINISTER OF MINERALS AND ENERGY)

FORM T:

THI	E MINER	RAL AN	D PETRO								
28	OF 2002)	AND Y	s a	i javis						
upoi	recom	mendati	Ministe on made	by th	e Mining	and and	Mineral	Develo	pment	Board	hereby
Terr	ns and co	nditions	•	. 67	1 T - 1-11	807			. 18		
The	applicant f	or a min	ing right, sh	all;		- 31 ***			· **		
	18V 4			E				¥ (1)			
1 1 1 1	granted in	n terms s	with section sub-sections	s 21(1)(a) –	(h), exerci	se the ri	ights stipula	ated under	sub-sect	ions 5(a)) – (d);
(b)	renewal o	d to the of a minir	exclusive ring right com	ght as stipo ply with the	ulated und provisions	er sub-s s of sub-	section 23(sections 22	1), on con 2(1) - (3);	dition the	applica	nt for the
(c)	be exem Developn or 24(b) o	nent Boa	m payment ard, whereby	of royaltie	es as per mmendatio	recomi n is bas	mendation sed either o	made by n the prov	the Mindisions of	erals an sub-sect	d Mining ion 24(a)
(d)	after beer	n granted	d a mining ri	ght, registe	r such righ	t in term	s of section	n 23(a).		- 50	
्रिक्षा	1.1.5		10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	· · · · · · · · · · · · · · · · · · ·	4 5/14			8, 1	1.		
Sign	ed at		on	the	day of						8 10
		15 26 Ja	F 50 - 120 2400 - E - 30	# G	W 10 # 400		M 5		,		
	E.							W and a			
(MIN	ISTER OF	MINER	ALS AND E	NERGY)		3 (34)	74 (44) 10 (4)	I :		# #	

FORM U:

APPLICATION IN TERMS OF SECTION 43(2) OF THE MINERAL AND PETROLEU	JM
RESOURCES DEVELOPMENT ACT,2002, (ACT NO 28 OF 2002), TO TRANSFER	
ENVIRONMENTAL LIABILITIES AND RESPONSIBILITIES TO QUALIFIED PERSO	NC

						n eg =			7 - 7 -

older of:		352	110-11	¥8 52	ϵ^{W}	2020			20
older of.	% & 0			X		54 M		100 E	M M (a More)
1.	Reconnaissa								ns at
2. 3.	Prospecting Mining Right				es g			37	
4.	Mining Perm	it No		;		Salle		(4)	98
5.	Reconnaisar	ce Permit N	0	;	W			1919	
6. 7.	Exploration F Production R	Right No	•••••	·····;		6			4
	Froduction	iigiit No		,		* 1	8, 0	83	,
(which	ever is applicab	ole)	Ja.		* 8				
mention	apply for the tr	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	apply for the tr ned permission mme as the cas prescribed.	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention	ned permission mme as the cas	, right or per	mit of the er	vironmental	managem	ent plan o	r environ	mental ma	nageme
mention program may be	ned permission mme as the case prescribed.	n, right or per se may be o	mit of the er	vironmental ibed closure	managem plan, to a p	ent plan o	r environ	mental ma	nageme
mention program may be	ned permission mme as the cas	n, right or per se may be o	mit of the er	vironmental ibed closure	managem plan, to a p	ent plan o	r environ	mental ma	nageme
mention program may be	ned permission mme as the case prescribed.	n, right or per se may be o	mit of the er	vironmental ibed closure	managem plan, to a p	ent plan o	r environ	mental ma	nageme
mention program may be	ned permission mme as the case prescribed.	n, right or per se may be o	mit of the er	vironmental ibed closure	managem plan, to a p	ent plan o	r environ	mental ma	nageme
mention program may be	ned permission mme as the case prescribed.	n, right or per se may be or	mit of the er r any prescri	vironmental ibed closure	managem plan, to a p	ent plan o person wit	r environ	mental ma	nageme

FORM V:

NOTICE FOR SUSPENSION / CANCELLATION:

OF A PROSPECTING RIGHT / MINING RIGHT / MINING PERMIT / MINING RETENTION PERMIT RECONNAISSANCE / TECHNICAL CO-OPERATION PERMIT / EXPLORATION RIGHT / PRODUCTION RIGHT

(in terms of Section 47 of the Mineral and Petroleum Resources Development Act, 2002, (Act No. 28 of 2002)

							\$355 E			
Permit / Rig	ht Number :									******
Name of Ap										
Permit / Rig	ht Type :			(0)				25 20		
Mineral Type	e:	•••••••	, .				F .			
Over Farms	/ Offshore i	Petroleum Li	cence Blo	ck	1 1	-	4.3		4	
50										
The state of the s										
1111 411						305				
	••••••	••••••	••••••		••••••	•••••				
(attach list of	farm names	if necessary	')	635			6			÷ 2
		*					-	**		
I, the undersi	igned,	9 t	18	. 5		70 700		10 Hz	3 32	16
*******************************	(Ministe	er of Minerals	and Energ			•••••••••	<u>.</u>	······································		
hereby suspe		the permit / ri	74	teros	rounds ·			e e e e e e e e e e e e e e e e e e e	90	72
		nducting pros			(8)	producti				- 4- 41
Act;	Holder is co	nducting pros	specing, ii	ming, expic	Jialion of	producti	on open	ations in cor	uravenuo	n to the
2. The	holder has b	oreached the	terms and	conditions	upon whi	ch the p	ermit or	right has be	en grante	d:
3. The	holder is no	t prospecting	, mining, e	xploring or	producing	g the mir	eral or p	etroleum re	source or	otimally;
4. The	holder is ha	s contravene	d the appro	oved enviro	nmental i	manager	nent pro	gramme; or		14
5. The required to be	holder has s e submitted i	submitted ina under the Act	ccurate, ind	correct or m	nisleading	ginforma	ition in c	onnection w	ith any m	atter
The Minister in not be susper	further invite nded / cance	s the holder, elled.	in terms of	sub-sectio	n 47(2)(c), to sho	w cause	why the affo	ected righ	t should
				-	-					
Dated at		•		on the		9 9	day	of	ř.	200
					***************************************		uay (200
Minister of M	linerals and	Energy				10.8	19			

FORM W:

WITHDRAWAL OF SUSPENSION NOTICE IN TERMS OF SECTION 47(5) OF THE ACT:

OF A PROSPECTING RIGHT / MINING RIGHT / MINING PERMIT / MINING RETENTION PERMIT

RECONNAISSANCE / TECHNICAL CO-OPERATION PERMIT / EXPLORATION RIGHT / PRODUCTION RIGHT

(in terms of Section 47(5) of the Mineral and Petroleum Resources Development Act, 2002, (Act No. 28 of 2002)

	Minister hereby gives notice of the withdrawal of the suspens sed in terms of sub-section 47(5) of the Mineral and Petroleu		
(a)	Name and surname of permit / right holder (in case of ju	ristic person, registration number):	•••
(b)	Physical address where the main business is carried out	ıt <u>.</u>	•
(c) :	The district where the property / land / offshore petroleur		
		a first to the first the second of the	the second
	linister therefore finds the holder to have satisfactorily disched holder has furnished compelling reasons for the withdrawal		
Minis	ter of Minerals and Energy	Date	90

FORM X:

NOTICE FOR PROHIBITION OR RESTRICTION OF GRANTING A RIGHT OR PERMIT IN TERMS OF SECTION 49 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002, (ACT NO. 28 OF 2002)

	18					#1		
a)	***************************************		••••••	situated a	ıt			S a
)	÷ * *		- 15 72		ıt			
11.0				175	ıt			
	······	+ 6			ıt			
	·····			+11	t	4.5		The second secon
	8 6 6		16		t			
		***************************************	***************************************	situateu a	A		************	
n t	he following terms and	conditions :				¥0		37 W
			Seem If the	. r. s		500	16	8 8
)						***************************************		
)			•••••••	······································				
			·	***************************************		•		
					8			
	4.	2.1			*****************	***************************************	*************	
fr	om		9					(7A) -2
1113	JIII	10	••••••	sut	ject to further	notice.	55 (85)	
	¥:	19 v	30 W					26
ne	ed at	on the	da	v of	tiki.			2
				,				
		3 5						£0
		19	49					#
	ster of Minerals and En						15	
	ster of Millerals and En	iergy)			92			
	8					111		
							17	
ohi	bition or Restriction to	the granting	of a prosp	ecting righ	t, mining rigi	nt or minin	g permi	t .
			(Section 1)	on 49 Notic	<u>e)</u>			
	*		9		er - 1		30 20	
e N ht,	finister of Minerals and E mining right or mining pe	nergy hereby ermit in, on or	give notice under the fo	of the prohi	bition or restri ties:	iction to the	granting	of a prospe
				situated at			5300594588888	

(e)	
(f)	situated at
	節
On the following terms and conditions:	
(a)	
(b)	
(a)	
(c)(d)	
(d)	
(e)	
(f)	<u> </u>
<i>V</i>	·
As from to	subject to further notice.
	en ^F
g a s ^a	
Signed at on the day	of
	and the second s
(MINISTER OF MINERALS AND ENERGY)	2 X

FORM Y:

	PRODUCTION OPERATIONS AFFECTING EMPLOYMENT PRODUCTION OPERATIONS AFFECTING EMPLOYMENT
of th	holder of a mining <u>or production right</u> (No) hereby gives notice in terms of section 52 49(1) ne Minerals and Petroleum Development Act, of the following negative economic conditions and the social nomic impact thereof.
(a)	Indications of the profitability levels (as required in terms of section 5249(a)
(b)	The percentage of the labour force to be retrenched in a period of 12 months: (as required in terms of sub-section 5249(b).
to m	holder is further required to attach documents that will illustrate the negative market conditions, measures taken inlimize the impact, audited balance sheets (one reflecting the profit and loss 12 months before the negative set conditions, the other reflecting the current profit and loss of the company) and any other documents that all disclosed or may be required to enable the Board to investigate).
Sian	ed atonof
Olgi	
2 Cap	acity Minister of Minerals and Energy
Prof	itability and curtailment of mining operations affecting employment (Section 52 Notice)
and	holder of a mining right (No) hereby give notice in terms of section 52(1) of the Minerals Petroleum Development Act, of the following negative economic conditions and the social nomic impact thereof.
`(c) Indications of the profitability levels (as required in terms of section 52(a))
(d) The percentage of the labour force to be retrenched in a period of 12 months
to m mark	holder is further required to attach documents that will illustrate the negative market conditions, measures taken inimize the impact, audited balance sheets (one reflecting the profit and loss 12 months before the negative set conditions, the other reflecting the current profit and loss of the company) and any other documents that lid be disclosed or may be required to enable the Board to investigate).
Sign	ed atonof

?.....? (Capacity)

FORM Z:

NOTICE BY HOLDER IN TERMS OF SECTION 54(1) OF THE ACT WHI	RE THE
HOLDER OF ANY RIGHT OR PERMIT IN TERMS OF THE ACT CANNO	
COMMENCE WITH OPERATION	

To:T	he Designated Officer	v - k			
				¥ 9	
		***************************************	· · · · · · · · · · · · · · · · · · ·	•••••	
•••••		••••••		••••••	
			1991 		90
I, the	undersigned,			., the holder	
of per	mit or right:-No :				
for	The Congression of the Congressi		(n	nineral/petroleum);	
prosp	y give notice in terms of section 54(1) of the Ac ecting, mining, reconnaissance, exploration or p ing grounds:				
(a)	refusal of entry to the relevant land;				
(b)	unreasonable demands in return for access to	the land;			
(c)	owner/occupier cannot be found to apply for ac (Delete which is not applicable)	cess.	\$ a		1
			10		
DATE	D AT ON THE DAY O		2001.	22	
			-		
01011	ATURE OF HOLDER			ngar a a ^k	

FORM AA:

NOTICE TO LAND OWNER IN TERMS OF SECTION 54(2) OF THE ACT WHERE THE HOLDER OF ANY RIGHT OR PERMIT IN TERMS OF THE ACT CANNOT COMMENCE WITH OPERATION

The L	andowner/Occu	ıpier	a a	* 4	
					·····
	••••			- 4, ,	
• ;*				9 S	
Kindly	/ note that I ha	ive received a Notice of	lated	20 in terms of se	action 54(1) of the
	732 (8)	240	ng permit / reconnaissance pe		and the second of the second o
			e cannot commence or condu		
50	ration	or	production	operations	on
•••••			·	due	to:
1.			8 V V		
ı. 2.			3 7 7		
3.					
4.		6)	······································		
30.5					200
notice	re nereby called	upon to make represe	entations on or before	200 re	garding this
You a	re hereby furthe	er notified of the rights b	peing held in terms of sections	5. 17. 23 and 27(5) of th	e Act of the
above	-mentioned hole	der as per annexure.			7
Take	further notice th	at you are contravening	g the following sections	of the A	ot.
				ž s	13 8
You a			teps in terms of the Act may be	97700 ADA TO 1678	
1.					
2.					
3.					
4.	******************	* *			(2)************************************
5.				·	27 27 5) E-
		*			****
			*	***	
DATE	D AT	ON THE .	DAY OF	20	00
3	El .	20 SE) (i)	a e .	
				22 14	W
Desig	nated Officer	\$0 68	8	網	100
DEGIC	ONAL MANAGE	Б	HOLDED OF	C DIOLET / DEC. ***	72
HEGIC	JNAL MANAGE	n	lie	F RIGHT / PERMIT	
			NO		

FORM BB:

EXPROPRIATION NOTICE IN TERMS OF SECTION 55 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002, (ACT NO 28 OF 2002)

	nergy, hereby gives notice of the expro 5(1) of the Minerals and Petroleum Deve						
(a)	situated at	title deed no					
(b)	situated at	title deed no					
(c)	situated at	title deed no					
(d)	situated at	title deed no					
(e)	situated at	title deed no					
(f)	situated at	title deed no					
Further, the Minister acknowledged the application of section 6, 7 and 9(1) of the Expropriation Act, 1975 (Act No. 63 of 1975), when exercising the powers to expropriate property vested to the Minister by section 52(1).							
Signed aton	of200						
- den - 🕶 ettern som har det med stelle det et en det et en en et en et en		80					
	3 T						
Minister of Minerals and Energy	······································						

FORM CC:

NOTICE IN TERMS OF SECTION 88(1) OF THE ACT REGARDING CONTRAVENTION OF PROVISIONS OF ACT

KINDLY TAKE NOTICE that you as holder of prospecting/mining right, mining/retention permit, reconnaissance permit, exploration right or production right or environmental management programme have contravened or are contravening the following provision/provisions of the Act:

1
2
3
370 Page 1970
4
You are hereby ordered to take immediate steps to rectify the contravention(s) on or before the
200.
<u>OR</u>
You are hereby ardared to append your proposition/which have been also as a few to the standard to the standar
You are hereby ordered to suspend your prospecting/mining /reconnaissance / exploration / production
operations as from the day of
instructions:
d
and the second s
2
3
4
DATED AT ON THE DAY OF
Designated Officer
*
ORIGINAL HEREOF RECEIVED ON THE DAY OF 200.
HOLDER OF THE RIGHT/PERMIT, MANAGER,
OFFICIAL, EMPLOYEE OR AGENT
HOLDER OF RIGHT / PERMIT NO///

FORM DD:

FORM IN TERMS OF SECTION 94 OF MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002 (ACT NO 15 OF 2002)

DEPARTMENT OF MINERALS AND ENERGY AUTHORIZATION IN TERMS OF SECTION 94 OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT	Under section 94 of the Mineral and Petroleum Resources Development Act, 2002, heraby, by virtue of the powers delegated to me, authorize			
Photograph	of the Department of Minerals and Energy to enter without any warrant, at ay time, without prior notice, upon any land or place, including any offshore installation, vehicle, vessel or aircraft, which may be necessary for the proper exercising of the powers or the performance of the duties or functions in the said Act.			
	MINISTER OFFICE STAMP			

FORM EE:

APPLICATION FOR A CLOSURE CERTIFICATE:

(in terms of Section 43(3) of the Mineral and Petroleum Resources Development Act, 2002)

NB: The application is to be made within 180 days of lapse / abandonment / cancellation / cessation /

relinquishment or rehabilition as contemplated in section 40 (3) of the Act. To: The Regional Manager 1. Permit / Right Number: 2. Name of Applicant : 3. Permit / Right Type : 4. Mineral Type: 5. Over Farms / Offshore Petroleum Licence Block (as per attached map) :..... 6. Reason for Application: (abandonment / cancellation / cessation / relinquishment / rehabilition completed) 7. Details of relevant area / operation 8. Application for the Transfer of Environmental liabilities and Responsibilities: If relevant, a separate submission must be appended to the application. 9. The following documents should accompany the application: A copy of the Closure Plan must be appended to the application (a) (b) A copy of the Final Environmental Management Programme Performance Assessment. This may precede or accompany the application for closure. Signature of Applicant Date

FORM FF:

CLOSURE CERTIFICATE:

(in terms of Section 43 of the Mineral and Petroleum Resources Development Act, 2002)

relinquishment or rehab					on / cessation /
Permit / Right Number	·:				
Name of Applicant :					
Permit / Right Type :	Reconnaissance	Permit / Explore	ation Right /	Production Righ	nt
Mineral Type :					E E E
Over Farms / Offshore				127	
				98	8
					+1:
		***************************************	**************		
I, the undersigned,				¥ 3	
	nd Energy) :	••••••	••••		
(a) hereby grant cl	osure to the above	permit / right hold	der. as envis	loned in section	n 43 of the Act :
(b) agree to the tra	ansfer of the followin			1	
as follows:	æ				@ ==
		90			and the same of th
2)					
3)					
4)					
to the following entity:		8	76		
Person / Company Nar	ne :				,
Company Registration	Number / Individua	al's ID Number :			
Address					
	•••••			••••••	

64		*		¥0	
Minister of Minerals an	d Enerav		Date		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,

STANDARD FORMAT OF PERMISSIONS, RIGHTS AND PERMITS IN TERMS OF THE MINERAL AND PETROLEUM RESOURCES DEVELOPMENT ACT, 2002 (ACT NO 28 OF 2002)

PROS		NG RIC	HT			2 3		禁				ű.	
I ET IT I	JEDER\	BE KNO	OWN.			0	6	05			3		
	iLnLD.	DE IQI	J. 1111.		- 12 - 13			8 5					
THAT	on	this	the		day	of				 ,	in ı	the before	yea
				Notary	Public,	duly	sworn	and ad	mitted,	residing	and	practis	ing a
			in the)				Pro	vince, l	Republic of	South A	Africa, ar	nd in the
presenc	e of the	subscrib	ing com	oetent witr	nesses pe	ersonali	ly came a	and appear	ed:	**		2500	
						33 - 33	æ	()‡	8		E		
(1)								.,		Regional			anage
	consoit	v ac tho	duly out	norised re			epartmer	nt of Minera	us and	Energy, a	nd as si	uch in hi	s or ne
	Capacii	y as the	uuiy auii	ionsed re	presenta	ive oi,					3 3		
	\times	4 a	- 8							(9)			
THE MII	NICTED	OF MINI	EDALC	AND ENE	PGV								
THE MI	NO I EN	Or Milla	LINALO	AND ENE	na i						(9	W	
(hereina	fter toge	ther with	his or	her succe	ssors in t	itle and	assigns	referred to	as "the	e Grantor")	, he	or she,	the sai
Appeare	er, being	g duly a	authorise	d thereto	under	and by	y virtue	of a Pow	er of	Attorney o	ranted	to him	or he
by				in his or h	er capac	ity as th	e Directo	or-General	of the L	Department	of Mine	erais and	Energ
of the G	rantor at	clarated	to him e	0	n the	day	OI	nd Engrav	00	_, acting	unde	r and b	y vinu
section	owers u	elegaleu of	the Min	eral and	Petroleu	m Reso	nurces D	nu Energy Developme	nt Act	2002 (Act	No. 15	of 2002).	
00011077	110			oral arra					,,,,,,,,,,				
AND	-				Ö.	50 10					0.0		
(0)	8			19				of o	nd oo	such in his	or hor	oonacit	v ac th
(2)	duly au	thorised	represe	ntative of:				OI, a	nu as:	Such in the	OI HOL	Capacit	y as in
3						8 3	AS				18		39
				1	221 (0.20)				28				
				(No				1			B	-	
				(140				/				탈	
(hereina	fter tog	ether wit	h its su	ccessors	in title a	nd assi	gns refe	rred to as	"the G	Grantee"),	he o	or she, t	he said
Appeare	er, being	duly au	thorised	thereto u	ınder an	d by vir	tue of a	Resolution	of D	irectors of	the <u>Gra</u>	antee pa	issed a
		on	the	day of _	hitad to n	as the	Notone	, which	Power	of Attorney ord in my f	and a	certified	copy c
wnich H	esolutio	n nave tr	nis day t	been exnii	dited to n	ne, the	Notary,	remain illec	or rec	oru in my r	TOLOCO	will the	, willian
11010011													
AND TH	IE APPE	EARERS	DECLA	RED THA	T:					7/4	9: 10	1	
WHERE	AS	A) 45	The S	tate is the	e custod	ian of t	the Natio	onal Minera	al and	Petroleum	Resou	rces,	
AND W	HEREAS	3	The G	arantee h	as applie	ed for a	Prospe	cting <i>Right</i>	in resi	pect of			_ in, o
55.5	19		and u	nder the <u>I</u>	and,	HOVELOUSE	_ ,	60 0 m	: ::::::::::::::::::::::::::::::::::::	2	et .	9.	16
AND W			11- 0			.		Ouantas II		and evalue	ine sisk	t to sec-	nost L
AND WI	HEREAS	5	tne G	rantor nas	agreed	to gran	i to the j	occordance	e sole	and exclus he terms a	nd con	i io pros ditione n	ecorde
			hereu		and unde		anu III 8	Locordance	WILL U	ne tenno a	ind COII	uniona n	Journe

NOW THEREFORE THE $\underline{\textit{GRANTOR}}$ GRANTS A PROSPECTING RIGHT SUBJECT TO THE FOLLOWING TERMS AND CONDITIONS:

1. Definitions

In this <u>prospecting right</u> the clause headings are inserted for reference purposes only and shall not govern or affect the meaning or interpretation thereof. The following words and expressions shall have the meaning assigned to them:

- 1.1. "the Ac"t shall mean the <u>Mineral and Petroleum Resources Development Act, 2002 (Act No. 15 of 2002)</u> or any subsequent law which amends or substitutes the Act;
- 1.2. "the EMP" shall mean the environmental management plan referred to in section of the Act;
- 1.3. "the Grantor" shall mean the Government of the Republic of South Africa in its Department of Minerals and Energy and shall include its successors and/or assigns, and wherever appropriate (in particular where the provisions of this <u>prospecting right</u> require the prior consent of the Grantor) the references <u>herein</u> to the Grantor shall be read as meaning the Minister or any person duly authorised by the Minister to act in the Minister's place and stead;
- 1.4. "the <u>Grantee"</u>, shall mean the <u>holder of a prospecting right</u> and shall include the administrators or assigns or successors in title of the <u>Grantee</u> or any syndicate or company or juristic or natural person which in any manner takes over or otherwise acquires the rights of the <u>Grantee</u> under this <u>right</u>.
- 1.5. "the Minister", shall mean the responsible Minister of State referred to in the Act, or any official of the Department of Minerals and Energy to whom the Minister's authority may be delegated from time to time;
- 1.6. "mineral", shall mean (the mineral(s) concerned to be specified at the time of negotiations with the <u>Grantee</u>);
- 1.7. "the Prospecting Area", shall mean the <u>land</u> referred to in clause 1 <u>hereof</u>, excluding those portions relinquished or abandoned from time to time;
- 1.8. "the Prospecting Work Programme", shall mean the Grantee's written recordal of the manner in which he / she / it intends to carry out the prospecting operations in the Prospecting Area, and shall include all amendments and supplementary submissions relating to the prospecting programme which may be made by the Grantee and approved in writing by the Grantor from time to time, all of which amendments and supplementary submissions shall form part of this prospecting right as Annexure _____ attached hereto;
- 1.9. "the <u>Manager</u>", shall mean the <u>Regional Manager</u> for the _____ Region of the Department of Minerals and Energy, appointed in terms of the Act;
- 1.10. "State", shall mean the Republic of South Africa;
- 1.11. "effective date", shall mean the date in terms of section 17(5) of the Act.

		TAI	BLE OF CONTENTS	100		X
ŧ	Head	ling			Clause	
	Prea	mble		18		
	Defi	nition		10.		2. 8
	Des	cription of Prospecting Area		30	1	
	Pros	pecting Rights		12	2	
100	Com	mencement, Duration and Renev	val		4	
20	Bros	pecting Fees			5	
			he Dreenseting Area		7	
		ndonment or Relinquishment of t				
		rity of Tenure and Optimal Explo			8	
8 1		nination of the Prospecting Area			9	
	Rep	orts to be Furnished			10	
	Kee	oing of Records			11	
10	Pros	pecting <i>Work</i> Programme			12	
		mnity			13	
£ .		th and Safety		9	14	
		sion, Transfer, Alienation			15	
					16	
		ndments				
- 2		t and Several Liability			17	
		er or Lenience			18	
	Arbi	tration			19	
	Envi	ronmental Protection	© (6)		20	
	Cos		31 30 39		21	E 33
		erability			23	25
		icilia citandi et executandi			24	
	Don	icina citanui et excoutanui			(0.0)	
		9 8				
4						
1.	Descri	tion of Prospecting Area				
The						
		g area shall comprise the following:				8
Certain:		- X				8
Situate:						
Measur	ing:					
		,1				4
which p	rospect	ng area is more fully described in t	he attached sketch plan/di	iagram, mar	ked Annexure	
			29		* MA	
2.	Prospe	cting Rights				
	971		8 B			
39	2.1.	The Grantor grants to the Grante	e, the sole and exclusive r	ight to prosi	pect fori	in, on and under
		the Prospecting Area;				835 St. 32(4)31
-	W. 0	the ricopeoting rica,				
19	0.0	this Prospecting Right shall cons	titute the Minister's written	annroval	and antitles the	e Grantee to all
	2.2.	this Prospecting <u>right</u> shall cons	titute trie Minister's Writter	approvar a	too to proceed	e Grantee to an
		the rights, and obligations in t	erms of the Act to enable	e the <u>Gran</u>	tee to brospec	and search for
		in, on and under the Pros	pecting Area.			
10		*	- 14			
					- 2	
3.	Comm	encement, Duration and Renewal				
		77	W			
	3.1.	This Prospecting Right will comr	nence on the effective d	ate and, un	less cancelled	or suspended
		earlier under section 47 of the A	ct will continue in force for	a period of	from the	effective date.
		54.1101 41140. <u>555.5101. 17 57 515 71</u>				
	3.2.	The Grantee shall within 90 da	ve reckaned from the of	fective date	e commence v	vith Prospecting
	3.2.	One with the Decement of Ar	vs reckoned from the en	rective date	enerations so	ntinuously and
(6)		Operations in the Prospecting Ar				
		actively in a manner and on a so		ce with the i	Prospecting w	ork Programme,
		and the approved EMP to the sat	staction of the Grantor.			
		a				
8	3.3.	Subject to section 18 this Prosp	ecting <i>Right</i> may be rene	wed once at	the request of	the Grantee for
	1000ER	a further period of not longer th	an years: and ti	he renewal	of this prost	ecting right is
		subject to the Grantee complying	na duly with section 19	of the Act -		
		CANICOL TO THE CHAINES COMPLY	IN WALL THAT DOUGHT IO	. III PAGE		

3.3.1.

3.3.2. the Grantee has complied with the Prospecting Work Programme referred to in sub-clause 9.2 or any amendment or extension thereof, and submits an extension thereon in respect of each renewal period together with its request for the renewal;

the *Grantee* has complied with the provisions of sub-clause 9.1 below.

- 3.3.3. the <u>Grantee</u> has complied with the approved EMP and has ensured the continued relevancy of the EMP on an updated version thereof, as well as the financial guarantee for rehabilitation; and
- 3.3.4. the <u>Grantee</u> submits to the Grantor a written application <u>on the prescribed form</u> for <u>the</u> renewal at least <u>90 days</u> prior to the expiry date of any period.

4. Prospecting Fees

4.1.3.

4.1.	In consideration for the <u>prospecting</u> rights granted to the <u>Grantee</u> under this Prospecting <u>Right</u> , the <u>Grantee</u> shall pay to the Grantor at the Office of the <u>Manager</u> , or to such other person and/or at such place as the Grantor may from time to time by written notice to the <u>Grantee</u> , elect, as from the <u>effective date</u> :	
	4.1.1.	an annual prospecting fee of R RAND) per hectare of the Prospecting Area payable yearly in advance, which shall increase annually on (ANNIVERSARY DATE) of each year by R (RAND) per hectare
æ	4.1.2.	The <u>Grantee</u> shall, in addition, pay to the Grantor a share of of the value of any minerals found, removed and disposed <u>off</u> in the course of prospecting operations on the Prospecting Area; payment of the said share of shall be made within 30 (thirty) days after the end of every quarter and shall be accompanied by a certificate or other document acceptable to the <u>Manager</u> , reflecting particulars of the value of minerals found, removed and disposed of on the Prospecting Area during the quarter in question; such quarterly payments shall be based:
e a		 on the selling price of the minerals (if applicable); in the case of minerals not sold, on the estimated value thereon; and after the subsequent sale (if applicable) of any minerals in respect of which payment has already been made on the estimated value thereof, on the difference between the estimated value and selling price

- 4.2. The prospecting fees referred to in sub-clause 4.1 shall be paid as follows:
 - 4.2.1. prospecting fees payable in respect of the first year of the prospecting period shall be paid to the <u>Manager</u> within fourteen (14) days following the date of <u>registration</u> of the prospecting <u>right in the name of the Grantee</u>; and

In the event of the prospecting period being extended in accordance with the provisions of

clause 3.3 above, R____per hectare of the Prospecting Area for the _____ year of the

- 4.2.2. prospecting fees payable in respect of each subsequent year of the prospecting period shall be paid within thirty (30) days following the date of commencement of each succeeding year of the prospecting period.
- 4.3. If the prospecting fees referred to in sub-clause 4.1 are not paid punctually on the due dates for payment, the <u>Grantee</u> shall be in <u>mora debitoris</u> and shall pay interest thereon at the rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act 1 of 1999) reckoned from the due date for payment <u>up to and including the</u> date <u>on which</u> payment <u>takes place</u>.

5. Abandonment or Relinquishment of the Prospecting Area

prospecting period.

- 5.1. Subject to the approval by the Grantor, the <u>Grantee</u> shall annually on the anniversary date of this Prospecting <u>Right</u>, by giving the Grantor not less than three (3) months prior notice in writing to that effect, abandon or relinquish a portion or portions of the Prospecting Area <u>if necessary</u>.
- 5.2. In abandoning or relinquishing the Prospecting Area or any portion thereof, the <u>Grantee</u> will from the date of such abandonment, be absolved from the obligations and liabilities as set out in the <u>prospecting</u> work programme in respect of the abandoned area, but not be absolved from those obligations or liabilities which may have accrued up to the date of such abandonment under the terms and conditions of this Prospecting <u>Right</u>, the <u>prospecting</u> work programme, the EMP and the Act.
- 5.3. With effect from the date the <u>Grantee</u> has so abandoned or relinquished <u>a</u> portion or portions of the Prospecting Area, the Grantor shall be entitled to grant any prospecting <u>rights</u> or mining rights or any <u>right or permit</u> referred to in the Act in, on, or under the portion or portions, so abandoned or relinquished, to any persons or persons.

5.4. Upon abandonment or relinquishment of the Prospecting Area or any portion thereof, the <u>Grantee</u> will furnish the Grantor with all prospecting results and/or information, as well as its general evaluation of, the geological, geophysical and borehole data in respect of such abandoned area insofar as it applies to _____ or any other mineral or minerals referred to in clause 8 such results and/or information may be made available to third parties.

6. Examination of the Prospecting Area

The Minister and any person or persons designated by him/her shall have the right to carry out or cause to be carried out any examination or inspection which may be deemed necessary <u>on</u> the Prospecting Area.

7. Reports to be Furnished

- 7.1. Within thirty (30) days after the end of each quarter, the <u>Grantee</u> shall furnish to the <u>Manager</u>, a report giving an account of the progress of its operations in the Prospecting Area <u>of which proper records</u> are <u>kept at its registered address or place of business</u>. Such returns must contain:
 - 7.1.1. a statement and sketch plan of the areas in which prospecting <u>operations</u> has been carried out;
 - 7.1.2. the number of boreholes drilled <u>as well as borehole core data and core-log data</u>; prospecting method applied, including but not limited to trenching, pits.
 - 7.1.3. a statement of the depth drilled and formations intersected and geological information gained from cores and/or cuttings in respect of each hole;
 - 7.1.4. a statement of any occurrences of _____ and other minerals of potential value encountered in the course of the said operations;
 - a statement of financial expenditure, excluding salaries and wages, incurred on prospecting operations;
 - 7.1.6. the co-ordinates of each hole location and the collar elevation of each hole in the LO system; and
 - 7.1.7. a statement reflecting rehabilitation work completed and of the rehabilitation work uncompleted and still to be done.
- 7.2. Within three (3) months after the end of each year in which this Prospecting <u>Right</u> is in force the <u>Grantee</u> shall furnish the Grantor with a full report together with plans, of the prospecting operations carried out by it in the Prospecting Area during that year or the period prior to such expiration or termination as the case may be, together with a detailed statement of prospecting expenditures incurred during such year.
- 7.3. The <u>Grantee</u> shall furnish the <u>Manager</u> with such other information, maps and plans, as to the progress of its operations in the Prospecting Area as the Grantor may from time to time require and which the <u>Grantee</u> can reasonably make available.
- 7.4. Within three (3) months after the termination and/or cancellation and/or abandonment of this Prospecting *Right*, the *Grantee* shall furnish the Grantor with all results of and its general evaluation of the geological, geophysical and borehole data relating to the Prospecting Area as it applies to _____ or any other mineral or minerals under sub-clause 8.2.
- 7.5. All information furnished by the <u>Grantee</u> to the Grantor in accordance with this clause and clause <u>8</u> below, may be made available for use by third parties on expiry of the Prospecting Period <u>in terms of the prospecting right</u>, termination of the Prospecting <u>Right</u> or voluntary abandonment <u>thereof</u> by the <u>Grantee</u> provided that the Grantor shall during the Prospecting Period hold as confidential all information supplied to it under this clause <u>as well as in terms of section 30 of the Act</u>.

8. Keeping of Records

- 8.1. The <u>Grantee</u> must keep, in a form as may be from time to time reasonably approved by the Grantor, current and accurate records of the drilling, deepening, plugging or abandonment of all boreholes and of any alterations in the casing thereof. Such records must contain particulars of the following matters:
 - 8.1.1. the site of and number assigned to every borehole;
 - 8.1.2. the subsoil and strata through which the borehole was drilled:
 - 8.1.3. the casing inserted in any borehole and any alterations to such casing;
 - 8.1.4. any ____ and/or other minerals of potential value encountered in any borehole;
 - 8.1.5. detailed results of all assay tests; and
 - 8.1.6. such other records as the Grantor may from time to time direct and which the *Grantee* can reasonably keep.
- 8.2. The <u>Grantee</u> must keep within the Republic of South Africa current and accurate geological, geophysical and drilling plans and maps relating to the Prospecting Area as may be necessary to preserve all information which the <u>Grantee</u> has about the geology of the Prospecting Area.
- 8.3. The <u>Grantee</u> shall deliver to the Grantor at the completion of each phase of the Prospecting <u>Work</u> Programme two copies of the said records, plans and maps referred to in sub-clauses 11.1 and 11.2.

9. Prospecting Work Programme

The Grantee must, to the satisfaction of the Grantor:

- over the period of this Prospecting <u>Right</u>, initiate and thereafter actively pursue operations designed to progressively prospect substantially over the Prospecting Area; and
- 9.2. carry out and meet the commitments set out in the Prospecting <u>Work</u> Programme annexed hereto marked **Annexure____**; <u>and</u>
- 9.3. it is recorded that the time sequences as set out in the Prospecting <u>Work</u> Programme will commence on the <u>effective date</u>.
- 9.4. Any revisions or amendments to the Prospecting <u>Work</u> Programme and budget must be submitted with the reasons therefor to <u>the Minister</u> for his/her written approval.

10. Indemnity

10.1. The <u>Grantee</u>, its successors and/or assigns, during the tenure of this Prospecting <u>Right</u> will take all such necessary and reasonable steps and do all such acts, matters and things and carry out their Prospecting Operations in such a manner as will adequately safeguard and protect the environment, Prospecting Area and any person or persons using or entitled to use the surface of the Prospecting Area from any damage caused by or through or in consequence of the exercise by the <u>Grantee</u> of its aforesaid right to prospect under this Prospecting <u>Right</u>, of any activity or activities associated with the <u>Grantee's</u> operations in the Prospecting Area, and will in so far as there is a legal liability, compensate such person or persons for any damage or losses, including but not limited to damage to the surface, to any crops or improvements, which such person or persons may suffer as a result of, arising from or in connection with the exercise of his/her rights under this Prospecting <u>Right</u> or of any act or omission in connection therewith.

The Grantor <u>is not or</u> will not be deemed to be a partner in any Prospecting Operations carried out in terms of this Prospecting <u>Right</u>. The Grantor will give notice to the <u>Grantee</u> and its aforesaid successors and/or assigns of any claim or occurrence in respect of which it proposes to invoke the provisions of this indemnity and will give the <u>Grantee</u> and its aforesaid successors and/or assigns the fullest opportunity to investigate, repudiate or otherwise deal with any matter that may arise from such claim. Provided always that nothing herein contained will be deemed to render the <u>Grantee</u> or its aforesaid successors and/or assigns liable for, or to indemnify the Grantor and/or its servants and employees against, any compensation or damages for or with respect to injuries or damage to persons or Prospecting Area directly resulting from any delict of the Grantor and/or its servants and employees during the currency of this Prospecting <u>Right</u>, or for or in respect of any claims, demands, proceedings, damages, costs, charges and expenses in respect thereof or in relation thereto.

10.2. The Grantor will hold harmless and indemnify the <u>Grantee</u> and its aforesaid successors and/or assigns from and against all claims, demands, proceedings, damages, costs, charges and expenses in respect of matters referred to in the proviso of sub-clause 13.1 of this clause.

11. Health and Safety

: i :

. · 340

The <u>Grantee</u> and its successors and/or assigns will and do hereby undertake and bind themselves during the currency of this Prospecting <u>Right</u> to observe and comply with all relevant provisions of the Mine Health and Safety Act, 29 1996 and regulations in terms thereof, or any subsequent law which amends or substitutes the Mine Health and Safety Act, 1996.

12. Cession, Transfer, Alienation

- 12.1. The <u>Grantee</u> shall not, without the Grantor's prior written consent be entitled to either wholly or in part <u>transfer or encumber this prospecting right in terms of section 11 of the Act</u>.
- 12.2. Unless the <u>Grantee</u> has obtained the prior written consent of the Grantor on such terms and conditions as the Grantor may determine (but subject always to the provisions of clause 12.3), the <u>Grantee</u> shall not:
 - 12.2.1. lease the rights granted to the Grantee in terms of this Prospecting Right;
 - 12.2.2. allow any other person to prospect for ---- in the Prospecting Area for such other person's own benefit and account;
 - 12.2.3. allow any other person to carry out prospecting operations for ---- in the Prospecting Area and to retain the whole or any part of the income or profits from the prospecting operations.
- 12.3. The <u>Grantee</u> may from time to time appoint one or more independent contractors to carry out any work in relation to prospecting operations, but the <u>Grantee</u> shall always remain liable for the compliance and observation of the provisions of the Act, the EMP and of the terms and conditions of this Prospecting <u>Right</u>.

13. Joint and Several Liability

The parties forming the <u>Grantee</u> are jointly and severally liable to the Grantor for all obligations of the <u>Grantee</u>, in respect of this Prospecting <u>Right</u> and all its Annexures.

14. Waiver or Lenience

No waiver of any of the Grantor's rights and no lenience granted or permitted by the Minister or the <u>Manager</u> from time to time in a form which does not constitute a formal amendment of this Prospecting <u>Right</u> shall prejudice the Minister's or <u>the Manager's</u> rights under this <u>right</u> or the Act.

15. Arbitration

If any difference or dispute arise between the parties to this Prospecting <u>Right</u> to the intention and meaning of any of the terms and conditions <u>thereof</u>, or any matter arising from this Prospecting <u>Right</u>, such difference or dispute may, after the parties have exhausted all possible attempts to resolve the dispute or difference, be referred by mutual agreement between the parties for determination by an arbitrator in terms of the Arbitration Act 42, 1965, or any subsequent law which amends or substitutes the Arbitration Act.

16. Environmental Protection

- 16.1. The <u>Grantee</u> hereby undertakes and binds itself to conduct all Prospecting Operations in accordance with the requirements of the Act and any other legislation in a manner that would facilitate the protection and conservation of the natural resources of the Republic of South Africa and of the environment in general;
- 16.2. Subject to the provisions of the Act, the <u>Grantee</u> shall on the expiration or termination of this Prospecting <u>Right</u> or on relinquishment of part of the Prospecting Area or on the voluntary abandonment of the Prospecting Area:
 - 16.2.1. remove all equipment and installations from such relinquished or abandoned Prospecting Area; and

perform all necessary site rehabilitation and shall take all other actions necessary to minimise hazards to human or other life or to the Prospecting Area(s) of others or the environment in general for time to come.

17. Costs

All costs flowing from this Prospecting Right direct or incidental, and cost of notarial execution and registration. deregistration (in cases or abandonment or relinquishment) including transfer duty, Value Added Tax at the prevailing rate and stamp duty, shall be paid by the Grantee.

18. Severabilty

Notwithstanding anything to the contrary, any provision of this Right which contravenes the Act or which is otherwise ultra vires, void, voidable, or unenforceable, shall be severable from the rest of this Right, such rest thus being and remaining of full force, effect and enforceability.

19. Domicilia citandi et executandi

19.1. The parties hereto choose the undermentioned addresses as their domicilia citandi et executandi for

	the purposes of writing and post- Grantor at:	serving any ed under p	y notice in t ere-paid reg	terms of the istered cov	is <u>right</u> a ver addre	and any no ssed to t	otice shal	l be valid or or deli	lly given if in vered to the
								*	
RIIGINE	SS ADDRESS:	- N			951	#		8 .	i)
DOSINE	33 ADDRESS.	707				2 22 AM - 22		¥ .	- 4
									90.00
POSTAL	ADDRESS:				•		70)
, GOTAL	ADDITESS.					*		33	
*	.						4		
and to th	e <u>Grantee</u> at:							20 8	
BUSINE	SS ADDRESS:						85 (6)		*
			-		1				
POSTAL	. ADDRESS:				E 8				
			2 0			ia _a	37	E &	
Thus done and significant undersigned witne	immediately notify remain in force.			* * *		14			
undersigned withe	esses:								
AS WITNESSES:	F6					4.		F- 1	
AO WITHESSES.					*				2
1		7 9	ş . F	ur: Ag	E 80	6	N _p		+ 30
2	*						*	9.1	
		for and on b	ehalf of the	Grantor	••••				
	(0)		CAL SHIP NEW YORK	V.			3 40		
Thus done and s witnesses:	igned at		on the	day d	of 200	on the	presence	of the u	undersigned
	6					10	28		
AS WITNESS:	*				38				
1									
	() ==						(4)	80	8.
2									

for and on behalf of the Grantee

FORM HH: MINING RIGHT

LET IT HEREBY BE MA	ADE KNOWN:			10
7 5	¥			8 31 29
THAT on this the	day of	in the year		, before me,
Notary by lawful authori	ty duly sworn and admitted, resi	iding and practising at	, in the	_ Province of South
Africa, and in the prese	nce of the subscribing competer	nt witnesses, personally car	me and appeared_	in his or
her capacity as the Re	egional Manager of the	Region of the Depar	tment of Minerals	and Energy of the
Republic of South Africa	a, he or she being duly authorise	ed thereto by	*	
4			£(== ==	- 번 설 - 번 작성
THE MINISTER OF MIL	NERALS AND ENERGY	¥		
		e a		
(hereafter together with	<u>his or her</u> successors in title ar	nd assigns called " <u>the Grad</u>	ntor")	
under a Power of Attorn	ey granted to him or her by	in his	or her capacity as	Director-General of
the Department of Mine	rals and Energy him or her bein	g <u>duly</u> authorised thereto l	by virtue of powers	delegated to him or
her by the Minister of M	linerals and Energy of the Repul	blic of South Africa (hereaf	ter together with h	is or her successors
in office and assigns)	on	in terms of section	of the Mine	ral and Petroleum
Resources Developm	ent Act, 2002 (Act No. 15 of 20	002)	51	2 2 29
i i			* *	
and also personally can	ne and appeared in his	or her capacity as a		
3, 8	V)	100	4.20	
125.07	Re	gistration No		
# SEC. 20			5.54	
(hereafter together with	is successors in title and assign	ns called)	16 28	32 W.C
			2	
he or she being duly au	thorised thereto by a	_ of the <i>Grantee</i> , which P	ower of Attorney a	nd a certified extract
from the minutes of the	meeting of the Directors of the	Grantee evidencing the abo	ove-mentioned res	olution have this day
been examined by me t	he Notary and now remain filed	in my Protocol,	; ==	
Description				
AND THE APPEARERS	DECLARED THAT	7	¥ (#	g 28
		*C_	7	
WHEREAS	The State is the custodian	of the National Mineral a	nd Petroleum Res	sources.
40 20 20		* s	19	- M
AND WHEREAS	the Grantee has conducted	prospecting operations for	on the prop	erty and wishes to
650 0	acquire a mining right in re	espect of the rights to	in, on and unde	r the <i>land</i> ,
				785 ⁽³ (2)
AND WHEREAS	the Grantor has agreed to	grant to the Grantee the	sole and exclusi	ve right to mine for
x = 2 = x = x	on the <i>land</i> in acc	cordance with the terms a	nd conditions reco	rded in this <u>mining</u>
\$) ((e) d	<u>right,</u>			

NOW THEREFORE THE GRANTOR GRANTS A MINING RIGHT SUBJECT TO THE FOLLOWING TERMS AND CONDITIONS:

	0.110		2002/07/2005	an emilian con-	
-			Def	-	
		9.5	LJER		ons

In this	mining right the following words and expressions shall have the meanings assigned to them:
1.1.	"the Act" shall mean the Mineral and Petroleum Resources Development Act, 2002 (Act 15 of 2002) or any subsequent law which amends or substitutes the Act.
1.2.	"mineral" shall mean
1.3.	"Business Plan" shall mean the <u>Grantee's</u> written recordal of the manner in which <u>he/she or it</u> intends to facilitate, implement and control the Mining <u>Work</u> Programme on a profitable basis, including, without limitation, the financing and financial management of operations, estimated mining expenditures, financial provision for rehabilitation under the Act, the making of provision for the payment of any taxes and the discharge of any other fiscal responsibilities and, in this regard, all amendments to and supplementary submissions on the Business Plan which may be made by the <u>Grantee</u> from time to time shall be deemed to be included in this expression.
1.4.	"the EMP" shall mean the environmental management programme referred to in the Act.
1.5.	"the <u>Grantor</u> " shall mean the Government of the Republic of South Africa in its Department of Minerals and Energy and shall include the successors in title and assigns of the <u>Grantor</u> , and wherever appropriate (in particular where the provisions of this <u>mining right</u> require the prior consent of the <u>Grantor</u>) the references in this <u>mining right</u> to the <u>Grantor</u> shall be as meaning the Minister or any person duly authorised by the Minister to act in the Minister's place and stead.
1.6.	"the <u>Grantee</u> " shall mean (Registration No and shall include the assigns or successors in title of the <u>Grantee</u> or any syndicate or juristic or natural person which in any manner takes over or otherwise acquires the rights of the <u>Grantee</u> under this <u>Mining Right</u> .
1.7.	"mine" when used shall bear the meaning assigned thereto in the Act.
1.8.	"the Mining Right Area" shall mean the portion or portions of the land referred to in clause 1
1.9.	"Mining <u>Work Programme</u> " shall mean the <u>Grantee's</u> written recordal of the manner in which the <u>Grantee</u> intends to mine, recover and win minerals in, on and under the <u>Mining Right</u> area, including without limitation, the mining methods to be employed, the machinery to be used, demarcations of where mining operations are to be conducted, demarcations of buildings, storerooms and conveyors,
	the manner of processing ore for the purposes of recovering and the manner in which the <u>Grantee</u> will comply with all regulations and legislation relevant to such mining activities and, in this regard, all amendments to and supplementary submissions on the Mining <u>Work</u> Programme which may be made by the <u>Grantee</u> from time to time shall be deemed to be included <u>herein</u> .

- 1.10. PPI' shall mean the figure representing the weighted average for all items of the Producer Price Index for mining and quarrying as reflected in the official Producer Price Index published for the relevant month by the Government's Central Statistical Service or, in the event of any such Producer Price Index being discontinued at any time, the weighted average of such other similar statistical index published from time to time by the Government (or other recognised body processing statistical information).
- 1.11. "initial period" shall mean the period referred to in clause 3.1.
- 1.12. "mining right period" shall mean the period of time referred to in clause 3 together with all periods for which this mining right may be renewed or extended under the provisions of clause 4.
- 1.13. "financial year" shall mean a complete financial year of the <u>Grantee</u> which, as the time of conclusion of this <u>mining right</u>, commences on ___/__/__ and ends on ___/__/_.
- 1.14. "<u>Manager</u>" shall mean the <u>Regional Manager</u> for the _____ Region of the Department of Minerals and Energy.
- 1.15. "Effective date", shall mean the date in terms of section 23(5) of the Act

TABLE OF CONTENTS

Heading	fix e	80	4 8	Clau	se
Preamble			9 - 200 C		
Definition <u>s</u>	\$9		385 88		3 K K
Description of the Mining Right Area	a	<i>8</i> =	02. N	1	- Ea
Grant of Mining Right				2	
Duration	14			3	
Renewal of Mining Right			# 5 # 54	4	
Consideration		11		- 5	
Restrictions and Obligations impos	ed on the <u>c</u>	Grantee		6	979
No Mortgage, Cession, Assignment	of Sub-lett	ing		7	20
Protection of Boreholes, Shafts, etc				9	- 1 0
Protection of Property and Persons				10	9 55
Inspection of Mining Right Area	es "Teste" ha	8	* *	11	# B+2
Cessation and termination of Mining	a Right	34 8 31		12	
Records and Returns				13	2 V 800
Payment of Compensation		(⊆ 100 (⊒ 90)		14	
Contravention	A			15	
Health and Safety		30 30		17	
Arbitration		E (6)		18	# VOT 19 (2)
Joint and Several Liability	B4	# 29	# 15 M	19	6
Domicilia citandi et executandi	10		N 22 222	21	. 00
Cost				22	¥.

1. Description of the Mining Right Area

The <i>land</i>	shall comprise the following:
Certain:	
Situate:	
Measuring	
which Mir	ing Right Area is more fully described in the attached Diagram/sketch plan marked Annexure
2. (irant of <u>Mining Right</u>
r V	the <u>Grantor</u> hereby grants to the <u>Grantee</u> and the <u>Grantee</u> accepts the sole and exclusive right to mine, and ecoverin, on and under the <u>Mining Right</u> Area for the <u>Grantee's</u> own benefit and account, and to deal ith, remove and sell or otherwise dispose of, subject to the terms and conditions of this <u>mining right</u> , see EMP and other provisions of the Act.
2	.1. Grants to the <u>Grantee</u> the consent referred to in section of the Act to mine for in, on, or under the <u>Mining Right</u> Area on the <u>Grantee's</u> own account and to dispose thereof.
3. E	uration
3	.1. This <u>mining right</u> shall commence on the effective date and shall thereafter endure for a period of
	years, subject to the provisions of clauses 4, 12 and 15.

4. Renewal of Mining Right

- 4.1. Any application for renewal shall be submitted in writing to the <u>Grantor</u> not later than <u>90 (ninety) days</u> prior to the date of expiry of the initial period.
- 4.2. The <u>Grantee</u> shall simultaneously with the application for renewal, submit a Business Plan and a Mining <u>Work</u> Programme indicating that mining and rehabilitation work will be conducted in accordance with the EMP or an amended version thereof during such renewal period.
- 4.3. Any renewal shall be subject to the approval of the <u>Grantor</u> in the <u>Grantor's</u> discretion, which approval may be made subject to such terms and conditions including environmental management requirements as the <u>Grantor</u> may in the <u>Grantor's</u> discretion deem necessary, it being recorded however that the latter discretion shall not be exercised in an unreasonable manner.

5. Consideration

5.1. In consideration for the rights granted to the <u>Grantee</u> under this <u>mining right</u>, the <u>Grantee</u> shall pay to the <u>Grantor</u> throughout the duration of this <u>mining right</u> the royalty referred to in clause 5.2, subject always to the minimum royalty provided for in clause 5.3.

- 5.2. An annual royalty equal to whichever of the following is the lesser shall be paid by the <u>Grantee</u> to the <u>Grantor</u>.
- 5.3. Notwithstanding the provisions of clause 5.2, a minimum annual royalty calculated in accordance with the following provisions shall be paid by the *Grantee* to the *Grantor*:
- 5.4. The balance (if any) of any royalty payable in terms of clause ____ remaining after application of the set-off referred to in clause ____ shall be paid annually in arrears within ____ days following the date of finalisation of the audited annual financial statements of the <u>Grantee</u> for the relevant year.
- 5.5. If the <u>Grantee</u> changes the <u>Grantee's</u> financial year during the currency of this <u>mining right period</u>, appropriate adjustments to the calculations provided for in clause 5 shall be made in order to ensure that neither party is prejudiced in consequence of such change.
- 5.6. If the royalties referred to in clauses 5.2 and 5.3 are not punctually paid on the dates on which they become payable, the <u>Grantee</u> shall be in mora debitoris and interest on all arrears amounts calculated at the uniform rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act 1 of 1999) and reckoned from the due date <u>for payment up to and including the date on which payment takes place</u> by the <u>Grantee</u> to the <u>Grantor</u>.
- 5.7. All payments in terms of this clause 5 shall be made by the <u>Grantee</u> to the <u>Grantor</u> at the offices of the <u>Manager</u>, provided that:
 - 5.7.1. The <u>Grantor</u> shall be entitled during the <u>mining right</u> period by means of written notice to the <u>Grantee</u> to nominate any other person or institution in South Africa to whom the royalty payments shall be made; and

6. Restrictions and Obligations imposed on the Grantee

- 6.1. Subject to the provisions of the Act, the <u>Grantee</u> and the <u>Grantee's</u> nominees, agents, contractors and employees shall have the right to enter upon and have access to any portion of the <u>Mining Right</u>. Area together with all such persons, plant, machinery and equipment as may be required for the purposes of prospecting or mining on the <u>Mining Right Area</u> and to carry out all such prospecting and mining operations on the <u>Mining Right</u> Area as the <u>Grantee</u> may in the <u>Grantee's</u> discretion determine, which rights of the <u>Grantee</u> shall (notwithstanding anything to the contrary in this <u>mining right</u>) be subject to the following special conditions:
 - 6.1.1. No mining operations shall be carried out on any land which is under cultivation or which is used as a garden, orchard, vineyard, nursery, plantation, forest, kraal or cemetery or within one hundred metres from any spring, well, borehole, stream, reservoir, dam, watercourse or waterworks, or within one hundred metres of any house, homestead or building unless security in such amount as the <u>Grantor</u> may have determined has been deposited with the <u>Regional Manager</u> for payment of any loss or damage arising directly or indirectly from such mining operations and for the renewal, reconstruction or re-erection of anything damaged or destroyed as a result of such mining operations.

- 6.1.2. The <u>Grantee</u> shall not trespass or enter into any homestead, house or its curtilage nor interfere with or prejudice the interests of the occupiers and/or owners of the surface of the <u>Mining Right</u> Area except to the extent to which such interference or prejudice is reasonably necessary for the purposes of enabling the <u>Grantee</u> properly to exercise the <u>Grantee's</u> rights under this <u>mining right</u>.
- 6.2. <u>Subject to section 44 of the Act</u>, no permanent buildings, except for buildings which are by law required to be of a permanent nature and except for buildings which the <u>Grantee</u> will remove from the <u>Mining Right</u> Area upon expiry of this <u>mining right</u>, may be erected on the <u>Mining Right</u> Area by the <u>Grantee</u> without the prior written consent of the registered owner of the surface of the <u>Mining Right</u> Area.
- 6.3. No new roads, railways, pipelines, power line or other similar infrastructure traversing the surface of the <u>Mining Right</u> Area shall be constructed on the <u>Mining Right</u> Area by the <u>Grantee</u> without the authorisation or permission from the authorities concerned.
- 6.4. The <u>Grantee</u> shall <u>conduct an environmental impact assessment and carry</u> out all operations on the <u>Mining Right</u> Area in accordance with the approved EMP and with the minimum amount of disturbance and damage to the environment.
- 6.5. Mining operations shall be conducted according to the Business Plan and in a manner which shall limit so far as may reasonably be possible any detrimental effect which such mining operations may have on existing developments and improvements on the <u>Mining Right</u> Area and surrounding areas and communities.
- 6.6. The <u>Grantee</u> must to the satisfaction of the <u>Grantor</u>, over the <u>mining right</u> period, initiate and actively conduct mining operations and carry out and meet the commitments set out in the approved Business Plan.
- 6.7. The <u>Grantee</u> undertakes to commence with mining operations within_____ years from the effective date. Failure to do so, this lease shall *ipso facto* lapse.

7. <u>No Mortgage</u>, Cession, Transfer, Alienation

- 7.1. The <u>Grantee</u> shall not, without the Grantor's prior written consent be entitled to either wholly or in part <u>transfer or encumber this mining right in terms of section 11 of the Act</u>.
- 7.2. Unless the <u>Grantee</u> has obtained the prior written consent of the Grantor on such terms and conditions as the Grantor may determine (but subject always to the provisions of clause 7.3), the <u>Grantee</u> shall not:
 - 7.2.1. lease the rights granted to the Grantee in terms of this Mining Right,
 - 7.2.2. allow any other person to <u>mine</u> for ---- in the <u>Mining</u> Area for such other person's own benefit and account;
 - 7.2.3. allow any other person to carry out <u>mining</u> operations for ---- in the <u>Mining</u> Area and to retain the whole or any part of the income or profits from the <u>mining</u> operations.

- 7.3. The <u>Grantee</u> may from time to time appoint one or more independent contractors to carry out any work in relation to <u>mining</u> operations, but the <u>Grantee</u> shall always remain liable for the compliance and observation of the provisions of the Act, the EMP and of the terms and conditions of this <u>Mining Right</u>.
- 8. Protection of Boreholes, Shafts, etc.

- 3,

No boreholes sunk by the <u>Grantee</u> during the currency of this <u>mining right</u> shall be sealed or closed up by the <u>Grantee</u> without the prior approval of the <u>Grantee</u>, but the <u>Grantee</u> shall fence and render safe all boreholes, shafts, openings and excavations in accordance with the provisions of the Act, the Mine Health and Safety Act, 1996 and any other applicable laws and regulations.

9. Protection of Property and Persons

The <u>Grantee</u> shall during the currency of this <u>mining right</u> take all such steps, do all such acts and things and conduct all prospecting and mining operations in such manner as may be required to ensure that the property and person of anyone who is entitled from time to time to make lawful use of the surface of the <u>Mining Right</u>. Area are fully protected against any damage caused by or as a result of the exercise of the rights granted under this <u>mining right</u> to the <u>Grantee</u> and, in this regard, the <u>Grantee</u> hereby indemnifies such persons against all such damage.

10. Inspection of Mining Right Area

The <u>Grantor</u> and any person duly authorised thereto in writing by the <u>Grantor</u> shall be entitled to inspect the <u>Mining Right</u> Area, the <u>Grantee's</u> mining operations and the execution of the approved EMP on the <u>Mining Right</u> Area at any time, and any instruction conveyed in writing by the <u>Grantor</u> to the <u>Grantee</u> requiring the proper performance by the <u>Grantee</u> of the <u>Grantee's</u> obligations under this <u>mining right</u> shall be put into effect by the <u>Grantee in terms of the Act</u>.

- 11. Cessation and termination of Mining Right
- 11.1 The Grantor shall be entitled to terminate this mining right in terms of section 47 of the Act.

Unless the contrary is agreed to in writing between the <u>Grantor</u> and the <u>Grantee</u> the following provisions shall apply on termination of this <u>mining right</u>:

11.2.1. The <u>Grantee</u> or any other person entitled thereto shall subject to the provisions of the Act, within of termination of this <u>mining right</u> remove all installations, machinery, plant and equipment from the <u>Mining Right</u> Area, but shall not remove or damage any machinery or assets used for the maintenance of underground works or installations or for the prevention of damage to the mine or underground works (which machinery and assets shall for all purposes be deemed to have been abandoned by the <u>Grantee</u>).

- 11.2.2. No compensation shall be payable to the <u>Grantee</u> in respect of such removal or in respect of such abandonment.
- 11.2.3. If any of the installations, machinery, plant or equipment which are not necessary for the maintenance of the mine and underground works are not removed within such period of ______, the _______, the ________ may (unless and extension of time for such removal is allowed) cause the same to be sold by public tender or public auction.
- 11.2.4. One month's notice of such sale shall be published in the Government Gazette and in a newspaper circulating in the Magisterial District of ___ and shall be given to the registered mortgagees, if any, of the *Mining Right* Area.
- 11.2.5. After the recovery of the costs of such sale and of any moneys which remain due to the <u>Grantor</u> under this <u>Mining Right</u>, the balance of the proceeds of such sale shall be paid to the <u>Grantee</u>.
- 11.2.6. As an alternative to such sale the <u>Grantor</u> shall have the right, upon written notice to the <u>Grantee</u>, to purchase all or any of the buildings, machines, installations, equipment or other property of the mine on the <u>Mining Right</u> Area at such price and on such conditions as may be agreed upon between the parties or, failing mutual agreement on any such conditions, the dispute shall be referred for determination to arbitration in accordance with the provisions of clause 18.

12. Records and Returns

- 12.1. The <u>Grantee</u> shall maintain all such books, plans and records in regard to mining on the <u>Mining Right</u> Area as may be required by the Act and the regulations promulgated thereunder, and shall furnish to the Minister such reports and documents as may be relevant to the rights of the <u>Grantor</u> under this agreement.
- 12.2. The <u>Grantee</u> shall furnish to the appropriate authorities all such returns as may from time to time be required under the Act and the regulations promulgated thereunder.
- 12.3. The <u>Grantee</u> shall on a quarterly basis, calculated from the commencement of this <u>mining right</u>, furnish the <u>Grantor</u> or the <u>Grantor's</u> nominee with a return reflecting the particulars of any primary geological exploration undertaken by the <u>Grantee</u> on the <u>Mining Right</u> Area.
- 12.4. The <u>Grantee</u> shall furthermore at the end of each year following commencement of this <u>mining right</u>, inform the <u>Grantor</u> in writing of any new developments and of the future prospecting and mining activities planned in connection with the exploitation/mining of _____ on the <u>Mining Right</u> Area.

13. Payment of Compensation

The <u>Grantor</u> shall not at any time be liable or responsible for the payment of compensation of whatsoever nature to the <u>Grantee</u>, the <u>Grantee's</u> successors-in-title or assigns, except for compensation for which the <u>Grantor</u> is liable in consequence of the breach by the <u>Grantor</u> of any of the provisions of this <u>mining right</u> or on account of any wrongful act or omission by or attributable to the <u>Grantor</u>.

14. Contravention

Should the <u>Grantee</u> breach or fail to observe any material term or condition of this <u>mining right</u>, and should the <u>Grantee</u> fail to remedy such breach or default within 60 (Sixty) days following the <u>Grantor's</u> written notice to do so, the <u>Grantor</u> shall be entitled to cancel this <u>mining right</u> forthwith <u>in terms of section 47 of the Act</u>, without prejudice to any other rights or remedies of the <u>Grantor</u>.

15. Health and Safety

The <u>Grantee</u> and its successors in title or assigns will and do hereby undertake and bind themselves to during the currency of this <u>Mining Right</u> to observe and comply with all relevant provisions of the Mine, Health and Safety Act 29, 1996.

16. Arbitration

- 16.1. Should any difference or dispute arise between the parties to this <u>mining right</u> as to the intention and meaning of any of the terms and conditions of this <u>mining right</u>, or as to any matter arising from this <u>mining right</u>, then the parties shall endeavour in good faith to make every effort to resolve the dispute on its merits by negotiations and shall, for that purpose, attend at least one meeting with each other. Such negotiations shall take place within 21 (twenty-one) days of the dispute arising unless the parties otherwise agree in writing and shall endure for no longer than 7 (seven) days from the date of commencement thereof.
- 16.2. If the negotiation referred to in 18.1 does not take place or does not result in a resolution of the dispute or difference within 7 (seven) days of the commencement thereof (or such extended period as the parties may agree in writing), any party claiming that a dispute has arisen shall give written notice to the other party of that fact and shall specify in that notice the nature of the dispute.
- 16.3. If the parties are unable to resolve the difference or dispute in question despite compliance with 18.1,18.2 and 18.3, then the dispute shall be referred for arbitration in terms of the Arbitration Act, 42, 1965, or any subsequent law which amends or substitutes the Arbitration Act.

17. Joint and Several Liability

Notwithstanding anything to the contrary, any provision of this <u>mining right</u> which contravenes the Act or which is otherwise <u>ultra vires</u>, void, voidable, or unenforceable, shall be severable from the rest of this <u>mining right</u>, such rest thus being and remaining of full force, effect and enforceability.

18. Domicilia citandi et executandi

18.1. The parties choose the undermentioned business addresses as their respective domicilia citandi et executandi and for all purposes arising from this <u>mining right</u>, in particular for the purposes of the serving of any notice in terms of this <u>mining right</u>, and any notice properly addressed to the undermentioned postal addresses of the parties shall be deemed to have been received by the addressee within 5 (five) if given in writing and posted by prepaid registered post addressed to the addressee at the relevant postal address:

18.1.1. In the case of the Grantor.

	Business Address:		r th sa	4		The blow
	· · · · · · · · · · · · · · · · · · ·			3 B 8		, 7 B S
	Postal Address:	12	10			
					•	**
	Tel:			ð		83
	Fax: 18.1.2. In the case of the <i>Grantee</i> :				8	W.
	Business Address:		19	in W		N
	Business Address.		\$ \$	tii		8
	Postal Address:	lg	85 W1	SEC SV	æ	a a
	8 6 0 <u>2</u>					
	Tel:		*		8	
	Fax:					
•	Notwithstanding anything to the contrar received by a party at any place other the adequate notice or communication to the such party's chosen domicilium citandi	han the ch	osen <i>domiciliun</i> ootwithstanding	n citandi et exe	<i>cutandi</i> sha	all constitute
	Either party shall be entitled from time t address furnished above after giving at the other party, failing which the above	least 14 (Fourteen) days	prior written no	otice of suc	

19. Cost

18.2.

18.3.

18.4.

the provisions of this clause.

The <u>Grantee</u> shall pay all costs and charges incurred by the <u>Grantee</u> in connection with the negotiation, preparation, execution and registration of this <u>mining right</u>.

Any written notice or communication contemplated in this clause which is forwarded by one party to the other by registered post will be reputably presumed to have been received by the addressee on the fourteenth day following the date of posting from an address within the Republic of South Africa to the addressee at the postal address of the addressee for the time being as determined in accordance with

Thus done and signed at	on the _	day of _	200_	_in the pre	sence	of the	undersigi	ned witne	sses:
				10					0)
AS WITNESSES:	29	e s ^a	70 70 34		13 13 45		w.		
1	5					55		2. 5.	
2			260,471,12 124)0			6-1 6-1		3
	for	and on beha	lf of the <u>G</u>	rantor			9		19 15
Thus done and signed at	on the	day of	_ of 200_	in the p	resence	of the	undersig	ned witn	esses:
AS WITNESSES:	02	1					ž.		21
AS WITHESSES:				0					- 8
1			×		, 4				
p	100		35	52		97 F	\$8	20	
2.	-	for and or	behalf of	the <u>Grant</u>	<u>ee</u>				8.

QUOD ATTESTOR

NOTARY

FORM II:

ĸ	ΛH	u	N	C	PE	RI	/IT
Ľ١	/	•	14	LJ.			

PKEAM	IBLE
WHERE	EAS, The State is the <u>custodian</u> of the right to in respect of the <u>land</u>
section	2 of the Mineral and Petroleum Resources Development Act, 2002, (hereinafter referred to as "the Act"; and
WHERE	EAS,, ID/Registration number
(hereina	after referred to as "the Grantee"), applied to mine which may occur on the land in
onstantial de la constantial	n, it is hereby agreed, in terms of the provisions of section 27 of the Act, by me,
	, Minister of Minerals and Energy (hereinafter referred to as "the
Minister	"), to grant to the Grantee, and the Grantee hereby accepts the right to mine for at his own
cost,	and to dispose thereof, on the <u>land</u> , approximately
	(not more than ,5) hectares in extent (hereinafter referred to as "the land"), as shown on sketch
pian No	, which sketch plan is filed in the office of the Regional Manager, subject to the provisions of
the Act	and the conditions set out hereinafter:
BEACO	NS .
1.	Recognizable, sturdy and clearly-visible beacons must be placed and maintained so that the mining area as described by the permit and sketch plan is clearly identified.
GRANT	ING OF RIGHTS
2.	The Grantee shall as soon as is possible, but within a period of not less than three months after the signing
	hereof by the Minister, apply for a mining permit as described in the Act and the permit hereby granted shall be
	valid for a period of years after the issuing thereof provided that should the Grantee not apply
	for a mining permit within the prescribed period of <u>three</u> months, this agreement shall <u>ipso facto</u> lapse.
3.	This mining permit may be renewed upon written application by the Grantee for a further period of one (1) year, provided:
#£	(a) The Grantee has conducted the mining operations in accordance with the mining plan.
ot 🔉	(b) The Grantee has complied with the provisions of the approved and has ensured the continued relevancy of the <i>Environmental Management Plan</i> or an updated version thereof, as well as financial
	guarantee for rehabilitation

UNDERTAKING BY GRANTEE

4. The Grantee undertakes to commence with mining operations/activities within <u>three</u> (3) months from the date of issue of the mining permit. Failure to do so, this agreement shall <u>ipso facto</u> lapse.

5. Mining operations shall be carried out in a manner and on a scale satisfactory to the Minister.

ABANDONMENT

6. The Grantee may, at any time, by notice in writing to the Minister, abandon the permit hereby granted or any portion of the land comprising the subject of such a permit, and thereupon it shall be deemed to have lapsed as from the date of such notice to the extent indicated therein.

CANCELLATION OR SUSPENSION OF MINING PERMIT BY THE MINISTER

ation and termination of Mining Permit

designation and termination of mining remint

7.1	The Minister shall be entitled to termina	te this mining right in terms of section 47 of the Act.
-----	---	---

7.2 The <u>Grantee</u> shall be entitled to terminate this <u>mining right</u> at any time on ______ prior written notice to the <u>Grantor</u>, which notice shall take effect as from the date of receipt thereof by the <u>Grantor</u>.

Unless the contrary is agreed to in writing between the <u>Grantor</u> and the <u>Grantee</u> the following provisions shall apply on termination of this <u>mining right</u>:

- 7.2.1. The <u>Grantee</u> or any other person entitled thereto shall subject to the provisions of the Act, within _______ of termination of this <u>mining right</u> remove all installations, machinery, plant and equipment from the <u>Mining Right</u> Area, but shall not remove or damage any machinery or assets used for the maintenance of underground works or installations or for the prevention of damage to the mine or underground works (which machinery and assets shall for all purposes be deemed to have been abandoned by the <u>Grantee</u>).
- 7.2.2. No compensation shall be payable to the <u>Grantee</u> in respect of such removal or in respect of such abandonment.
- 7.2.4. One month's notice of such sale shall be published in the Government Gazette and in a newspaper circulating in the Magisterial District of ___ and shall be given to the registered mortgagees, if any, of the <u>Mining Right</u> Area.
- 7.2.5. After the recovery of the costs of such sale and of any moneys which remain due to the <u>Grantor</u> under this <u>Mining Right</u>, the balance of the proceeds of such sale shall be paid to the <u>Grantee</u>.
- 7.2.6. As an alternative to such sale the <u>Grantor</u> shall have the right, upon written notice to the <u>Grantee</u>, to purchase all or any of the buildings, machines, installations, equipment or other property of the mine on the <u>Mining Right</u> Area at such price and on such conditions as may be agreed upon between the parties or, failing mutual agreement on any such conditions, the dispute shall be referred for determination to arbitration in accordance with the provisions of clause 18.

NOTE

Clauses 8 and 9 are specifically applicable to <u>diamonds</u> <u>only</u> should any other mineral/s be involved, the clauses must be deleted or amended accordingly

CONSIDERATION PAYABLE

- 8. The Grantee shall pay to the State, in addition to such amounts as are payable at any time under any law providing for taxation on the profits of diamond mining, at the office of the Department in the region where the operation is taking place as from the date of issuing of a mining permit by the Minister to the Grantee -
 - (a) a royalty of 5% of the proceeds of the diamonds found in the course of mining operations on the land. Payment of the said royalty shall be made within 30 (thirty) days after the end of every quarter and shall be accompanied by a certificate or other document acceptable to the Minister, reflecting particulars of the proceeds and sales of diamonds. Such quarterly payments shall be based -
 - (i) on the selling price of the diamonds;
 - (ii) in the case of diamonds not yet sold, on the estimated value thereof; and
 - (iii) after the subsequent sale of any diamonds in respect of which payment has been made in terms of subparagraph (ii) above, on the difference between the estimated value and the selling price; and
 - (b) a minimum royalty of R...... for year one and R..... for year two. Such minimum royalties may be offset against actual royalties payable for each year in which the royalties are payable.

If the said payments are not made on the due date, the Grantee shall be in mora and interest thereon shall be charged at the "prescribed rate" as defined in section 80 of the Public Finance Management Act, 1999 (Act 1 of 1999), reckoned from the due date to the date on which such payment is made.

9. The Minister may at any time direct that the value of any diamonds found on the land during mining operations be determined by someone appointed by the Minister, and the Grantee shall not dispose of such diamonds at a price below the value so determined, without the consent of the Minister.

KEEPING OF RECORDS

10. The Grantee shall keep all such books, accounts, plans and records as may reasonably be deemed by the Minister to be necessary for conveying a full account and record of his operations on the land and such books, accounts, plans and records shall at all reasonable times be open to the inspection of the Minister or any officer authorised thereto by him/her. The Grantee shall furthermore promptly submit, at the intervals stated in Schedule A to this agreement, the documents required in the said schedule.

INSPECTIONS

11. The Minister or any other officer appointed by him/her shall have the right to carry out, or cause to be carried out, any investigation which may be deemed necessary on the land pertaining to the mining permit.

GRANTEE NOT EXEMPTED FROM OBLIGATIONS

12. In the event of this <u>mining permit</u> ceasing to exist for any reason, the Grantee shall not be exempted from any of his unfulfilled obligations <u>in terms hereof or the Act</u>.

STATE NOT LIABLE

13. Subject to the provisions of the Act, the State is not liable for the payment of any compensation whatsoever for any installation or structure erected on or over the site or for any improvement made by the Grantee on the site after conclusion of his mining operations and evacuation of the land.

CESSION, TRANSFER, ALIENATION

14. The rights granted in terms of this <u>mining permit</u> may not be dealt with as described in section 27(7)(b) of the Act, <u>but may be encumbered or mortgaged only for the purpose of funding or financing of the mining operations with the consent of the Minister.</u>

GRANTING OF RIGHTS IN RESPECT OF OTHER MINERALS

Subject to the provisions of the Act, the Minister reserves the right to grant a <u>mining permit</u> to any other person to mine or to prospect for any other mineral which may occur on the land and which is not included <u>herein</u>: Provided that such <u>other mining</u> permit shall not be granted without <u>the</u> prior consultation between the Minister and the Grantee.

PROTECTION OF PROPERTY AND PERSONS

16. The Grantee shall take all such necessary steps and do all such acts, matters and things and carry out his mining operations in such manner as will fully and adequately safeguard and protect the rights, property and person of any person or persons occupying, using or present on the land or adjacent land, from any damage caused by or through or in consequence of the exercise by the Grantee of his mining operations pertaining to the mining permit.

INDEMNITY

17. The Grantee hereby undertakes to hold harmless and indemnifies the State and/or the servants of the State, of and from any injury, loss (including loss of life) or damage, costs, charges and expenses which may be suffered or sustained by any person or persons whomsoever as a result of the mining operations carried out by the Grantee and shall also on demand refund to the State and/or the servants of the State all costs, charges and expenses which the State and/or the servants of the State may be put to or sustain in connection with or arising out of any claims made against the State and/or the servants of the State by any such person or persons, it being specially agreed that the State shall in no way be deemed to be a partner in the mining operations. The State shall give notice to the Grantee of any occurrence in respect of which he intends to benefit by this indemnity and shall afford the Grantee full facilities for investigating, contesting or otherwise dealing at his discretion with any matter arising in respect thereof.

WAIVER OR LENIENCE

No waiver or lenience that the Minister may grant or allow towards the Grantee in connection with the latter's obligations shall prejudice the rights of the Minister in terms <u>hereof</u> or the Act. Such waiver or lenience shall furthermore, not be regarded as an abandonment by the Minister of any of his/her rights.

DO		

19.	Any notice, instru	ctions, claim of	account which	is required o	r may be give	en to the Gran	ntee in term	s <u>hereof</u> shall
	be valid if given in	writing and for	warded by prep	aid registere	ed post to the	Grantee at :	排 (表)	
	D 20	¥-11			8	4		
								\$ S
					30 30	.0		
								8 E
	or in any such pla			Africa as the	Grantoo ma	v in writing a	ive the Min	istor nation of
						COLUMN PROPERTY TO THE		
	hereafter. The Gr	rantee hereby c	nooses <u>domicii</u>	ium çıtandi e	et executanui	at the lastine	nuonea aad	iress.
THUS	DONE AND SIGNE	D at	on	this	dav of	27		. 20 in the
	nce of the undersigne						- 1	.
picoci	ioe of the underloigh	od competent w		¥0		75 60		李
	*			2	8	. 4		
)9								
					M	IINISTER:		
AS WI	TNESS:							3 9 9
			(6)				20 ===	15
1	********************************	10		VI 19	a W	36		2
	*				65	¥8		40
2	V		127			Ø	*	10
6 1,			- E			48		
		'e		5.				
l,		V.	, hereby	accept the	above condit	ions set out h	erein.	7
ma*i		100	•	and the property of the second				
							整	(2)
Signer	l at	on this	the day	of	20	in the nree	ence of the	undereinned
		011 11110	uio uay	01		., iii iiie pies	once or the	dilderaigned
compe	tent witnesses.							
			18		<u> </u>	RANTEE		
			2) (3	al .		an a gas u		kr 40
AS WI	TNESSES:			6.		A		
	## ## ## ## ## ## ## ## ## ## ## ## ##		\$3				49	
1,					0.0	20		
\$8	· ·				369			9
2	***************************************				86		10	

SCHEDULE A

DOCUMENTS TO BE LODGED AT THE DEPARTMENT

The Grantee shall furnish to the Minister quarterly returns reflecting particulars of:

- (i) The number and dimensions of all pits, trenches, shafts, tunnels or other excavations made, together with a description of the geological formations encountered.
- (ii) Details of boreholes drilled, i.e., depth and strata traversed.
- (iii) The quantity of waste and gravel or ore excavated and the quantity of gravel or ore treated.
- (iv) The number of diamonds found, together with their mass (in carats) and the value.
- (v) The number of workers employed.
- (vi) Expenditure incurred in respect of -
 - (a) salaries and wages;
 - (b) stores; and
 - (c) equipment, machinery and vehicles purchased and details of any other expenditure.

N.B. "NIL" RETURNS WITH THE REASONS THEREFOR MUST BE FORWARDED WHERE APPLICABLE.)

FORM JJ:

RECONNAISSANCE PERMIT

-	rea	-	-	-
-	rea	ш	ш	ш

- 1. Definitions
- 2. Description of the Permit Area
- 3. Duration
- 4. Rights and Obligations of the Grantee
- 5. Fees
- 6. Technical Advisory Committee
- 7. Minister's Right to Cancel or Suspend the Reconnaissance Permit
- 8. No guaranteed right to other minerals
- 9. Examination of the Permit Area and Operations
- 10. Reports and Samples to be Furnished
- 11. Reconnaissance Work Programme and Budget
- 12. Black Economic Empowerment
- 13. Good international practice, pollution, navigation, fishing and conservation
- 14 Local labour, equipment, materials and contractors
- 15. Tax
- 16. Accounts and Audits
- 17. Customs duties
- 18. Exchange Control
- 19. Indemnity and insurance
- 20. Health and Safety
- 21. Data and information to be kept confidential and limitations on public announcements
- 22. Cession, Transfer, Alienation
- 23. Law and interpretation
- 24. Vis Major
- 25. Amendments
- 26. Waiver of Lenience
- 27. Arbitration
- 28. Environmental Protection
- 29. Costs and VAT
- 30. Severability

31. Notices, domicilia citandi et executandi and Representatives

ANNEXURES

- A Map and description of the Permit Area
- B Reconnaissance Work Programme
- C Account to which payments to the Grantor are to be made

SPECIMEN AGREEMENT FOR A RECONNAISSANCE PERMIT

entered into between

THE REPUBLIC OF SOUTH AFRICA

(hereinafter referred to as "the Grantor") and herein represented by the Chief Executive of South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited, he being duly authorised by a Resolution of Directors of Agency passed at Cape Town on [] 2000 [], Agency acting herein on behalf of the Minister of Minerals and Energy of the Republic of South Africa (hereinafter referred to as "the Minister"), by virtue of the provisions of section 70(1) read with section 66(1)(d) of the Mineral and Petroleum Resources Development Act, 2002 (Act No. xx of 2002)

	and	
	a a sa	e v
a 9	Registration No. [
hereinafter referred to	as "the Grantee") a company incorporated and registe	red in
nder the laws of	and having its registered address at	and herei
epresented by	, he being duly authorised by virtue of a res	solution of directors of Grantee passed a
to		
-00		

(the Agency and Grantee are jointly referred to as "Parties" and individually as "Party")

WHEREAS

- In terms of section 66(1)(a) of the Mineral and Petroleum Resources Development Act, 2002, Act No. ____ of 2002
 ("the Act") it is the function of the Agency to promote exploration for petroleum in the Republic of South Africa;
- In terms of section 66(1)(d) of the Act the Minister is authorised to issue reconnaissance permits for the purpose of
 promoting exploration for petroleum within the onshore and offshore territory of the Republic of South Africa, the
 latter in such maritime zones over which the State has jurisdiction, including but not limited to the territorial
 waters, the continental shelf, and the exclusive economic zone, all as envisaged in the Sea-Shore Act, 1935
 (Act No. 21 of 1935), and in the Maritime Zones Act, 1994 (Act No. 15 of 1994);
- The Grantor and the Grantee have jointly identified certain offshore areas where reconnaissance work can be
 expected substantially to increase the understanding of the underlying geology and which reconnaissance work
 can reasonably be expected to attract the interest of the industry;

- The Grantor has agreed to issue to the Grantee a Reconnaissance Permit by way of and being this Reconnaissance
 Permit Agreement to enable it to carry out certain reconnaissance work on an area ("the Permit Area") more
 fully described in clause 2 below in accordance with the terms and conditions recorded hereunder.
- The Grantor has appointed South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited as its designated representative for the purposes of this agreement

NOW THEREFORE THE PARTIES AGREE AS FOLLOWS:

Definitions

The definitions in the Act shall apply also to this Agreement and additionally the following words and expressions shall have the meaning assigned to them:

List of definitions to be fleshed out as text approaches finalisation

- 1.1. "the Act" shall mean the Mineral and Petroleum Resources Development Act, 2002 (Act No. ... of 2002) or any subsequent law which amends or substitutes the Act;
- 1.2. "Affiliate" of a Party shall mean a company which owns, or is owned by, or which is owned by an entity which owns that Party; "owns" and "owned" mean in this definition the ownership directly or indirectly of more than 50 % of the voting shares or members' interests;
- 1.3 "the Agency" shall bear the meaning attributed thereto in the description of the Parties above'
- 1.4. "Effective Date", shall mean the date on which this reconnaissance permit agreement is signed by the last Party to sign it;
- 1.5. "the Grantee", shall bear the meaning attributed thereto in the definition of the Parties above;
- 1.6. "the Grantor" shall mean the State;
- 1.12. "the Minister", shall bear the meaning attributed thereto in the description of the Parties above;
- 1.8. "the Permit Area" means the area referred to in clause 2 of this Agreement, excluding parts relinquished or abandoned from time to time;
- 1.10. "the Reconnaissance Work Programme", shall mean the Grantee's written recordal envisaged in regulation 6 made in terms of the Act of the manner in which it intends to carry out the Reconnaissance Operations in the Permit Area, and shall include all amendments and supplementary submissions relating to the Reconnaissance Work Programme which may be made by the Grantee and approved in writing by the Agency from time to time;
- 1.11. State, shall mean the Republic of South Africa;

Reference to the singular of a defined term includes a reference to it in the plural and vice versa and reference to any gender includes a reference to all other genders.

Description of the Permit Area

The Permit Area comprises approximately ______ km² in the West / South / East* Coast Region offshore, being the area of sea including the seabed, identified as such by co-ordinates on the map attached as Annexure A hereto.

3. Duration

3.1 Unless this permit is suspended, cancelled or abandoned it shall, subject to acceptance thereof by Grantee, where provided at the foot of this permit, be valid from the Effective Date for a period of [] months thereafter.*

- * In terms of section 70(4)(b) of the Act, this period may not exceed one year.
- 3.2 In terms of sections 70(4)(d) and (e) of the Act, this permit is not transferable and not renewable.
- 3.3. Subject to satisfactory completion of the Reconnaissance Work Programme, the Grantee shall for a period of 9 years from the expiry of this Permit (Marketing Period) have the exclusive right to license the use of the Data to third parties through contracts that will preserve the confidentiality of the data and on terms and conditions that are customary in the industry, and retain all consequent revenues.
- 3.4. During the Marketing Period as defined in 3.3. the Agency shall have the right to use all such data and reports for internal evaluation of the petroleum potential of the Permit Area, and to disclose portions of the said data and reports to bona fide potential licensees of the data and/or potential applicants for a Technical Co-operation Permit or Exploration Right or Production Right in relation to the portion of the Permit Area to which the disclosed data relates. On expiry of the Marketing Period the Agency shall be free to display, disclose, sell, trade or otherwise dispose of the Data and reports without restriction.

4. Rights and Obligations of the Grantee

- 4.1 The Agency hereby in terms of section 70(1) read with section 66(1)(d) of the Act issues to the Grantee and the Grantee hereby accepts the Reconnaissance Permit constituted by this Agreement, conferring the non-exclusive right to carry out the Reconnaissance Operations in the Permit Area as set out in the Reconnaissance Work Programme attached hereto as, marked Annexure B,
- 4.2 The Grantee acknowledges that all technical data, reports, maps, samples and information generally (defined in clause 13.6 as "the Data") that arise from the carrying out of the reconnaissance work are the property of the Agency.
- 4.3 The Grantee shall have the right to:
- 4.3.1 have access to and enter upon the Permit Area with such persons, aircraft, animals, vehicles, boats, ships, vessels, machinery, equipment, instruments or material as may be necessary for the purpose of carrying out reconnaissance operations upon the Permit Area as may be necessary for or incidental to the aforesaid purposes;

The Grantee's Operations shall be limited to those reasonably necessary and pertinent to the methods and techniques set out in the Reconnaissance Work Programme (Annexure B) and any combination or combinations of such work and all such auxiliary or related work or activities as may be useful in connection with such Operations; for the absence of doubt, such Operations shall not include any form of drilling.

However, the Grantee shall not, without the Minister's prior written consent, carry out Reconnaissance Operations over:

- (a) areas used or reserved under any law for any Government or public purpose;
- (b) areas within 5 (five) nautical miles seaward of the low water mark (as defined in section 1 of the Sea-Shore Act, 1935 (Act No. 21 of 1935)), nor in the areas shown as reserved areas on the attached sketch plan, Annexure A.
- 4.3 Without detracting from the Grantee's further obligations in terms of this Reconnaissance Permit Agreement or in terms of the Act the Grantee shall:
- 4.3.1 actively conduct Reconnaissance Operations in respect of Petroleum on the Permit Area in accordance with the Reconnaissance Work Programme;
- 4.3.2 comply with the terms and conditions of this Reconnaissance Permit Agreement and the relevant provisions of the Act;
- 4.3.3 pay the prescribed reconnaissance fee to the Agency;
- 4.3.4 submit, timeous, complete and correct summary progress reports to the Agency on a monthly basis;
- 4.3.5 supply to the Agency digital and, where appropriate, paper copies of all data generated, as soon as possible after completion of each phase of Operations or projects.
- 5. Fees

- All fees, considerations and the like which are payable by Grantee to the Agency and/or the Grantor in terms of this Reconnaissance Permit shall be paid by deposit into the account of the Agency at the bank and account indicated in Annexure C, or such other bank and account as the Agency may from time to time give notice of. The onus for timely payment rests solely with the Grantee; the Agency will not normally issue invoices.
- 5.2. If any fees, considerations and the like payable by the Grantee to the Agency in terms of this Reconnaissance Permit are not paid punctually on or before the due dates for payment, the Grantee shall be in *mora debitoris* and shall pay interest thereon at the rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act No. 1 of 1999) reckoned from the due date for payment to the date of actual payment.
- 5.3. Regardless of any relinquishment of parts of the Permit Area or early termination of this Reconnaissance Permit for whatever reason, no portion of the fees referred to in clause 5.1 shall be refundable.

6. Technical Advisory Committee

- 6.1. The Agency and the Grantee shall as soon as possible after the Effective Date establish a committee to be known as the Technical Advisory Committee which shall consist of:
 - (a) a chairman and one other person appointed by the Agency; and
 - (b) two other persons appointed by the Grantee.
- 6.2. The Agency and the Grantee may appoint by notice an alternative member to act in the place of either or both of their representatives on the Technical Advisory Committee during his or her absence or incapacity to act as a member of the Committee. When an alternate member acts in the place of any member he or she shall have the powers and perform the duties of such member.
- 6.3. Without prejudice to the rights and obligations of the Grantee in relation to the management of Operations the functions of the Technical Advisory Committee shall be:
 - to review all Operations carried out by the Grantee and to provide the Grantee advice and recommendations with respect thereto;
 - to review and approve any proposed reconnaissance work programme to be submitted by the Grantee to the Agency in terms of clause 14;
 - (c) to review the accounting of expenditure and the maintenance of operating records and reports kept in connection with Operations; and
 - (d) to offer advice to the Grantee in order to promote the efficient carrying out of Operations.
- 6.4. Meetings of the Technical Advisory Committee shall be held at least semi-annually. The location of such meetings shall be in Cape Town, South Africa unless otherwise unanimously agreed by the members of the committee.
- 6.5. Three members of the Technical Advisory Committee shall form a quorum for a meeting of the Committee, provided that at least one representative of Grantee and one representative of the Agency is present.
- 6.6. Any member of the Technical Advisory Committee shall have the right to call any expert to any meeting of the Committee to advise the Committee on any matter of a technical nature requiring expert advice.
- 6.7. The Chairman of the Committee shall cause minutes of each meeting to be kept and circulated to the members.

7. Minister's Right to Cancel or Suspend the Reconnaissance Permit

- 7.1 It is recorded that the Minister is empowered to cancel or suspend this Reconnaissance Permit in the circumstances envisaged in and in accordance with the provisions of section 85 read with section 44 of the Act.
- 7.2. In the event of this Reconnaissance Permit being cancelled in accordance with the provisions of clause 7.1, the Grantee will not be absolved from its obligations or liabilities which have accrued up to the date of such cancellation.

7.3. The cancellation of this Reconnaissance Permit in terms of clause 7.1 will be without prejudice to the Grantor's or the Agency's other rights under this Reconnaissance Permit or in statutory or common law of the Republic of South Africa, including its right to recover any damages it may have suffered or may suffer, its right to claim specific performance or its right to claim ancillary relief.

8. No guaranteed right to other minerals

8.1. Should any mineral or minerals be discovered in the course of Reconnaissance Operations, the Grantee shall be bound to report such discovery to the Minister. The Grantee may thereafter, subject to any prior rights, apply for and, if granted the required prospecting rights by the State, prospect for such mineral or minerals.

9. Examination of the Permit Area and Operations

- 9.1 In terms of sections 86 to 89 of the Act, all read with section 65 of the Act, the Minister may authorise inspections and exercise of related powers more fully therein set forth.
- 9.2 Upon request by the Agency during offshore operations of Grantee, Grantee shall provide free transport between the Grantee's onshore base and the offshore facilities as well as accommodation on the offshore facilities for two representatives of the Agency.

10. Reports and Samples to be Furnished

- 10.1 Grantee shall timeously provide information and reports consistent with the Reporting Guidelines of the Agency unless agreed otherwise in writing. This reporting shall include, *inter alia*:"
- 10.2. The Grantee shall keep the Agency currently advised of all major developments taking place during the course of Operations and shall furnish the Agency with Data (as defined in clause 10.6) and other available information, reports, assessments and interpretations relating to Operations as the Agency may reasonably require.
- 10.3. The Grantee shall during the currency of this Permit provide the Agency with a summary report of all Operations during each month within 15 days of the end of that month.
- 10.4. Within 30 (thirty) days of the end of each quarter the Grantee shall submit to the Agency a report reflecting, for the relevant quarter:
 - the numbers/names of companies which have visited the Grantee's data room;
 - (ii) the work done and money expended on Operations; and
 - (iii) a statement reflecting rehabilitation work completed and of the rehabilitation work uncompleted.
- 10.5. The Grantee shall furnish the Agency with such other information, maps and plans, as to the progress of its operations in the Permit Area as the Agency may from time to time require and which the Grantee can reasonably make available.
- 10.6. Copies of all maps, magnetic tapes and all other geological and geophysical information and interpretation obtained by the Grantee in the course of carrying out Operations (hereinafter referred to as "Data") shall be given to the Agency as soon as they are acquired or prepared.
- 10.7. The Grantee may export such Data for processing or laboratory examination or analysis by the Grantee or by third parties or storage outside of the Republic of South Africa, provided that copies of equivalent quality have first been delivered to the Agency and provided that the Agency's prior written approval has been obtained.
- 10.8. Within three (3) months after the termination and/or cancellation and/or abandonment of this Reconnaissance Permit, the Grantee shall furnish the Agency with all results of Operations in-the Permit Area.
- 10.9. Nothing in this clause contained shall be construed as requiring the Grantee to disclose any of its proprietary technology or that of its Affiliates.
- 10.10 The above provisions are additional to and without detraction from the provisions of Regulations 7 and 8 made in terms of the Act.

11. Reconnaissance Work Programme and Budget

- 11.1. The Grantee must, to the satisfaction of the Grantor over the period of this Reconnaissance Permit, initiate and thereafter actively pursue operations designed tocarry out and meet the commitments set out in the Reconnaissance Work Programme annexed hereto marked Annexure B.
- 11.2. It is recorded that the time sequences as set out in the Reconnaissance Work Programme will commence on the Effective Date.

12. Black Economic Empowerment

The Grantee acknowledges that in order to give effect to Section 2(d) of the Act, the Grantor is desirous of expanding opportunities for historically disadvantaged persons (as defined in the Act) to enter the upstream petroleum industry and to benefit from the exploitation of petroleum resources in the Republic of South Africa and as such Grantee undertakes to use its best endeavours to seek historically disadvantaged South African partners to participate in this reconnaissance permit.

13. Good international practice, pollution, navigation, fishing and conservation

In addition to the provisions of the Act, the Grantee shall comply with the following special obligations.

- The Grantee shall maintain all installations erected for the safety of Operations in good repair and condition and shall execute all Operations in a proper and workmanlike manner in accordance with methods customarily used in good international practice and, without prejudice to the generality of the foregoing, the Grantee shall take all steps practicable in order to prevent pollution of the terrestrial or marine environment.
- The Grantee shall take immediate action with respect to notices given by the Agency of any event described in clause 13.1. In determining the occurrence of such event, the Agency shall adopt the standards and thresholds of the Department of Minerals and Energy of the Grantor.
- The Grantee shall immediately report the release of any petroleum into the environment to the Agency. This procedure shall be reflected in the oil spill contingency plan required in terms of the Prevention and Combating of Pollution of the Sea Act (Act no. 6 of 1981).
- The Grantee shall leave the seafloor free of significant obstruction and shall notify the relevant government agencies that the Permit Area is free of obstruction or, if it is not, the location, nature and extent of any obstruction resulting from its Operations.
- The Grantee shall give timeous notice of Operations to relevant government agencies and to commercial fishing associations and shall take all reasonable steps to minimise interference with navigation and fishing.

14 Local labour, equipment, materials and contractors

- 14.1. In furtherance of section 2(f) of the Act, in carrying out Operations the Grantee shall, to the extent reasonably possible, employ South African citizens having appropriate qualifications.
- 14.2. Not later than three months from the Effective Date, the Grantee shall meet with the Agency to discuss and agree on a programme for the recruitment, training and employment of South African citizens during Operations.
- 14.3 The Grantee may employ a person who has necessary qualifications and experience and is not a South African citizen if the required skills are not available in the South African labour market. The subsequent availability of qualified candidates will be reviewed annually.
- 14.4 The Grantee:
- 14.4.1 and its contractors shall use and purchase equipment, machinery, materials, instruments, supplies and accessories (all referred to as "Goods") manufactured, produced or available in the Republic of South Africa for use in Operations under this Reconnaissance Permit, provided that such Goods are reasonably comparable with like goods manufactured, produced or available outside the Republic of South Africa as to landed cost, quantity and quality and can be made available at the time when and the place where required;
- 14.4.2 shall to the extent reasonable, utilise contractors in the Republic of South Africa where services of comparable standards with those obtained elsewhere are available from such contractors at

competitive prices and on competitive terms and can be made available at the time when and the place where required.

15. Tax

Except to the extent exempted, Grantee shall, throughout the term of this Reconnaissance Permit, be liable for income tax payments to Grantor on the annual taxable income derived by it from Operations and transactions in connection therewith in accordance with the Income Tax Act.

It is however recorded that the Minister has procured that the Minister of Finance has issued a directive in terms of section 36(7F) of the Income Tax Act, that the provisions of that section (which relates to "ring-fencing") shall not apply to Grantee's Operations by virtue of this Reconnaissance Permit.

16. Accounts and Audits

The Grantee shall maintain such accounting records as will properly support all expenditure and receipts of reconnaissance operations under this reconnaissance permits.

Such records or copies thereof shall be provided at Petroleum Agency's request for the purpose of confirming the minimum expenditure obligations under the reconnaissance work programme. Agency shall have the right to appoint an auditor for the purpose of confirming that these obligations have been met. Such confirmation, if any, shall be completed within two years after the end of the calendar year in which the expenditures or receipts occurred.

17. Customs duties

The Minister records having procured the following from his/her colleague the Minister of Finance, namely:

- 17.1 The Grantee may during the continuance of this Reconnaissance Permit and subject to the provisions of international trade agreements which may be in force, import into the Republic of South Africa or clear from any customs and excise warehouse under rebate of full customs duty (as defined in the Customs and Excise Act, 1964) or export from the Republic of South Africa such Goods (excluding refined Petroleum products for use in road vehicles and distillate and residual fuel oil for any purpose and goods imported by the Grantee for the personal use of its personnel who are South African residents) as are required for use in Operations and which are not obtainable or available from stocks of Goods manufactured in the Republic of South Africa in terms of clause 20.4.1; provided however that the Grantee shall not sell, transfer, dispose of or use or retain for use except in Operations under this Reconnaissance Permit any Goods obtained free of customs duty without the permission of the Commissioner for Customs and Excise; such duty as may be assessed by the Commissioner for Customs and Excise shall be payable to him/her in respect of any such Goods sold, transferred, disposed of or used or retained for any use other than in Operations except if such Goods are sold, transferred or disposed of to any person in the Republic of South Africa entitled to a like rebate of customs duty or if such Goods are exported from the Republic of South Africa by the Grantee.
- 17.2 The Grantee shall have the right, subject to the applicable laws and regulations, to employ such contractors as in its opinion may be necessary for Operations; such contractors shall be exempted from the payment of customs duties in the same manner as the Grantee is exempted under this Reconnaissance Permit as long as their work is pursuant to their contract with the Grantee.
- 17.3 The Grantee's non-South African personnel may import household and personal effects into South Africa free of customs duty and may export such effects freely, but disposal in the Republic of South Africa of any such household and personal effects shall be subject to approval by the Commissioner for Customs and Excise and shall be subject to customs duties.

18. Exchange Control

- 18.1 The Minister records having procured the following from his/her colleague the Minister of Finance.
 - 18.1.1 The Grantee shall have the free availability of the currency generated from the export of Petroleum which the Grantee can place directly to a bank account within the Republic of South Africa or abroad.
 - 18.1.2 The Grantee shall have the free disposition and right to freely exchange South African Rand whether received from the sales of Petroleum in the national market for foreign currency or

received from other activities or events related to this Exploration Right. The Grantee has the right to deposit such currency directly into bank accounts in the Republic of South Africa or abroad.

- 18.1.3 The Grantee shall have the right to maintain, control and operate bank accounts in any currency, inside the Republic of South Africa and abroad; to have the control and free use of these accounts and to maintain and freely dispose of the funds of these accounts without any restrictions. The Grantee shall not be obliged to convert any foreign currencies into South African Rand.
- 18.1.4 Without prejudice to what is mentioned above, the Grantee shall have the right to dispose, distribute, send or retain abroad, without any restrictions, the Grantee's annual net profits.
- 18.1.5 The Grantee may use South African banks to exchange the currencies referred to in clause 18.1.2.
- 18.2 The exchange rate of the South African Rand will be determined by the free market.
- 18.3 If other exchange rules are established while this Reconnaissance Permit is in place, the Grantee may elect to have such other exchange rules apply to Operations. To the extent that such other exchange rules do not affect rules set out herein, the rules set out herein shall continue to apply to Operations.
- The Grantee may, after electing to have such other exchange rules apply to Operations, elect to return to the rules set out herein.
- The Grantee may acquire abroad materials, equipment and services and pay for them in foreign currency. The Grantee may obtain financing abroad through loans and advances to pay for costs of activities connected with this Reconnaissance Permit.
- 18.6 The Grantee shall provide periodic reports as reasonably requested by the Reserve Bank.
- In the event that the ratio of borrowings from South African financial institutions to foreign investment exceeds the ratio prescribed in Exchange Control Regulations promulgated by Government Gazette Notice R1111 of December 1961 as amended, the Grantee shall notify the Reserve Bank and advise the Reserve Bank as to any steps the Grantee intends to take to reduce this ratio, or request approval, which shall not be unreasonably withheld, to exceed this ratio.

19. Indemnity and insurance

- 19.1. The Grantee, its successors and/or assigns, during the tenure of this Reconnaissance Permit, will take all such necessary and reasonable steps and do all such acts, matters and things and carry out their Reconnaissance Operations in such a manner as will adequately safeguard and protect the environment, the Permit Area and any person or persons using or entitled to use the surface of the Permit Area from any damage caused by or through or in consequence of the exercise by the Grantee of its aforesaid right to explore under this Reconnaissance Permit, of any activity or activities associated with the Grantee's operations in the Permit Area, and will in so far as there is a legal liability, compensate such person or persons for any damage or losses, including but not limited to damage to the surface, to any crops or improvements, which such person or persons may suffer as a result of, arising from or in connection with the exercise of the Grantee's rights under this Reconnaissance Permit or of any act or omission in connection therewith..
- 19.2. The Grantee and its aforesaid successors and/or assigns will and do hereby undertake and bind themselves to hold harmless and indemnify the Grantor and the Agency and/or their servants and employees against any claim or claims which may be instituted by any person as a result of any injury, loss (including loss of life), damage, cost, charges, and expenses which may be suffered or sustained by any person or persons whomsoever, as a result of, or which may in any way whatsoever be associated with the exercise by the Grantee's successors and/or assigns of the right to conduct reconnaissance under this Reconnaissance Permit, and must also on request refund to the Grantor and/or Agency and/or their servants and employees all costs, charges and expenses which the Grantor and/or its servants and employees may be put to or sustain in connection with or arising out of any such claim made by such person or persons.
- 19.3. It is explicitly agreed that the Grantor and/or the Agency will not be deemed to be a partner in any Operations carried out in terms of this Reconnaissance Permit. The Grantor and/or the Agency will give notice to the Grantee and its aforesaid successors and/or assigns of any claim or occurrence in respect of which it proposes to invoke the provisions of this indemnity and will give the Grantee and its aforesaid successors and/or assigns the fullest opportunity to investigate, repudiate or otherwise deal with any matter that may arise from such claim. Provided always that nothing herein contained will be deemed to render the Grantee or its aforesaid successors and/or assigns liable for, or to indemnify the Grantor and/or the Agency and/or their servants and employees against, any compensation or damages for or with respect to injuries or damage to persons or the Permit Area directly resulting from any delict of the Grantor and/or its servants and employees during the currency of this Reconnaissance Permit, or for or in respect of any claims, demands, proceedings, damages, costs, charges and expenses in respect thereof or in relation thereto.

((There is no such proviso))

- 19.4 The Grantee shall during the term of this Reconnaissance Permit obtain and maintain for and in relation to Operations, insurance to indemnify the Agency, the Grantor and the Grantee against:
 - (a) loss or damage to any or all assets being used in connection with Operations;
 - (b) loss or damage for which the Grantee may be liable caused by pollution in the course of or as a result of Operations;
 - (c) loss or destruction of property or damage to property as well as bodily injury, sickness or death suffered by any third party for which the Grantee may be liable;
 - (d) the cost of removing wrecks and cleaning-up operations pursuant to an accident in the course of or as a result of Operations;
 - (e) The Grantee's liability to its employees, consultants and agents engaged in Operations;
 - (f) any other risk of whatever nature as is customary to insure against in the international petroleum industry in accordance with good international oil-field practices.
- The amount insured against, the type of insurance referred to in clause 19.5 above and the terms of such insurance shall be determined in accordance with good international oil-field practices.

20. Health and Safety

- 20.1 The Grantee and its successors and/or assigns will and do hereby undertake and bind themselves during the currency of this Reconnaissance Permit to observe and comply with all relevant provisions of the Mine Health and Safety Act, 1996 (Act no. 29 of 1996) and regulations in terms thereof, or any subsequent law which amends or substitutes the Mine Health and Safety Act, 1996.
- 20.2 Without detraction from sections 42 and 43 of the Act, in the event of any emergency or accident arising from Operations affecting the environment, the Grantee shall forthwith notify the Agency accordingly and take such action as may be prudent and necessary in accordance with good international oil-field practice in such circumstances.

21. Data and information to be kept confidential and limitations on public announcements

In addition to the provisions of section 83 of the Act .:

- 21.1. The Data (as defined in clause 10.3) and all programmes, tests, analyses, results, books, statements, records, returns, plans and information which the Grantee is or may from time to time be required to furnish under the provisions of this Reconnaissance Permit, shall be supplied at the expense of Grantee. Each Party shall keep confidential the Data and such information and shall not disclose it to any person without the consent of the other Party except:
 - (a) where disclosure is required by any law;
 - (b) to an Affiliate, provided such Affiliate maintains confidentiality as provided in this clause 21;
 - to the extent such Data and information is required to be furnished pursuant to any legal proceedings or because of any order of any court binding upon a Party;
 - (d) to prospective or actual contractors, consultants and attorneys employed by any Party where disclosure of such Data or information is essential to such contractor's, consultant's or attorney's work;
 - to a bona fide prospective transferee of Party's interest (including an entity with whom a Party or its Affiliate are conducting bona fide negotiations directed toward a merger, consolidation or the sale of a majority of its or an Affiliate's shares);
 - (f) to a bank or other financial institution to the extent appropriate to a Party arranging for funding;
 - (g) to the extent such Data and information must be disclosed pursuant to any rules or requirements of any government or stock exchange having jurisdiction over such Party or its Affiliates;
 - to its respective employees for the purposes of Operations, subject to the Party taking customary precautions to ensure such Data and information is kept confidential; and

(i) where any Data or information, through no fault of a Party, has become published in the public domain;

Disclosure as pursuant to clauses 22.1 (d), (e) and (f) shall not be made unless prior to such disclosure the disclosing Party has obtained a written undertaking from the intended recipient to keep the Data and information strictly confidential.

21.3 Except as may be required by applicable securities law or stock exchange requirements, no public announcement shall be made by any Party unless the other has been furnished with a copy and has given prior approval, which prompt approval shall not be unreasonably withheld. When a public announcement or statement becomes required by law or necessary or desirable because of impending danger to or loss of life, damage to property or pollution resulting from Operations, either Party is authorised to issue and make such announcement or statement without prior notice or the prior approval of the other where such prior notice and/or approval is not practical in the circumstances. In such a case, the Party making the announcement or statement shall promptly furnish the other Party with a copy of such announcement.

22. Cession, Transfer, Alienation

22.1. In terms of s70(4)(d) of the Act, this Reconnaissance Permit is not transferable, and furthermore the Grantee shall not be entitled to sub-let, alienate, sell, cede, assign, transfer, delegate or mortgage, its rights and obligations under this Reconnaissance Permit, either wholly or in part.

22.2. The Grantee shall not:

- (a) lease the rights granted to the Grantee in terms of this Reconnaissance Permit; or
- (b) allow any other person to conduct reconnaissance for petroleum in the Permit Area for such other person's own benefit and account; or
- (c) allow any other person to carry out reconnaissance operations for petroleum in the Permit Area.
- 22.3. The Grantee may from time to time appoint one or more independent contractors to carry out any work in relation to carry out reconnaissance operations, but the Grantee shall always remain liable for the compliance and observation of the provisions of the Act, the Environmental Management Programme and of the terms and conditions of this Reconnaissance Permit.

23. Law and interpretation

Subject to section 4 of the Act, and section 27 (Arbitration) of this Reconnaissance Permit, this Reconnaissance Permit, the interpretation thereof and any dispute arising thereunder or associated therewith shall be governed by and determined in accordance with the laws of the Republic of South Africa.

In the event of any inconsistency between this Reconnaissance Permit and the Act, the Act shall prevail.

In this agreement the clause headings are inserted for reference purposes only and shall not govern or affect the meaning or interpretation thereof.

24. Vis Major

- 24.1. Failure on the part of the Grantee to fulfil any of the terms and conditions provided for in this Reconnaissance Permit and to conform to the terms and provisions of the Act and the terms and provisions of such other laws and regulations as may be applicable shall not be deemed to be a breach of this Reconnaissance Permit on the part of the Grantee, in so far as such failure results from any act, cause, thing or event outside the control of the Grantee including, without limitation to the generality of the aforegoing, acts of God, war, insurrection, civil commotion, blockade, strikes, flood, storm, lightning, fire or earthquake. Financial inability, ordinary hardship and inconvenience, however caused or arising, shall not be a vis major circumstance.
- 24.2. If the Grantee by reason of the provisions of clause 24.1 above is prevented from fulfilling its obligations or of enjoying its rights under this Reconnaissance Permit, it shall immediately notify Petroleum Agency in writing to that effect and shall take all reasonable steps to remove the cause thereof and to investigate the consequences. The Grantee shall promptly notify Petroleum Agency as soon as conditions of vis major no longer prevent the Grantee from carrying out its obligations and following such notice shall resume Operations as soon as reasonably practicable.

25. Amendments

The provisions of this Reconnaissance Permit replace all prior negotiations, representations and other communication between the parties hereto and may be amended only in writing between the parties.

26. Waiver or Lenience

No waiver of any of the Agency's or the Grantor's rights and no lenience granted or permitted by the Minister or the Agency from time to time in a form which does not constitute a formal amendment of this Reconnaissance Permit shall prejudice the Minister's or the Agency's rights under this right or the Act and shall not constitute grounds for Grantee to expect that the Minister or the Agency as the case may be will continue to waive its/their rights or grant or permit lenience in the future.

27. Arbitration

- 27.1. All disputes relating to or arising out of this Reconnaissance Permit shall be finally settled under the rules of Conciliation and Arbitration of the International Chamber of Commerce by three arbitrators appointed in accordance with such rules. Subject to clause 27.2, the venue of the arbitration shall be London, England and the proceedings and records of the proceedings shall be in the English language.
- 27.2. The venue of the arbitration shall be Cape Town, South Africa if the amount in dispute is less than US\$500 0000,00 (FIVE HUNDRED THOUSAND UNITED STATES DOLLARS). However, if the relief sought is declaratory or specific relief, then the venue shall be as provided in clause 27.1.
- 27.3. Notwithstanding any such difference or dispute or reference to arbitration, the Parties shall, to the extent possible, proceed with the carrying out of their obligations under this Reconnaissance Permit, subject to the provisions of thereof.
- 27.4. The Parties irrevocably agree that the decision in those arbitration proceedings:
 - (a) shall be binding on them; and
 - (b) shall be carried into effect; and
 - (c) can be made an order of any Court of competent jurisdiction.

28. Environmental Protection

- 28.1. The Grantee hereby undertakes and binds itself to conduct all Reconnaissance Operations in accordance with the requirements of the Act and any other legislation, and the Environmental Management Programme submitted by the Grantee in terms of section 36(1) read with section 65 of the Act and which has been approved by the Minister, in a manner that would facilitate the protection and conservation of the natural resources of the Republic of South Africa and of the environment in general.
- 28.2. Subject to the provisions of the Act, the Grantee shall on the expiration or termination of this Reconnaissance Permit:
 - (a) remove all equipment and installations from the Permit Area; and
 - (b) perform all necessary site rehabilitation and shall take all other actions necessary to minimise hazards to human or other life or the environment in general for time to come.

29. Costs and VAT

- 29.1. All expenses, fees, costs stamp and transfer duties flowing from this Reconnaissance Permit direct or incidental, and Value Added Tax at the prevailing rate and stamp duty, shall be paid by the Grantee.
- 29.2. All remittances and payments to the Agency in terms of this Reconnaissance Permit are exclusive of Value Added Tax (VAT).

30. Severability

Any provision of this Reconnaissance Permit which is in any way unenforceable or which may render it unenforceable or which may contravene the applicable laws or regulations of any jurisdiction to which this Reconnaissance Permit is subject shall be deemed, to the extent of such unenforceability or contravention, severable from this Reconnaissance Permit and of no force or effect and shall not affect any other provision of this Reconnaissance Permit and its enforceability or legality in all other respects.

31. Notices, domicilia citandi et executandi and Representatives

All notices, requests and reports provided for herein shall be in writing and shall be delivered either by hand to an authorised representative of the receiving Party, or sent by courier or telefax to the following addresses in the Republic of South Africa, provided that if given by telefax a copy thereof shall be sent immediately by prepaid registered airmail:

Grantor:	*	
Minister of Minerals and Energy	g	
Physical address :	Postal address :	
Mineralia Centre Cnr Visagie & Andries Streets PRETORIA	Private Bag X59 PRETORIA 0001	
Tel number : (27) 12 317 9000	Fax number : (27) 12 322 3416	s e 2 ×
PETROLEUM AGENCY:		* ************************************
Attention : Chief Executive		
Physical address:	Postal address :	*_
Petroleum House 151 Frans Conradie Drive PAROW Western Cape	P O Box 1174 PAROW 7499	H G
Tel number : (27) 21 938 3500	Fax number : (27) 21 938 3520	
GRANTEE :		42
Attention : Mr	or	8 K ₈
Physical address :	Postal address:	70 %
Tel number :	Fax number :	_
		in the same care

Either Party may change its address to such other address as it may notify the other Party of on at least 30 (thirty) days prior notice.

All notices, requests and reports sent by telefax shall be deemed to have been received within 12 (twelve) hours of transmission. Those delivered by hand or sent by courier shall be deemed to have been received at the time of actual delivery.

Each Party also chooses the physical address specified above as its domicilium citandi et executandi for all purposes arising under this Reconnaissance Permit, including service of process.

The Grantee shall within 7 (SEVEN) calendar days of the Effective Date of this Reconnaissance Permit give notice to the Agency in which it appoints one representative with whom Petroleum Agency may

deal concerning this Reconnaissance Permit. Such representative shall continue to represent the Grantee unless the Grantee notifies the Agency of a change of representative.

Thus done and sig	ned at	on th	e day of		20 in th	e presence of the
undersigned witnes	ses:	7 - 2			Ware and American	
		2 4 4 4 4 4		5.8 1		
AS WITNESSES:	39		0			ere E
				00		
1			81 N	2 B		
* *				a	3	
2		***************************************			*	
		for and on behalf	of the Agency	A #		
	379	E W				
Thus done and sig	rned et	on the	day of	*	00 :- *-	
undersigned witnes	ses:	On the	uay or		20 in the	presence of the
			W.		+	
AS WITNESSES:	70	32		A contract to		F
			¥ 14	37	2 P 10 W P	
1,	•••••	- a			50	32 ₁₀
	ta p		8.	# # # # # # # # # # # # # # # # # # #		
2		for and o	n behalf of the	Grantee		
		.0. 4.14	ni bonan oi mo	Granice		
			ANNEXURE A		*	
(d)	%. 1	MAP AND DESCR	A 7:	E PERMIT AREA	V	
					™ pr	8
	# **	,	NNEXURE B		· ·	
		RECONNAISSA	NCE WORK P	ROGRAMME		
ří.		4	NNEXURE C	12 m	15	NT BY
	ACCOUNT TO		and the second s	DANITOD 400		85 UF
	ACCOUNT TO	O WHICH PAYME!	VIS TO THE G	HANTOH ARE	TO BE MADE	
% 2	The amounts Agency at:	payable to the Age	ency will be dep	osited by the Gr	antee into the acc	count of the
	Bank:	ABSA				3%
	Branch Branch code		7.7.7.			
	Account no.		···			a 11 101
	In the case of p	ayments originatin	g in the USA th	ne following furth	er information will	be of assistance:
e ti					N.	
- 1a	PAYMENT TO	DE MADE 10;	1 9	Harris Bank	(International	
	Full name of B	ank	20 e	Corporation		
a o ⁵	Address			3 Times Sq	uare	
	9 845-2			New York		
		4		N1	10000	42 808

SWIFT Address

HATRUS33

Account Number

49-0-2945-7

For Credit of

ABSA Bank Limited

International Banking Services:

Corporate Pretoria

South Africa

For Further Credit **Under Reference**

CEF (PTY) LTD

SEF 8245Y55

The remitting Bank must confirm payments by

MT100 to:

ABSA Bank Limited

IBS Corporate

Pretoria

South Africa

SWIFT Code

For Attention

ABSAZAJJACPT

Amanda Reyneke

Kindly confirm payments per Fax to CEF (PTY) LTD Attention Dolf Jonker/Kobus Geertsema Fax No. (27)11 784 5494

In all cases, the Grantee is required to deliver proof of payment to the Agency within 3 working days of effecting the payment.

FORM KK:

TECHNICAL CO-OPERATION PERMIT

Index

P	ream	ble

- 1. Definitions
- 2. Description of the Permit Area
- 3. Duration
- 4. Rights and Obligations of the Grantee
- 5. Fees
- 6. Technical Advisory Committee
- 7. Minister's Right to Cancel or Suspend the Technical Co-operation Permit
- 8. No guaranteed right to other minerals
- 9. Examination of the Permit Area and Operations
- 10. Reports and Samples to be Furnished
- 11. Technical Work Programme and Budget
- 12. Black Economic Empowerment
- 13. Good international practice, pollution, navigation, fishing and conservation
- 14 Local labour, equipment, materials and contractors
- 15. Existing Data
- 16. Data and information to be kept confidential and limitations on public announcements
- 17. Cession, Transfer, Allenation
- 18. Law and interpretation
- 19. Vis Major
- 20. Amendments
- 21. Waiver of Lenience
- 22. Arbitration
- 23. Costs and VAT
- 24. Severability
- 25. Notices, domicilia citandi et executandi and Representatives

ANNEXURES

- A Map and description of the Permit Area
- B Technical Work Programme
- C Account to which payments to the Agency/Grantor are to be made

D Existing Data

SPECIMEN AGREEMENT FOR A TECHNICAL CO-OPERATION PERMIT

entered into between

THE REPUBLIC OF SOUTH AFRICA

(hereinafter referred to as "the Grantor") and herein represented by its Minister of Minerals and Energy (hereinafter referred to as "the Minister"), she/he being duly authorised by virtue of the provisions of section 70(1) read with section 66(1)(d) of the Mineral and Petroleum Resources Development Act, 2002 (Act No. xx of 2002)

and

H		
3	Registration No. []	
(hereinafter referred to as	"the Grantee") a company incorporated and registered in	
under the laws of	and having its registered address at	and herein
represented byto	he being duly authorised by virtue of a resolution of	directors of Grantee passed at
	in respect of Block	
	REPUBLIC OF SOUTH AFRICA	

(the Grantee and the Grantee are jointly referred to as "Parties" and individually as "Party")

WHEREAS

- In terms of section 66(1)(a) of the Mineral and Petroleum Resources Development Act, 2002 (Act No. ____ of 2002)
 ("the Act") it is the function of the Agency to promote exploration for petroleum in the Republic of South Africa;
- In terms of section 66(1)(d) of the Act the Minister is authorised to issue Technical Co-operation permits on behalf of the Minister for the purpose of promoting exploration for petroleum within the onshore and offshore territory of the Republic of South Africa, the latter in such maritime zones over which the State has jurisdiction, including but not limited to the territorial waters, the continental shelf, and the exclusive economic zone, all as envisaged in the Sea-Shore Act, 1935 (Act No. 21 of 1935), and in the Maritime Zones Act, 1994 (Act No. 15 of 1994) to which end it will undertake all necessary marketing, promotion, advertising and publicity;
- The Grantee has identified an area which it believes may hold significant potential for discovery of petroleum (the Permit Area, more fully described in clause 2 below) and desires to have the opportunity to study, analyse and interpret the Existing Data (as more fully described in clause 2) relating to the Permit Area;
- The Grantor has agreed to issue to the Grantee a Technical Co-operation Permit by way of and being this Technical Co-operation Permit Agreement to enable the Grantee to carry out certain studies, analyses and interpretations ("the Technical Work") on the Existing Data which can be expected substantially to increase the understanding of the underlying geology and petroleum potential of the Permit Area in accordance with the terms and conditions recorded hereunder.
- In recognition of this contribution to the increased understanding of the geology and petroleum potential of the Permit Area that will result from diligent execution of the Technical Work by the Grantee, the Grantor undertakes

to negotiate with no other entity for the granting of any form of Right relating to petroleum over any portion of the Permit Area for the duration of this Permit and for so long thereafter as the Grantor in its sole discretion considers that negotiation between the Parties over the terms of an Exploration Right relating to some portion or all of the Permit Area are progressing satisfactorily.

 The Grantor has appointed South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited as its designated representative for the purposes of this agreement

NOW THEREFORE THE PARTIES AGREE AS FOLLOWS:

1. Definitions

The definitions in the Act shall apply also to this Agreement and additionally the following words and expressions shall have the meaning assigned to them:

- 1.1. "the Act" shall mean the Mineral and Petroleum Resources Development Act, 2002 (Act No. ... of 2002) or any subsequent law which amends or substitutes the Act;
- 1.2. "Affiliate" of a Party shall mean a company which owns, or is owned by, or which is owned by an entity which owns that Party; "owns" and "owned" mean in this definition the ownership directly or indirectly of more than 50 % of the voting shares or members' interests;
- 1.3 "the Agency" shall mean South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited, known as Petroleum Agency SA;
- 1.4. "Effective Date", shall mean the date on which this Technical Co-operation permit agreement is signed by the last Party to sign it or such later date being the date upon which the data, as listed in Annexure C (other than the seismic field data referred to in clause 4.1.4), are received by Grantee;
- 1.5. "the Grantee", shall bear the meaning attributed thereto in the definition of the Parties above;
- 1.6. "the Grantor" shall mean the State:
- 1.7. "the Minister", shall bear the meaning attributed thereto in the description of the Parties above;
- 1.8. "the Permit Area" means the area referred to in clause 2 of this Agreement, excluding parts relinquished or abandoned from time to time;
- 1.9. "the Technical Work Programme", shall mean the Grantee's written recordal of the manner in which it intends to carry out the Technical Work in relation to Existing Data, and shall include all amendments and supplementary submissions relating to the Technical Work Programme which may be made by the Grantee and approved in writing by the Grantor from time to time;
- 1.10 "State", shall mean the Republic of South Africa;

Reference to the singular of a defined term includes a reference to it in the plural and vice versa and reference to any gender includes a reference to all other genders.

2. Description of the Permit Area

The Permit Area comprises approximately _____ km² in the West / South / East* Coast Region offshore, being the area of land or sea including the seabed, identified as such by co-ordinates on the map attached as Annexure A hereto.

3. Duration

3.1 Unless this permit is suspended, cancelled or abandoned it shall, subject to acceptance thereof by Grantee, where provided at the foot of this permit, be valid from the Effective Date for a period of [] months thereafter.*

- * In terms of section 72(4)(b) of the Act, this period may not exceed one year.
- 3.2 In terms of sections 72(4)(c) and (d) of the Act, this permit is not transferable and not renewable.

4. Rights and Obligations of the Grantee

- 4.1 The Grantor hereby in terms of section 72(1) read with section 66(1)(d) of the Act issues to the Grantee and the Grantee hereby accepts the Technical Co-operation Permit constituted by this Agreement, conferring the exclusive right to negotiate the terms of an Exploration Right to be granted over some or all of the Permit Area, subject to satisfactory completion of the Technical Work Programme attached hereto as Annexure B. For the avoidance of doubt it is recorded that this exclusive right to negotiate the terms of such an Exploration Right does not imply that the Parties must reach agreement on the said terms and if the Parties fail to reach agreement on the aforesaid terms then no Exploration Right will be awarded to the Grantee regardless of satisfactory conclusion of the Technical Work Programme.
- 4.2 The Grantee acknowledges that all technical data, reports, maps, samples and information generally (defined in clause 10.4 as "the Data") that arise from the carrying out of the Technical Work are the property of the Grantor and that on termination of this permit agreement through expiry, lapse, abandonment or howsoever else, the Grantor shall be free to sell, give, disclose or otherwise dispose of the Data without restriction.
- 4.3 The Grantee shall have no special right of access to the Permit Area arising from this permit agreement.
- 4.4 Without detracting from the Grantee's further obligations in terms of this Technical Co-operation Permit Agreement or in terms of the Act the Grantee shall:
- 4.4.1 actively carry out the Technical Work Programme in respect of petroleum on the Permit Area;
- 4.4.2 comply with the terms and conditions of this Technical Co-operation Permit Agreement and the relevant provisions of the Act;
- 4.4.3 pay the prescribed technical co-operation fee to the Grantor;
- 4.4.4 submit, timeous, complete and correct summary progress reports to the Grantor on a monthly basis; and
- 4.4.5 supply to the Grantor digital and, where appropriate, paper copies of the Data as soon as possible after completion of each phase of the Technical Work.

5. Fees

- 5.1 All fees, considerations and the like which are payable by Grantee to the Grantor and/or the Grantor in terms of this Technical Co-operation Permit Agreement shall be paid by deposit into the account of the Agency at the bank and account indicated in Annexure C, or such other bank and account as the Grantor may from time to time give notice of. The onus for timely payment rests solely with the Grantee; the Grantor will not normally issue invoices.
- 5.2. If any fees, considerations and the like payable by the Grantee to the Grantor in terms of this Technical Cooperation Permit are not paid punctually on or before the due dates for payment, the Grantee shall be in mora debitoris and shall pay interest thereon at the rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act No. 1 of 1999) reckoned from the due date for payment to the date of actual payment.
- 5.3. Regardless of any relinquishment of parts of the Permit Area or early termination of this Technical Cooperation Permit Agreement for whatever reason, no portion of the fees referred to in clause 5.1 shall be refundable.

6. Technical Advisory Committee

- 6.1. The Grantor and the Grantee shall as soon as possible after the Effective Date establish a committee to be known as the Technical Advisory Committee which shall consist of:
 - (a) a chairman and one other person appointed by the Grantor; and
 - (b) two other persons appointed by the Grantee.

- 6.2. The Grantor and the Grantee may appoint by notice an alternative member to act in the place of either or both of their representatives on the Technical Advisory Committee during his or her absence or incapacity to act as a member of the Committee. When an alternate member acts in the place of any member he or she shall have the powers and perform the duties of such member.
- 6.3. Without prejudice to the rights and obligations of the Grantee in relation to the management of the Technical Work the functions of the Technical Advisory Committee shall be:
 - to review all Technical Work carried out by the Grantee and to provide the Grantee advice and recommendations with respect thereto;
 - to review any proposed Technical Work Programme to be submitted by the Grantee to the Grantor in terms of clause 11;
 - (c) to review the maintenance of operating records and reports kept in connection with the Technical Work; and
 - (d) to offer advice to the Grantee in order to promote the efficient carrying out of the Technical Work.
- 6.4. Meetings of the Technical Advisory Committee shall be held at least semi-annually. The location of such meetings shall be in Cape Town, South Africa unless otherwise unanimously agreed by the members of the committee.
- 6.5. Three members of the Technical Advisory Committee shall form a quorum for a meeting of the Committee, provided that at least one representative of Grantee and one representative of the Grantor is present.
- 6.6. Any member of the Technical Advisory Committee shall have the right to call any expert to any meeting of the Committee to advise the Committee on any matter of a technical nature requiring expert advice.
- 6.7. The Chairman of the Committee shall cause minutes of each meeting to be kept and circulated to the members.

7. Minister's Right to Cancel or Suspend the Technical Co-operation Permit

- 7.1 It is recorded that the Minister is empowered to cancel or suspend this Technical Co-operation Permit in the circumstances envisaged in and in accordance with the provisions of section 85 read with section 44 of the Act.
- 7.2. In the event of this Technical Co-operation Permit being cancelled in accordance with the provisions of clause 7.1, the Grantee will not be absolved from its obligations or liabilities which have accrued up to the date of such cancellation.
- 7.3. The cancellation of this Technical Co-operation Permit in terms of clause 7.1 will be without prejudice to the Grantor's or the Grantor's other rights under this Technical Co-operation Permit or in statutory or common law of the Republic of South Africa, including its right to recover any damages it may have suffered or may suffer, its right to claim specific performance or its right to claim ancillary relief.

8. No guaranteed right to other minerals

8.1. Should any mineral or minerals be discovered in the course of the Technical Work, the Grantee shall be bound to report such discovery to the Minister. The Grantee may thereafter, subject to any prior rights, apply for and, if granted the required prospecting rights by the State, prospect for such mineral or minerals.

9. Examination of the Permit Area and Operations

In terms of sections 86 to 89 of the Act, all read with section 65 of the Act, the Minister may authorise inspections and exercise of related powers more fully therein set forth.

10. Reports and Samples to be Furnished

- 10.1 For the duration of the Study the Grantee shall submit to the Grantor:
- 10.1.1 written progress and status reports on progress with the seismic reprocessing starting from 3 (three) months after receipt by the Grantee of the data listed in Annexure D, and monthly thereafter general progress and status reports; and

- 10.1.2 within 30 (thirty) days from the expiry of this permit agreement, provide the Grantor with two copies of the final report (which shall be a comprehensive assessment of the studies and interpretations), together with copies of the results of any reprocessing of seismic data, together with film images and digital records. Such final report shall be presented and delivered by the Grantee to the Grantor at its offices in Parow, South Africa.
- 10.3. The Grantee shall furnish the Grantor with such other information, maps and plans, as to the progress of the Technical Work in relation to the Permit Area as the Grantor may from time to time require and which the Grantee can reasonably make available.
- 10.4. Copies of all maps, magnetic tapes and all other geological and geophysical information and interpretation obtained by the Grantee in the course of carrying out the Technical Work (hereinafter referred to as "Data") shall be given to the Grantor as soon as they are acquired or prepared.
- 10.5. The Grantee may export such Data for processing or laboratory examination or analysis by the Grantee or by third parties or storage outside of the Republic of South Africa, provided that copies of equivalent quality have first been delivered to the Grantor and provided that the Grantor's prior written approval has been obtained.
- 10.6. Within three (3) months after the termination and/or cancellation and/or abandonment of this Technical Cooperation Permit, the Grantee shall furnish the Grantor with all results of the Technical Work in relation to the Permit Area.
- 10.7. Nothing in this clause contained shall be construed as requiring the Grantee to disclose any of its proprietary technology or that of its Affiliates.
- 10.8 The above provisions are additional to and without detraction from the provisions of Regulations 7 and 8 made in terms of the Act.
- The Grantor shall have the right to use all data and reports for internal evaluation of the petroleum potential of the Permit Area, and to disclose portions of the said data and reports to bona fide potential licensees of the data and/or potential applicants for a Technical Co-operation Permit or Exploration Right or Production Right in relation to the portion of the Permit Area to which the disclosed data relates.

11. Technical Work Programme and Budget

- 11.1. The Grantee must, to the satisfaction of the Grantor over the period of this Technical Co-operation Permit, initiate and thereafter diligently pursue activities intended to meet the commitments set out in the Technical Work Programme annexed hereto and marked Annexure B.
- 11.2. It is recorded that the time sequences as set out in the Technical Work Programme will commence on the Effective Date.

12. Black Economic Empowerment

The Grantee acknowledges that in order to give effect to Section 2(d) of the Act, the Grantor is desirous of expanding opportunities for historically disadvantaged persons (as defined in the Act) to enter the upstream petroleum industry and to benefit from the exploitation of petroleum resources in the Republic of South Africa and as such Grantee undertakes to use its best endeavours to seek historically disadvantaged South African partners to participate in this Technical Co-operation permit and any Exploration Right that may flow therefrom.

13. Good international practice

In addition to the provisions of the Act, the Grantee undertakes to execute all Technical Work in a proper and workmanlike manner in accordance with methods customarily used in good international practice.

14 Local labour, equipment, materials and contractors

- 14.1. In furtherance of section 2(f) of the Act, in carrying out the Technical Work the Grantee shall, to the extent reasonably possible, employ South African citizens having appropriate qualifications.
- 14.2 The Grantee:
- 14.2.1 and its contractors shall use and purchase equipment, machinery, materials, instruments, supplies and accessories (all referred to as "Goods") manufactured, produced or available in the Republic of South Africa for use under this Technical Co-operation Permit, provided that such Goods are reasonably

comparable with like goods manufactured, produced or available outside the Republic of South Africa as to landed cost, quantity and quality and can be made available at the time when and the place where required;

14.2.2 shall to the extent reasonable, utilise contractors in the Republic of South Africa where services of comparable standards with those obtained elsewhere are available from such contractors at competitive prices and on competitive terms and can be made available at the time when and the place where required.

15 Existing Data

The	Grantor	will	procure	that	the	Agency	will,	for	а	conside	ration	of	US\$
(t	Inited S	States I	Dollars)	exclusi	ve of t	the c	costs of di	igital m	edia	(magnetic tapes, etc.
and :	shipping co	ost, ma	ake availal	ble for i	use so	lely by th	he Gran	ntee th	he e	xisting da	ta as i	ndica	ted in (a) to (f) below
The	considerat	ion sh	all be paid	to the	Grant	or within	n 30 (th	nirty) (days	from the	Effect	ive D	ate. Costs for digital
med	a and ship	pping	will be ch	arged a	at cost	plus 10	0 % an	d sha	II be	paid wit	thin 30	(thir	ty) days of receipt of
invoi	ce by the	Grante	e. The G	rantee	agree	s that w	hile the	Gran	ntor	has made	all re	ason	able efforts to ensur
the c	uality of th	e exis	ting data,	the Gra	antor s	hall not	be liabl	e for	the i	ntegrity, a	accura	cv or	quality of the existin
data.	The Gran	ntor ac	cepts no	liability	for the	results	of any	decis	ions	or action	s of th	e Gra	antee arising from the
Gran	tee's use o	of the e	existing da	ta. Su	ch data	a shall in	clude,	but as	s ava	ailable no	t be lin	nited	to the following:

- (a) one copy of all available borehole measurements and analyses including sample logs, wireline logs, gas chromatograms, micropalaeontological, petrographical, and geochemical data and / or reports (including core descriptions) as well as one digital tape copy, where available (in industry-standard format), of all wireline logs from all wells drilled in the Exploration Area;
- (b) one copy of each technical report of the wells drilled in the Exploration Area, regional reports for the West / South / East Coast and general reports for RSA offshore operations, as appropriate to the Exploration Area;
- (c) one copy of each seismic line which intersects the Exploration Area, at the vertical scale of 2.5 inches per second (two-way time) or as imaged, as well as one copy of [______] km of lines to be selected by the Grantee at a scale of 5 inches per second (two-way time). The Grantor will also provide location data for all seismic lines in SEG-P-1, UKOOA or ASCII format, in digital form;
- (d) field data for up to [] km of seismic lines in the Exploration Area as available and as selected by the Grantee. This data set shall be defined by the Grantee after its preliminary evaluation of the copies of seismic lines described in (c) above. The Grantee will provide the Grantor with copies of the reprocessed data as both digital records in industry standard format and a single paper copy. Where such field tapes are available only in a format other than digital records on 9-track half-inch tape (or more modern format) the Grantor will make available such original field data tapes so that the Grantee can transcribe the data into a current industry-standard format. The Grantee shall provide the Grantor with a copy of the data so transcribed onto media of an industry-standard format which is current at the time, and which the Grantor is equipped to read and, at the Grantor's request, return the original tapes to the Grantor at the Grantee's sole cost, risk and expense.
- (e) stacked and migrated data for up to [] km of seismic lines, as available, as selected by the Grantee;
- (e) access to available sets of conventional cores, sidewall cores, drilling cuttings from the wells in the Exploration Area for sampling; and

Wherever possible documents will be provided as digital images in a format appropriate to the document.

Additional field data tapes and stacked and migrated data tapes, over and above that detailed in clauses (d) and (e) above will be made available upon request by Grantee provided that Grantee shall reimburse the Grantor for the reasonable and customary costs incurred in preparing such tapes.

The Grantor will co-operate with the Grantee at the Grantee's request by assisting in resolving technical problems relating to the data (but not interpretation) upon terms and conditions then to be agreed.

All the aforesaid data, studies and reports (hereinafter referred to as the "Existing Data") provided to the Grantee are owned by the Grantor and are of considerable commercial value to it. Such Existing Data shall remain the property of the Grantor and shall, on termination of this Exploration Right, be returned at Grantee's cost to the Grantor. The Grantee will hold the Existing Data confidential between itself and the Grantor in the manner described in clause 16.1 in relation to the Data with the exception of items 16.1(e) and 16.1(f), in which instances the Grantee must first acquire The Grantor's prior written approval, and 16.1(b)

16. Data and information to be kept confidential and limitations on public announcements

In addition to the provisions of section 83 of the Act,:

- 16.1. The Data (as defined in clause 10.4) and all programmes, tests, analyses, results, books, statements, records, returns, plans and information which the Grantee is or may from time to time be required to furnish under the provisions of this Technical Co-operation Permit, shall be supplied at the expense of Grantee. Each Party shall keep confidential the Data and such information and shall not disclose it to any person without the consent of the other Party except:
 - (a) where disclosure is required by any law;
 - (b) to an Affiliate, provided such Affiliate maintains confidentiality as provided in this clause 16;
 - to the extent such Data and information is required to be furnished pursuant to any legal proceedings or because of any order of any court binding upon a Party;
 - (d) to prospective or actual contractors, consultants and attorneys employed by any Party where disclosure of such Data or information is essential to such contractor's, consultant's or attorney's work;

 - (f) to the extent such Data and information must be disclosed pursuant to any rules or requirements of any government or stock exchange having jurisdiction over such Party or its Affiliates;
 - (g) to its respective employees for the purposes of Operations, subject to the Party taking customary precautions to ensure such Data and information is kept confidential;
 - (h) where any Data or information, through no fault of a Party, has become published in the public domain;
 and

Disclosure as pursuant to clauses 16.1 (d) and (e) shall not be made unless prior to such disclosure the disclosing Party has obtained a written undertaking from the intended recipient to keep the Data and information strictly confidential.

Except as may be required by applicable securities law or stock exchange requirements, no public announcement shall be made by any Party unless the other has been furnished with a copy and has given prior approval, which prompt approval shall not be unreasonably withheld. When a public announcement or statement becomes required by law or necessary or desirable because of impending danger to or loss of life, damage to property or pollution resulting from Operations, either Party is authorised to issue and make such announcement or statement without prior notice or the prior approval of the other where such prior notice and/or approval is not practical in the circumstances. In such a case, the Party making the announcement or statement shall promptly furnish the other Party with a copy of such announcement.

17. Cession, Transfer, Alienation

- 17.1. In terms of section 72(3)(c) of the Act, this Technical Co-operation Permit is not transferable, and furthermore the Grantee shall not be entitled to let, sub-let, alienate, sell, cede, assign, transfer, delegate or mortgage, its rights and obligations under this Technical Co-operation Permit, either wholly or in part without the prior written approval of the Minister.
- 17.2 The Minister hereby approves that the Grantee may cede, transfer, let, sub-let, assign or otherwise dispose of or mortgage, either wholly or in part, its rights and obligations under this Exploration Right, to an Affiliate but only if the Grantee's guarantor has bound itself to guarantee the obligations of the transferee. The Grantee shall advise the Grantor by notice within 10 days of effecting such transaction.
- 17.3. The Grantee may from time to time appoint one or more independent contractors to carry out any portion of the Technical Work, but the Grantee shall always remain liable for the compliance and observation of the provisions of the Act and of the terms and conditions of this Technical Co-operation Permit.

18. Law and interpretation

18.1 Subject to section 4 of the Act, and section 22 (Arbitration) of this Technical Co-operation Permit, this Technical Co-operation Permit, the interpretation thereof and any dispute arising thereunder or associated therewith shall be governed by and determined in accordance with the laws of the Republic of South Africa.

- 18.2 In the event of any inconsistency between this Technical Co-operation Permit and the Act, the Act shall prevail.
- 18.3 In this agreement the clause headings are inserted for reference purposes only and shall not govern or affect the meaning or interpretation thereof.

19. Vis Major

- 19.1. Failure on the part of the Grantee to fulfil any of the terms and conditions provided for in this Technical Cooperation Permit and to conform to the terms and provisions of the Act and the terms and provisions of such
 other laws and regulations as may be applicable shall not be deemed to be a breach of this Technical Cooperation Permit on the part of the Grantee, in so far as such failure results from any act, cause, thing or event
 outside the control of the Grantee including, without limitation to the generality of the aforegoing, acts of God,
 war, insurrection, civil commotion, blockade, strikes, flood, storm, lightning, fire or earthquake. Financial
 inability, ordinary hardship and inconvenience, however caused or arising, shall not be a vis major
 circumstance.
- 19.2. If the Grantee by reason of the provisions of clause 19.1 above is prevented from fulfilling its obligations or of enjoying its rights under this Technical Co-operation Permit, it shall immediately notify the Grantor in writing to that effect and shall take all reasonable steps to remove the cause thereof and to investigate the consequences. The Grantee shall promptly notify the Grantor as soon as conditions of vis major no longer prevent the Grantee from carrying out its obligations and following such notice shall resume the Technical Work as soon as reasonably practicable.

20. Amendments

The provisions of this Technical Co-operation Permit replace all prior negotiations, representations and other communication between the parties hereto and may be amended only in writing between the parties.

21. Waiver of Lenience

No waiver of any of the Agency's or the Grantor's rights and no lenience granted or permitted by the Minister or the Agency from time to time in a form which does not constitute a formal amendment of this Technical Co-operation Permit shall prejudice the Minister's or the Agency's rights under this right or the Act and shall not constitute grounds for Grantee to expect that the Minister or the Agency as the case may be will continue to waive its/their rights or grant or permit lenience in the future.

22. Arbitration

- 22.1. All disputes relating to or arising out of this Technical Co-operation Permit shall be finally settled under the rules of Conciliation and Arbitration of the International Chamber of Commerce by three arbitrators appointed in accordance with such rules. Subject to clause 22.2, the venue of the arbitration shall be London, England and the proceedings and records of the proceedings shall be in the English language.
- 22.2. The venue of the arbitration shall be Cape Town, South Africa if the amount in dispute is less than US\$500 0000,00 (FIVE HUNDRED THOUSAND UNITED STATES DOLLARS). However, if the relief sought is declaratory or specific relief, then the venue shall be as provided in clause 27.1.
- 22.3. Notwithstanding any such difference or dispute or reference to arbitration, the Parties shall, to the extent possible, proceed with the carrying out of their obligations under this Technical Co-operation Permit, subject to the provisions of thereof.
- 22.4. The Parties irrevocably agree that the decision in those arbitration proceedings :
 - (a) shall be binding on them; and
 - (b) shall be carried into effect; and
 - (c) can be made an order of any Court of competent jurisdiction.

23. Costs and VAT

- All expenses, fees, costs stamp and transfer duties flowing from this Technical Co-operation Permit direct or 23.1. incidental, and Value Added Tax at the prevailing rate and stamp duty, shall be paid by the Grantee.
- 23.2. All remittances and payments to the Grantor in terms of this Technical Co-operation Permit are exclusive of Value Added Tax (VAT).

24. Severability

Any provision of this Technical Co-operation Permit which is in any way unenforceable or which may render it unenforceable or which may contravene the applicable laws or regulations of any jurisdiction to which this Technical Co-operation Permit is subject shall be deemed, to the extent of such unenforceability or contravention, severable from this Technical Co-operation Permit and of no force or effect and shall not affect any other provision of this Technical Co-operation Permit and its enforceability or legality in all other respects.

25. Notices, domicilia citandi et executandi and Representatives

All notices, requests and reports provided for herein shall be in writing and shall be delivered either by hand to ollowing addresses in sent immediately by

Grantor:		
Minister of Minerals and Energy		£:
Physical address:		Postal address :
Mineralia Centre Cnr Visagie & Andries Streets		Private Bag X59 PRETORIA
PRETORIA Gauteng Province		0001
Tel. number : (27) 12 317 9000		Fax number : (27) 12 322 3416
PETROLEUM AGENCY SA:		· · · · · · · · · · · · · · · · · · ·
Attention : Chief Executive		
Physical address :		Postal address :
Petroleum House 151 Frans Conradie Drive PAROW Western Cape Province		P O Box 1174 PAROW 7499
Tel. number : (27) 21 938 3500	a 9	Fax number : (27) 21 938 3520
GRANTEE:		# 2
LIMITED		
Attention : Mr or		
Physical address:	8	Postal address :
		7
		· · · · · · · · · · · · · · · · · · ·
Tel. number :		Fax number :

Either Party may change its address to such other address as it may notify the other Party of on at least 30 (thirty) days prior notice.

All notices, requests and reports sent by telefax shall be deemed to have been received within 12 (twelve) hours of transmission. Those delivered by hand or sent by courier shall be deemed to have been received at the time of actual delivery.

Each Party also chooses the physical address specified above as its domicilium citandi et executandi for all purposes arising under this Technical Co-operation Permit, including service of process.

The Grantee shall within 7 (SEVEN) calendar days of the Effective Date of this Technical Co-operation Permit give notice to the Grantor in which it appoints one representative with whom the Grantor may deal concerning this Technical Co-operation Permit. Such representative shall continue to represent the Grantee unless the Grantee notifies the Grantor of a change of representative.

Tage to sent the contract of

And the second of the second s	dera, i i- a gaganaka			9 100 V	92 30 W W	
Thus done and signed at	w 184 . 2	on the		. F. 1957.	NO. 10 10 10 10 10 10 10 10 10 10 10 10 10	e presence of the
	* * * * * * * * * * * * * * * * * * * *		7 × 4 ×	Mai at	22 E 12 10	26 18
AS WITNESSES:	*			-, -,	#) 	
				i .	1 500 16 ¹	
1	K (4	g 1 6 8 6	39 (2)	W 3	**	
2		Mark t	at 10 2 8 10			
A 20 10000000000000000000000000000000000		behalf of the			-4, -	
		11.5	060 (6	*		
Thus done and signed at undersigned witnesses:	73 OM DO N O	on theda	ay of	***	_20 in the	presence of the
		7	g e	£3	- II. 0	W.
AS WITNESSES:	444		g = 12	78		4
1					2 ⁴⁷⁰	
•						
4.	fo	r and on beha	If of the Grant	 ee		
				* T		

ANNEXURE A MAP AND DESCRIPTION OF THE PERMIT AREA

ANNEXURE B TECHNICAL WORK PROGRAMME

In order to achieve the objectives as set out in the Technical Co-operation Permit Agreement to which this Technical Work Programme forms Annexure B the Grantee shall, within the term of this Agreement:

- carry out a study and evaluation as detailed in clause 3 of the petroleum potential of the Permit Area through integrated interpretation of all the geological and geophysical data provided by the Agency relating to the Permit Area. This process will result in the completion of a report which identifies areas within the Permit Area that exhibit, in the opinion of the Grantee, the highest petroleum I potential (hereinafter referred to as the "Study"). The Study shall include:
- 1.1.1 a comprehensive geological evaluation of samples and other data provided by the Agency pursuant to clause 15 of the main agreement. Such studies will include a petrographic study, a tectonic study, and source-rock and thermal modelling studies of the Permit Area.;

- 1.1.2 a comprehensive geophysical study including the reprocessing of at least [] km of seismic data, and special processing as the Grantee deems necessary;
- 1.1.3 interpretation of the seismic data and map selected horizons;
- 1.1.4 preparation of a geological model of the Permit Area, incorporating results of geological and geophysical studies referred to in clauses 1.1 and 1.2; and
- 1.1.5 preparation a technical report on the results of the Study, making recommendations for future exploration work.
- 2 All Studies, including but not limited to the geophysical, geological and engineering work, seismic reprocessing, interpretation, and evaluation will be executed by the Grantee at its sole cost, risk and expense and without any cost to the Grantor.
- 3 The Grantee shall conduct such Studies utilising its own resources and such third party contractors as it deems necessary.

ANNEXURE C

ACCOUNT TO WHICH PAYMENTS TO AGENCY ARE TO BE MADE

The amounts payable to the Agency will be deposited by the Grantee into the account of the Agency at:

In the case of payments originating in the USA the following further information will be of assistance:

PAYMENT TO BE MADE TO:

Full name of Bank

Address

Harris Bank International

Corporation

3 Times Square

New York

New York, 10036

SWIFT Address

HATRUS33

Account Number

49-0-2945-7

For Credit of

ABSA Bank Limited

International Banking Services:

Corporate

Pretoria South Africa

For Further Credit

Under Reference

CEF (PTY) LTD

SEF 8245Y55

The remitting Bank must confirm payments by

MT100 to:

ABSA Bank Limited

IBS Corporate

Pretoria

South Africa

SWIFT Code

For Attention

ABSAZAJJACPT

XXXXXXXXXXX

In all cases, the Grantee is required to deliver proof of payment to the Grantor within 3 working days of effecting the payment.

ANNEXURE D

In terms of clause 10 of the Technical Co-operation Permit Agreement to which this Annexure D is attached, the Grantor will make available the following Data:

FORM LL: EXPLORATION RIGHT

INDEX

n	rea	 -1	-

30.

31.

Law

Grantee's Rights and Obligations Ensured

Prean	ble	
1.	Definitions	
2.	Grantor's Agent: Petroleum Agency	
3.	Description of Exploration Area	16.2
4.	Exploration Right	19 10
5.	Ancillary Rights and Obligations	
6.	Commencement, Duration and Renewal	797
7.	Fees and Royalties	
8.	Technical Advisory Committee	
9.	Minister's Right to Cancel or Suspend the Exploration Right	100
10.	Abandonment or Relinquishment of the Exploration Area	
11	No guaranteed right to other minerals	128
12.	Examination of the Exploration Area	7
13.	Records and samples	
14.	Reports	
15.	Exploration Work Programme and Budget	
16.	Discoveries and testing	85
17.	Black Economic Empowerment	
18.	Good international oil-field practice	
19.	Existing Data	
20.	Upstream Training Trust	8
21.	Local labour, equipment, materials and contractors	
22.	Тах	
23.	Accounts and Audits	
24.	Customs duties	
25.	Exchange Control	
26.	Indemnity and insurance	
27.	Health and Safety	
28.	Data and information to be kept confidential and limitations on public announces	nent
29.	Cession, Transfer, Alienation	

- 32. Vis Major
- 33. Amendments
- 34. Unitisation
- 35. Waiver or Lenience
- 36. Arbitration
- 37. Environmental Protection
- 38. Costs and VAT
- 39. Exclusive right to Acquire a Production Right
- 40. Grantor's Option
- 41. Gas
- 42. Entire AgreementAct Prevails
- 43. Severability
- 44. Notices, domicilia citandi et executandi and Representatives
- 45. REGISTRATION

ANNEXURES

- A Plan and description of the Exploration Area
- B Periodic Relinquishments
- C Account to which payments to Grantor are to be made
- D Contributions to the Upstream Training Trust
- E Exploration Work Programme
- F Financial Guarantee
- G Capital Allowance Schedule
- H Schedule of Experts (accounting)
- I Production Right Agreement
- J Operating Agreement

SPECIMEN AGREEMENT FOR AN EXPLORATION RIGHT

			ii ii	PROTOCOL NO	<u> </u>
LET IT	HEREBY BE KN	IOWN:	ti sa		
	X	\$5	torrators	T	
THAT	on this the	_ day of	Notany Public of	year I wo thousand a	d residing and practising
(<) before file, .	in the	, INOIGHY I dono, c	Province, Repub	lic of South Africa, and in
the pre	esence of the subs	day ofin thescribing competent witnesse	es personally came a	nd appeared:	8
		035 W			
(1)			C	nief Executive of Sc	outh African Agency for
117		etroleum Exploration and Enorised representative of:			
	製				a to
1		THE REP	UBLIC OF SOUTH A	FRICA	
being Energy and P	duly authorised th y on etroleum Resourc	h its successors in title and ereto under and by virtue of the common in terms of second act; 2002 (75(1) of the said Act.	f the powers delegat ction 97(1) read with	ed to him or her by the sections 65(2)(a) and	e Minister of Minerals and 65(2)(b)(vi) of the Mineral
		¥6	AND		
		14		a.*	£:
(2)		, a		of, and as such in h	nis or her capacity as the
	duly authorised	representative of:			
	F				
		100 mm			
		(No/_)	2
being	duly authorised	n its successors in title and thereto under and by v the day of	irtue of a Resolution	"the Grantee"), he on of Directors of	or she, the said Appearer, the Grantee passed at
certifie	ed copies of which frecord in my Prot	delegation and of which Rotocol with the Minute hereof	esolution have this da	y been exhibited to me	e, the Notary, and remain
		想			E
AND T	HE APPEARERS	DECLARED THAT:			
WHER	REAS	Resources Development	Act, 2002 (Act No. said Act) in respect	of 2002), the Gran	ne Mineral and Petroleum tor may grant exploration efined) within the area of
AND V	VHEREAS	the Grantee wishes to o the property referred to i		Right in respect of Pe	stroleum in, on and under
AND V	VHEREAS	the Grantor has agreed terms and conditions rec	o grant such Explora orded hereunder,	tion Right to the Grant	ee in accordance with the
					2
MOM	THEREFORE THE	PARTIES AGREE HERET	O:	*	19
1.	Definitions				

In this agreement the clause headings are inserted for reference purposes only and shall not govern or affect the meaning or interpretation thereof. The definitions in the Act and regulations shall apply also to this agreement and additionally the following words and expressions shall have the meaning assigned to them:

1.1. "the Act" shall mean the Mineral and Petroleum Resources Development Act, 2002 (Act No. ... of 2002) or any subsequent law which amends or substitutes the Act;

- 1.2. "Affiliate" of a Party means a company which owns, or is owned by, or which is owned by an entity which owns that Party; "owns" and "owned" mean in this definition the ownership directly or indirectly of more than 50 % of the voting shares or members' interests;
- 1.3. "Effective Date", shall mean the date on which this exploration right agreement is executed notarially by the last Party so to execute;
- 1.4. "the Exploration Area", shall mean the property referred to in clause 2 of this agreement, excluding those portions relinquished or abandoned from time to time;
- 1.5. "Exploration Operations", shall mean any operation, activity or matter that relates to the exploration for Petroleum in the Exploration Area by the Grantee and/or its contractors;
 - Exploration Work Programme" shall mean the Grantee's written recordal of ...
- 1.6. "the Grantee", shall have the meaning attributed thereto in the description of the Parties above and shall as appropriate mean all Grantee Parties;
- 1.7 "Grantee Party" means each cessionary, transferee, lessee, sub-lessee or assignee of rights, duties and obligations under this Exploration Right pursuant to the provisions of clause 28 hereof;
- 1.8. "the Grantor" shall have the meaning attributed thereto in the description of the Parties above;
- 1.9. "the Income Tax Act" shall mean the Income Tax Act, 1962 (Act No. 58 of 1962) as amended or substituted;
- 1.10. "installation" shall have the meaning assigned to it in the Mine Health and Safety Act, 1996 (Act no. 29 of 1996) as amended;

"Natural Oil" shall have the meaning assigned to it in the Income Tax Act, and is the same as petroleum as defined in the Act;

- "Petroleum Agency" means the South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited (No. 1999/015715/07);
- 1.12. "Quarter" means a three-month period of a year beginning on 1st January, 1st April, 1st July or 1st October of any year;
- 1.13. "State" shall mean the Grantor:

Reference to the singular of a defined term includes a reference to it in the plural and vice versa and reference to any gender includes a reference to all other genders.

- 2. Grantor's Agent : Petroleum Agency
- 2.1. Without detracting from section 66 of the Act the Grantor records that it has appointed the Petroleum Agency as the Grantor's agent in regard to all matters relating to this Exploration Right, excluding matters concerning health and safety save in regard to the approval of exploration rights and production rights.
- 2.2. Without detracting from the generality of clause 2.1:
- 2.2.1. the Grantee shall be obliged to pass all data, including but not limited to information contemplated in section 14 of the Act and this Exploration Right and samples and borehole cores, and interpretations of all the aforegoing, to the Petroleum Agency, and all of which data, information, samples, borehole cores, and interpretations, shall be passed in the format and in the medium as may be required by the Petroleum Agency in writing from time to time.
- 2.2.2. The Grantor hereby appoints the Petroleum Agency or its representatives as the Grantor's representative or representatives on the Technical Advisory Committee provided for in clause 8 hereof.

5.

3. Description of Exploration Area

The Exploratio	n Area shall	comprise	the	following:
----------------	--------------	----------	-----	------------

Block		
Region	260	
Measuring:		km ²

which Block is more fully described in the attached sketch plan/diagram, marked Annexure A____

4. Exploration Right

The Grantor hereby in terms of section 75 of the Act grants to the Grantee, which hereby accepts, this exploration right conferring the sole and exclusive right to explore for petroleum in, on and under the Exploration Area on the terms and conditions set forth herein and subject to the provisions of the Act.

5. Ancillary Rights and Obligations

- 5.1. Without limiting the generality of clause 4 or detracting from Grantee's rights in terms of sections 5 and 77 of the Act, the Grantee shall have the right to:
 - 5.1.1. enter the Exploration Area together with the Grantee's employees, and bring on to the Exploration Area any plant, machinery and equipment and build, construct and lay down any surface or underground infrastructure which may be required for the purposes of exploration;
 - 5.1.2. explore for the Grantee's own account on or under the Exploration Area for petroleum;
 - 5.1.3. remove and dispose of any petroleum found during the course of exploration;
 - 5.1.4. subject to the National Water Act, 1998 (Act No. 36 of 1998), use water from any natural spring, lake, river or stream, situated on, or flowing through, the Exploration Area or from any excavation previously made and used for exploration or production purposes, or sink a well or borehole required for use relating to exploration on the Exploration Area;
 - 5.1.5. in terms of section 77(1)(a) of the Act, apply for and be granted a production right in respect of petroleum on the Exploration Area.
- 5.2. Without detracting from the Grantee's further obligations in terms of this Exploration Right or in terms of the Act, the Grantee shall:
 - 5.2.1. not, without the Minister's prior written consent, carry out Exploration Operations over:
 - 5.2.1.1. areas used or reserved under any law for any Government or public purpose;
 - 5.2.1.2. areas within five nautical miles seaward of the low water mark (as defined in section 1 of the Sea-Shore Act, 1935 (Act No. 21 of 1935)), nor in the areas shown as reserved areas on the attached sketch plan, Annexure A;
 - continuously and actively conduct exploration operations in accordance with the approved Exploration Work Programme;
 - 5.2.3. comply with the terms and conditions of this exploration right and the relevant provisions of the Act:
 - 5.2.4. comply with the requirements of the approved Environmental Management Programme;
 - 5.2.5. pay the prescribed fees to the Grantor.
- 5.3. No provision contained in this Exploration Right shall constitute a stipulation for the benefit of any person or entity who is not a Party to this Exploration Right capable of acceptance by such person, except the Petroleum Agency.

6. Commencement, Duration and Renewal

- 6.1. This Exploration Right will commence on the Effective Date and, unless terminated earlier under any provisions hereof, will continue in force for an Initial Period of _____ [not more than 3 years] from the Effective Date.
- 6.2. The Grantee shall within 30 (thirty) days reckoned from the Effective Date, commence with Exploration Operations in the Exploration Area and shall thereafter conduct such operations continuously and actively in accordance with the Exploration Work Programme and the approved Environmental Management Programme.
- 6.3. This Exploration Right shall be renewed by the Minister on application in terms of section 76(1) of the Act by the Grantee for a maximum of 3 (three) further Periods not exceeding 2 years each if the Grantee has complied with the:
 - 6.3.1. terms and conditions of this Exploration Right and is not in contravention of any relevant provision of the Act;
 - 6.3.2. Exploration Work Programme; and
 - 6.3.3. requirements of the approved Environmental Management Programme.
- 6.4. At the end of each Period the Grantee shall relinquish the relevant portion of the original extent of the Exploration Area as indicated in Annexure B and in the manner described in clause 10.
- 6.5. For the purposes of calculating the relinquishment required at the end of any period, any portion of the Exploration Area which is the subject of an accepted application for a production right or situated within a production area shall be considered as having been relinquished from the date of the application;
- 6.6. Any relinquishment in terms of this clause 6 or clause 10 shall be without prejudice to any obligation incurred by the Grantee in respect of the Exploration Area prior to the date of relinquishment and such relinquishment shall not affect the obligations of the Grantee under clause 15 (Exploration Work Programme and Budget) of this Exploration Right.

7. Fees and Royalties

- 7.1 The Grantee shall pay to the Grantor the prescribed exploration fees and the prescribed royalties in respect of petroleum removed and disposed of during the course of the Grantee's Exploration Operations, as envisaged in section 77 of the Act
 - 7.1.1. an annual exploration fee of R____(_____RAND) per km² of the Exploration Area payable yearly in advance, which shall increase annually on the anniversary of the Effective Date by the increase in the Consumer Price Index (All Items, Metropolitan and Other Urban Areas) as published from time to time by Statistics South Africa
 - 7.1.2. a royalty of _____% of the value of any petroleum or minerals found, removed and disposed of in the course of exploration operations on the Exploration Area; payment of the said share shall be made within 30 (thirty) days after the end of every Quarter and shall be accompanied by a certificate or other document acceptable to the the Grantor, reflecting particulars of the value of petroleum and/or minerals found, removed and disposed of on the Exploration Area during the Quarter in question; such quarterly payments shall,
 - (a) in the case of petroleum,
 - (i) be made on a fair market value of the Petroleum produced, calculated at the boundary of the mine. "Fair market value" shall, except as provided in (ii) below, be the actual realised price used to determine gross income for income tax reporting purposes under the Income Tax Act. The currency of payment will be either SA Rand or US Dollars.
 - (ii) In the event that all sales during a Quarter are to an Affiliate of Grantee then fair market value for the purposes of calculating royalty will be determined utilising a basket of similar products on offer world-wide. Otherwise, the fair market value for purposes of calculating royalty on sales to an Affiliate will be the average weighted by volume of the actual realised price of sales to third parties during such Quarter.

- (b) In the case of other minerals it will be:
- based on the selling price of the minerals (if applicable);
- in the case of minerals not sold, on the estimated value thereon; and
- after the subsequent sale (if applicable) of any minerals in respect of which payment has already been made on the estimated value thereof, on the difference between the estimated value and selling price.
- 7.2. In terms of the money bill the Grantee shall additionally pay to the Grantor a data maintenance fee equal to 1% of the fair market value, at the boundary of the mine, of petroleum produced from the Exploration Area, saved and sold.
- 7.2.1 "Fair market value" shall be as defined in 7.1.2. and shall, except as provided in clause 7.3.2 below, be the actual realised price used to determine gross income for income tax reporting purposes under the Income Tax Act. The currency of payment will be either SA Rand or US Dollars.
- 7.2.2 In the event that all sales during a Quarter are to an Affiliate of Grantee then fair market value will be determined utilising a basket of similar products on offer world-wide. Otherwise, the fair market value in respect of sales to an Affiliate will be the average weighted by volume of the actual realised price of sales to third parties during such Quarter.
- 7.3. The exploration fees and data maintenance fees referred to in clause 7.1 and 7.2 shall be paid as follows:
 - 7.3.1. exploration fees shall be paid within 30 (thirty) days following the Effective Date and each subsequent anniversary thereof for the duration of this Exploration Right; and
 - 7.3.2. data maintenance fees payable in respect of each Quarter shall be paid within thirty (30) days following the end of each such Quarter for the duration of this Exploration Right.
- 7.4. If any fees or royalties referred to in clauses 7.1 or 7.2 are not paid punctually on or before the due dates for payment, the Grantee shall be in *mora debitoris* and shall pay interest thereon at the rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act No. 1 of 1999) reckoned from the due date for payment to the date of actual payment.
- 7.5. Regardless of any relinquishment of parts of the exploration area or early termination of this Exploration Right for whatever reason, no portion of the fees referred to in clauses 7.1 and 7.2 shall be refundable.
- All fees and royalties which are payable by Grantee to the Grantor in terms of the Exploration Right shall be paid by deposit into the account of the Grantor at the bank and account indicated in Annexure C, or such other bank and account as the Grantor may from time to time give notice of. The onus for timely payment rests solely with the Grantee; the Grantor will not normally issue invoices. It should also be noted that the Upstream Training Trust is independent of the Grantor and payments to it should be made separately to the bank and account indicated in Annexure D, or such other bank and account as the Upstream Training Trust may from time-to-time give notice of.

8. Technical Advisory Committee

- 8.1. the Grantor and Grantee shall as soon as possible after the Effective Date establish a committee to be known as the Technical Advisory Committee which shall consist of:
 - (a) a chairman and one other person appointed by the Grantor; and
 - (b) two other persons appointed by Grantee.

The membership of the Technical Advisory Committee may be enlarged to include one representative of each of Grantee Parties. The Grantor may in this instance enlarge its representation to equal that of all the Grantee Parties.

8.2. The Grantor and any Grantee Party may appoint by notice an alternative member to act in the place of its representative on the Technical Advisory Committee during his or her absence or incapacity to act as a member of the Committee. When an alternate member acts in the place of any member he or she shall have the powers and perform the duties of such member.

- 8.3. Without prejudice to the rights and obligations of Grantee in relation to the management of Operations the functions of the Technical Advisory Committee shall be:
 - to review all Operations carried out by Grantee and to provide Grantee advice and recommendations with respect thereto;
 - (b) to review and approve any proposed exploration work programme and budgets to be submitted by Grantee to the Grantor in terms of clause 14 and to monitor the implementation of any appraisal programmes submitted by Grantee to the Grantor in terms of clause 16;
 - (c) to review any appraisal programmes submitted by Grantee to the Grantor in terms of clause 16 and any development plan which Grantee proposes to submit in connection with an application for a Production Right in terms of clause 39;
 - to review the accounting of expenditure and the maintenance of operating records and reports kept in connection with Operations; and
 - (e) to offer advice to Grantee in order to promote the efficient carrying out of Operations.
- 8.4. All meetings of the Technical Advisory Committee shall be held at least semi-annually. The location of such meetings shall be in Cape Town, South Africa unless otherwise unanimously agreed by its members, and the Grantee shall provide the the Grantor with copies of all documentation and presentation materials 7 (seven) days prior to the meeting.
- 8.5. Three members of the Technical Advisory Committee shall form a quorum for a meeting of the Committee, provided that at least one representative of Grantee and one representative of the Grantor is present.
- 8.6. Any member of the Technical Advisory Committee shall have the right to call any expert to any meeting of the Committee to advise the Committee on any matter of a technical nature requiring expert advice.
- 8.7. The Chairman of the Committee shall cause minutes of each meeting to be kept and circulated to the members.

9. Minister's Right to Cancel or Suspend the Exploration Right

- 9.1 It is recorded that the Minister is empowered to cancel or suspend this Exploration Right in the circumstances envisaged in and in accordance with the provisions of section 85 read with section 44 of the Act.
- 9.2. In the event of this Exploration Right being cancelled in accordance with the provisions of clause 9.1, the Grantee will not be absolved from its obligations or liabilities which have accrued up to the date of such cancellation.
- 9.3. The cancellation of this Exploration Right in terms of clause 9.1 above will be without prejudice to the Grantor's other rights under this Exploration Right or in statutory or common law of the Republic of South Africa, including its right to recover any damages it may have suffered or may suffer, its right to claim specific performance or its right to claim ancillary relief.

10. Abandonment or Relinquishment of the Exploration Area

- 10.1. The Grantee may, subject to sub-clauses 10.2., 10.3., 10.4. and 10.5., abandon this right in its entirety or relinquish any portion or portions of the Exploration Area at any time.
- 10.2. All areas relinquished in terms of clause 6 or this clause 10 shall cease to be part of the Exploration Area as from the end of the relevant period or on the date identified so notified in conformance with sub-clause 5.5 of this Exploration Right, whichever is the sooner and provided that:
 - 10.2.1 Any part of the Exploration Area relinquished under sub-clause clause 5.5 shall be deemed to have been relinquished for the purposes of the determination of the relinquishment next required to be made by the Grantee;
 - 10.2.2 The Grantee shall relinquish such parts of the Exploration Area in such a way that the area relinquished is a single area:

- bounded by lines running either due North and South or due East and West and coinciding with integral multiples of 01 minutes of longitude or latitude; and
- (ii) approximately rectangular with the length not more than three times the width of the narrowest portion of the area to be relinquished;
- 10.3 Subject to the submission of a clear technical motivation by the Grantee, the Grantor may in its sole discretion relax the requirement in 10.2.2 above that the relinquished area be a single area.
- 10.4 Grantee may, subject to the terms of clause 10.2.2, by notice in writing addressed and delivered to the Grantor, relinquish any part of the Exploration Area from a date not less than six months from the date on which such notice was delivered to the Grantor.
- 10.5. In abandoning or relinquishing the Exploration Area or any portion thereof, the Grantee will not be absolved from those obligations or liabilities, either in the form of exploration work or otherwise, which may have accrued under the terms and conditions of this Exploration Right, the Exploration Work programme, the Environmental Management Programme and the Act up to the date of such abandonment or relinquishment.
- 10.6. With effect from the date the Grantee has so abandoned or relinquished a portion or portions of the Exploration Area, the Grantor shall be entitled to grant any exploration rights or production rights or any of the rights and permits referred to in the Act in, on, or under the portion or portions, so abandoned or relinquished, to any persons or persons.
- 10.7. Upon abandonment or relinquishment of the Exploration Area or any portion thereof, the Grantee will furnish the Grantor with all exploration results and/or information and interpretations, as well as its evaluation of the geological, geophysical and borehole data in respect of such abandoned or relinquished area or areas insofar as it applies to petroleum or any other mineral or minerals referred to in clause 11. Such results and/or information may be made available to third parties.
- 10.8. Upon abandonment or relinquishment of the Exploration Area or any part thereof the Grantee shall apply for an Exoneration Certificate in terms of section 40 of the Act.

11. No guaranteed right to other minerals

- 11.1. The Grantee shall not have the exclusive right to acquire a prospecting right, mining permit or retention permit in respect of any mineral or minerals (excluding petroleum) in, on and under the Exploration Area.
- 11.2. Should any mineral or minerals be discovered in the course of Exploration Operations, the Grantee shall nevertheless be bound to report such discovery to the Minister. The Grantee may thereafter, subject to any prior rights, apply for and, if granted the required prospecting rights by the State, prospect for such mineral or minerals.

12. Examination of the Exploration Area

- 12.1. The Minister may authorise inspections and exercise of related powers as set out in the Act
- 12.2. Upon request by the Grantor during offshore operations of Grantee, Grantee shall provide free transport between the Grantee's onshore base and the offshore facilities as well as accommodation on the offshore facilities for authorised officers.

13. Records and samples

13.1. The Grantee must keep, in a form as may be from time to time reasonably approved by the Grantor, current and accurate records of all Exploration Operations. Such records must include, inter alia, details of drilling, completion, plugging or abandonment of all boreholes, drill-stem tests, well logs, all other geological and geophysical information and interpretation, maps, digital records obtained by the Grantee in the course of carrying out Operations (hereinafter referred to as "Data") and any equipment lost overboard or down hole, relating to the Exploration Area as may be necessary to preserve all information which the Grantee has about the geology of the Exploration Area, and such other records as the Grantor may from time to time direct and which the Grantee can reasonably keep.

13.2. The Grantee shall at its own expense save and correctly label a representative portion of each sample of cores and cuttings from the earth or sea bed and strata encountered in drilling wells, such samples to be taken at regular intervals in accordance with good international oilfield practice, and samples of any Petroleum or other Minerals of potential value recovered in any of the Grantee's wells in the Exploration Area, to be forwarded promptly to the Grantor at the Grantee's expense. All samples acquired by the Grantee for its own purpose shall be made available for inspection at any reasonable time by the Grantor.

Before the Grantee discards any core and/or cuttings and samples, the Grantor shall, by prior notice, be afforded an opportunity of obtaining the same and these will be delivered to it at Grantee's expense.

- 13.3. The Grantee may export such Data for processing or laboratory examination or analysis by the Grantee or by third parties or storage outside of the Republic of South Africa, provided that representative samples equivalent in quality or, where such material is capable of being reproduced, copies of equivalent quality have first been delivered to the Grantor and provided that the Grantor's prior written approval has been obtained.
- 13.4. The Grantee shall deliver to the Grantor, at the Grantee's expense, copies of all data as defined in 12.1 above as soon as they are acquired or prepared in accordance with the the Grantor's reporting guidelines.

14. Reports

- 14.1. The Grantee shall keep the Grantor currently advised of all major developments taking place during the course of Operations and shall furnish the Grantor with Data (as defined eleswhere) and other available information, reports, assessments and interpretations relating to Operations as the Grantor may reasonably require in accordance with the Grantor's reporting guidelines.
- 14.2. The Grantee shall provide the Grantor with a summary report of all Operations on a monthly basis.
- 14.3. Within 30 (thirty) days of the end of each Quarter the Grantee shall submit to the Grantor a report reflecting, for the relevant Quarter, details of all Operations, including but not limited to:
 - (i) the numbers of local and expatriate persons employed;
 - (ii) the work done and money expended on Operations; and
 - (iii) the site and depth of every well drilled or being drilled and the formations penetrated and particulars regarding any occurrence of petroleum and/or any other mineral of potential value encountered.
 - (iv) a statement of compliance with the approved Environmental Management Programme.
- 14.4. Nothing in clause 13 or this clause 14 shall be construed as requiring the Grantee to disclose any of its proprietary technology or that of its Affiliates.
- 14.5. The Grantee shall furnish the Grantor with such other reports as to the progress of its operations in the Exploration Area as the Grantor may from time to time require and which the Grantee can reasonably make available.
- 14.6. Within three (3) months after the termination and/or cancellation and/or abandonment of this Exploration Right, the Grantee shall furnish the Grantor with all results of and its interpretations and general evaluation of the geological, geophysical and borehole data relating to the Exploration Area as it applies to petroleum or any other mineral or minerals under sub-clause 11.2.
- 14.7 The Grantee shall provide copies of all hydrographic information and data obtained by the Grantee or its contractors and sub-contractors, during marine operations undertaken under this Exploration Right to the Grantor and the Chief Hydrographer of the South African Navy.
- 14.8. All information furnished by the Grantee to the Grantor in accordance with this clause and clause 13 above, may be made available by the Grantor for use by third parties on expiry of the Exploration Period, termination of the Exploration Right or relinquishment or abandonment by the Grantee.

14.9 The provisions of this clause 14 are additional to and without detraction from the provisions of Regulation 14 made in terms of the Act.

15. Exploration Work Programme and Budget

- 15.1. The Grantee must, to the satisfaction of the Grantor:
 - 15.1.1. over the period of this Exploration Right, initiate and thereafter actively pursue operations designed to progressively explore substantially over the Exploration Area;
 - 15.1.2. carry out and meet the commitments set out in the Exploration Work Programme annexed hereto marked Annexure E;
 - 15.1.3. not less than three months prior to each anniversary of the Effective Date prepare and submit to the Grantor an amended and revised exploration work programme and budget setting forth the exploration operations which the Grantee proposes to carry out during the twelve months following the aforesaid anniversary and the cost thereof.
- 15.2. Any amended and revised exploration work programme and budget submitted to the Grantor shall be consistent with the relevant Minimum Work Obligation set out in Annexure E and with the requirements set out in clause 37 relating to environmental protection.
- 15.3. Any revisions or amendments to the aforesaid Exploration Work Programme and budget must be submitted to Petroelum Agency for approval.
- 15.4. In the event that the work carried out by the Grantee, as reviewed from time to time, is less than that specified in the Exploration Work Programme and insufficient reason has been given by the Grantee to the Grantor for the failure to carry out the full Exploration Work Programme, then the Grantor may in terms of section 76(3) of the Act refuse an application for a renewal of this Exploration Right in terms of clause 6.3 hereof.
- 15.5. If at the end of any sub-period, the Grantee has not completed the Minimum Work Obligation for that sub-period, then the Grantee shall (where the amount spent is less than the estimated expenditure for such Minimum Work Obligation) pay the Grantor the difference (if any) between the said estimated expenditure in respect of the Minimum Work Obligation for the relevant sub-period and the actual amount which the Grantee expended on the Minimum Work Obligation.
- 15.6. If the Grantee performs more exploration work than the Minimum Work Obligation for the Initial Period, the First Renewal Period or the Second Renewal Period, then the Grantee can carry forward the excess work together with any associated excess expenditure and apply it against the Minimum Work Obligation for the next sub-period if an application for renewal of this Exploration Right for such next sub-period is granted but not beyond that next sub-period. Appraisal work or expenditure in respect of appraisal work shall not be taken into account in establishing whether there has been excess expenditure or exploration work as aforesaid.
- 15.7. It is recorded that the time sequences as set out in the Exploration Work Programme will commence on the Effective Date.
- 15.8. As set out in section 84 of the Act, each Grantee Party shall secure and deliver to the Grantor within 14 (fourteen) days of Effective Date a financial guarantee, appropriate to its proportionate interest in this Exploration Right in the format indicated in Annexure F.

Discoveries and testing

- 16.1. When a discovery of Petroleum is made in the Exploration Area, the Grantee shall -
 - (a) promptly inform the Grantor by notice of the fact that such discovery has been made;
 - (b) forthwith cause tests to be made in connection with such discovery in order to determine whether it is, in the Grantee's opinion, commercial. Before testing each occurrence of Petroleum, the Grantee shall give notice of testing to the Grantor and shall provide, free of charge, to the representatives of the Grantor access and transportation to the drilling rig and reasonable facilities there in order to witness such tests; and
 - (c) within 60 days after having completed such tests, furnish the Grantor with a report containing an evaluated result of such tests.

- 16.2. The testing and measuring of Petroleum for the purpose of establishing the existence of a commercial discovery of Petroleum shall be carried out in accordance with good international oil-field practice.
- 16.3. If the Grantee considers, after providing the report referred to in clause 16.1.(c), that the discovery of Petroleum could be commercial, then the Grantee shall forthwith take reasonable steps to appraise the discovery and submit the appraisal programme which it intends to undertake to the Grantor for its agreement. the Grantor shall review such appraisal programme and within 30 days of receipt thereof either agree to it or give Grantee any comments which Grantor may have with respect to it. If the Grantor fails to respond with comments within such time, it shall be deemed to have agreed to the appraisal programme.
- 16.4. The Grantee shall, within one year from the date on which the Grantor is advised of the appraisal programme or such longer period as may be appropriate under the circumstances address and deliver to the Grantor a full report containing particulars of the results of the appraisal programme, including particulars and preliminary estimates relating to:
 - the location and depth of petroleum bearing structures;
 - (ii) the composition of petroleum;
 - (iii) the estimated recoverable reserves of petroleum; and
 - (iv) the estimated daily production potential of petroleum; and
 - a preliminary estimate of the expenditure of development and production operations, including the expenditure on transportation of petroleum, based on an outline design for the development of the discovery of petroleum.
- 16.5. the Grantor may, at any time after delivery of the report and estimates referred to in clauses 16.4 request the Grantee to supply such further particulars relating to such reports as the Grantor deems necessary and the Grantee shall comply in writing with such request within 30 days from the date of delivery of such request.

17. Black Economic Empowerment

The Grantee acknowledges that in order to give effect to section 2(d) of the Act, Grantor is desirous of expanding opportunities for historically disadvantaged persons (as defined in the Act) to enter the upstream petroleum industry and to benefit from the exploitation of Petroleum resources in South Africa and as such Grantee undertakes to use its best endeavours to seek historically disadvantaged South African partners to participate in this Exploration Right for at least a 10 % undivided interest. Such participation shall be governed by the principles contained in Appendix D and an associated Operating Agreement.

18. Good international oil-field practice

In addition to the provisions of the Act, the Grantee shall comply with the following special obligations.

- 18.1. The Grantee shall maintain all installations in good repair and condition and shall execute all Operations in a proper and workmanlike manner in accordance with methods of exploration used in good international oil-field practice and, without prejudice to the generality of the foregoing, the Grantee shall take all steps practicable in order to prevent:
 - (a) the escape or waste of petroleum discovered in the Exploration Area;
 - (b) damage to petroleum-bearing strata;
 - (c) the entrance of uncontrolled water through wells to petroleum-bearing strata;
 - (d) the escape of petroleum into any waters or aquifer in or in the vicinity of the Exploration Area;
 and
 - (e) pollution of the terrestrial or marine environment.
- 18.2. The Grantee shall immediately inform the the Grantor and other relevant government bodies of the occurrence of any event described in clauses 18.1(a) through 18.1(e) above and shall take immediate remedial action.

- 18.3. The Grantee shall take immediate action with respect to notices given by the Grantor of any event described in clauses 18.1(a) through 18.1(e) above.
- 18.4. The Grantee shall give all notices and take all actions described in the relevant oil-spill contingency plan required in terms of the Prevention and Combating of Pollution of the Sea Act (Act no. 6 of 1981).
- 18.5. The Grantee shall protect all wells drilled under this Exploration Right. Upon completion or abandonment of a well the Grantee shall remove from the drill site all guide bases and other substantial equipment so as to leave the environment free of significant obstruction, except where the relevant well is intended to be used as a production well. The Grantee shall notify the relevant government agencies that the location is free of obstruction or, if it is not, the location, nature and extent of any obstruction.
- 18.6. No gas produced from the Exploration Area, except in the case of short term flaring necessary for testing or other operational reasons, may be flared unless the Grantor's prior written approval, which approval shall not be unreasonably withheld, has been obtained.
- 18.7. The Grantee shall give timeous notice to relevant government agencies and to interested and affected parties as to all Exploration Operations and shall take all reasonable steps to minimise interference with rights of other users.

19. Existing Data

- (a) one copy of all available borehole measurements and analyses including sample logs, wireline logs, gas chromatograms, micropalaeontological, petrographical, and geochemical data and / or reports (including core descriptions) as well as one digital tape copy, where available (in industry-standard format), of all wireline logs from all wells drilled in the Exploration Area;
- (b) one copy of each technical report of the wells drilled in the Exploration Area, regional reports for the West / South / East Coast and general reports for RSA offshore operations, as appropriate to the Exploration Area;
- (c) one copy of each seismic line which intersects the Exploration Area, at the vertical scale of 2.5 inches per second (two-way time) or as imaged, as well as one copy of 500 km of lines to be selected by the Grantee at a scale of 5 inches per second (two-way time). Petroleum Agency will also provide location data for all seismic lines in SEG-P-1, UKOOA or ASCII format, in digital form;
- (d) field data for up to ______ km of seismic lines in the Exploration Area as available and as selected by the Grantee. This data set shall be defined by the Grantee after its preliminary evaluation of the copies of seismic lines described in (c) above. The Grantee will provide Petroleum Agency with copies of the reprocessed data as both digital records in industry standard format and a single paper copy. Where such field tapes are available only in a format other than digital records on 9-track half-inch tape (or more modern format) Petroleum Agency will make available such original field data tapes so that the Grantee can transcribe the data into a current industry-standard format. The Grantee shall provide Petroleum Agency with a copy of the data so transcribed onto media of an industry-standard format which is current at the time, and which Petroleum Agency is equipped to read and, at Petroleum Agency's request, return the original tapes to Petroleum Agency at the Grantee's sole cost, risk and expense.
- (e) stacked and migrated data for up to _____km of seismic lines, as available, as selected by the Grantee;
- (e) access to available sets of conventional cores, sidewall cores, drilling cuttings from the wells in the Exploration Area for sampling; and

Wherever possible documents will be provided as digital images in a format appropriate to the document.

Additional field data tapes and stacked and migrated data tapes, over and above that detailed in clauses (d) and (e) above will be made available upon request by Grantee provided that Grantee shall reimburse Petroleum Agency for the reasonable and customary costs incurred in preparing such tapes.

Petroleum Agency will co-operate with the Grantee at the Grantee's request by assisting in resolving technical problems relating to the data (but not interpretation) upon terms and conditions then to be agreed.

All the aforesaid data, studies and reports (hereinafter referred to as the "Existing Data") provided to the Grantee are owned by Petroleum Agency and are of considerable commercial value to it. Such Existing Data shall remain the property of Petroleum Agency and shall, on termination of this Exploration Right, be returned at Grantee's cost to Petroleum Agency. The Grantee will hold the Existing Data confidential between itself and Petroleum Agency in the manner described in clause 27.1 in relation to the Data with the exception of items 27.1(f) and 27.1(k), in which instances the Grantee must first acquire Petroleum Agency prior written approval, and 27.1(b)

20. Upstream Training Trust

The Grantee shall pay the amounts set out in Annexure D to the independent Upstream Training Trust, to be used in the training, education, and obtaining of practical experience for South African personnel in a manner to be mutually agreed by the Trustees.

21. Local labour, equipment, materials and contractors

- 21.1. In furtherance of section 2(f) of the Act, in carrying out Operations the Grantee shall, to the extent reasonably possible, employ South African citizens having appropriate qualifications.
- 21.2. Not later than three months from the Effective Date, the Grantee shall meet with the Grantor to discuss and agree on a programme for the recruitment, training and employment of South African citizens during the exploration phase of Operations.
- 21.3. The Grantee may employ a person who has necessary qualifications and experience and is not a South African citizen if the required skills are not available in the South African labour market. The subsequent availability of qualified candidates will be reviewed annually.
- 21.4. The Grantee:
- 21.4.1. and its contractors shall use and purchase equipment, machinery, materials, instruments, supplies and accessories (all referred to as "Goods") manufactured, produced or available in the Republic of South Africa for use in Operations under this Exploration Right, provided that such Goods are reasonably comparable with like goods manufactured, produced or available outside the Republic of South Africa as to landed cost, quantity and quality and can be made available at the time when and the place where required;
- 21.4.2. shall to the extent reasonable, taking cognisance of government policies, utilise contractors in the Republic of South Africa where services of comparable standards with those obtained elsewhere are available from such contractors at competitive prices and on competitive terms and can be made available at the time when and the place where required.

22. Tax

- 22.1. The Grantee shall maintain a permanent establishment in the Republic of South Africa and arrange its affairs in such a manner that all the profits from production of petroleum under this Exploration Right will be attributable to that establishment.
- 22.2. Except to the extent exempted, the Grantee shall throughout the term of this Exploration Right be liable for income tax payments to the State on the annual taxable income derived by it from the production of Natural Oil in accordance with the Income Tax Act as amended from time to time. In no event, however, shall such tax exceed, in respect of taxable income derived during any year of assessment from the production of Natural Oil in the Exploration Area, the sum of an amount of the "normal tax" as defined in Chapter II, Part 1, Section 5 of the Income Tax Act (hereinafter called the "basic tax"), calculated on the full amount of such taxable income at a maximum tax rate of 35 %. The Grantee is hereby expressly exempted and shall not be liable, for the payment of any "additional normal tax" under Chapter II, Part 1, Section 5(2A) of the Income Tax Act.
- 22.3. South African income taxes shall be calculated in United States Dollars. For South African tax reporting purposes only, all account balances necessary to compute the US Dollar tax liability shall be converted to Rand at the rate determined under clause 25 (Exchange Control). Such conversion shall occur at the end of each taxation year. Taxes may be paid in either Rand or US Dollars at the Grantee's election. Payments made in

US Dollars for any taxation year shall be converted to Rand in the same manner described in this sub-clause 22.3.

- 22.4. The Grantee shall maintain a permanent establishment in the Republic of South Africa and arrange its affairs in such a manner that all the profits from production of petroleum under this Exploration Right will be attributable to that establishment.
- 22.5. Any payments to the State in terms of this clause 22 shall constitute the Grantee's only commitment to the State, or any political sub-division thereof, for taxes or levies, of any kind or nature, on the income or profits derived by the Grantee from the production of petroleum under this Exploration Right. Specifically, but without limiting the provisions contained in the preceding sentence, the Grantee shall not, other than as set out in clause 7, be liable to the State for:
 - 22.5.1. any other taxes or levies of a similar nature, calculated in relation to taxable income, profits or turnover arising out of Operations in the production and disposal of petroleum under this Exploration Right;
 - 22.5.2. any share of profit by the State or any other form of consideration in respect of the activities performed by it pursuant to this Exploration Right;
 - 22.5.3. non-resident shareholders' tax or any other dividend withholding tax, in respect of dividends declared or paid in favour of shareholders who are not resident in the Republic of South Africa out of profits derived from the production of petroleum under this Exploration Right, or any tax on the remittance outside the Republic of South Africa of profits derived from the aforesald production of petroleum;
 - 22.5.4. distributed or undistributed profits tax, including the Secondary Tax on Companies imposed under Part VII, Section 64B, et. seq. of the Income Tax Act, in respect of profits derived from the production of petroleum.
- 22.6. In calculating the taxable income referred to in this clause 22, the deductions from income shall be as granted in the Income Tax Act as at the date on which the 1977 amendments thereto were promulgated, and shall in any event include:
 - 22.6.1. the costs expended by the Grantee itself on Exploration Operations within the Exploration Area in any tax year or part thereof, and
 - 22.6.2. under current working costs, also the costs of transportation of petroleum from the Exploration Area to the marine terminal or such refinery in the Republic as the Grantee may elect, as the case may be, and as capital ranking for redemption:
 - the costs of laying pipeline from the points referred to above in this sub-clause 22.6.2. or to any onshore processing facility of the Grantee;
 - (b) the costs incurred by the Grantee itself in connection with the viability of the relevant undertaking and the design, procurement, management (including also project management), transport and construction of the constituent parts (from and after raw material stage and including also the piles and other foundations) of any marine or onshore receiving installations erected or to be erected on the Exploration Area or onshore with a view to exploitation of petroleum discovered or found in the Exploration Area, including also the costs of training of personnel for any purpose in connection with such installations, at any time prior to the successful commissioning of such installations but excluding any assets belonging to another taxpayer, and
 - (c) a capital allowance equal to 12 % (twelve per cent) compound interest per annum on the total amount of unredeemed capital expenditure, ranking for redemption, calculated from the end of the month during which each such cost was incurred, until it be redeemed and calculated, as far as this can be done, according to the provisions of sections 26(3) and (4) of the Mining Rights Act, 1967, an example of which is attached hereto as Annexure G.
- 22.7. Expenditure shall not be disallowed as a deduction for income tax purposes because it has been incurred by the Grantee outside the Republic of South Africa.
- 22.8. Production activities taking place in or in connection with an area or areas covered by one Production Right and its associated pipeline and onshore receiving installations up to the boundary of the local refinery, if any, shall be deemed to have taken place in respect of a single mine provided that assets belonging to another taxpayer will be disregarded.

- 22.9. In determining the taxable income or profits in respect of the Exploration Area, the phrase "mining for Natural Oil" shall be interpreted as including all exploration, development and production activities and such taxable income or profits shall be taxed as set out in this clause 22.
- 22.10. Should the Grantee's offshore facilities be linked to an onshore plant also belonging to the Grantee, the offshore facilities and pipelines as well as the onshore receiving facilities and pipelines (as well as all related structures and foundations) up to the boundary of the onshore plant will be regarded as a mine and the income therefrom taxed as herein set out, while the rest of the onshore facilities will be regarded as a factory and the income therefrom taxed in accordance with the income tax laws of the Republic of South Africa as amended from time to time, the gross income of the Grantee from the combined mine and factory respectively being in such proportion as the Grantee and the Commissioner for Inland Revenue agree or, failing such agreement, as determined by any of the firms or entities referred to in Annexure H (Schedule of Experts) hereof at the request of the Commissioner, unless the Grantee and the said Commissioner agree in writing on another firm or entity.

23. Accounts and Audits

Landers A

The Grantee shall maintain in South Africa such accounting records as will properly support all expenditure and receipts of Operations under this Exploration Right.

Such records or copies thereof shall be provided at the Grantor's request for the purpose of confirming the minimum expenditure obligations under the Exploration Work Programme. The Grantor shall have the right to appoint an auditor for the purpose of confirming that these obligations have been met. Such confirmation, if any, shall be completed within two years after the end of the calendar year in which the expenditures or receipts occurred.

24. Customs duties

The Minister records having procured the following from the Minister of Finance, namely:

- 24.1. The Grantee may during the continuance of this Exploration Right and subject to the provisions of international trade agreements which may be in force, import into the Republic of South Africa or clear from any customs and excise warehouse under rebate of full customs duty (as defined in the Customs and Excise Act, 1964) or export from the Republic of South Africa such Goods (excluding refined Petroleum products for use in road vehicles and distillate and residual fuel oil for any purpose and goods imported by the Grantee for the personal use of its Carrièrea. personnel who are South African residents) as are required for use in Operations and which are not obtainable or available from stocks of Goods manufactured in the Republic of South Africa in terms of clause 21.4.1; provided however that the Grantee shall not sell, transfer, dispose of or use or retain for use except in Operations under this Exploration Right any Goods obtained free of customs duty without the permission of the Commissioner for Customs and Excise; such duty as may be assessed by the Commissioner for Customs and Excise shall be payable to him/her in respect of any such Goods sold, transferred, disposed of or used or retained for any use other than in Operations except if such Goods are sold, transferred or disposed of to 1 M. 12 1 - 1 - 2 any person in the Republic of South Africa entitled to a like rebate of customs duty or if such Goods are exported from the Republic of South Africa by the Grantee.
- 24.2. The Grantee shall have the right, subject to the applicable laws and regulations, to employ such contractors as in its opinion may be necessary for Operations; such contractors shall be exempted from the payment of customs duties in the same manner as the Grantee is exempted under this Exploration Right as long as their work is pursuant to their contract with the Grantee.
 - 24.3. The Grantee's non-South African personnel may import household and personal effects into South Africa free of customs duty and may export such effects freely, but disposal in the Republic of South Africa of any such household and personal effects shall be subject to approval by the Commissioner for Customs and Excise and shall be subject to customs duties.

Exchange Control

- 25.1. The Minister records having procured the following from the Minister of Finance.
 - 25.1.1. The Grantee shall have the free availability of the currency generated from the export of petroleum produced under this Exploration Right which the Grantee can place directly to a bank account within the Republic of South Africa or abroad.

- 25.1.2. The Grantee shall have the free disposition and right to freely exchange South African Rand whether received from the sales of petroleum in the national market for foreign currency or received from other activities or events related to this Exploration Right. The Grantee has the right to deposit such currency directly into bank accounts in the Republic of South Africa or abroad.
- 25.1.3. The Grantee shall have the right to maintain, control and operate bank accounts in any currency, inside the Republic of South Africa and abroad; to have the control and free use of these accounts and to maintain and freely dispose of the funds of these accounts without any restrictions. The Grantee shall not be obliged to convert any foreign currencies into South African Rand.
- 25.1.4. Without prejudice to what is mentioned above, the Grantee shall have the right to dispose, distribute, send or retain abroad, without any restrictions, the Grantee's annual net profits.
- 25.1.5. The Grantee may use South African banks to exchange the currencies referred to in clause 25.1.2.
- The exchange rate of the South African Rand will be determined by the free market.
- 25.3. The Grantee may acquire abroad materials, equipment and services and pay for them in foreign currency. The Grantee may obtain financing abroad through loans and advances to pay for costs of activities connected with this Exploration Right.
- 25.4. The Grantee shall provide periodic reports as reasonably requested by the Reserve Bank.
- 25.5. In the event that the ratio of borrowings from South African financial institutions to foreign investment exceeds the ratio prescribed in Exchange Control Regulations promulgated by Government Gazette Notice R1111 of December 1961 as amended, the Grantee shall notify the Reserve Bank and advise the Reserve Bank as to any steps the Grantee intends to take to reduce this ratio, or request approval, which shall not be unreasonably withheld, to exceed this ratio.

26. Indemnity and insurance

- 26.1. The Grantee, its successors and/or assigns, will, during the tenure of this Exploration Right, take all such necessary and reasonable steps and do all such acts, matters and things and carry out their Exploration Operations in such a manner as will adequately safeguard and protect the environment, the Exploration Area and any person or persons using or entitled to use the surface of the Exploration Area from any damage caused by or through or in consequence of the exercise by the Grantee of its aforesaid right to explore under this Exploration Right, of any activity or activities associated with the Grantee's operations in the Exploration Area, and will in so far as there is a legal liability, compensate such person or persons for any damage or losses, including but not limited to damage to the surface, to any crops or improvements, which such person or persons may suffer as a result of, arising from or in connection with the exercise of his/her rights under this Exploration Right or of any act or omission in connection therewith.
- 26.2. The Grantee and its aforesaid successors and/or assigns will and do hereby undertake and bind themselves to hold harmless and indemnify the Grantor and/or its servants and employees against any claim or claims which may be instituted by any person as a result of any injury, loss (including loss of life), damage, cost, charges, and expenses which may be suffered or sustained by any person or persons whomsoever, as a result of, or which may in any way whatsoever be associated with the exercise by the Grantee, it's successors and/or assigns of the right to explore under this Exploration Right, and must also on request refund to the Grantor and/or its servants and employees all costs, charges and expenses which the Grantor and/or its servants and employees may be put to or sustain in connection with or arising out of any such claim made by such person or persons.
- 26.3. It is explicitly agreed that the Grantor will not be deemed to be a partner in any Exploration Operations carried out in terms of this Exploration Right. The Grantor will give notice to the Grantee and its aforesaid successors and/or assigns of any claim or occurrence in respect of which it proposes to invoke the provisions of this indemnity and will give the Grantee and its aforesaid successors and/or assigns the fullest opportunity to investigate, repudiate or otherwise deal with any matter that may arise from such claim. Provided always that nothing herein contained will be deemed to render the Grantee or its aforesaid successors and/or assigns liable for, or to indemnify the Grantor and/or its servants and employees against, any compensation or damages for or with respect to injuries or damage to persons or the Exploration Area directly resulting from any delict of the Grantor and/or its servants and employees during the currency of this Exploration Right, or for or in respect of any claims, demands, proceedings, damages, costs, charges and expenses in respect thereof or in relation thereto.

- 26.4. The Grantor will hold harmless and indemnify the Grantee and its aforesaid successors and/or assigns from and against all claims, demands, proceedings, damages, costs, charges and expenses in respect of matters referred to in the proviso of sub-clause 25.1 of this clause.
- 26.5. The Grantee shall during the term of this Exploration Right obtain and maintain for and in relation to Operations, insurance to indemnify the Grantor, the Petroleum Agency and the Grantee against:
 - (a) loss or damage to any or all assets being used in connection with Operations;
 - (b) loss or damage for which the Grantee may be liable caused by pollution in the course of or as a result of Operations;
 - (c) loss or destruction of property or damage to property as well as bodily injury, sickness or death suffered by any third party for which the Grantee may be liable;
 - the cost of removing wrecks and cleaning-up operations pursuant to an accident in the course of or as a result of Operations;
 - (e) the Grantee's liability to its employees, consultants and agents engaged in Operations;
 - (f) any other risk of whatever nature as is customary to insure against in the international petroleum industry in accordance with good international oilfield practices.
- 26.6. The amount insured against, the type of insurance referred to in clause 26.5 above and the terms of such insurance shall be determined in accordance with good international oilfield practices.
- 26.7. The Grantee acknowledges that the Grantor cannot guarantee that the Grantee will at all times be in a position to exercise the rights granted in terms of this Exploration Right and that in certain instances a conflict may arise with the holders of other mineral and / or fishing rights which overlap with the Exploration Area. The Grantee hereby indemnifies and holds the Grantor harmless from and against all claims, liabilities and costs incurred by the Grantee should the rights of the Grantee in terms of this Exploration Right not be exercisable by a reason of such a conflict with other rights holders within the Exploration Area.

27. Health and Safety

- 27.1. The Grantee and its successors and/or assigns will and do hereby undertake and bind themselves during the currency of this Exploration Right to observe and comply with all relevant provisions of the Mine Health and Safety Act (Act no. 29 of 1996) as amended and regulations in terms thereof, or any subsequent law which amends or substitutes the Mine Health and Safety Act, 1996.
- 27.2. Without detraction from sections 42 and 43 of the Act, in the event of:
 - 27.2.1. an emergency or accident arising from Operations affecting the environment, Grantee shall forthwith notify the Grantor accordingly; or
 - 27.2.2. any fire or oil spill in the Exploration Area or from any facilities connecting the Exploration Area with the land, Grantee shall promptly implement the relevant contingency plan; or
 - 27.2.3. any other emergency or accident arising from Operations affecting the environment, Grantee shall take such action as may be prudent and necessary in accordance with good international oil-field practice in such circumstances.

28. Data and information to be kept confidential and limitations on public announcements

In addition to the provisions of section 83 of the Act and Regulation 14 made in terms of the Act:

- 28.1. The Data (as defined in clause 13.1) and all programmes, tests, analyses, results, books, statements, records, returns, plans and information which the Grantee is or may from time to time be required to furnish under the provisions of this Exploration Right, shall be supplied at the expense of Grantee. Each Party shall each keep confidential the Data and such information and shall not disclose it to any person without the consent of the other Party except:
 - (a) where disclosure is required by any law;
 - (b) when this Exploration Right has expired or has been terminated;
 - (c) to an Affiliate, provided such Affiliate maintains confidentiality as provided in this clause 28;

- (d) to the extent such Data and information is required to be furnished pursuant to any legal proceedings or because of any order of any court binding upon a Party;
- to prospective or actual contractors, consultants and attorneys employed by any Party where disclosure of such Data or information is essential to such contractor's, consultant's or attorney's work;
- (f) to a bona fide prospective transferee of Party's interest (including an entity with whom a Party or its Affiliate are conducting bona fide negotiations directed toward a merger, consolidation or the sale of a majority of its or an Affiliate's shares);
- (g) to a bank or other financial institution to the extent appropriate to a Party arranging for funding;
- to the extent such Data and information must be disclosed pursuant to any rules or requirements of any government or stock exchange having jurisdiction over such Party or its Affiliates;
- to its respective employees for the purposes of Operations, subject to each Party taking customary precautions to ensure such Data and information is kept confidential;
- (j) where any Data or information, through no fault of a Party, has become published in the public domain; and
- (k) where any relinquishment or abandonment has taken place (but only as to Data and information with regard to the relinquished or abandoned areas).

Disclosure as pursuant to clauses 28.1.(f) and 28.1.(g) shall not be made unless prior to such disclosure the disclosing Party has obtained a written undertaking from the recipient Party to keep the Data and information strictly confidential.

- 28.2. Each Party shall own the Data as defined in 13.1 above in its possession, whether original or a copy, and after termination of this Exploration Right or relinquishment of any portion of the Exploration Area each Party may sell, trade, license and otherwise disclose or dispose of such Data relating to the areas no longer included in the Exploration Area.
- 28.3. Except as may be required by applicable securities law or stock exchange requirements, no public announcement shall be made by any Party unless the other has been furnished with a copy and has given prior approval, which prompt approval shall not be unreasonably withheld. When a public announcement or statement becomes required by law or necessary or desirable because of impending danger to or loss of life, damage to property or pollution resulting from Operations, either Party is authorised to issue and make such announcement or statement without prior notice or the prior approval of the other where such prior notice and approval is not practical in the circumstances. In such a case, the Party making the announcement or statement shall promptly furnish the other Party with a copy of such announcement.

29. Cession, Transfer, Alienation

- 29.1. This Exploration Right may not be ceded, transferred, let, sub-let, assigned or otherwise disposed of without the written consent of the Minister in terms of section 11 of the Act.
- 29.2. The Minister hereby approves that the Grantee may cede, transfer, let, sub-let, assign or otherwise dispose of or mortgage, either wholly or in part, its rights and obligations under this Exploration Right, to an Affiliate but only if the Grantee's guarantor has bound itself to guarantee the obligations of the transferee. The Grantee shall advise the Grantor by notice within 10 days of effecting such transaction.

30. Law

Subject to section 4 of the Act, and sections 36 (Arbitration) and 21 (Tax) of this Exploration Right this Exploration Right, the interpretation thereof and any dispute arising thereunder or associated therewith shall be governed by and determined in accordance with the laws of the Republic of South Africa.

31. Grantee's Rights and Obligations Ensured

31.1. The Minister approves and adopts the terms and conditions of this Exploration Right on behalf of the Grantor, and undertakes on behalf of the State to do and perform all acts and things which are or may be required to be done or performed to give full effect to this Exploration Right in accordance with its

- provisions, and guarantees that the contractual rights and obligations of the Grantee under this Exploration Right will not be altered without prior written consent of the Grantor and the Grantee.
- 31.2. The Minister, on behalf of the Grantor, agrees and commits to the Grantee for the duration of this Exploration Right, to maintain the stability of the provisions of this Exploration Right as such provisions exist as of the Effective Date of this Exploration Right.
- 31.3. Any renewal of this Exploration Right granted by the Minister in terms of section 76 shall be on the same terms and conditions as existed immediately prior to the renewal.

32. Vis Major

- 32.1. Any act, cause, thing or event outside the control of the Grantee including, but not limited to, acts of God, war, insurrection, civil commotion, blockade, strikes, flood, storm, lightning, fire or earthquake which prevents the Grantee from fulfilling its obligations under this Exploration Right, the Act or the regulations, or such other laws and regulations as may be applicable, may be declared a vis major event and such failure on the part of the Grantee shall not constitute breach. Financial inability, ordinary hardship and inconvenience, however caused or arising, shall not be a vis major circumstance.
- 32.2. If the Grantee by reason of the provisions of clause 32.1 above is prevented from fulfilling its obligations or of enjoying its rights under this Exploration Right, it shall immediately notify the Grantor in writing to that effect and shall take all reasonable steps to remove the cause thereof and to investigate the consequences. The Grantee shall promptly notify the Grantor as soon as conditions of vis major no longer prevent the Grantee from carrying out its obligations and following such notice shall resume Operations as soon as reasonably practicable.

33. Amendments

The provisions of this Exploration Right replace all prior negotiations, representations and other communication between the parties hereto and may be amended only in writing between the parties.

34. Unitisation

- 34.1. A situation may arise where rights held by holders under two or more production rights and/or mining leases and/or exploration rights and or prospecting leases or sub-leases extend over different areas which geologically form part of the same petroleum-bearing area. In such circumstance the Grantor may by notice in writing require the holders of such rights to prepare a scheme for the production of that petroleum-bearing area as a unit to be submitted to the Grantor within the period specified in the said notice which shall be not less than 90 (ninety) days.
- 34.2. A scheme so prepared shall maximise the exploitation and/or prospecting of the unitised area for the benefit of all holders of interests in the respective mining and/or prospecting sub-leases.
- 34.3. If no scheme is submitted within the period specified in the notice or such further period as the Grantor may approve, or if the scheme submitted is not acceptable to the Grantor, the Grantor may appoint a committee consisting of industry specialists to advise the Grantor.
- 34.4. Any committee so appointed shall after having considered any representation made by the Grantee submit a scheme to the Minister as soon as is practical.
- 34.5. The Minister may, if he/she is satisfied that any scheme submitted in terms of sub-clause 34.2. or 34.4. is practical and fair and equitable to all parties concerned, confirm such scheme, and it shall thereupon become binding upon all parties.
- 34.6. If the Grantee fails to carry out any provision of the scheme which is binding upon him, the Minister may by notice in writing require the Grantee to do so within the period specified in the notice and if the Grantee fails to comply with the notice the Minister may terminate the exploration right or production right as the case may be.

35. Waiver or Lenience

No waiver of any of the Grantor's rights and no lenience granted or permitted by the Grantor from time to time in a form which does not constitute a formal amendment of this Exploration Right shall prejudice the Grantor's rights under this right or the Act and shall not constitute grounds for Grantee to expect that the Grantor will continue to waive its/their rights or grant or permit lenience in the future.

36. Arbitration

- 36.1. All disputes relating to or arising out of this Exploration Right shall be finally settled under the rules of Conciliation and Arbitration of the International Chamber of Commerce by three arbitrators appointed in accordance with such rules. Subject to clause 36.2, the venue of the arbitration shall be London, England and the proceedings and records of the proceedings shall be in the English language.
- 36.2. The venue of the arbitration shall be Cape Town, South Africa if the amount in dispute is less than US\$500 0000,00 (FIVE HUNDRED THOUSAND UNITED STATES DOLLARS). However, if the relief sought is declaratory or specific relief, then the venue shall be as provided in clause 36.1.
- 36.3. Notwithstanding any such difference or dispute or reference to arbitration, the Parties shall, to the extent possible, proceed with the carrying out of their obligations under this Exploration Right, subject to the provisions of this Exploration Right.
- 36.4. The Parties irrevocably agree that the decision in those arbitration proceedings :
 - 36.4.1. shall be binding on them;
 - 36.4.2. shall be carried into effect; and
 - 36.4.3. can be made an order of any Court of competent jurisdiction.

37. Environmental Protection

- 37.1. The Grantee hereby undertakes and binds itself to conduct all Exploration Operations in accordance with the requirements of the Act and any other legislation, and the Environmental Management Programme submitted by Grantee in support of its application for this right and approved by Grantor, in a manner that would facilitate the protection and conservation of the natural resources of the Republic of South Africa and of the environment in general;
- 37.2. Subject to the provisions of the Act, the Grantee shall on the expiration or termination of this Exploration Right or on relinquishment of part of the Exploration Area or on the voluntary abandonment of the Exploration Area:
 - remove all equipment and installations from such relinquished or abandoned Exploration Area; and
 - 37.2.2. perform all necessary site rehabilitation and shall take all other actions necessary to minimise hazards to human or other life or to the Exploration Area(s) of others or the environment in general for time to come.
- 37.3. The Grantee may from time to time appoint one or more independent contractors to carry out any work in relation to exploration operations, but the Grantee shall always remain liable for the compliance and observation of the provisions of the Act, the Environmental Management Programme and of the terms and conditions of this Exploration Right.
- 37.4. The Grantee shall within 14 (fourteen) days of the Effective Date provide to the Grantor the financial provision for remediation of environmental damage as set out in regulations 32 and 33 and the approved Environmental Management Programme

38. Costs and VAT

- 38.1. All expenses, fees, costs stamp and transfer duties flowing from this Exploration Right direct or incidental, and the costs of notarial execution and registration, deregistration (in cases or abandonment or relinquishment) including transfer duty, Value Added Tax at the prevailing rate and stamp duty, shall be paid by the Grantee.
- 38.2. The Grantee shall pay Value Added Tax (VAT) at the prevailing rate on all remittances and payments to the Grantor or Petroleum Agency in terms of this Exploration Right as applicable.

39. Exclusive right to Acquire a Production Right

- 39.1. In terms of section 77(1)(a) of the Act but subject to compliance with the further provisions of the Act, the Grantee has the exclusive right to apply for and be granted a production right in respect of petroleum on the Exploration Area.
- 39.2. The Grantor hereby grants to the Grantee the exclusive right to acquire a production right over such part or parts of the Exploration Area as the Grantee may at any time during the currency of this Exploration Right select on the terms and conditions set out in 39.3 below.
- 39.3. Any such production right acquired shall be in accordance with the terms and conditions of the Production Right attached hereto as Annexure I.
- 39.4. The application for and granting of and duration of any such Production Right, applications for renewal of such Production Right, and the rights and obligations of the holder of such Production Right, are governed by, inter alia, sections 78, 79, 80 and 81 of the Act and the relevant regulations.
- 39.5. For the avoidance of doubt, the Grantee shall be entitled to exercise this exclusive right not just once but as many times as can be justified during the currency of the Exploration Right by the results of exploration operations.
- 39.6. In respect of each application for a Production Right the Grantee shall submit a development programme which the Grantee intends to undertake. Such development programme shall:
 - (i) subject to clause 34, relate exclusively to the Production Area applied for;
 - (ii) ensure the maximum efficient recovery of the petroleum;
 - (iii) be compiled in accordance with sound engineering, economic, safety and environmental principles recognised in good international oil-field practice;
 - (iv) address the following matters:
 - required production equipment, processing facilities (if any) and storage facilities;
 - feasible alternatives for transportation of petroleum, including pipelines;
 - onshore and offshore installations required;
 - production profiles;
 - cost estimates;
 - safety and environmental measures;
 - employment and training programme;
 - flaring provisions;
 - overall project economics; and
 - (v) include an environmental impact assessment and management programme.
- 39.7. In the event of the Grantee electing to exercise its exclusive right, as set out in 39.1 above the extent (size) of the selected area referred to in sub-clause 39.1 above, shall be subject to approval by the Granter.

40 Grantor's Option

- 40.1. The Grantor shall have the option, exercisable within 60 (SIXTY) days of the date of acceptance of an application for a Production Right, by written notice to the Grantee, to elect to retain or transfer to a wholly owned Affiliate of the Grantor up to a maximum of 1/10th (ONE TENTH) undivided interest in the Production Right.
- 40.2. Should the Grantor so elect, any application by the Grantee for a Production Right in respect of such Mining Block shall be deemed to have been made in respect of an undivided share which excludes such retained undivided interest, and the Production Right shall be in respect of such residual undivided share.
- 40.3. The Grantor shall not be liable for any prior exploration costs in respect of the undivided interest so retained by it but shall be liable for its undivided interest share of all costs incurred under the Production Right henceforth, as provided in the Operating Agreement, Annexure J hereto.
- 40.4. Should the Grantor retain an undivided ownership interest in the Production Right, the Grantor and the Grantee shall enter into the Operating Agreement (Annexure J hereto)

contemporaneously with the grant of the Production Right. The Accounting Procedure (Exhibit A of Annexure J hereto) shall then apply between the parties to the Operating Agreement.

- 40.5. In the event that, at the time of granting of a Production Right, there is more than one Grantee Party and that an Operating Agreement (and associated Accounting Procedure) is already in effect between the Grantee Parties then the Grantee may invite the Grantor to agree to become a party to that agreement rather than the Operating Agreement described in clause 40.3. and the Grantor shall use its best endeavours to comply provided that the terms and conditions offered are not less favourable to the Grantor than those in the Operating Agreement described in clause 40.3.
- 40.6. If the Grantor elects not to retain an undivided interest in the Production Right, the Production Right shall relate to the whole of the undivided interest in the Production Right.

41 Gas

If the Grantee discovers Petroleum, the economic development of which can only be accomplished if Gas produced as the primary or secondary product is sold commercially, then the Grantee shall at its option, exercisable upon notice to the Grantor and the Grantor, have a period of five years (Gas Market Development Period) from the date of granting of the relevant Production Right in which to study the discovery, locate markets and determine the commerciality of the discovery. The Grantor shall grant said Production Right subject to above, the implementation of which is contingent upon the Grantee determining during such five year period that the discovery can be commercially developed and produced. Prior to the expiry of the said five years the Grantee shall confirm by notice to the Grantor that the discovery can be developed and produced commercially and proceed with implementation of the Development Programme, failing which the aforesaid Production Right shall be deemed terminated. The Gas Market Development Period may, at the discretion of the Grantor, be extended to a maximum of 10 (ten) years.

42. Entire AgreementAct Prevails

Subject to the Act, this Exploration Right and the Appendices attached hereto (those Appendices being and forming an integral part of this Exploration Right) contain the entire agreement between the Parties and supersedes all their prior understandings, agreements and negotiations of whatsoever nature.

In the event of any inconsistency between this Exploration Right and the Act, the Act shall prevail.

43. Severability

Any provision of this Exploration Right which is in any way unenforceable or which may render it unenforceable or which may contravene the applicable laws or regulations of any jurisdiction to which this Exploration Right is subject shall be deemed, to the extent of such unenforceability or contravention, severable from this Exploration Right and of no force or effect and shall not affect any other provision of this Exploration Right and its enforceability or legality in all other respects.

44. Notices, domicilia citandi et executandi and Representatives

All notices, requests and reports provided for herein shall be in writing and shall be delivered either by hand to an authorised representative of the receiving Party, or sent by courier or telefax to the following addresses in the Republic of South Africa, provided that if given by telefax a copy thereof shall be sent immediately by prepaid registered airmail:

Grantor:

Minister of Minerals and Energy

Physical address:

Postal address:

Mineralia Centre Cnr Visagie & Andries Streets PRETORIA Private Bag X59 PRETORIA

Tel number: (27) 12 317 9000

Fax number: (27) 12 322 3416

S \$ 5 7 3	PETROLEUM AGE	NCY:			1 14	W.	Sec.	
	Attention : Chief Ex	ecutive	³⁰ 60					2
	Physical address :		8	8 8	Postal addre	ess:		. 4
	Petroleum House 151 Frans Conradie PAROW	Drive		ूच ह अ. ८.	P O Box 117 PAROW 7499	' 4		
<i>a</i>	Western Cape							10
	Tel number : (27) 2	938 3500	88	Fax nur	mber : (27) 21	938 3520		а
	GRANTEE:				180)*)		
		LIMITED			76			
±	Attention : Mr	<u> </u>	or					
	Physical address:	² n e	8 a	18	Postal addre	ss:	3 96	
30			¥4 75					8
6 Se S	Tel number :		e = 4	Fax nur	nber:			
= S_M			2	1 ax mai		 -		2
	Each Party also for all purposes Grantee shall w notice to the G	chooses the arising under within 7 (SEV arantor in wh	physical add this Explora EN) calendation it appoi	dress specition Right, ar days of interesting interesting in the contract of t	including servi the Effective epresentative	ice of proces Date of this with whom	s. Explora the Gra	tion Right give
we, na	concerning this Grantee notifies	Exploration I the Grantor of	Right. Such	represent	ative shall cor	ntinue to rep	resent (Grantee unless
10 - 12						E		₩**
45. Reg	istration				O. O.		\$ 20°	¥ %
	erms of section 77(2)(ing Titles Office within					on Right for	registrat	ion at the
20			41	*				XI
Thus done a	nd signed at	5) -arc 51	on the	day of		20	in the r	resence of the
undersigned			on		1			roccinos or ano
AS WITNES	5E5:							30
1	0		88		540 6			
1			*					
2	******************************	33 5						
		for and on b	ehalf of the	Grantor	* .	*	31 [36	

QUOD ATTESTOR

Ø.			NC	TARY PUBLIC	8	. /	
Thus done and sig	gned at	on the			20	n the presenc	e of the
AS WITNESSES:		. 1 .			8 × a	3 *	
2		for and	on behalf of t	ne <i>Grantee</i>		52	
	* si		QUOD ATTI		a a E		
s c			NO	TARY PUBLIC			
			ANNEXURE	A			
	PLA	N AND DESCRIPT	TION OF THE	EXPLORATION	AREA	5.	
		2 8	To be inseted	i	14	20	10
14		· 1	ANNEXURE	B	4,4	S	
		PERIODIC	C RELINQUIS	SHMENTS		* 4	
A	reas to be reline	quished		2			
(a	At the e	nd of the Initial Per f the Exploration A	iod Grantee s rea	shall relinquish n	ot less than	_% of the origin	nal
(b		nd of the First Ren f the Exploration A	ewal Period (rea relinquish	Grantee shall included to not less th	rease the propo an%	ortion of the orig	ginal
(c) At the electron	nd of the Second F f the Exploration A	Renewal Perio rea relinquish	od Grantee shall led to not less th	increase the pr an%	oportion of the	original
- (4)	ž.		NNEXURE (\$ \$ B	4.	teg a	
a a .	ACCOUNT TO	O WHICH PAYME	NTS TO THE	GRANTOR ARE	TO BE MADE		5 0
	The amounts Petroleum Age	payable to the Gra ency at:	ntor will be de	eposited by the C	Grantee into the	account of the	
	Bank: Branch Branch code Account no.	ABSA	•••	2 (##) (#) (#)	2 S		
	For payments further informa	originating in the Ution may be neces	ISA the follow sary	ing	2	*	
	PAYMENT TO	BE MADE TO:	x	0	300000 VO 200 CERS 90	18 N	

Full name of Bank

Address

Harris Bank International Corporation 3 Times Square New York

New York, 10036 **SWIFT Address** HATRUS33 Account Number 49-0-2945-7 For Credit of **ABSA Bank Limited** International Banking Services: Corporate Pretoria South Africa For Further Credit CEF (PTY) LTD **Under Reference** SEF 8245Y55 The remitting Bank must confirm payments by MT100 to: **ABSA Bank Limited IBS** Corporate Pretoria South Africa **SWIFT Code ABSAZAJJACPT** For Attention Amanda Reyneke

Kindly confirm payments per Fax to CEF (PTY) LTD Attention *Dolf Jonker/Kobus Geertsema* Fax No. (27)11 784 5494

In all cases, the Grantee is required to deliver proof of payment to the Grantor within 3 working days of effecting the payment.

ANNEXURE D

CONTRIBUTIONS TO THE UPSTREAM TRAINING TRUST

1)	within Dollars	30 d	ays of	the	Effec	tive Da	te - US\$		—	<u> </u>	- 7	1	United Stat
)	within (30	days	of	the	first Inited S	anniversary tates Dollars);	of	the	Effective	Date	-,	US\$
)							d anniversary tates Dollars);		the	Effective	Date		US\$
)	within (30	days	of			anniversary tates Dollars);		the	Effective	Date	•	US\$
)	within (anniversary tates Dollars);		the	Effective	Date	•	US\$
)	within	30	days	of			anniversary tates Dollars);		the	Effective	Date	•	US\$

13	(c)	within	30	davs	of	the	e third	anı	niversary	, of	th	e I	Effective	a Da	te		US \$	ı.	
	(-)	(,-	:59		United	State	s Dollars	s);	•							S. t.	V
*			250									š	2	40					
35	(g)	within (30	days	OT .	the	seven United S	th a State:	nniversa s Dollars	ıry (of t	ne	Effectiv	e Da	ate	•	US \$	·	
230	(h)	within	30	days	of				niversar		f th	ne	Effective	e Da	ite	•	US\$		
60 10	S	(-			United 8	States	s Dollars	s);									
et Vigi	- 0.2	The a		ts paya	able	will b	e depo	sited	by the C	arant	ee ir	nto t	he acco	unt of	the	Up	strea	ım Tra	aining
1,		Bank:		AB	SA		Ξ_{ij}				6								
	3	Branc	h					•				, in ;							
4 7 2		Branc																	
	19												4						
		n all case of effectir				s rec	quired to	o deli	ver proo	r of p	aym	ent	to the 1	rust w	/ithii	n th	ree w	orkin!	g days
18		1 N				œ	ΑN	INEX	URE E										
E		g g 2			FXI	PI O	RATION	N WO	RK PRO	OGR	AMN	1F							((6.1)
									7.11.7.7.7.4.		******		3						
The Grante	e shall	carry out	the w	ork sp	ecific	ed he	ereunde	er. Sı	uch work	k, wh	ich s	shall	exclude	e any	app	rais	al wo	ork, fo	r any
F	(a)						Grante		all:								*		
a 11		(i)	com	olete,th	ne re	-proc	cessing	of at	least		9	_ kr	n of exis	sting 2	2D :	seis	mic (data, a	and
	## ##	(ii)	com	olete th	ne re	-proc	cessing	of at	least			_ kr	n ² of ex	isting	ЗD	sei	smic	data,	and
	05 85	(iii)				GF			cessing										
a s	23	(iV)	comp data,		ne ac	quis	ition an	d pro	cessing	of _			squa	are kile	ome	tre	s of 3	D seis	smic
	1%	(v)							nimum d rk Obliga						Per	iod)
W.	25 26 27	8					4			*									24
	1528	which	shall	be the	Mini	mum	Work	Oblig	ation for	the	Initia	l Pe	riod.						*
		The a					of the A	Aimina	um Mad	. Ob	li			d .a.b			-41		
		US\$_	cpena	nure in	resp	_ (or the N	vairairia	um Wor	K OD	ilgat	Uni	ted Stat	tes Do	llar	s).	Suma	ilea ic	be
	<i>4</i> . \						15 12	_					100				107	90	
	(b)						15		antee sh										
*		0.650	STAROUR.			iecērion.			cessing				x.**						el St.,,,
	- Fi	(ii)							nimum d rk Obliga				irst Rer			iod)
	54 E								um Worl									od is	
(3) (4)	(c)	During	the S	Second	Ren	ewa	l Period	d, the	Grantee	sha	II:								
31 Ki 1 Ki	*	(1)							minimu res) each		pth	of _	-	m	E v				6
10		which s	hall b	e the l	Vinin	num	Work C	Obliga	ation for	the S	eco	nd F	Renewal	Perio	d.	3 8		9	

	The expenditure in respect of the Minimum W estimated to be US \$(
		- W	
(d)	During the Third Renewal Period, the Grantee	shall:	85
	(i) drillexploratory wells to a minimum me	num depth of m tres) each,	
- 2	which shall be the Minimum Work Obligation I	or the Third Renewal Period.	
	The expenditure in respect of the Minimum W	ork Obligation for the Third Renewal	

ANNEXURE F

FINANCIAL GUARANTEE

A financial guarantee acceptable to the Grantor in the form of one of the alternatives incorporated as examples under Appendix B to the regulations, to be attached

ANNEXURE G

CAPITAL ALLOWANCE SCHEDULE

Actual capital expenditure qualifying for re	FIRST YEAR OF	Unredeemed balance of Capital Expenditure (C/I	E) for the purpose of calculating
		the Capital Allowance (C/A)	
ncurred during the year	R 2 000 000	Incurred C/A for the year	R 2 000 000
C/A for the year	R 120 000	(at 12 %, pro rata for a portion of a year*, say)	R 120 000
Carried forward	R 2 120 000		R 2 120 000
	SECOND YEAR O	F PRE-PRODUCTION ACTIVITIES	
Balance brought forward	R 2 120 000	Balance brought forward	R 2 120 000
ncurred during the year **	R 2 000 000	Incurred during the year **	R 1 940 000
C/A for the year	_ R 370 800		R 4 060 000
The second secon		C/A for the year (at, say 12 %)	R 370 800
Carried forward	R 4 490 800		<u>R 4 430 800</u>

C/E is calculated at 12 % (or other applicable percentage) from the end of the month during which it was incurred until the end of the tax year; in this example, however, it is supposed that all the qualifying C/E was incurred just before the end of the 6th month of the tax year

Interest and other finance charges on loans do not qualify for the C/A

FIRST YEAR OF PRODUCTION

1.	Calculation of taxable income	22 E S	3. Calculation of C/A in determining the balance of C/E for the purposes of the C/A
	Gross income R 3 000 000		A Property of the Appropriate Control Section 1
	Working costs R 3 740 000		Unredeemed balance C/A at 12 % *
	Working loss R 740 000		
	Redemption of C/E R 6 626 896		Balance brought forward R 4 430 800 R 531 696
	Assessed loss R 7 366 896		C/E incurred R 1 500 000 R 90 000
	A3363360 1033	48	Possible C/E included in
2.	C/E qualifying for redemption		working costs but on which
۷.	C/E qualifying for recemption	. 6	
22	Hdd	29 W	C/A is allowed R 240 000 R 14 400
	Unredeemed balance b/f R 4 490 800	70 89	
	Incurred during the year R 1 500 000	(5)	C/A for the year <u>R 636 096</u> → <u>R 636 096</u>
	C/A for the year R 636 096		R 6 806 896
	Total allowed for redemption R 6 626 896	2 2	
		59	<u>Deduct</u>
	Thus no balance carried forward		C/E allowed for redemption:
		*	amount deducted in the
00		10	calculation of profits is
i.		800 0	R6 626 895 but the deduction
			is restricted under section 26(3)
			to the amount of the "profit",
	70 CC (20 CC)))))))))))))))))))))))))))))))))))		i.e. R NIL
		2	Unredeemed balance
į.	E		carried forward R 6 806 896
Š	80 88 and		Carried IO Ward
-		* (Or other applicable percentage

1.	Calculation of taxable income	lnee		SECOND YEAR OF PRODUCTION
	Calculation of taxable incomes	ioss		 Calculation of C/A in determining the balance of C/E for the purposes of the C/A
	Gross income		R 7 000 000	U/A
	Working cost Working profit		R 3 500 000 R 3 500 000	Unredeemed balance C/A at 12 %
	Deduct			Balance brought forward R 6 806 896 R 816 828 C/E incurred R 1 000 000 R 60 000
	Redemption of C/E Assessed loss b/f	R 1 900 828	D a annua .	Possible C/E included in working costs but which
	Assessed loss carried forward	R 7 366 896	R 9 267 724 R 5 767 724	qualifies for C/A R 400 000 R 24 000 C/A for the year R 900 828 → R 900 828
2.	C/E qualifying for redemption			R 9 107 724
	Balance b/f C/E incurred for the year	W	N I L R 1 000 000	Deduct C/E allowed for redemption as calculated below * R 3 400 000
	C/A for the year	96	R 900 828 R 1 900 828	Unredeemed balance carried forward B 5 707 724
			ž.	g ²
	~			1. 0/5
			(建)	* <u>C/E allowed for redemption</u> Deemed deduction [sec 26(4)] is R9 107 724
			18 (48) (48)	but the deduction is restricted under sec 26(3) to the amount of the "profit" i.e. Working profit R 3 500 000
				Plus
				C/E included in working costs but which can for the purposes of the C/A be regarded as
	W.			C/E R 400 000
			(a)	R 3 900 000
	8			Deduct So much of the loss brought forward which
				is not ascribable to the C/E (see below) # R 500 000 R3 400 000
35			48	# The portion of the loss brought forward not
			8. 34	resulting from the deduction of C/E
				Assessed loss brought forward R 7 366 896 Actual C/E
				(redemption allowance) R 6 626 896 C/E in working costs R 240 000 R 6 866 896
				R 500 000

STAATSKOERANT, 6 DESEMBER 2002

ANNEXURE H

SCHEDULE OF EXPERTS (ACCOUNTING)

To be agreed between the parties at the time.

ANNEXURE I

PRODUCTION RIGHT AGREEMENT

A production right agreement based on the specimen document published as Appendix C part 4 to the regulations under the Act, to be negotiated between the parties and attached.

ANNEXURE J

OPERATING AGREEMENT

An operating agreement based on the Association of International Petroleum Negotiators Model form Operating Agreement to be negotiated between the parties and attached

30.

31.

Cession, Transfer, Alienation

Law and interpretation

FORM MM: OPERATION RIGHT

INDEX

Pream	amble	
1.	Definitions	
2.	Grantor's Agent: The petroleum Agency	
3.	Description of Production Area	*
4.	Production Right	23 21
5.	Ancillary Rights and Obligations	
6.	Commencement, Duration and Renewal	
7	Fees and Royalties	8
8.	Technical Advisory Committee	
9.	Minister's Right to Cancel or Suspend the Production Right	* *
10.	Abandonment or Relinquishment of the Production Area	
11,	No guaranteed right to other minerals	
12.	Examination of the Production Area	er e
13.	Records and samples	E S
14.	Reports	
15.	Production Work Programme and Budget	
16.	Production of Petroleum	E 6 8
17.	Discoveries and testing	et or
18.	Black Economic Empowerment	a a a
19.	Good international oil-field practice	2 E E
20.	Existing Data	# #**
21.	Upstream Training Trust	e t
22.	Local labour, equipment, materials and contractors	1965
23.	Тах	n *
24.	Accounts and Audits	B.
25.	Customs duties	
26.	Exchange Control	2
27.	Indemnity and insurance	
28.	Health and Safety	
29.	Data and information to be kept confidential and limitations of	n public announcements

- 32. Grantee's Rights and Obligations Ensured
- 33. Vis Major
- 34. Amendments
- 35. Unitisation
- 36. Waiver of Lenience
- 37. Arbitration
- 38. Environmental Protection
- 39. Costs and VAT
- 40. Further Developments
- 41. Grantor's Option
- 42. Entire Agreement
- 43. Severability
- 44. Notices, domicilia citandi et executandi and Representatives
- 45. Registration

ANNEXURES

- A Plan and description of the Production Area
- B Account to which payments are to be made
- C Contributions to the Upstream Training Trust
- D Production Work Programme
- E Financial Guarantee
- F Black Investment
- G Capital Allowance Schedule
- H Schedule of Experts
- I Operating Agreement (40.3)(40.4 x 2)

SPECIMEN AGREEMENT FOR A PRODUCTION RIGHT

SE - 50					PROTOCOL N	10	/
	18						
LET IT HE	REBY BE KN	OWN:					98
THAT on t	this the	in the	nt witnesses ne	in the Notary Public, o	year Two thousaduly sworn and ad Province, Rond appeared:	ind and mitted, residing epublic of Sou	ig and practising uth Africa, and in
the present	oc of the duba	onbing competer	ni wanesses pe	roomany came a	на арреатеа.	0	
		etroleum Explora orised represent		, C tation (Proprieta	hief Executive of ary)(Limited), and	South Africas such in his	an Agency for or her capacity
			THE REPUBLI	C OF SOUTH A	FRICA		¥2
being duly Energy on and Petrol	authorised the	ereto under and	by virtue of the erms of section Act; 2002 (Ac	powers delegat 97(1) read with	as "the Grantor"), ed to him or her b sections 65(2)(a) 2), the said Minis	y the Minister and 65(2)(b)(v	of Minerals and vi) of the Mineral
			e a	AND		W.	£7
(2) <u>d</u> u	uly authorised	representative o	, a f:		of, and as such	in his or her	capacity as the
						10 W	*
		(No)		8 8
being duly	authorised	its successors in thereto under a the day of _	and by virtue	of a Resolution	s "the Grantee"), on of Directors	he or she, the of the Grai	e said Appearer, ntee passed at
		delegation and o		tion have this da	ay been exhibited	to me, the No	tary, and remain
AND THE	APPEARERS	DECLARED TH	AT:				
WHEREAS	5	Resources De	velopment Act, ned in the said	2002 (Act No.	s 65(2)(a) and 75 of 2002), the of petroleum (as	Grantor may	grant production
AND WHE	REAS		ishes to obtain ed to in clause		ght in respect of P	etroleum in, o	n and under the
AND WHE	REAS		s agreed to gra		tion Right to the G	rantee in acco	ordance with the

NOW THEREFORE THE PARTIES AGREE HERETO:

1. Definitions

The definitions in the Act and regulations shall apply also to this agreement and additionally the following words and expressions shall have the meaning assigned to them:

- 1.1. "the Act" shall mean the Mineral and Petroleum Resources Development Act, 2002 (Act No. ... of 2002) or any subsequent law which amends or substitutes the Act;
- 1.2. "Affiliate" of a Party means a company which owns, or is owned by, or which is owned by an entity which owns that Party; "owns" and "owned" mean in this definition the ownership directly or indirectly of more than 50 % of the voting shares or members' interests;
- 1.3. "Effective Date", shall mean the date on which this Production right agreement is executed notarially by the last Party so to execute;
- 1.4. "the Production Area", shall mean the property referred to in clause 2 of this agreement, excluding those portions relinquished or abandoned from time to time;
- 1.5. "Production Operations", shall mean any operation, activity or matter that relates to the Production for Petroleum in the Production Area by the Grantee, its successor and/or assign and/or its contractors;
- 1.6. "the Production Work Programme", shall mean the Grantee's written recordal of the manner in which it intends to carry out the Production operations in the Production Area, and shall include all amendments and supplementary submissions relating to the Production programme which may be made by the Grantee and approved in writing by the Grantor from time to time, all of which amendments and supplementary submissions shall form part of this Production Right as Annexure D attached hereto, and shall include and incorporate the Development Plan, approved by the Grantor, submitted in support of the Grantee's application for this Production Right;
- 1.7. "the Grantee", shall have the meaning attributed thereto in the description of the Parties above and shall as appropriate mean all Grantee Parties;
- 1.8 "Grantee Party" means each cessionary, transferee, lessee, sub-lessee or assignee of rights, duties and obligations under this Production Right pursuant to the provisions of clause 28 hereof;
- 1.9. "the Grantor" shall have the meaning attributed thereto in the description of the Parties above; Exploration and Exploitation (Proprietary) Limited (No. 1999/015715/07);
- 1.10. "the Income Tax" shall mean the Income Tax Act, 1962 (Act No. 58 of 1962) as amended or substituted;
- 1.11. "installation" shall have the meaning assigned to it in the Mine Health and Safety Act, 1996 (Act no. 29 of 1996) as amended;
- 1.12. "Petroleum Agency" means the South African Agency for Promotion of Petroleum Exploration and Exploitation (Proprietary) Limited (No. 1999/015715/07);
- 1.13 "Quarter" shall mean a three-month period of a year beginning on 1st January, 1st April, 1st July or 1st October of any year;
- 1.14. "State" shall mean the Grantor;

Reference to the singular of a defined term includes a reference to it in the plural and vice versa and reference to any gender includes a reference to all other genders.

2 Grantor's Agent : The Petroleum Agency

2.1. Without detracting from section 66 of the Act the Grantor records that it has appointed the Petroleum Agency as the Grantor's agent in regard to all matters relating to this Production Right, excluding matters concerning health and safety save in regard to the approval of exploration rights

and production rights.

- 2.2. Without detracting from the generality of clause 2.1:
- 2.2.1. the Grantee shall be obliged to pass all data, including but not limited to information contemplated in section 14 of the Act and this Production Right and samples and borehole cores, and interpretations of all the aforegoing, to the Petroleum Agency, and all of which data, information, samples, borehole cores, and interpretations, shall be passed in the format and in the medium as may be required by the Petroleum Agency in writing from time to time.
- 2.2.2. The Grantor hereby appoints the Petroleum Agency or its representatives as the Grantor's representative or representatives on the Technical Advisory Committee provided for in clause 8 hereof.

3. Description of Production Area

The Production Area shall comprise the following:

Block
Region
Measuring: ____km²

which Block is more fully described in the attached plan and description, marked Annexure A.

4. Production Right

The Grantor hereby in terms of sections 79 and 81 of the Act grants to the Grantee, which hereby accepts, this Production right conferring the sole and exclusive right to explore for, remove and dispose of petroleum found in, on and under the Production Area on the terms and conditions set forth herein and subject to the provisions of the Act.

Ancillary Rights and Obligations

- 5.1. Without limiting the generality of clause 4 or detracting from Grantee's rights in terms of sections 5 and 81 of the Act, the Grantee shall have the right to:
 - 5.1.1. enter the Production Area together with the Grantee's employees, and bring on to the Production Area any plant, machinery and equipment and build, construct and lay down any surface or underground infrastructure which may be required for the purposes of Production; provided that:
 - 5.1.1.1. onshore land (i.e. land above the high-water mark, as defined in section 1 of the Sea-Shore Act, 1935 (Act No. 21 of 1935)) may be used for the said purposes if (1) the permission of the owner thereof, where such owner is not the State or, in terms of the Sea-Shore Act, 1935, the Minister, has first been obtained and if (2) the owner thereof will be compensated by the Grantee for actual damage caused to such land; and
 - 5.1.1.2. cables or pipelines may be laid over or under the said Sea-shore or the Sea or the Continental Shelf and structures may be erected on the Production Area with relevant permission of the Petroleum Agency in terms of the Act.
 - 5.1.2. explore for the Grantee's own account on or under the Production Area for petroleum;
 - 5.1.3. subject to the National Water Act, 1998 (Act No. 36 of 1998), use water from any natural spring, lake, river or stream, situated on, or flowing through, the Production Area or from any excavation previously made and used for exploration or production purposes, or sink a well or borehole required for use relating to Production on the Production Area;
 - 5.1.4. in terms of section 80 of the Act, apply for and be granted a renewal of this Production Right.
 - 5.2. Without detracting from the Grantee's obligations in terms of this Production Right or in terms the Act the Grantee shall:

- continuously and actively conduct Operations in accordance with the Production Work Programme;
- comply with the terms and conditions of this Production Right and the relevant provisions of the Act;
- 5.2.3. comply with the requirements of the approved environmental management programme; and
- 5.2.4. pay the prescribed fees to the Grantor.
- 5.3. No provision contained in this Production Right shall constitute a stipulation for the benefit of any person or entity who is not a Party to this Production Right capable of acceptance by such person, except the Petroleum Agency.

6. Commencement, Duration and Renewal

- 6.1. This Production Right will commence on the Effective Date and, unless terminated earlier under any provisions hereof, will continue in force for a period of _____ [not more than 30 years] from the Effective Date
- 6.2. The Grantee shall within 30 (thirty) days reckoned from the Effective Date, commence with Production Operations in the Production Area and shall thereafter conduct such operations continuously and actively in accordance with the Production Work Programme and the approved Environmental Management Programme.
- 6.3. This Production Right shall be renewed by the Minister on application in terms of section 80 of the Act by the Grantee if the Grantee has complied with the:
 - terms and conditions of this Production Right and is not in contravention of any relevant provision of the Act;
 - 6.3.2. Production Work Programme; and
 - 6.3.3. requirements of the approved Environmental Management Programme.

7. Fees and Royalties

Year Table

- 7.1. The Grantee shall pay to the Grantor the prescribed production fees and the prescribed royalties in respect of petroleum removed and disposed of during the course of the Grantee's Production Operations, as envisaged in section 81(2)(e)
 - 7.1.1. an annual production fee of R____(____RAND) per km² of the Production Area payable yearly in advance, which shall increase annually on the anniversary of the Effective Date by the increase in the Consumer Price Index (All Items, Metropolitan and Other Urban Areas) as published from time to time by Statistics South Africa
 - 7.1.2. a royalty of _____ of the value of any petroleum or minerals found, removed and disposed of in the course of Production operations from the Production Area; payment of the said share shall be made within 30 (thirty) days after the end of every Quarter and shall be accompanied by a certificate or other document acceptable to the Grantor, reflecting particulars of the value of petroleum and/or minerals found, removed and disposed of on the Production Area during the Quarter in question; such quarterly payments shall,
 - (a) in the case of petroleum,
 - (i) be made on a fair market value of the Petroleum produced, calculated at the boundary of the mine. "Fair market value" shall, except as provided in (ii) below, be the actual realised price used to determine gross income for income tax reporting purposes under the Income Tax Act. The currency of payment will be either SA Rand or US Dollars.

- (ii) In the event that all sales during a Quarter are to an Affiliate of Grantee then fair market value for the purposes of calculating royalty will be determined utilising a basket of similar products on offer world-wide. Otherwise, the fair market value for purposes of calculating royalty on sales to an Affiliate will be the average weighted by volume of the actual realised price of sales to third parties during such Quarter.
- (b) In the case of other minerals it will be:
- based on the selling price of the minerals (if applicable);
- in the case of minerals not sold, on the estimated value thereon; and
- after the subsequent sale (if applicable) of any minerals in respect of which payment has already been made on the estimated value thereof, on the difference between the estimated value and selling price.
- 7.2. In terms of section 66(i) of the Act the Grantee shall additionally pay to the Grantor a data maintenance fee equal to 1% of the fair market value, at the boundary of the mine, of petroleum produced, removed from the Production Area, and disposed of.
- 7.2.1. "Fair market value" shall be as defined in 7.1.2. and shall, except as provided in clause 7.2.2 below, be the actual realised price used to determine gross income for income tax reporting purposes under the Income Tax Act. The currency of payment will be either SA Rand or US Dollars.
- 7.2.2. In the event that all sales during a Quarter are to an Affiliate of Grantee then fair market value will be determined utilising a basket of similar products on offer world-wide. Otherwise, the fair market value in respect of sales to an Affiliate will be the average weighted by volume of the actual realised price of sales to third parties during such Quarter.
- 7.3. The production fees and data maintenance fees referred to in clause 7.1 and 7.2 shall be paid within 30 (thirty) days after the end of each Quarter following the Effective Date.
- 7.4. If any fees or royalties referred to in clauses 7.1 or 7.2 are not paid punctually on or before the due dates for payment, the Grantee shall be in *mora debitoris* and shall pay interest thereon at the rate prescribed in terms of section 80 of the Public Finance Management Act, 1999 (Act No. 1 of 1999) reckoned from the due date for payment to the date of actual payment.
- 7.5. Regardless of any relinquishment of parts of the Production Area or early termination of this Production Right for whatever reason, no portion of the fees referred to in clauses 7.1 and 7.2 shall be refundable.
- 7.6. All fees and royalties which are payable by Grantee to the Grantor in terms of the Production Right shall be paid by deposit into the account of the Grantor at the bank and account indicated in Annexure B, or such other bank and account as the Grantor may from time to time give notice of. The onus for timely payment rests solely with the Grantee; the Grantor will not normally issue invoices. It should also be noted that the Upstream Training Trust is independent of the Grantor and payments to it should be made separately to the bank and account indicated in Annexure C, or such other bank and account as the Upstream Training Trust may from time-to-time give notice of.

8. Technical Advisory Committee

- 8.1. The Grantor and Grantee shall as soon as possible after the Effective Date establish a committee to be known as the Technical Advisory Committee which shall consist of:
 - (a) a chairman and one other person appointed by the Grantor; and
 - (b) two other persons appointed by Grantee.

The membership of the Technical Advisory Committee may be enlarged to include one representative of each of Grantee Parties. The Grantor may in this instance enlarge its representation to equal that of all the Grantee Parties.

- 8.2. The Grantor and any Grantee Party may appoint by notice an alternative member to act in the place of its representative on the Technical Advisory Committee during his or her absence or incapacity to act as a member of the Committee. When an alternate member acts in the place of any member he or she shall have the powers and perform the duties of such member.
- 8.3. Without prejudice to the rights and obligations of the Grantee in relation to the management of Operations the functions of the Technical Advisory Committee shall be:
 - to review all Operations carried out by the Grantee and to provide the Grantee advice and recommendations with respect thereto;
 - (b) to review and approve any proposed Production Work Programme and budgets, and any propose amendments thereof, which are to be submitted by the Grantee to the Grantor in terms of clause 15 and to monitor the implementation of any appraisal programmes submitted by Grantee to the Grantor in terms of clause 16;
 - (c) to review any appraisal programmes to be submitted by Grantee to the Grantor in terms of clause 16 and any development plan which Grantee proposes in relation to any further discoveries of petroleum in the Production Area;
 - (d) to review the accounting of expenditure and the maintenance of operating records and reports kept in connection with Operations: and
 - (e) to offer advice to the Grantee in order to promote the efficient carrying out of Operations.
- 8.4. All meetings of the Technical Advisory Committee shall be held at least semi-annually. The location of such meetings shall be in Cape Town, South Africa unless otherwise unanimously agreed by its members, and the Grantee shall provide the the Grantor with copies of all documentation and presentation materials 7 (seven) days prior to the meeting.
- 8.5. Three members of the Technical Advisory Committee shall form a quorum for a meeting of the Committee, provided that at least one representative of the Grantee and one representative of the Grantor is present.
- 8.6. Any member of the Technical Advisory Committee shall have the right to call any expert to any meeting of the Committee to advise the Committee on any matter of a technical nature requiring expert advice.
- 8.7. The Chairman of the Committee shall cause minutes of each meeting to be kept and circulated to the members.

9. Minister's Right to Cancel or Suspend the Production Right

1.3

7 1 -

.135

- 9.1. It is recorded that the Minister is empowered to cancel or suspend this Production Right in the circumstances envisaged in and in accordance with the provisions of section 85 read with section 44 of the Act.
 - 9.2. In the event of this Production Right being cancelled in accordance with the provisions of clause 8.1, the Grantee will not be absolved from its obligations or liabilities which have accrued up to the date of such cancellation.
 - 9.3. The cancellation of this Production Right in terms of clause 8.1 above will be without prejudice to the Grantor's other rights under this Production Right or in statutory or common law of the Republic of South Africa, including its right to recover any damages it may have suffered or may suffer, its right to claim specific performance or its right to claim ancillary relief.

10. Abandonment or Relinquishment of the Production Area

- 10.1. The Grantee may, subject to sub-clauses 10.2. and 10.3., abandon this right in its entirety or relinquish any portion or portions of the Production Area at any time.
- 10.2. In abandoning or relinquishing the Production Area or any portion thereof, the Grantee will not be absolved from those obligations or liabilities, either in the form of Production work or otherwise, which may have accrued under the terms and conditions of this Production Right, the Production Work programme, the Environmental Management Programme and the Act up to the date of such abandonment or relinquishment.

10.3. The Grantee may, by notice in writing addressed and delivered to the Grantor, abandon or relinquish any part of the Production Area from a date not less than six months from the date on which such notice was delivered to the Grantor.

All such abandoned or relinquished areas shall cease to be part of the Production Area on the date identified so notified provided that the area(s) relinquished are:

- (i) bounded by lines running either due North and South or due East and West and coinciding with integral multiples of 01 minutes of longitude or latitude; and
- approximately rectangular with the length not more than three times the width of the narrowest portion of the area to be relinquished.
- 10.4. With effect from the date the Grantee has so abandoned or relinquished a portion or portions of the Production Area, the Grantor shall be entitled to grant any exploration rights or production rights or any of the rights and permits referred to in the Act in, on, or under the portion or portions, so abandoned or relinquished, to any persons or persons.
- 10.5. Upon abandonment or relinquishment of the Production Area or any portion thereof, the Grantee will furnish the Grantor with all exploration and production results and/or information and interpretations, as well as its evaluation of the geological, geophysical and borehole data in respect of such abandoned or relinquished area or areas insofar as it applies to petroleum or any other mineral or minerals referred to in clause 10. Such results and/or information may be made available to third parties.
- 10.6 Upon abandonment or relinquishment of the Production Area or any part thereof the Grantee shall apply for an Exoneration Certificate in terms of section 40 of the Act.

11. No guaranteed right to other minerals

- 11.1. The Grantee shall not have the exclusive right to acquire a prospecting right, mining right, mining permit or retention permit in respect of any mineral or minerals in, on and under the Production Area.
- 11.2. Should any mineral or minerals be discovered in the course of Production Operations, the Grantee shall nevertheless be bound to report such discovery to the Grantor. The Grantee may thereafter, subject to any prior rights, apply for and, if granted the required prospecting rights by the State, prospect for such mineral or minerals.

12. Examination of the Production Area

- 12.1. The Minister may authorise inspections and exercise of related powers as set out in the Act
- 12.2. Upon request by the Granter during offshore operations of Grantee, Grantee shall provide free transport between the Grantee's onshore base and the offshore facilities as well as accommodation on the offshore facilities for authorised officers.

13. Records and samples

- 13.1. The Grantee must keep, in a form as may be from time to time reasonably approved by the Grantor, current and accurate records of all Production Operations. Such records must include, inter alia, details of drilling, completion, plugging or abandonment of all boreholes, drill-stem tests, well logs, all other geological and geophysical information and interpretation, maps, digital records obtained by the Grantee in the course of carrying out Operations (hereinafter referred to as "Data") and any equipment lost overboard or down hole, relating to the Production Area as may be necessary to preserve all information which the Grantee has about the geology of the Production Area, and such other records as the Grantor may from time to time direct and which the Grantee can reasonably keep.
- 13.2. The Grantee shall at its own expense save and correctly label a representative portion of each sample of cores and cuttings from the earth or sea bed and strata encountered in drilling wells, such samples to be taken at regular intervals in accordance with good international oilfield practice, and samples of any petroleum or other minerals of potential value recovered in any of the Grantee's wells in the Production Area, to be forwarded promptly to the Grantor at the Grantee's expense. All samples acquired by the Grantee for its own purpose shall be made available for inspection at any reasonable time by the Grantor.

Before the Grantee discards any core and/or cuttings and samples, the Grantor shall, by prior notice, be afforded an opportunity of obtaining the same and these will be delivered to it at Grantee's expense.

- 13.3. The Grantee may export such Data for processing or laboratory examination or analysis by the Grantee or by third parties or storage outside of the Republic of South Africa, provided that representative samples equivalent in quality or, where such material is capable of being reproduced, copies of equivalent quality have first been delivered to the Grantor and provided that the Grantor's prior written approval has been obtained.
- 13.4. The Grantee shall deliver to the Grantor, at the Grantee's expense, copies of all data as defined in 13.1 above as soon as they are acquired or prepared in accordance with the Grantor's reporting guidelines.

14. Reports

- 14.1. The Grantee shall keep the Grantor currently advised of all major developments taking place during the course of Operations and shall furnish the Grantor with Data (as defined eleswhere) and other available information, reports, assessments and interpretations relating to Operations as the Grantor may reasonably require in accordance with the Grantor's "Reporting Guidelines".
- 14.2. The Grantee shall provide the Grantor with a summary report of all Operations on a monthly basis.
- 14.3. Within 30 (thirty) days of the end of each Quarter the Grantee shall submit to the Grantor a report reflecting, for the relevant Quarter, details of all Operations, including but not limited to:
 - the numbers of local and expatriate persons employed;
 - (ii) the work done and money expended on Operations;
 - (iii) the site and depth of every well drilled or being drilled and the formations penetrated and particulars regarding any occurrence of petroleum and/or any other mineral of potential value encountered; and
 - (iv) a statement of compliance with the approved Environmental Management Programme.
- 14.4. Nothing in clause 13 or 14 shall be construed as requiring the Grantee to disclose any of its proprietary technology or that of its Affiliates.
- 14.5. The Grantee shall furnish the Grantor with such other reports as to the progress of its operations in the Production Area as the Grantor may from time to time require and which the Grantee can reasonably make available.
- 14.6. Within three (3) months after the termination and/or cancellation and/or abandonment of this Production Right, the Grantee shall furnish the Grantor with all results of and its interpretations and general evaluation of the geological, geophysical and borehole data relating to the Production Area as it applies to petroleum or any other mineral or minerals under sub-clause 10.2.
- 14.7. The Grantee shall provide copies of all hydrographic information and data obtained by the Grantee or its contractors and sub-contractors, during marine operations undertaken under this Production Right to the Grantor and the Chief Hydrographer of the South African Navy.
- 14.8. All information furnished by the Grantee to the Grantor in accordance with this clause 14 and clause 13 above, may be made available by the Grantor for use by third parties on expiry or termination of the Production Right or relinquishment or abandonment by the Grantee.
- 14.9. The provisions of this clause 14 are additional to and without detraction from the provisions of Regulation 14 made in terms of the Act.

15. Production Work Programme and Budget

- 15.1. The Grantee must, to the satisfaction of the Grantor:
 - 15.1.1. over the period of this Production Right, initiate and thereafter actively pursue operations designed to produce petroleum optimally;

- 15.1.2. carry out and meet the commitments set out in the Production Work Programme annexed hereto marked Annexure D;
- 15.1.3. not less than three months prior to each anniversary of the Effective Date prepare and submit to the Grantor an amended and revised Production Work Programme and budget setting forth the Production Operations which the Grantee proposes to carry out during the twelve months following the aforesaid anniversary and the cost thereof.
- 15.2 Any amended and revised Production Work Programme and budget submitted to the Grantor shall be consistent with the requirements set out in clause 37 relating to environmental protection.
- 15.3 Any revisions or amendments to the aforesaid Production Work Programme and budget must be submitted to Petroelum Agency for approval.
- 15.4. In the event that the work carried out by the Grantee, as reviewed from time to time, is less than that specified in the Production Work Programme and insufficient reason has been given by the Grantee to the Grantor for the failure to carry out the full Production Work Programme, then the Grantor may in terms of section 80(3) of the Act refuse an application for a renewal of this Production Right in terms of clause 6.3 hereof.
- 15.5. It is recorded that the time sequences as set out in the Production Work Programme will commence on the Effective Date.
- 15.6 As set out in section 84 of the Act, each Grantee Party shall secure and deliver to the Grantor within 14 (fourteen) days of Effective Date a financial guarantee, appropriate to its proportionate interest in this Production Right in the format indicated in Annexure E.

16. PRODUCTION OF PETROLEUM

- 16.1. If petroleum produced by the Grantee is of quality which can be utilised by local refineries and provided the price for such petroleum is not greater than the price for petroleum available to local refiners from other sources with due consideration to the quality, product yield and other factors which relate to price, then the Grantee shall be entitled to dispose of its entire production to local refiners forming part of the "private sector" as defined from time to time by the Minister provided, however, that if the demand for petroleum by the said refiners in the private sector in the Republic of South Africa shall at the time or any time thereafter be supplied entirely from indigenous sources, then the Grantee shall be entitled to dispose of a portion of its production to the said refiners based upon the ratio which the Grantee's share of the average monthly production of petroleum from the Production Area shall bear to the total average monthly production of petroleum from all such indigenous sources. Such allocation shall be re-determined periodically but not more frequently than once every three months.
- The Grantee shall be entitled to export petroleum and/or processed products produced from the Production Area which are not sold under sub-clause 16.1 hereof in the Republic of South Africa.
- 16.3 No export duty or similar export charges shall be levied on petroleum and/or processed products exported by the Grantee in terms of clause 16.2.
- Nothing in this clause 16 shall be deemed to require the Grantee to refine any portion of its production of petroleum before seiling the same in or outside the Republic of South Africa and the Grantee shall be entitled, subject to obtaining prior State approval, to erect a refinery for the processing of petroleum produced by it, if such refinery can be justified.

No provision in this clause 16 shall prejudice the right of the Grantee to use freely and exempt from taxes specified in clause 2 below, petroleum produced under the terms of this Production Right, in Operations, including its injection in or return to any producing formation.

17. Discoveries and testing

- 17.1. When a further discovery of Petroleum is made in the Production Area, the Grantee shall -
 - (a) promptly inform the Grantor by notice of the fact that such discovery has been made;
 - (b) forthwith cause tests to be made in connection with such discovery in order to determine whether it is, in the Grantee's opinion, commercial. Before testing each occurrence of Petroleum, the Grantee shall give notice of testing to the Grantor and shall provide, free of charge, to the representatives of the Grantor access and transportation to the drilling rig and reasonable facilities there in order to witness such tests; and
 - (c) within 60 days after having completed such tests, furnish the Grantor with a report containing an evaluated result of such tests.
- 17.2. The testing and measuring of Petroleum for the purpose of establishing the existence of a commercial discovery of Petroleum shall be carried out in accordance with good international oil-field practice.
- 17.3. If the Grantee considers, after providing the report referred to in clause 17.1.(c), that the discovery of petroleum could be commercial, then the Grantee shall forthwith take reasonable steps to appraise the discovery and submit the appraisal programme which it intends to undertake to the Grantor for its agreement. The Grantor shall review such appraisal programme and within 30 days of receipt thereof either agree to it or give Grantee any comments which Grantor may have with respect to it. If the Grantor fails to respond with comments within such time, it shall be deemed to have agreed to the appraisal programme.
- 17.4. The Grantee shall, within one year from the date on which the Grantor is advised of the appraisal programme or such longer period as may be appropriate under the circumstances address and deliver to the Grantor a full report containing particulars of the results of the appraisal programme, including particulars and preliminary estimates relating to:
 - (i) the location and depth of petroleum bearing structures;
 - (ii) the composition of petroleum;
 - (III) the estimated recoverable reserves of petroleum;
 - (iv) the estimated daily production potential of petroleum; and
 - a preliminary estimate of the expenditure of development and production operations, including the expenditure on transportation of petroleum, based on an outline design for the development of the discovery of petroleum.
- 17.5. the Grantor may, at any time after delivery of the report and estimates referred to in clause 17.4 request the Grantee to supply such further particulars relating to such reports as the Grantor deems necessary and the Grantee shall comply in writing with such request within 30 days from the date of delivery of such request.

18. Black Economic Empowerment

The Grantee acknowledges that in order to give effect to section 2(d) of the Act, Grantor is desirous of expanding opportunities for historically disadvantaged persons (as defined in the Act) to enter the upstream petroleum industry and to benefit from the exploitation of petroleum resources in South Africa and as such Grantee undertakes to use its best endeavours to seek historically disadvantaged South African partners to participate in this Production Right for at least a 10 % undivided interest. Such participation shall be governed by the principles contained in Annexure F and an associated Operating Agreement.

19. Good international oil-field practice

In addition to the provisions of the Act, the Grantee shall comply with the following special obligations.

- 19.1. The Grantee shall maintain all installations in good repair and condition and shall execute all Operations in a proper and workmanlike manner in accordance with methods of Production used in good international oil-field practice and, without prejudice to the generality of the foregoing, the Grantee shall take all steps practicable in order to prevent:
 - (a) the escape or waste of petroleum discovered in the Production Area;
 - (b) damage to petroleum-bearing strata;
 - (c) the entrance of uncontrolled water through wells to petroleum-bearing strata;
 - (d) the escape of petroleum into any waters or aquifer in or in the vicinity of the Production Area;
 and
 - (e) pollution of the terrestrial or marine environment.
- 19.2. The Grantee shall immediately inform the Grantor and other relevant government bodies of the occurrence of any event described in clauses 19.1(a) through 19.1(e) above and shall take immediate remedial action.
- 19.3. The Grantee shall take immediate action with respect to notices given by the Grantor of any event described in clauses 18.1(a) through 18.1(e) above.
- 19.4. The Grantee shall give all notices and take all actions described in the relevant oil-spill contingency plan oil-spill contingency plan required in terms of the Prevention and Combating of Pollution of the Sea Act (Act no. 6 of 1981).
- 19.5. The Grantee shall protect all wells drilled under this Production Right. Upon completion or abandonment of a well the Grantee shall remove from the drill site all guide bases and other substantial equipment so as to leave the environment free of significant obstruction, except where the relevant well is intended to be used as a production well. The Grantee shall notify the relevant government agencies that the location is free of obstruction or, if it is not, the location, nature and extent of any obstruction.
- 19.6. No gas produced from the Production Area, except in the case of short term flaring necessary for testing or other operational reasons, may be flared unless the Grantor's prior written approval, which approval shall not be unreasonably withheld, has been obtained.
- 19.7. The Grantee shall give timeous notice to relevant government agencies and to interested and affected parties as to all Production Operations and shall take all reasonable steps to minimise interference with the rights of other users.

Existing Data

- - (a) one copy of all available borehole measurements and analyses including sample logs, wireline logs, gas chromatograms, micropalaeontological, petrographical, and geochemical data and / or reports (including core descriptions) as well as one digital tape copy, where available (in industry-standard format), of all wireline logs from all wells drilled in the Production Area;
 - (b) one copy of each technical report of the wells drilled in the Production Area, regional reports for the West / South / East Coast and general reports for RSA offshore operations, as appropriate to the Production Area;
 - (c) one copy of each seismic line which intersects the Production Area, at the vertical scale of 2.5 inches per second (two-way time) or as imaged, as well as one copy of 500 km of lines to be selected by the Grantee

at a scale of 5 inches per second (two-way time). The Grantor will also provide location data for all seismic lines in SEG-P-1, UKOOA or ASCII format, in digital form;

- (d) field data for up to _____ km of seismic lines in the Production Area as available and as selected by the Grantee. This data set shall be defined by the Grantee after its preliminary evaluation of the copies of seismic lines described in (c) above. The Grantee will provide the Grantor with copies of the reprocessed data as both digital records in industry standard format and a single paper copy. Where such field tapes are available only in a format other than digital records on 9-track half-inch tape (or more modern format) the Grantor will make available such original field data tapes so that the Grantee can transcribe the data into a current industry-standard format. The Grantee shall provide the Grantor with a copy of the data so transcribed onto media of an industry-standard format which is current at the time, and which the Grantor is equipped to read and, at the Grantor's request, return the original tapes to the Grantor at the Grantee's sole cost, risk and expense.
- (e) stacked and migrated data for up to _____km of seismic lines, as available, as selected by the Grantee;
- (e) access to available sets of conventional cores, sidewall cores, drilling cuttings from the wells in the Production Area for sampling; and

Wherever possible documents will be provided as digital images in a format appropriate to the document.

Additional field data tapes and stacked and migrated data tapes, over and above that detailed in clauses (d) and (e) above will be made available upon request by the Grantee provided that Grantee shall reimburse the Grantor for the reasonable and customary costs incurred in preparing such tapes.

The Grantor will co-operate with the Grantee at the Grantee's request by assisting in resolving technical problems relating to the data (but not interpretation) upon terms and conditions then to be agreed.

All the aforesaid data, studies and reports (hereinafter referred to as the "Existing Data") provided to the Grantee are owned by the Grantor and are of considerable commercial value to it. Such Existing Data shall remain the property of the Grantor and shall, on termination of this Production Right, be returned at the Grantee's cost to the Grantor. The Grantee will hold the Existing Data confidential between itself and the Grantor in the manner described in clause 29.1 in relation to the Data with the exception of items 29.1(f) and 29.1(k), in which instances the Grantee must first acquire the Grantor's prior written approval, and 29.1(b).

20.2. In the event that the Grantee already has possession of Existing Data in terms of an Exploration Right or Prospecting Sub-lease from which this Production Right arises, then the Grantee shall be entitled to retain that portion of the Existing data already in its possession for the duration of this Production Right and shall treat the same as though it were acquired in terms of clause 20.1.

Upstream Training Trust

117

The Grantee shall pay the amounts set out in Annexure C to the independent Upstream Training Trust, to be used in the training, education, and obtaining of practical experience for South African personnel in a manner to be mutually agreed by the Trustees.

22. Local labour, equipment, materials and contractors

- 22.1. In furtherance of section 2(f) of the Act, in carrying out Operations the Grantee shall, to the extent reasonably possible, employ South African citizens having appropriate qualifications.
- 22.2. Not later than three months from the Effective Date, the Grantee shall meet with the Grantor to discuss and agree on a programme for the recruitment, training and employment of South African citizens during the Production Operations.
- 22.3. The Grantee may employ a person who has necessary qualifications and experience and is not a South African citizen if the required skills are not available in the South African labour market. The subsequent availability of qualified candidates will be reviewed annually.

22.4. The Grantee:

22.4.1. and its contractors shall use and purchase equipment, machinery, materials, instruments, supplies and accessories (all referred to as "Goods") manufactured, produced or available in the Republic of South Africa for use in Operations under this Production Right, provided that such Goods are reasonably comparable with like goods manufactured, produced or available outside the Republic of South Africa as to

landed cost, quantity and quality and can be made available at the time when and the place where required;

- 22.4.2. shall to the extent reasonable, taking cognisance of government policies, utilise contractors in the Republic of South Africa where services of comparable standards with those obtained elsewhere are available from such contractors at competitive prices and on competitive terms and can be made available at the time when and the place where required.
- 23. Tax
- 23.1. The Grantee shall maintain a permanent establishment in the Republic of South Africa and arrange its affairs in such a manner that all the profits from production of petroleum under this Production Right will be attributable to that establishment.
- 23.2. Except to the extent exempted, the Grantee shall throughout the term of this Production Right be liable for income tax payments to the State on the annual taxable income derived by it from the production of Natural Oil in accordance with the Income Tax Act as amended from time to time. In no event, however, shall such tax exceed, in respect of taxable income derived during any year of assessment from the production of Natural Oil in the Production Area, the sum of an amount of the "normal tax" as defined in Chapter II, Part 1, Section 5 of the Income Tax Act (hereinafter called the "basic tax"), calculated on the full amount of such taxable income at a maximum tax rate of 35 %. The Grantee is hereby expressly exempted and shall not be liable for the payment of any "additional normal tax" under Chapter II, Part 1, Section 5(2A) of the Income Tax Act.
- 23.3. South African income taxes shall be calculated in United States Dollars. For South African tax reporting purposes only, all account balances necessary to compute the US Dollar tax liability shall be converted to Rand at the rate determined under clause 26 (Exchange Control). Such conversion shall occur at the end of each taxation year. Taxes may be paid in either SA Rand or US Dollars at the Grantee's election. Payments made in US Dollars for any taxation year shall be converted to Rand in the same manner described in this sub-clause 23.3
- 23.4. The Grantee shall maintain a permanent establishment in the Republic of South Africa and arrange its affairs in such a manner that all the profits from production of petroleum under this Production Right will be attributable to that establishment.
- 23.5. Any payments to the State in terms of this clause 23 shall constitute the Grantee's only commitment to the State, or any political sub-division thereof, for taxes or levies, of any kind or nature, on the income or profits derived by the Grantee from the production of petroleum under this Production Right. Specifically, but without limiting the provisions contained in the preceding sentence, the Grantee shall not, other than as set out in clause 7, be liable to the State for:
 - 23.5.1. any other taxes or levies of a similar nature, calculated in relation to taxable income, profits or turnover arising out of Operations in the production and disposal of petroleum under this Production Right;
 - 23.5.2. any share of profit by the State or any other form of consideration in respect of the activities performed by it pursuant to this Production Right;
 - 23.5.3. non-resident shareholders' tax or any other dividend withholding tax, in respect of dividends declared or paid in favour of shareholders who are not resident in the Republic of South Africa out of profits derived from the production of petroleum under this Production Right, or any tax on the remittance outside the Republic of South Africa of profits derived from the aforesaid production of petroleum;
 - 23.5.4. distributed or undistributed profits tax, including the Secondary Tax on Companies imposed under Part VII, Section 64B, et. seq. of the Income Tax Act, in respect of profits derived from the production of petroleum.
- 23.6. In calculating the taxable income referred to in this clause 23, the deductions from income shall be as granted in the Income Tax Act as at the date on which the 1977 amendments thereto were promulgated, and shall in any event include:
 - 23.6.1. the costs expended by the Grantee itself on Production Operations within the Production Area In any tax year or part thereof, and
 - 23.6.2. under current working costs, also the costs of transportation of petroleum from the Production Area to the marine terminal or such refinery in the Republic as the Grantee may elect, as the case may be, and as capital ranking for redemption:

- the costs of laying pipeline from the points referred to above in this sub-clause 23.6.2. or to any onshore processing facility of the Grantee;
- (b) the costs incurred by the Grantee itself in connection with the viability of the relevant undertaking and the design, procurement, management (including also project management), transport and construction of the constituent parts (from and after raw material stage and including also the piles and other foundations) of any marine or onshore receiving installations erected or to be erected on the Production Area or onshore with a view to exploitation of petroleum discovered or found in the Production Area, including also the costs of training of personnel for any purpose in connection with such installations, at any time prior to the successful commissioning of such installations but excluding any assets belonging to another taxpayer, and
- (c) a capital allowance equal to 12 % (twelve per cent) compound interest per annum on the total amount of unredeemed capital expenditure, ranking for redemption, calculated from the end of the month during which each such cost was incurred, until it be redeemed and calculated, as far as this can be done, according to the provisions of sections 26(3) and (4) of the Mining Rights Act, 1967, an example of which is attached hereto as Annexure G.
- 23.7. Expenditure shall not be disallowed as a deduction for income tax purposes because it has been incurred by the Grantee outside the Republic of South Africa.
- 23.8. Production activities taking place in or in connection with an area or areas covered by one Production Right and its associated pipeline and onshore receiving installations up to the boundary of the local refinery, if any, shall be deemed to have taken place in respect of a single mine provided that assets belonging to another taxpayer will be disregarded.
- 23.9. In determining the taxable income or profits in respect of the Production Area, the phrase "mining for Natural Oil" shall be interpreted as including all exploration, development and production activities and such taxable income or profits shall be taxed as set out in this clause 23.
- 23.10. Should the Grantee's offshore facilities be linked to an onshore plant also belonging to the Grantee, the offshore facilities and pipelines as well as the onshore receiving facilities and pipelines (as well as all related structures and foundations) up to the boundary of the onshore plant will be regarded as a mine and the income therefrom taxed as herein set out, while the rest of the onshore facilities will be regarded as a factory and the income therefrom taxed in accordance with the income tax laws of the Republic of South Africa as amended from time to time, the gross income of the Grantee from the combined mine and factory respectively being in such proportion as the Grantee and the Commissioner for Inland Revenue agree or, failing such agreement, as determined by any of the firms or entities referred to in Annexure H (Schedule of Experts) hereof at the request of the Commissioner, unless the Grantee and the said Commissioner agree in writing on another firm or entity.

24. Accounts and Audits

Grantee shall maintain in South Africa such accounting records as will properly support all expenditure and receipts of Operations under this Production Right.

Such records or copies thereof shall be provided at the Grantor's request for the purpose of confirming any minimum expenditure obligations under the Production Work Programme. The Grantor shall have the right to appoint an auditor for the purpose of confirming that these obligations have been met. Such confirmation, if any, shall be completed within two years after the end of the calendar year in which the expenditures or receipts occurred.

25. Customs duties

The Minister records having procured the following from the Minister of Finance, namely:

25.1. The Grantee may during the continuance of this Production Right and subject to the provisions of international trade agreements which may be in force, import into the Republic of South Africa or clear from any customs and excise warehouse under rebate of full customs duty (as defined in the Customs and Excise Act, 1964) or export from the Republic of South Africa such Goods (excluding refined Petroleum products for use in road vehicles and distillate and residual fuel oil for any purpose and goods imported by the Grantee for the personal use of its personnel who are South African residents) as are required for use in Operations and which are not obtainable or available from stocks of Goods manufactured in the Republic of South Africa in terms of clause22.4.1; provided however that the Grantee shall not sell, transfer, dispose of or use or retain for use except in Operations under this Production Right any Goods obtained

free of customs duty without the permission of the Commissioner for Customs and Excise; such duty as may be assessed by the Commissioner for Customs and Excise shall be payable to him/her in respect of any such Goods sold, transferred, disposed of or used or retained for any use other than in Operations except if such Goods are sold, transferred or disposed of to any person in the Republic of South Africa entitled to a like rebate of customs duty or if such Goods are exported from the Republic of South Africa by the Grantee.

- 25.2. The Grantee shall have the right, subject to the applicable laws and regulations, to employ such contractors as in its opinion may be necessary for Operations; such contractors shall be exempted from the payment of customs duties in the same manner as the Grantee is exempted under this Production Right as long as their work is pursuant to their contract with the Grantee.
- 25.3. The Grantee's non-South African personnel may import household and personal effects into South Africa free of customs duty and may export such effects freely, but disposal in the Republic of South Africa of any such household and personal effects shall be subject to approval by the Commissioner for Customs and Excise and shall be subject to customs duties.

26. Exchange Control

- 26.1. The Minister records having procured the following from the Minister of Finance:
 - 26.1.1. The Grantee shall have the free availability of the currency generated from the export of petroleum produced under this Production Right which the Grantee can place directly to a bank account within the Republic of South Africa or abroad.
 - 26.1.2. The Grantee shall have the free disposition and right to freely exchange South African Rand whether received from the sales of Petroleum in the national market for foreign currency or received from other activities or events related to this Production Right. The Grantee has the right to deposit such currency directly into bank accounts in the Republic of South Africa or abroad.
 - 26.1.3. The Grantee shall have the right to maintain, control and operate bank accounts in any currency, inside the Republic of South Africa and abroad; to have the control and free use of these accounts and to maintain and freely dispose of the funds of these accounts without any restrictions. The Grantee shall not be obliged to convert any foreign currencles into South African Rand.
 - 26.1.4. Without prejudice to what is mentioned above, the Grantee shall have the right to dispose, distribute, send or retain abroad, without any restrictions, the Grantee's annual net profits.
 - 26.1.5. The Grantee may use South African banks to exchange the currencies referred to in clause 26.1.2.
- 26.2. The exchange rate of the South African Rand will be determined by the free market.
- 26.3. The Grantee may acquire abroad materials, equipment and services and pay for them in foreign currency. The Grantee may obtain financing abroad through loans and advances to pay for costs of activities connected with this Production Right.
- 26.4. The Grantee shall provide periodic reports as reasonably requested by the Reserve Bank.
- 26.5. In the event that the ratio of borrowings from South African financial institutions to foreign investment exceeds the ratio prescribed in Exchange Control Regulations promulgated by Government Gazette Notice R1111 of December 1961 as amended, the Grantee shall notify the Reserve Bank and advise the Reserve Bank as to any steps the Grantee intends to take to reduce this ratio, or request approval, which shall not be unreasonably withheld, to exceed this ratio.

27. Indemnity and insurance

27.1. The Grantee, its successors and/or assigns, will, during the tenure of this Production Right, take all such necessary and reasonable steps and do all such acts, matters and things and carry out their Production Operations in such a manner as will adequately safeguard and protect the environment, the Production Area and any person or persons using or entitled to use the surface of the Production Area from any damage caused by or through or in consequence of the exercise by the Grantee of its aforesaid right to produce petroleum under this Production Right, of any activity or activities associated with the Grantee's operations in the Production Area, and will in so far as there is a legal liability, compensate such person or persons for any damage or losses, including but not limited to damage to

the surface, to any crops or improvements, which such person or persons may suffer as a result of, arising from or in connection with the exercise of his/her rights under this Production Right or of any act or omission in connection therewith.

- 27.2. The Grantee and its aforesaid successors and/or assigns will and do hereby undertake and bind themselves to hold harmless and indemnify the Grantor and/or its servants and employees against any claim or claims which may be instituted by any person as a result of any injury, loss (including loss of life), damage, cost, charges, and expenses which may be suffered or sustained by any person or persons whomsoever, as a result of, or which may in any way whatsoever be associated with the exercise by the Grantee, it's successors and/or assigns, of the right to produce petroleum under this Production Right, and must also on request refund to the Grantor and/or its servants and employees all costs, charges and expenses which the Grantor and/or its servants and employees may be put to or sustain in connection with or arising out of any such claim made by such person or persons.
- 27.3. It is explicitly agreed that the Grantor will not be deemed to be a partner in any Operations carried out in terms of this Production Right. The Grantor will give notice to the Grantee and its aforesaid successors and/or assigns of any claim or occurrence in respect of which it proposes to invoke the provisions of this indemnity and will give the Grantee and its aforesaid successors and/or assigns the fullest opportunity to investigate, repudiate or otherwise deal with any matter that may arise from such claim. Provided always that nothing herein contained will be deemed to render the Grantee or its aforesaid successors and/or assigns liable for, or to indemnify the Grantor and/or its servants and employees against, any compensation or damages for or with respect to injuries or damage to persons or the Production Area directly resulting from any delict of the Grantor and/or its servants and employees during the currency of this Production Right, or for or in respect of any claims, demands, proceedings, damages, costs, charges and expenses in respect thereof or in relation thereto.
 - 27.4. The Grantor will hold harmless and indemnify the Grantee and its aforesaid successors and/or assigns from and against all claims, demands, proceedings, damages, costs, charges and expenses in respect of matters referred to in the proviso of sub-clause 27.1 of this clause.
 - 27.5. The Grantee shall during the term of this Production Right obtain and maintain for and in relation to Operations, insurance to indemnify the Granter and the Grantee against:
 - (a) loss or damage to any or all assets being used in connection with Operations;
 - (b) loss or damage for which the Grantee may be liable caused by pollution in the course of or as a result of Operations;
 - (c) loss or destruction of property or damage to property as well as bodily injury, sickness or death suffered by any third party for which the Grantee may be liable:
 - (d) the cost of removing wrecks and cleaning-up operations pursuant to an accident in the course of or as a result of Operations;
 - (e) the Grantee's liability to its employees, consultants and agents engaged in Operations; and
 - (f) any other risk of whatever nature as is customary to insure against in the international petroleum industry in accordance with good international oil-field practices.
 - 27.6. The amount insured against, the type of insurance referred to in clause 27.5. above and the terms of such insurance shall be determined in accordance with good international oil-field practices.
 - 27.7. The Grantee acknowledges that the Grantor cannot guarantee that the Grantee will at all times be in a position to exercise the rights granted in terms of this Production Right and that in certain instances a conflict may arise with the holders of other mineral and/or fishing rights which overlap with the Production Area. The Grantee hereby indemnifies and holds the Grantor harmless from and against all claims, liabilities and costs incurred by the Grantee should the rights of the Grantee in terms of this Production Right not be exercisable by a reason of such a conflict with other rights holders within the Production Area.

28. Health and Safety

- 28.1. The Grantee and its successors and/or assigns will and do hereby undertake and bind themselves during the currency of this Production Right to observe and comply with all relevant provisions of the Mine Health and Safety Act (Act no. 29 of 1996) as amended and regulations in terms thereof, or any subsequent law which amends or substitutes the Mine Health and Safety Act, 1996.
- 28.2. Without detraction from sections 42 and 43 of the Act, in the event of:

- an emergency or accident arising from Operations affecting the environment, Grantee shall forthwith notify the Grantor accordingly; or
- 28.2.2. any fire or oil spill in the Production Area or from any facilities connecting the Production Area with the land, Grantee shall promptly implement the relevant contingency plan; or
- 28.2.3. any other emergency or accident arising from Operations affecting the environment, Grantee shall take such action as may be prudent and necessary in accordance with good international oil-field practice in such circumstances.

29. Data and information to be kept confidential and limitations on public announcements

In addition to the provisions of section 83 of the Act and Regulation 14 made in terms of the Act:

- 29.1. The Data (as defined in clause 13.1) and all programmes, tests, analyses, results, books, statements, records, returns, plans and information which the Grantee is or may from time to time be required to furnish under the provisions of this Production Right, shall be supplied at the expense of Grantee. Each Party shall each keep confidential the Data and such information and shall not disclose it to any person without the consent of the other Party except;
 - (a) where disclosure is required by any law;
 - (b) when this Production Right has expired or has been terminated;
 - (c) to an Affiliate, provided such Affiliate maintains confidentiality as provided in this clause 29;
 - to the extent such Data and information is required to be furnished pursuant to any legal proceedings or because of any order of any court binding upon a Party;
 - to prospective or actual contractors, consultants and attorneys employed by any Party where disclosure of such Data or information is essential to such contractor's, consultant's or attorney's work;
 - (f) to a bona fide prospective transferee of Party's interest (including an entity with whom a Party or its Affiliate are conducting bona fide negotiations directed toward a merger, consolidation or the sale of a majority of its or an Affiliate's shares);
 - (g) to a bank or other financial institution to the extent appropriate to a Party arranging for funding;
 - (h) to the extent such Data and information must be disclosed pursuant to any rules or requirements of any government or stock exchange having jurisdiction over such Party or its Affiliates:
 - to its respective employees for the purposes of Operations, subject to each Party taking customary precautions to ensure such Data and information is kept confidential;
 - (j) where any Data or information, through no fault of a Party, has become published in the public domain; and
 - (k) where any relinquishment or abandonment has taken place (but only as to Data and information with regard to the relinquished or abandoned areas).

Disclosure as pursuant to clauses 29.1.(f) and 29.1.(g) shall not be made unless prior to such disclosure the disclosing Party has obtained a written undertaking from the recipient Party to keep the Data and information strictly confidential.

- 29.2. Each Party shall own the Data as defined in 13.1 above in its possession, whether original or a copy, and after termination of this Production Right or relinquishment of any portion of the Production Area each Party may sell, trade, license and otherwise disclose or dispose of such Data relating to the areas no longer included in the Production Area.
- 29.3. Except as may be required by applicable securities law or stock exchange requirements, no public announcement shall be made by any Party unless the other has been furnished with a copy and has given prior approval, which prompt approval shall not be unreasonably withheld. When a public announcement or statement becomes required by law or necessary or desirable because of impending danger to or loss of life, damage to property or pollution resulting from Operations, either Party is authorised to issue and make such announcement or statement without prior notice or the prior approval of the other where such prior notice and approval is not practical in the circumstances. In such a case, the Party making the announcement or statement shall promptly furnish the other Party with a copy of such announcement.

30. Cession, Transfer, Alienation

- 30.1. This Production Right may not be ceded, transferred, let, sub-let, assigned or otherwise disposed of without the written consent of the Minister in terms of section 11 of the Act.
- 30.2. The Minister hereby approves that the Grantee may cede, transfer, let, sub-let, assign or otherwise dispose of or mortgage, either wholly or in part, its rights and obligations under this Production Right, to an Affiliate but only if the Grantee's guarantor has bound itself to guarantee the obligations of the transferee. The Grantee shall advise the Grantor by notice within 10 days of effecting such transaction.

31. Law and interpretation

Subject to section 4 of the Act, and clauses 37 (Arbitration) and 23 (Tax) of this Production Right this Production Right, the interpretation thereof and any dispute arising thereunder or associated therewith shall be governed by and determined in accordance with the laws of the Republic of South Africa.

In this agreement the clause headings are inserted for reference purposes only and shall not govern or affect the meaning or interpretation thereof.

32. Grantee's Rights and Obligations Ensured

- 32.1. The Minister approves and adopts the terms and conditions of this Production Right on behalf of the Grantor, and undertakes on behalf of the State to do and perform all acts and things which are or may be required to be done or performed to give full effect to this Production Right in accordance with its provisions, and guarantees that the contractual rights and obligations of the Grantee under this Production Right will not be altered without prior written consent of the Grantor and the Grantee.
- 32.2. The Minister, on behalf of the Grantor, agrees and commits to the Grantee for the duration of this Production Right, to maintain the stability of the provisions of this Production Right as such provisions exist as of the Effective Date of this Production Right.

Vis Major

- 33.1. Any act, cause, thing or event outside the control of the Grantee including, but not limited to, acts of God, war, insurrection, civil commotion, blockade, strikes, flood, storm, lightning, fire or earthquake which prevents the Grantee from fulfilling its obligations under this Production Right, the Act or the regulations, or such other laws and regulations as may be applicable, may be declared a vis major event and such failure on the part of the Grantee shall not constitute breach. Financial inability, ordinary hardship and inconvenience, however caused or arising, shall not be a vis major circumstance.
- 33.2. If the Grantee by reason of the provisions of clause 33.1 above is prevented from fulfilling its obligations or of enjoying its rights under this Production Right, it shall immediately notify the Grantor in writing to that effect and shall take all reasonable steps to remove the cause thereof and to investigate the consequences. The Grantee shall promptly notify the Grantor as soon as conditions of vis major no longer prevent the Grantee from carrying out its obligations and following such notice shall resume Operations as soon as reasonably practicable.

34. Amendments

The provisions of this Production Right replace all prior negotiations, representations and other communication between the parties hereto and may be amended only in writing between the parties.

35. Unitisation

35.1. A situation may arise where rights held by holders under two or more production rights and/or mining leases and/or exploration rights and/or prospecting leases or sub-leases extend over different areas which geologically form part of the same petroleum-bearing area. In such circumstance the Grantor may by notice in writing require the holders of such rights to prepare a scheme for the production of that petroleum-bearing area as a unit to be submitted to the Grantor within the period specified in the said notice which shall be not less than 90 (ninety) days. Where such petroleum-bearing area extends

- over an area over which the State's rights have not been alienated then the Grantor shall be deemed to be the holder of an exploration right or a production right over such area.
- 35.2. A scheme so prepared shall maximise the exploitation and/or prospecting of the unitised area for the benefit of all holders of interests in the respective mining and/or prospecting sub-leases.
- 35.3. If no scheme is submitted within the period specified in the notice or such further period as the Grantor may approve, or if the scheme submitted is not acceptable to the Grantor, the Grantor may appoint a committee consisting of industry specialists to advise the Grantor.
- 35.4. Any committee so appointed shall after having considered any representation made by the Grantee submit a scheme to the Minister as soon as is practical.
- 35.5. The Minister may, if he/she is satisfied that any scheme submitted in terms of sub-clauses 35.2. or 35.4. is practical and fair and equitable to all parties concerned, confirm such scheme, and it shall thereupon become binding upon all parties.
- 35.6. If the Grantee fails to carry out any provision of the scheme which is binding upon him, the Minister may by notice in writing require the Grantee to do so within the period specified in the notice and if the Grantee fails to comply with the notice the Minister may terminate the exploration right or production right as the case may be.

36. Waiver of Lenience

No waiver of any of the Grantor's rights and no lenience granted or permitted by the Grantor from time to time in a form which does not constitute a formal amendment of this Production Right shall prejudice the Grantor's rights under this right or the Act and shall not constitute grounds for the Grantee to expect that the Grantor will continue to waive its rights or grant or permit lenience in the future.

37. Arbitration

- 37.1. All disputes relating to or arising out of this Production Right shall be finally settled under the rules of Conciliation and Arbitration of the International Chamber of Commerce by three arbitrators appointed in accordance with such rules. Subject to clause 37.2, the venue of the arbitration shall be London, England and the proceedings and records of the proceedings shall be in the English language.
- 37.2. The venue of the arbitration shall be Cape Town, South Africa if the amount in dispute is less than US\$500 0000,00 (FIVE HUNDRED THOUSAND UNITED STATES DOLLARS). However, if the relief sought is declaratory or specific relief, then the venue shall be as provided in clause 37.1.
- 37.3. Notwithstanding any such difference or dispute or reference to arbitration, the Parties shall, to the extent possible, proceed with the carrying out of their obligations under this Production Right, subject to the provisions of this Production Right.
- 37.4. The Parties irrevocably agree that the decision in those arbitration proceedings :
 - 37.4.1. shall be binding on them;
 - 37.4.2. shall be carried into effect; and
 - 37.4.3. can be made an order of any Court of competent jurisdiction.

38. Environmental Protection

- 38.1. The Grantee hereby undertakes and binds itself to conduct all Production Operations in accordance with the requirements of the Act and any other legislation, and the Environmental Management Programme submitted by Grantee in support of its application for this right and approved by Grantor, in a manner that would facilitate the protection and conservation of the natural resources of the Republic of South Africa and of the environment in general.
- 38.2. Subject to the provisions of the Act, the Grantee shall on the expiration or termination of this Production Right or on relinquishment of part of the Production Area or on the voluntary abandonment of the Production Area:
 - 38.2.1. remove all equipment and installations from such relinquished or abandoned Production Area; and

- 38.2.2. perform all necessary site rehabilitation and shall take all other actions necessary to minimise hazards to human or other life or to the Production Area(s) of others or the environment in general for time to come.
- 38.3. The Grantee may from time to time appoint one or more independent contractors to carry out any work in relation to Production operations, but the Grantee shall always remain liable for the compliance and observation of the provisions of the Act, the Environmental Management Programme and of the terms and conditions of this Production Right.
- 38.4. The Grantee shall within 14 (fourteen) days of the Effective Date provide to the Grantor the financial provision for remediation of environmental damage as set out in regulations 32 and 33 and the approved Environmental Management Programme.

39. Costs and VAT

240

- 39.1. All expenses, fees, costs stamp and transfer duties flowing from this Production Right direct or incidental, and the costs of notarial execution and registration, deregistration (in cases or abandonment or relinquishment) including transfer duty, Value Added Tax at the prevailing rate and stamp duty, shall be paid by the Grantee.
- 39.2. The Grantee shall pay Value Added Tax (VAT) at the prevailing rate on all remittances and payments to the Grantor or Petroleum Agency in terms of this Production Right as applicable.

40. Further Developments

- In the event of that, having discovered further deposits of petroleum in the Production Area, the Grantee considers that such further discovered petroleum can be developed for commercial production and wishes to carry out such development then the Grantee shall submit a development programme for approval by the Grantor. Such development programme shall:
 - (i) subject to clause 35., relate exclusively to the Production Area applied for ;
 - (ii) ensure the maximum efficient recovery of the petroleum;
 - (iii) be compiled in accordance with sound engineering, economic, safety and environmental principles recognised in good international oil-field practice;
 - (iv) address the following matters:
 - required production equipment, processing facilities (if any) and storage facilities:
 - feasible alternatives for transportation of petroleum, including pipelines;
 - onshore and offshore installations required;
 - production profiles;
 - cost estimates;
 - safety and environmental measures;
 - employment and training programme;
 - flaring provisions;
 - overall project economics; and
 - (v) include an environmental impact assessment and management programme.

41. Grantor's Option

- 41.1. Should the Grantor have elected, in terms of Regulation 22(3), to retain and transfer an undivided interest in this Production Right the application by the Grantee for this Production Right in respect of such Mining Block shall be deemed to have been made in respect of an undivided share which excludes such retained undivided interest, and the Production Right shall be in respect of such residual undivided share.
- 41.2. The Grantor shall not be liable for any prior exploration costs in respect of the undivided interest so retained by it but shall be liable for its undivided interest share of all costs incurred under the Production Right henceforth, as provided in the Operating Agreement, Annexure I hereto.

- 41.3. Should the Grantor retain an undivided ownership interest in the Production Right, the Grantor and the Grantee shall enter into the Operating Agreement (Annexure I hereto) contemporaneously with the grant of the Production Right. The Accounting Procedure (Exhibit A of Annexure I hereto) shall then apply between the parties to the Operating Agreement.
- 41.4. In the event that, at the time of granting of a Production Right, there is more than one Grantee Party and that an Operating Agreement (and associated Accounting Procedure) is already in effect between the Grantee Parties then the Grantee may invite the Grantor to agree to become a party to that agreement rather than the Operating Agreement described in clause 41.3. and the Grantor shall use its best endeavours to comply provided that the terms and conditions offered are not less favourable to the Grantor than those in the Operating Agreement described in clause 41.3.
- 41.5. If the Grantor elects not to retain an undivided interest in the Production Right, the Production Right shall relate to the whole of the undivided interest in the Production Right.

42. Entire Agreement; The Act Prevails

Subject to the Act, this Production Right and the Appendices attached hereto (those Appendices being and forming an integral part of this Production Right) contain the entire agreement between the Parties and supersedes all their prior understandings, agreements and negotiations of whatsoever nature.

In the event of any inconsistency between this Production Right and the Act, the Act shall prevail.

43. Severability

Any provision of this Production Right which is in any way unenforceable or which may render it unenforceable or which may contravene the applicable laws or regulations of any jurisdiction to which this Production Right is subject shall be deemed, to the extent of such unenforceability or contravention, severable from this Production Right and of no force or effect and shall not affect any other provision of this Production Right and its enforceability or legality in all other respects.

44. Notices and Representatives

All notices, requests and reports provided for herein shall be in writing and shall be delivered either by hand to an authorised representative of the receiving Party, or sent by courier or telefax to the following addresses in the Republic of South Africa, provided that if given by telefax a copy thereof shall be sent immediately by prepaid registered airmail:

Grantor:

Minister of Minerals and Energy

Physical address:

Postal address :

Mineralia Centre

Cnr Visagie & Andries Streets

PRETORIA

Private Bag X59 PRETORIA

0001

Tel number : (27) 12 317 9000

Fax number: (27) 12 322 3416

PETROLEUM AGENCY:

Attention: Chief Executive

Physical address:

Postal address:

P O Box 1174

Petroleum House 151 Frans Conradie Drive PAROW Western Cape

PAROW 7499

Tel number : (27) 21 938 3500

Fax number: (27) 21 938 3520

GRANTEE:

re in the		LIMITED			+ 1
	***************************************	0 570 *			
	Attention : Mr	or		d ^w	
and the second	Physical address :		Postal addi	ress :	
+ 120 c g .	II		, 55121 2221	6 a	
Sa di B	14 per 15		-		
va a nebad			3.5		
	Tal number:		For number :		114
	Tel number :	<u> </u>	Fax number :	V)	
Eithe other	er Party may change r Party of on at least	its address to such other ac 30 (thirty) days prior notice.	ddress in the Republic	of South Africa as it may no	otify the
of tra	otices, requests and ansmission. Those o tual delivery.	reports sent by telefax shall delivered by hand or sent by	be deemed to have by courier shall be deem	een received within 12 (twe ned to have been received a	lve) hours at the time
		s the physical address spe his Production Right, including		micilium citandi et executa	ındi for all
Gran Right	itor in which it appo	SEVEN) calendar days of the object of the control of the continue to represe the continue to represe	h whom the Grantor r	may deal concerning this F	Production
ys. 8	u.S		Đ		8
45. Regi	stration	W	93	S S	
7.1 					
		(a) of the Act, the Grantee meriod specified in the said se		ion Right for registration at t	the Mining
			Ø		
.00	*	#5 ⊕			
		on the	day of	20 in the prese	nce of the
undersigned	witnesses:			3E	
	0 S	3	8		8
AS WITNESS	SES:	9			
4	C.	120	3.		ar ar
1					
			_e sa		
2				18 34	
		for and on behalf of the G	irantor		
Thus done a	nd signed at	on theday	of	20 on the preser	ice of the
undersigned	witnesses:				
				e *** a	
AS WITNESS	SES:	8 5			
1					*
		e e Mo			
2		for and on behal	f of the Grantee	.0%	
	\$ \$	4 4	ATTESTOR	#	
	34 gg (9	18 N			
ita				_	30

ANNEXURE A

PLAN AND DESCRIPTION OF THE PRODUCTION AREA

To be inserted

ANNEXURE D

PRODUCTION WORK PROGRAMME

To be negotiated as part of the application process

ANNEXURE B

ACCOUNT TO WHICH PAYMENTS TO THE GRANTOR ARE TO BE MADE

The amounts payable to the Grantor will be deposited by the Grantee into the account of the Petroleum Agency at:

In the case of payments originating from outside the republic of South Africa the following further information will be of assistance:

For payments originating in the USA the following further information may be necessary

PAYMENT TO BE MADE TO:

Full name of Bank

Address

Harris Bank International

Corporation

3 Times Square

New York

New York, 10036

SWIFT Address

HATRUS33

Account Number

49-0-2945-7

For Credit of

ABSA Bank Limited

International Banking Services:

Corporate Pretoria South Africa

For Further Credit Under Reference CEF (PTY) LTD SEF 8245Y55

The remitting Bank must confirm payments by

MT100 to:

ABSA Bank Limited IBS Corporate

Pretoria South Africa

SWIFT Code For Attention ABSAZAJJACPT Amanda Reyneke Kindly confirm payments per Fax to CEF (PTY) LTD Attention *Dolf Jonker/Kobus Geertsema* Fax No. (27)11 784 5494

In all cases, the Grantee is required to deliver proof of payment to the the Grantor within 3 working days of effecting the payment.

ANNEXURE D

CONTRIBUTIONS TO THE UPSTREAM TRAINING TRUST

Dollars	30 (s);	days of	the	Effective Date -	US\$			(United
within	30	days	of	the first and United States	niversary Dollars);	of	the	Effective	Date	-	US	\$
within (30	days	of	the second a	nniversary Dollars);	of	the	Effective	Date	(14)	US	\$
within (30	days	of	the third and United States	niversary Dollars);	of.	the	Effective	Date	÷	US	\$
within (30	days	of	the fourth an United States	niversary Dollars);	of	the	Effective	Date	•	US	\$
within	30	days	of	the fifth ann United States	iversary o	of	the	Effective	Date	•	US	\$
within :	30 da	ys of th	ne six	th anniversary of United States	the Effectiv	e D	ate -	US	\$			
within (30	days	of	the third and United States	niversary	of	the	Effective	Date		USS	.
within	30	days	of	the seventh aUnited States i	nniversary Dollars);	of	the	Effective	Date	6) 1 4 0	USS	.
within (30	days	of	the eighth an United States I	niversary Dollars);	of	the	Effective	Date		USS	B
	oun	s payal	ble w	ill be deposited b	y the Grante	ee ii	nto the	e account c	f the U	pst	ream	Trainin
The arr												

In all cases, the Grantee is required to deliver proof of payment to the Trust within three working days of effecting the payment.

ANNEXURE E

FINANCIAL GUARANTEE

A financial guarantee acceptable to the Grantor in the form of one of the alternatives incorporated as examples under Appendix B to the regulations, to be attached

ANNEXURE F

BLACK INVESTMENT

- Pursuant to the provisions of clause 18, the State is desirous of encouraging black investment in the upstream sector of the petroleum industry in South Africa.
- In furtherance of this goal, the Grantee shall make available not less than 1/10 (one tenth) undivided interest share in the Production Right for the participation of companies or groups of companies having at least 51% of their shareholding owned by black persons of South African nationality (Black Investor). No such company or group of companies shall hold less than 1/20 (one twentieth) undivided interest in the Production Right.
- 3 As a pre-requisite to the issue of any Production Right the Grantee shall demonstrate to the Grantor that:
 - (a) it has used its best endeavours to seek black investment partners to participate in the Production Right;
 - (b) it has negotiated in good faith the terms and conditions of a Deed of Cession or Participation Agreement and that such terms and conditions conform to the following:
 - (i) In the event that the agreement is concluded prior to the Grantee incurring any expenditure relating to fulfilment of the minimum work obligation as described in any preceding exploration right or prospecting sub-lease then no premium shall be payable.
 - (ii) In the event that the agreement is concluded after the Grantee has incurred expenditure relating to fulfilment of the minimum work obligation as described in any preceding exploration right or prospecting sub-lease but prior to the declaration of a discovery then Black Investor shall reimburse the Grantee its proportionate share, in accordance with its interest in the said exploration right or prospecting sub-lease as the case may be, of such expenditure together with a premium equal to 100% (one hundred per cent) of Black Investor's proportionate share of such expenditure.
 - (iii) In the event that the agreement is concluded after declaration of a discovery then Black Investor shall reimburse the Grantee with its proportionate share, in accordance with its interest in the exploration right or prospecting sub-lease, of expenditure incurred by the Grantee in its fulfilment of the obligations set out in the exploration right or prospecting sub-lease as the case may be, together with a premium equal to 400 % (four hundred per cent) of Black Investor's proportionate share of such expenditure.

As from the effective date of the agreement Black Investor shall contribute to the expenditure incurred in fulfilment of the obligations contained in the exploration right or prospecting sub-lease, in accordance with its interest in the said exploration right or prospecting sub-lease, as determined by the operating agreement between the companies, the terms and conditions of which shall, except as determined in (i) to (iii) above, be the same as between Grantee and any other company holding an interest in the exploration right or prospecting sub-lease as the case may be.

	FIRST YE	AR OF PRE-P	RODUCTION ACTIVITIES	
Actual capital expenditure qualifying for redemption	מי		Unredeemed balance of Capital Expenditure (C/E the Capital Allowance (C/A)) for the purpose of calculating
Incurred during the year	R 2 000 000		Incurred C/A for the year	R 2 000 000
C/A for the year	R 120 000	* a**	(at 12 %, pro rata for a portion of a year*, say)	R 120 000
Carried forward	R 2 120 000			R 2 120 000
	W.			
	SECOND Y	EAR OF PRE-	PRODUCTION ACTIVITIES	
Balance brought forward	R 2 120 000		Balance brought forward	R 2 120 000
Incurred during the year **	R 2 000 000	е .	Incurred during the year **	R 1 940 000
C/A for the year	R 370 800			R 4 060 000
* e			C/A for the year (at, say 12 %)	R 370 800
Carried forward	R 4 490 800	Ele S	The second secon	R 4 430 800
		· ·	, A	

C/E is calculated at 12 % (or other applicable percentage) from the end of the month during which it was incurred until the end of the tax year; in this example, however, it is supposed that all the qualifying C/E was incurred just before the end of the 6th month of the tax year Interest and other finance charges on loans do not qualify for the C/A

FIRST YEAR OF PRODUCTION

. Calculation of taxable income	Calculation of C/A in determining the balance of C/E for the purposes of the C/A
Gross income R 3 000 000 Working costs R 3 740 000	Unredeemed balance C/A at 12 % *
Working loss R 740 000 Redemption of C/E R 6 626 896 Assessed loss R 7 366 896	Balance brought forward R 4 430 800 R 531 696 C/E incurred R 1 500 000 R 90 000
c. C/E qualifying for redemption	Possible C/E included in working costs but on which
Unredeemed balance b/f R 4 490 800 Incurred during the year R 1 500 000	C/A is allowed R 240 000 <u>R 14 400</u>
C/A for the year R 636 096 Total allowed for redemption R 6 626 896	C/A for the year <u>R 636 096</u> → <u>R 636 096</u>
Thus no balance carried forward	Deduct C/E allowed for redemption: amount deducted in the
	calculation of profits is R6 626 895 but the deduction is restricted under section 26(3)
	to the amount of the "profit", i.e. R NIL
	Unredeemed balance carried forward R 6 806 896
	* Or other applicable percentage

SECOND YEAR OF PRODUCTION

1.	Calculation of taxable inc	ome/loss		
	Gross income			R 7 000 000
	Working cost			R 3 500 000
-5	Working profit			R 3 500 000
02	Deduct	61	13	
V	Redemption of C/E	R 1 900 828		
	Assessed loss b/f	R 7 366 896	R 9 267	724
22 EX	Assessed loss carried forw		R 5 767	724
2.	C/E qualifying for redemp	otion		
	Balance b/f		NIL	* .
	C/E incurred for the year			R 1 000 000
39	C/A for the year			R 900 828
	126			R 1 900 828
58	0-F 	5%		

3.	Calculation of C/A in determ the C/A	inin	g the balance of C/E for the purposes of
A to	Unredeemed balance		C/A at 12 %
	Balance brought forward		R 6 806 896 R 816 828
	C/E incurred Possible C/E included in working costs but which		R 1 000 000 R 60 000
	qualifies for C/A		R 400 000 R 24 000
4 8	C/A for the year		R 900 828 → R 900 828 R 9 107 724
+ 1	Deduct		*
	C/E allowed for redemption		
#	as calculated below Unredeemed balance	*	<u>R 3 400 000</u>
	carried forward		R 5 707 724

C/E allowed for redemption

Deemed deduction [sec 26(4)] is R9 107 724
but the deduction is restricted under sec 26(3)
to the amount of the "profit" i.e. Working profit

R 3 500 000

Plus

C/E included in working costs but which can for the purposes of the C/A be regarded as

R 400 000 R 3 900 000

Deduct

So much of the loss brought forward which is not ascribable to the C/E (see below) #

R 500 000

R 3 400 000

The portion of the loss brought forward not resulting from the deduction of C/E

Assessed loss brought forward

R 7 366 896

Actual C/E

(redemption allowance)
C/E in working costs

R 6 626 896

R 240 000 R 6 866 896

R 500 000

ANNEXURE H

SCHEDULE OF EXPERTS (ACCOUNTING)

To be agreed between the parties as part of the application process.

ANNEXURE I

OPERATING AGREEMENT

An operating agreement based on the Association of International Petroleum Negotiators Model form Operating Agreement to be negotiated between the parties and attached.

STANDARD FORMS FOR FINANCIAL PROVISION FOR THE REMEDIATION OF ENVIRONMENTAL DAMAGE

FORM NN:	5	W W	±67 (30)	
STANDARD	FORM FOR APP	ROVED CONTRIBUTIONS T	O A DEDICATED TR	RUST FUND

			DEED OF 1	RUST		4			er 1			
				W .		iii.			3			10
	(%)	a	e.C						130			
						2.5						
Made	e and	entered into by and between		15				1,41		60		
.,		, , , , , , , , , , , , , , , , , , , ,			*				. 2		37	
-		· · · · · · · · · · · · · · · · · · ·										
(here	einaft	ter referred to as the "Founder	"	- 125			35	21	18 10			9
									N N			
AND							ig.					
104			20			SEA13						
(the	name	e(s) of the Trustee(s))	7.	60 di 1				33				
•		, , ,		32 9		53			73	- 3		
		# P	18	11 5	17420		85			15. 10.	8	
			20				0000			500 00		
1.		INITIONS his Agreement, the following e	umragaiana ha	wa tha	monning	acciar	and to t	hom an	d coans	to ever	eeione	e ehali
	have	e corresponding meanings,	unless it appe	ars othe	rwise fror	n the c	context:	nem <u>an</u>	u cogna	te expit	200110) Silali
		"Beneficiary(ies) Account(s)"						records	of the T	rust to b	e main	ıtained
	***	and administered by the Tru- which are required by this do	stees in respe	ect of the	e Benefic	ary or	each of	the Be	neficiarie	s as the	case n	nay be
	1.2	"Statutory Obligations" mean	16.4	s descri	ibed in cla	uses 2	2.2 and	2.3 of th	e record	al. claus	e 2.	
		"the Acts and Regulations"										lations
	2,0	thereunder, being inter alia, the Atmospheric Pollution Pre	the Mineral an evention Act, 1	d Petro 1965, the	leum Res e Nationa	ources Wate	Develor Act, 19	pment / 98, and	Act, 2002 any legi	Act No slation a	o 15 of nd regu	2002),
10	1 4	which may be imposed from		control	tne mana	igeme	nt of all	tne imp	acts of m	ining act	iviues.	
		"the Beneficiary(ies)" means										
	1.5	"the Founder" means Republic of South Africa whi	oh le ongago	d in min	, a	comp	any inc	orporate	ed in ten	ms of th	e laws Provi	or tne ince at
18		Republic of South Africa will	_ (and	th		100 111			11	Pro	vince	at
			_). ` ` `			N.		JIC-21-352				
	1.6	"the	Rehabilitat	ion Trus	t" means	a Trus	t create	d in terr	ns of the	provision	ns of cla	ause 3
í es	-	hereof.										
	1.7		·····		.,							
1 ,		and REHABILITATION TRUST.	jointly, w	no nave	e agreed	o be t	ne tirst t	trustees	or the _			-
	1 2	"the Commissioner" means t	ho Commissio	nor for	the South	Africa	n Reve	nua Sar	vice se r	oforred to	n in the	South
	1.0	African Revenue Service Act		niei ioi	ine South	Airica	iii i ievei	ilue Sei	vice as it	cicirea u	J III 1110	Count
	19	"the Minister" means the Mini		ls and F	neray.							
	1.,,	and minimizer models and minimize							\$ D			
							B =				8	
2.	REC	ORDAL								100		
		The Founder is engaged in m										
	2.2	The Beneficiary(ies) are legarespective mining operations	ally obliged to in terms of th	carry o	out rehab ind Regul	ilitatior ations.	and to	prever	it and co	ntrol pol	llution a	t their
	2.3	For the purpose of making						iary's(ie	s') obliga	ations a	nd conf	tingent

liabilities, the Founder is desirous of creating a Rehabilitation Trust to receive, hold and disburse such moneys as may be contributed by the Founder for the purpose of covering the costs, excluding the costs of on-going rehabilitation and pollution control, in terms of the Acts and Regulations on the closure of each of its mining

operations, and where applicable, any post closure costs.

2.4 The Founder wishes to create a Trust for the benefit of the Beneficiary(ies) and for this purpose wishes to donate R100.00 (one hundred rand) to the Trustees of the Trust.

THE TRUST AND ITS OBJECTS

- 3.2 The main object of the Trust is to act as the financial provider for expenditure which the Beneficiary(ies) are required to undertake in order to comply with the Statutory Obligations.
- 3.3 The Fund shall be a body corporate having perpetual succession and be capable of acquiring and disposing of and owning property and assets and contracting in its own name and suing and being sued in its own name. The rights and obligations of the Fund shall vest in it independently of its beneficiary and/or contributing company.

4. DONATION

- 4.1 The Founder hereby irrevocably donates the sum of R100.00 (one hundred rand) <u>in cash</u> to the Trustees in trust, who in their capacity as such, hereby accepts such donation for the purposes and subject to the conditions of this Trust.
- 4.2 The donation made in terms of clause 4.1 hereof shall immediately vest in the Trustees but always subject to the terms of this Deed.

5. ADMINISTRATION OF THE TRUST

- 5.1 The Trust shall be administered by Trustees appointed by the Founder.
- 5.2 The Trustees shall not receive any remuneration from the Trust for their services.
- 5.3 There shall at all times be not less than two nor more than _____ Trustees.
- 5.4 The Trustees shall hold office until they resign or are deemed to have vacated their offices in terms of clause 6.
- 5.5 The Trustees shall control and administer the Trust, and shall perform and discharge the duties incumbent on them hereunder.

RESIGNATION OF TRUSTEES

- 6.1 A Trustee may resign at any time on giving notice in writing to the remaining Trustees of his intention to do so and shall in any case be deemed to have vacated his office if:
 - 6.1.1 He/she is no longer employed by ______ or any of its subsidiaries; or
 - 6.1.2 the Founder should resolve in writing that he should be removed from office.

Should any vacancy thus occur, the Founder shall be entitled forthwith to fill such vacancy.

7. CHAIRMAN AND VOTING

- 7.1 The Trustees shall from time to time nominate one of their number to be Chairman of the Trust and he shall continue to be Chairman until he resigns from that office or ceases to be a Trustee.
- 7.2 Should the Chairman be absent at any meeting of the Trustees, the Trustees pre-sent shall appoint one of their number as Chairman to preside at that meeting.
- 7.3 Questions arising at meeting of Trustees shall be decided by a simple majority of votes each Trustee personally present having one vote, and in the case of equality of votes the Chairman of such meeting shall have a casting vote.

8. SECRETARY

The Trustees shall appoint a Secretary (which may be a company), who shall not be remunerated with monies from the Trust. The Secretary shall, whenever required to do so by any one Trustee, convene a meeting of the Trustees and be responsible for the minutes to be kept of all meetings and decisions of the Trustees.

9. MEETINGS AND QUORUM

- 9.1 The Trustees shall meet from time to time to transact the business of the Trust and the necessary quorum for any such meeting shall be two Trustees personally present.
- 9.2 Reasonable notice of every such meeting shall be given to each Trustee and all meetings of the Trustees shall, unless otherwise decided by the Trustees, be held in _______
- 9.3 A resolution signed by all the Trustees shall have the same effect as if it had been passed at a duly constituted meeting.

10. TRUSTEES' LIABILITY

- 10.1 The Trustees shall not be answerable or accountable for any loss arising out of their administration of the Trust, or in respect of any depreciation of any investment of the Trust, unless such loss or depreciation shall occur through their wilful misconduct or gross negligence in which event only the ∓rustees concerned and not all the Trustees shall be jointly and severally liable.
 - 10.2 The Trustees shall have no responsibility or liability for the efficacy of the measures taken by it in terms of clause 15 or for the sufficiency of contributions and amounts paid by the Founder of the Trust in terms of clause 14.
 - 10.3 The Trustees shall not have any liability whatsoever in respect of the creation of any of the conditions referred to in the recordal in clause 2 caused by the operations of the Beneficiary(ies) or in respect of any claims arising from the presence of such conditions.
 - 10.4 The Beneficiary(les) indemnify the Trustees against claims made against the Trustees or any of them arising from the loss or depreciation referred to in this clause 10 in respect of its monies (other than as a result of wilful misconduct or gross negligence on the part of the Trustees) or arising as a result of the measures taken on its behalf proving to be ineffective and/or the contributions and/ or amounts obtained from it proving to be insufficient.

11. SECURITY BY THE TRUSTEES

The Trustees shall not be required to lodge security with the Master of the <u>High</u> Court or any other official or authority in terms of the Trust Property Control Act No. 57 of 1988 or any other law requiring that security be lodged with any official or authority for the due performance of their duties hereunder.

12. TRUSTEES' POWERS

- 12.1 The Trustees shall have general control over the funds of the Trust and shall strive to attain the main object for which the Trust is established.
- 12.2 The Trustees shall have plenary powers to enable them to achieve the main object of the Trust.
- 12.3 The Trustees shall receive, hold, apply or disburse in terms of this agreement such monies as may be contributed to the Trust in terms of this agreement by the Founder and funds or assets received from any other source approved by the Commissioner, together with the net income thereon. Except as may be other-wise provided herein, the Trustees shall not distribute any of its profits or gains to any person and shall use the funds solely for the objects for which the Trust has been established.
- 12.4 The Trustees shall not engage in any trade or undertaking nor shall they participate in any business or the affairs of the Beneficiary(ies) or its associates, or provide any financial assistance or services or facilities other than such as required in terms of clause 12.3.
- 12.5 The management of the affairs and all the powers of the Trust shall vest in the Trustees and without derogating from the generality of the foregoing, the Trustees shall have full legal power:
 - 12.5.1 to invest, realise and re-invest the contributions made to the Trust by the Founder and the net income accrued thereon on such terms and at such rates of interest as the Trustees shall in their absolute discretion decide provided that, except as otherwise agreed in writing, prior to the making of such investment, by the Commissioner, they shall be limited to making investments with deposit-taking institutions as defined in the Deposit-taking Institutions Act No. 94 of 1990 as amended, and shall not lend to nor be permitted to invest directly or indirectly monies of the Trust with the Beneficiary(ies) of the Trust or any of its affiliated companies;
 - 12.5.2 to appropriate firstly the net income defined in 13.6 and secondly the contributions and/or amounts paid in terms of clause 14 for the measures taken for compliance with the Statutory Obligations of the Beneficiary(ies);
 - 12.5.3 to institute any legal action for the recovery of monies owing to the Trust and to prosecute, compromise, settle or withdraw any such action;
 - 12.5.4 to execute against the corporeal, incorporeal, movable and immovable property of any of the Trust's judgment debtors;
 - 12.5.5 to execute all documents for and on behalf of the Trust and in this regard to delegate their authority so to execute to two of the Trustees or to one of the Trustees and the Secretary of the Trust; and
 - 12.5.6 generally to perform all acts connected with any of the Trust's affairs.

13. TRUST'S BOOKS OF ACCOUNT AND ACCOUNTING PROCEDURES

- 13.1 The Trustees shall cause proper books of account to be kept for the Trust and shall appoint independent auditors to report on the financial statements for each financial year of the Trust.
- 13.2 The financial statements of the Trust for each financial year (which shall be reckoned from to ________) shall be forwarded by the Trustees to the Beneficiary(ies) of the Trust and to the Commissioner and the Regional Manager within six calendar months after the end of each financial year of the Trust.

- 13.3 The Trustees shall open a banking account in the name of the Trust which shall be operated upon by the joint signatures of one of the Trustees and the secretary, or another duly appointed authorised joint signatory.
- 13.4 All documents (other than documents relating to banking) requiring to be signed or executed on behalf of the Trust shall be so signed or executed by the person or persons authorised thereto by resolution of the Trustees.
- 13.5 All costs, charges and expenses of administering the Trust shall be chargeable to and be borne by the Trust. Such costs, charges and expense shall be paid out of income of the Trust.
- 13.6 The balance of the gross income remaining in any financial year of the Trust after deducting such costs, charges and expenses shall be the net income for the financial year, but if such costs, charges and expenses exceed the income, then the excess shall be the net loss for that financial year.
- 13.7 Such net income or net loss, as the case may be, shall be transferred (credited/ debited) to the account(s) of the Beneficiary(ies).
 Should there be a multiple of Beneficiaries, the net income or net loss shall be allocated to the respective accounts on a proportionate basis.
- 13.8 The Trustees shall not be permitted to distribute, except as may otherwise be provided herein, any of the funds of the Trust to any person and shall utilise the Trusts solely for investment or the objects for which the Trust has been established.
- 13.9 No surplus funds will be refunded to the Founder or any other person (natural or juristic) but will be dealt with in accordance with paragraph 17.2

14. ESTIMATES OF COST OF COMPLIANCE WITH THE STATUTORY OBLI-GATIONS AND CONTRIBUTIONS TO THE TRUST

- 14.1 Annually but at least ninety days before the end of each of their respective financial years, each of the Beneficiaries shall cause written estimates to be prepared, by suitably qualified persons, of the probable cost of measures, excluding those measures which are of an ongoing nature, that will be required to be taken in order to comply with the Statutory Obligations of such Beneficiary(ies) in respect of any defined period or mining activities in which a Beneficiary(ies) is engaged. The estimates may be made separately in respect of each duty, requirement or function of each group of duties, requirements, or functions as may be necessary or convenient. The estimates shall be certified as being fair and shall thereafter be forwarded together with the calculations per clause 14.2 for approval to the Commissioner, and a copy thereof to the Regional Manager.
- 14.2 The Founder shall before the end of the Founder's financial year concerned pay into the bank account of the Trust the approved contribution towards the estimated cost of implementing the measures so approved. The contributions shall be calculated in accordance with the following formula:

(<u>A - B</u>) C

where: A : the total current estimates as envisaged in 14.1

B: the total net balances of the account in respect of the Beneficiary's(ies') account as at the date of the said current estimates

 C: the estimated remaining number of year's life of each of a Beneficiary's (ies') mining operations

- 14.3 The Beneficiary(ies) of the Trust shall be entitled at any time to revise its/their estimates in the light of new or changed circumstances and the Commissioner or the Regional Manager shall likewise have the right to require the Beneficiary(ies) at any time to submit new estimates. Such revised or new estimates shall be certified and approved as provided in 14.1.
- 14.4 The Trustees may require payment from the Founder of a further contribution calculated mutatis mutandis as stipulated in 14.2 should new or revised estimates submitted in terms of 14.3 reflect a shortfall in the total amounts standing to the credit of the Beneficiary's(ies') accounts.

15. COMPLIANCE WITH THE STATUTORY OBLIGATIONS

- 15.1 The Trustees undertake to ensure that when the Statutory Obligations, as stated herein, have been properly carried out by the Beneficiary(ies) in accordance with the requirements laid down by the Regional Manager, the funds will be placed at the disposal of such Beneficiary(ies). This undertaking shall be a stipulation in favour of the Regional Manager and be enforceable by him.
- 15.2 In fulfilling the beforementioned undertaking the Trustees shall not utilise moneys standing to the credit of one beneficiary "for measures needed to comply with the statutory obligations of another beneficiary", and accordingly expenditure by the Trustees on measures needed to comply with the statutory obligations of any beneficiary shall be limited to monies standing to the credit of such beneficiary of the Fund.
- 15.3 Any disbursement of moneys shall be endorsed by the Regional Manager.
- 15.4 Should there be a balance standing to the credit of any Beneficiary(ies) after all the measures required to be taken in order to comply with its statutory obligations have been executed to the satisfaction of the Trustees

and the Commissioner, the said balance shall be credited to the remaining Beneficiary's(ies') ac-counts *pro rata* to the balance of each of such remaining Beneficiary's(ies') ac-counts as at the end of the month in which all the said measures were so executed, unless the Beneficiary(ies) shall have elected that the balance be paid to a body referred to in 17.

16. CESSATION OF MINING ACTIVITIES AND SHORTFALL

- 16.1 Should a Beneficiary(ies) decide to terminate its mining operations (other than a temporary cessation of such operations) and/or should the Beneficiary(ies) go into liquidation prior to them having complied with all of the Statutory Obligations which they have, it shall, not earlier than three months and not later than one month prior to taking any steps for the winding up of such mining operations or going into liquidation, as the case may be, have final estimates prepared of the probable cost of compliance with such outstanding Statutory Obligations which shall be certified and approved as provided in 14.1.
- 16.2 Should the total amount of the final estimates as so approved exceed the total amounts standing to the credit of a Beneficiary's(ies') account, the Founder shall forthwith pay to the Trust the shortfall.

17. WITHDRAWAL FROM THE FUND

- 17.1 The Founder may withdraw from the Trust giving to the Trustees 3 (three) months' written notice of the intention to do so, provided that the permission of the Regional Manager and the Commissioner for such withdrawal is first obtained. In such case, the Trustees undertake to pay the Regional Manager, on demand, the moneys available in the Beneficiary's(ies') accounts which are required to fulfil the Beneficiary's(ies') rehabilitation responsibilities.
- 17.2 Should any one of the beneficiaries of the Fund cease to be under the administration and technical control of
 it shall withdraw from the Fund as soon as reasonably possible
 provided that if any of the statutory obligations which it has remain unsatisfied at the time it is to withdraw, it
 shall obtain the written approval of the Regional Manager and the Commissioner to such withdrawal. Funds
 standing to the credit of the beneficiary must be dealt with in accordance with sub-clause (3).
- 17.3 Should, at the time of the withdrawal, it be that all its obligations have not been met to the satisfaction of the relevant authorities, any funds standing to the credit of the withdrawing beneficiary will (subject to clause 13.9) be dealt with in a manner agreed to in writing between the beneficiary and the Regional Manager.

18. TERMINATION OF THE TRUST

- 18.1 The Trust may only be terminated after all the Beneficiary's(ies') obligations including any post closure obligation set out in terms of the Acts and Regulations have been met to the satisfaction and approval of the Regional Manager as well as the Commissioner.
- 18.2 Should any monies and/or other assets remain after the requirements of sub-paragraph 17.1 have been met, those monies and other assets are to be given or transferred to a company, society or other association of persons or Trust which fulfil the requirements of section 10(1)(cH) of the Income Tax Act, 1962, and is approved by the Commissioner.

19. VARIATION OF THIS AGREEMENT

The provisions of this agreement may from time to time be amended by a resolution of the Founder and after the prior approval of the Commissioner and the Regional Manager.

THUS done and signed inFounder.		- 100	on the	day of		by the
AS WITNESSES:		* #				
1		# *		G	9	
2.			For and	on behalf of		
# # #						
THUS done and signed in Trustee.			on the	day of	2 V	by the
AS WITNESSES:		8			47	67
1.		± 0	į.			
2.				12 B		
* 1	.02	*	For and	on behalf of		

		5 ¹² 8 5		1	
THUS done and signed in		_ on the	day of		by the
AS WITNESSES:					
1.			(96)		
2	81		, , , , , , , , , , , , , , , , , , , 		
		For and	d on behalf of		
				_	
THUS done and signed in		on the	day of		by the
Trustee.					
AS WITNESSES:					
1. 2.			16		a).
2					
2.		For and	d on behalf of	11 12	A company
		#			
THUS done and signed in	2	_ on the	day of	7 6	by the
AS WITNESSES:	10		1	**	9
1.		27			
2.			l b-b-K-if	-	98
			i on behalf of		
1 4.					
	*	8	W 20	**	
THUS done and signed inTrustee.		_ on the	day of		by the
AS WITNESSES:					•
16 m at 25		8			
1.					
2.	禁 篇	For and	on behalf of	-	
				-	

FORM OO: STANDARD FORM FOR A WRITTEN GUARANTEE FROM A BANK OR FINANCIAL INSTITUTION

(BANK OR GUARANTOR'S LETTER HEAD)

	20		40	** **						*	
The Min						î		7))Al (d		
		·····					18 W B				185
				339	2		51			a 8	
***********					9		300	9			53
		,		10			80	92			81 12
	***********	÷			5. 		£ 3	* 15		t.	
					E						
		,	*		Ŧ					10	50
12			25					li a		. 8	
Sir							38	11.4			
				(%)				*.	8 3	St 40	61 8
FINANC	CIAL GI	JARANTEE FOR THE	REHABIL	ITATION	OF LAN	D DISTU	JRBED	BY MINI	NG (EX	ECUTION	I OF
ENVIRO	ONMENT	TAL MANAGEMENT PRO	GRAMME)			E.			90	
								B 44	74		5 5
200							. D	D		Dill	iah ia
1 ,	Conce	rning the responsibility in ent on	terms of	the Mine	rai and r	etroleun/ her	n Hesou	oferred to	eiopmeni ae "tha n	nine owne	cri is
	executo	e the environmental mana	gement n	rogramme	approved	in terms	s of the	provisions	of the s	aid Act fo	or the
		nown as									
	magist	erial district of			P	rovince .				,	I/We
			in my/our	capacity/	capacities	3	8 2 5		EQ. 100	10,000	of
					é	and as	duly	authorized	d repres	sentative/	s of
2.	The guexecus pay to Rso and default or if hi	able to you for the purpose arantor, who hereby waive ssionis et divisionis, the you the said sum of	es the advance meaning a	antages of and the con 	the excep sequence ir opinion of program	mental montal mo	n numer th is known on receipt retion) the he ceas n terms	ent progra ratae pect wn to the g at of a writt ne mine or es mining, of the Ins	guarantor ten claim wner fails prospect	from you for remaing opera acts whice	to do ins in itions,
s		nent. The said claim may									
3.	with all	id amount of R the provisions of the said ount was appropriated and	environme	ental manag	gement pr	rogramme	e, will giv	e account	u, after h	aving com uarantor o	nplied If how
4.	This ur	ndertaking is neither negoti	iable nor t	ransferable	, and -	景 巻	81 6 81			* v	
	(a)	must be returned to the	guarantor	when givin	g account	t to the g	uarantor	in terms o	of clause	3 above,	
	(b)	shall lapse on the granti	ng of a clo	sure certifi	cate in ter	rms of the	e Act and	.		. = =	ia.
W. 2	(c)	shall not be construed a guaranteed amount.	as placing	any other	responsit	oility on t	he guara	antor othe	r than the	e paying o	of the
4.5 26	(c)		as placing	any other	responsit	oility on t	he guara	antor othe	r than the	e paying o	of t

The guarantor reserves the right to withdraw from this guarantee after having given you at least three months' written notice in advance of his intention to do so.

Yours faithfully

(SIGNATURE)

ADDRESS:

DATE:
PLEASE NOTE: (1) No amendments and/or additions to the wording of this guarantee will be accepted.

(2) The address of the guarantee must be stated clearly.

(3) This guarantee must be returned to:

Ма	de an	d entered into by and between	S 3 8 S		st t	8 8	58 B					
(he	reinaf	ter referred to as the "Founder")	- A 14		M (0)38		V1477					
1,,,		ion rollinear to do the Troulinear,										
			9									
<u>AN</u>	<u>D</u>	(A)			50							
		4						沒				
					63							
(the	nam	e(s) of the Trustee(s))	- 3			(8) (8)						
							35					
			9.					15				
2.	DEE	INITIONS										
4.	In ti	his Agreement, the following expressions erwise from the context:	have the mea	ining respe	ctively set o	pposite the	em unless	s it appears				
60	1.2	"Beneficiary(ies) Account(s)" mean sepa and administered by the Trustees in res which are required by this document. "Statutory Obligations" mean the obligation	pect of the Bendons described in	eficiary or e	ach of the E 2 and 2.3 of	Seneficiaries the recorda	s as the c	ase may be 2.				
		"the Acts and Regulations" means all t thereunder, being inter alia, the Mineral Pollution Prevention Act, 1965, the Natio imposed from time to time to control the r	I and Petroleun nal Water Act, nanagement of	n Resource 1998, and a all the impa	s Developm any legislation acts of mining	ent Act, 20 on and regu g activities.	002, the A lations wh	Atmospherionich may be				
	1.5	"the Beneficiary(ies)" means	£									
	1.6	"the Beneficiary(ies)" means "the Founder" means Republic of South Africa with lin Province a	nited liability	, a compa which is	ny incorpora engaged (and th	ited in term in mining ne	ns of the activitie	laws of the es in the				
		Province at	_).	*			*****					
	1.7	"the Rehabilita hereof.	ation Trust" mea	ıns a Trust (created in te	rms of the p	provisions	of clause 3				
	1.8	.8 "the Trustees" mean jointly, who have agreed to be the first trustees of the										
		and jointly, REHABILITATION TRUST.	who have agree	ed to be the	first trustee	s of the						
	1.9	"the Commissioner" means the Commissioner for the South African Revenue Service as referred to in the South African Revenue Service Act, 1997.										
	1.10	0 "the Minister" means the Minister of Mine	erals and Energy	<i>/</i> .			W					
		\$ E				39 50	-:					
3.	REC	CORDAL	£(20.								
	2.2	The Founder is engaged in mining activiti	es consisting of					36				
		The Beneficiary(ies) are legally obliged respective mining operations in terms of t	to carry out reh	abilitation a	and to preve	ent and con	itrol pollut	tion at their				
	2.4	For the purpose of making provision for liabilities, the Founder is desirous of creat may be contributed by the Founder for rehabilitation and pollution control, in temperations, and where applicable, any po	or the discharge ting a Rehabilita the purpose o rms of the Acts	e of the Botton Trust to f covering and Regul	receive, ho the costs, e	ld and disbe	urse such	moneys as of on-going				
3	2.5	The Founder wishes to create a Trust for R100.00 (one hundred rand) to the Truste	the benefit of the	e Beneficia	ry(ies) and f	or this purp	ose wishe	s to donate				
4.	THE	TRUST AND ITS OBJECTS	32		sati (f)							
	3.2	There is hereby created with the appro- Department of Minerals and Energy and Act, 1962, a Trust called the	val of the Com of the Trustees	in accordan	nce with sect	ion 10(1)(cl	levenue S H) of the I	Service, the ncome Tax				
	3.4	The main object of the Trust is to act a required to undertake in order to comply	s the financial	provider for	expenditure		Beneficia	ary(ies) are				

3.5 The Fund shall be a body corporate having perpetual succession and be capable of acquiring and disposing of and owning property and assets and contracting in its own name and suing and being sued in its own name. The rights and obliga-tions of the Fund shall vest in it independently of its beneficiary and/or contribu-ting company.

DONATION

- 4.2 The Founder hereby irrevocably donates the sum of R100.00 (one hundred rand) to the Trustees in trust, who in their capacity as such, hereby accepts such dona-tion for the purposes and subject to the conditions of this Trust.
- 4.3 The donation made in terms of clause 4.1 hereof shall immediately vest in the Trustees but always subject to the terms of this Deed.

6. ADMINISTRATION OF THE TRUST

- 5.2 The Trust shall be administered by Trustees appointed by the Founder.
- 5.3 The Trustees shall not receive any remuneration from the Trust for their services.
- 5.4 There shall at all times be not less than two nor more than _____ Trustees
- 5.5 The Trustees shall hold office until they resign or are deemed to have vacated their offices in terms of clause 6.
- 5.6 The Trustees shall control and administer the Trust, and shall perform and dis-charge the duties incumbent on them hereunder.

7. RESIGNATION OF TRUSTEES

- 6.2 A Trustee may resign at any time on giving notice in writing to the remaining Trustees of his intention to do so and shall in any case be deemed to have vaca-ted his office if:
 - 6.1.2 He/she is no longer employed by ______ or any of its subsidiaries; or
 - 6.1.3 the Founder should resolve in writing that he should be removed from office.

Should any vacancy thus occur, the Founder shall be entitled forthwith to fill such vacancy.

8. CHAIRMAN AND VOTING

- 7.2 The Trustees shall from time to time nominate one of their number to be Chairman of the Trust and he shall continue to be Chairman until he resigns from that office or ceases to be a Trustee.
- 7.3 Should the Chairman be absent at any meeting of the Trustees, the Trustees pre-sent shall appoint one of their number as Chairman to preside at that meeting.
- 7.4 Questions arising at meeting of Trustees shall be decided by a simple majority of votes each Trustee personally present having one vote, and in the case of equality of votes the Chairman of such meeting shall have a casting vote.

9. SECRETARY

The Trustees shall appoint a Secretary (which may be a company), who shall not be remunerated with monies from the Trust. The Secretary shall, whenever required to do so by any one Trustee, convene a meeting of the Trustees and be responsible for the minutes to be kept of all meetings and decisions of the Trustees.

10. MEETINGS AND QUORUM

- 9.2 The Trustees shall meet from time to time to transact the business of the Trust and the necessary quorum for any such meeting shall be two Trustees personally present.
- 9.3 Reasonable notice of every such meeting shall be given to each Trustee and all meetings of the Trustees shall, unless otherwise decided by the Trustees, be held in ________.
- 9.4 A resolution signed by all the Trustees shall have the same effect as if it had been passed at a duly constituted meeting.

11. TRUSTEES' LIABILITY

- 10.2 The Trustees shall not be answerable or accountable for any loss arising out of their administration of the Trust, or in respect of any depreciation of any invest-ment of the Trust, unless such loss or depreciation shall occur through their wil-ful misconduct or gross negligence in which event only the Trustees concerned and not all the Trustees shall be jointly and severally liable.
- 10.3 The Trustees shall have no responsibility or liability for the efficacy of the mea-sures taken by it in terms of clause 15 or for the sufficiency of contributions and amounts paid by the Founder of the Trust in terms of clause 14.
- 10.4 The Trustees shall not have any liability whatsoever in respect of the creation of any of the conditions referred to in the recordal in clause 2 caused by the opera-tions of the Beneficiary(ies) or in respect of any claims arising from the pre-sence of such conditions.

10.5 The Beneficiary(ies) indemnify the Trustees against claims made against the Trustees or any of them arising from the loss or depreciation referred to in this clause 10 in respect of its monies (other than as a result of wilful misconduct or gross negligence on the part of the Trustees) or arising as a result of the mea-sures taken on its behalf proving to be ineffective and/or the contributions and/ or amounts obtained from it proving to be insufficient.

12. SECURITY BY THE TRUSTEES

The Trustees shall not be required to lodge security with the Master of the Supreme Court or any other official or authority in terms of the Trust Property Control Act No. 57 of 1988 or any other law requiring that security be lodged with any official or authority for the due performance of their duties hereunder.

13. TRUSTEES' POWERS

- 12.2 The Trustees shall have general control over the funds of the Trust and shall strive to attain the main object for which the Trust is established.
- 12.3 The Trustees shall have plenary powers to enable them to achieve the main object of the Trust.
- 12.4 The Trustees shall receive, hold, apply or disburse in terms of this agreement such monies as may be contributed to the Trust in terms of this agreement by the Founder and funds or assets received from any other source approved by the Commissioner, together with the net income thereon. Except as may be other-wise provided herein, the Trustees shall not distribute any of its profits or gains to any person and shall use the funds solely for the objects for which the Trust has been established.
- 12.5 The Trustees shall not engage in any trade or undertaking nor shall they partici-pate in any business or the affairs of the Beneficiary(ies) or its associates, or provide any financial assistance or services or facilities other than such as re-quired in terms of clause 12.3.
- 12.6 The management of the affairs and all the powers of the Trust shall vest in the Trustees and without derogating from the generality of the foregoing, the Trustees shall have full legal power:
 - 12.5.2 to invest, realise and re-invest the contributions made to the Trust by the Founder and the net income accrued thereon on such terms and at such rates of interest as the Trustees shall in their absolute discretion decide provided that, except as otherwise agreed in writing, prior to the making of such investment, by the Commissioner, they shall be limited to making investments with deposit-taking institutions as defined in the Deposit-taking Institutions Act No. 94 of 1990 as amended, and shall not lend to nor be permitted to invest directly or indirectly monies of the Trust with the Beneficiary(ies) of the Trust or any of its affiliated com-panies;
 - 12.5.3 to appropriate firstly the net income defined in 13.6 and secondly the contributions and/or amounts paid in terms of clause 14 for the measures taken for compliance with the Statutory Obligations of the Benefi-clary(les);
 - 12.5.4 to institute any legal action for the recovery of monies owing to the Trust and to prosecute, compromise, settle or withdraw any such action;
 - 12.5.5 to execute against the corporeal, incorporeal, movable and immovable property of any of the Trust's judgment debtors;
 - 12.5.6 to execute all documents for and on behalf of the Trust and in this regard to delegate their authority so to execute to two of the Trustees or to one of the Trustees and the Secretary of the Trust; and
 - 12.5.7 generally to perform all acts connected with any of the Trust's affairs.

14. TRUST'S BOOKS OF ACCOUNT AND ACCOUNTING PROCEDURES

- 13.2 The Trustees shall cause proper books of account to be kept for the Trust and shall appoint independent auditors to report on the financial statements for each financial year of the Trust.
- The financial statements of the Trust for each financial year (which shall be reckoned from being bein
- 13.4 The Trustees shall open a banking account in the name of the Trust which shall be operated upon by the joint signatures of one of the Trustees and the secre-tary, or another duly appointed authorised joint signatory.
- 13.5 All documents (other than documents relating to banking) requiring to be signed or executed on behalf of the Trust shall be so signed or executed by the person or persons authorised thereto by resolution of the Trustees.
- 13.6 All costs, charges and expenses of administering the Trust shall be chargeable to and be borne by the Trust. Such costs, charges and expense shall be paid out of income of the Trust.
- 13.7 The balance of the gross income remaining in any financial year of the Trust after deducting such costs, charges and expenses shall be the net income for the financial year, but if such costs, charges and expenses exceed the income, then the excess shall be the net loss for that financial year.

- 13.8 Such net income or net loss, as the case may be, shall be transferred (credited/ debited) to the account(s) of the Beneficiary(ies).
 Should there be a multiple of Beneficiaries, the net income or net loss shall be allocated to the respective accounts on a proportionate basis.
- 13.9 The Trustees shall not be permitted to distribute, except as may otherwise be provided herein, any of the funds of the Trust to any person and shall utilise the Trusts solely for investment or the objects for which the Trust has been established.
- 13.10 No surplus funds will be refunded to the Founder or any other person (natural or juristic) but will be dealt with in accordance with paragraph 17.2

15. ESTIMATES OF COST OF COMPLIANCE WITH THE STATUTORY OBLI-GATIONS AND CONTRIBUTIONS TO THE TRUST

- 14.2 Annually but at least ninety days before the end of each of their respective financial years, each of the Beneficiaries shall cause written estimates to be prepared, by suitably qualified persons, of the probable cost of measures, exclu-ding those measures which are of an ongoing nature, that will be required to be taken in order to comply with the Statutory Obligations of such Beneficiary(ies) in respect of any defined period or mining activities in which a Beneficiary(ies) is engaged. The estimates may be made separately in respect of each duty, requirement or func-tion of each group of duties, requirements, or functions as may be necessary or convenient. The estimates shall be certified as being fair and shall thereafter be forwarded together with the calculations per clause 14.2 for approval to the Commissioner, and a copy thereof to the Regional Manager.
- 14.3 The Founder shall before the end of the Founder's financial year concerned pay into the bank account of the Trust the approved contribution towards the esti-mated cost of implementing the measures so approved. The contributions shall be calculated in accordance with the following formula:

(<u>A - B</u>) C

where: A: the total current estimates as envisaged in 14.1

B: the total net balances of the account in respect of the Beneficiary's(ies') account as at the date of the said current estimates

: the estimated remaining number of year's life of each of a Beneficiary's(ies') mining operations

- 14.4 The Beneficiary(ies) of the Trust shall be entitled at any time to revise its/their estimates in the light of new or changed circumstances and the Commissioner or the Regional Manager shall likewise have the right to require the Beneficiary(ies) at any time to submit new estimates. Such revised or new estimates shall be certified and approved as provided in 14.1.
- 14.5 The Trustees may require payment from the Founder of a further contribution calculated mutatis mutandis as stipulated in 14.2 should new or revised esti-mates submitted in terms of 14.3 reflect a shortfall in the total amounts standing to the credit of the Beneficiary's(ies') accounts.

16. COMPLIANCE WITH THE STATUTORY OBLIGATIONS

- 15.2 The Trustees undertake to ensure that when the Statutory Obligations, as stated herein, have been properly carried out by the Beneficiary(ies) in accordance with the requirements laid down by the Regional Manager, the funds will be placed at the disposal of such Beneficiary(ies). This undertaking shall be a stipulation in favour of the Regional Manager and be enforceable by him.
- 15.3 In fulfilling the beforementioned undertaking the Trustees shall not utilise moneys standing to the credit of one beneficiary "for measures needed to comply with the statutory obligations of another beneficiary", and accordingly expenditure by the Trustees on measures needed to comply with the statutory obligations of any beneficiary shall be limited to monies standing to the credit of such beneficiary of the Fund.
- 15.5 Any disbursement of moneys shall be endorsed by the Regional Manager.
- 15.6 Should there be a balance standing to the credit of any Beneficiary(ies) after all the measures required to be taken in order to comply with its statutory obligations have been executed to the satisfaction of the Trustees and the Commissio-ner, the said balance shall be credited to the remaining Beneficiary's(ies') ac-counts pro rata to the balance of each of such remaining Beneficiary's(ies') ac-counts as at the end of the month in which all the said measures were so executed, unless the Beneficiary(ies) shall have elected that the balance be paid to a body referred to in 17.

20. CESSATION OF MINING ACTIVITIES AND SHORTFALL

16.2 Should a Beneficiary(ies) decide to terminate its mining operations (other than a temporary cessation of such operations) and/or should the Beneficiary(ies) go into liquidation prior to them having complied with all of the Statutory Obliga-tions which they have, it shall, not earlier than three months and not later than one month prior to taking any steps for the winding up of such mining opera-tions or going into liquidation, as the case may be,

have final estimates prepared of the probable cost of compliance with such outstanding Statutory Obligations which shall be certified and approved as provided in 14.1.

36.3 Should the total amount of the final estimates as so approved exceed the total amounts standing to the credit of a Beneficiary's (ies') account, the Founder shall forthwith pay to the Trust the shortfall.

21. WITHDRAWAL FROM THE FUND

- 17.4 The Founder may withdraw from the Trust giving to the Trustees 3 (three) months' written notice of the intention to do so, provided that the permission of the Regional Manager and the Commissioner for such withdrawal is first obtained. In such case, the Trustees undertake to pay the Regional Manager, on demand, the moneys available in the Beneficiary's(ies') accounts which are required to fulfil the Beneficiary's(ies') rehabilitation responsibilities.
- 17.5 Should any one of the beneficiaries of the Fund cease to be under the administration and technical control of it shall withdraw from the Fund as soon as reasonably possible provided that if any of the statutory obligations which it has remain unsatisfied at the time it is to withdraw, it shall obtain the written approval of the Regional Manager and the Commissioner to such withdrawal. Funds standing to the credit of the beneficiary must be dealt with in accordance with sub-clause (3).
- 17.6 Should, at the time of the withdrawal, it be that all its obligations have not been met to the satisfaction of the relevant authorities, any funds standing to the credit of the withdrawing beneficiary will (subject to clause 13.9) be dealt with in a manner agreed to in writing between the beneficiary and the Regional Manager.

22. TERMINATION OF THE TRUST

- 18.3 The Trust may only be terminated after all the Beneficiary's(ies') obligations including any post closure obligation set out in terms of the Acts and Regula-tions have been met to the satisfaction and approval of the Regional Manager as well as the Commissioner.
- 18.4 Should any monies and/or other assets remain after the requirements of sub-paragraph 17.1 have been met, those monies and other assets are to be given or transferred to a company, society or other association of persons or Trust which fulfil the requirements of section 10(1)(cH) of the Income Tax Act, 1962, and is approved by the Commissioner.

23. VARIATION OF THIS AGREEMENT

The provisions of this agreement may from time to time be amended by a resolution of the Founder and after the prior approval of the Commissioner and the Regional Manager.

THUS done and signed in	g (f	on the	day of		38 S	by the
Founder.		-		#		
AS WITNESSES:		W 15	6			20
3.	28		4			
4.		<i>E</i> 1		0. 81W)		
	20	Fo	and on behalf of			
		.				
TUIS done and dened in	\$0 50	41			546	h
THUS done and signed in Trustee.	*	_ on the	day of _			by the
AS WITNESSES:						
				2		
3			(4)			
4.	#	Foi	and on behalf of	((2)	8
						8
			¥	* **		
THUS done and signed in		on the	day of	W)		_ by the
Trustee.		46				
AS WITNESSES:			all di	417		Ħ
3.			近 青 一覧	4.		2

4.	* *						
	<u> </u>		For	and on behalf	of		3,3
THUS done and signed in			on the	day	of		by the
AS WITNESSES:					8 0	14	a seg
3.	•3				# W		2 **
4.			For	and on behalf	of		
# &			(2		*********		
THUS done and signed in Trustee.	-24		on the	day o	of		by the
AS WITNESSES:			5.9				
3. 4.		#					20
	i	a In	For	and on behalf	of		
*			· · · · · · · · · · · · · · · · · · ·	29		4	
THUS done and signed in Trustee.			on the	day d	of		by the
AS WITNESSES:	61 612		- 3				<u> </u>
3.	85			95			
4.			For	and on behalf	of		
線 後 24 - 克			1607		range on Trade de Cort ia (Cit		

FORM PP: DEPOSITS PLACED WITH THE DEPARTMENT

MEMORANDUM OF AGREEMENT

Made and entered into by and between

and

the Minister of Minerals and Energy of the Republic of South Africa he/she being authorised thereto by virtue of the powers vested in him/her in terms of the Mineral and Petroleum Resources Development Act ("the Minister")

	VII 185			25		
(ID No)	* *			(i)		
("the HOLDER")		22		*	10	
WHEREAS the HOLDER is the h	older of a reconna	issance nermi	ssion (No)	or prospecting	right(No)	or mining right
(No) or mining permit (No						
terms of the Mineral and Petrole						
on the farm	, N	lagisterial Distr	rict of	("the prope	erty")	

AND WHEREAS in terms of the Act, the rehabilitation of land concerned in any prospecting, mining, reconnaissance, exploration or production operations shall be carried out by the holder of such right or permit -

- in accordance with an approved environmental management programme or plan:
- (ii) as an integral part of the mining operations;
- (iii) simultaneously with such operations, unless otherwise determined; and

AND WHEREAS the MINISTER requires in terms of the Act, that the HOLDER has the means and ability and can make the necessary provision to prospect or mine optimally and safely on the property, to rehabilitate the land concerned satisfactorily and to carry out the environmental management programme or plan;

AND WHEREAS the adequacy of the HOLDER'S financial provision includes that the MINISTER shall be satisfied that -

- the HOLDER will have the financial means to fulfil the environmental requirements of the Act;
- (i) (ii) there will be sufficient financial provision for the final closure of the mine/operation:
- (iii) such funds remain available to the MINISTER to complete rehabilitation work as required in the environmental management plan or programme in the case of demise, sequestration or abscondence of the HOLDER or any failure of the HOLDER to fulfil the environmental requirements of the EMP and that such funds be protected from incorporation into the estate of the HOLDER in case of the circumstances described above; and
- (iv) the financial provision made to fulfil the requirements of the environmental management plan or programme, shall be utilized solely for rehabilitation until a closure certificate in terms of the Act has been issued:

AND WHEREAS the MINISTER and the HOLDER concur that the method of financial provision will be by the depositing of an acceptable amount of money in an account with the MINISTER where it will be held as guarantee and pledge for the aforesaid purposes;

NOW THEREFORE THESE PRESENT WITNESS-

That the parties hereto have contracted and agreed, as they do hereby contract and agree with one another as follows:

As security for the proper and timeous performance by the HOLDER of all his obligations concerning rehabilitation work conducted in terms of the approved environmental management programme or plan, the HOLDER hereby pledges to the MINISTER all right title and interest whatever which the HOLDER may have to and arising out of monies paid into the account with the MINISTER (hereinafter referred to as the account) where it will be held as guarantee and pledge that rehabilitation will be completed and the environmental management programme or plan carried out satisfactorily.

In order to effect the pledge of the monies to be paid into the account, the HOLDER shall simultaneously with the signing hereof or prior hereto, deliver and pay into the account with the MINISTER an amount of R......

3.

Payment from the account shall only be made for rehabilitation work done in terms of the environmental management programme or plan and after the MINISTER has certified that such work has been completed. If the MINISTER deems it necessary for rehabilitation purposes, the MINISTER may authorise prior payment from the account for such purposes. Payment of the HOLDER or any person doing rehabilitation work in terms of the environmental management programme or plan, can be authorised by the MINISTER from the account. The MINISTER will be entitled to ensure that sufficient funds always remain available to perform the HOLDER'S rehabilitation obligations.

4.

The HOLDER must annually, to the satisfaction of the MINISTER and in consultation with an expert acceptable to the MINISTER, determine the quantum of pecuniary provision which is required in terms of the regulations promulgated under the Act. If the quantum differs from the amount originally deposited in terms of clause 2, the amount must be increased or decreased accordingly. If the quantum so determined is more than the amount originally deposited by the HOLDER, the HOLDER must within 30 days of such determination pay the difference into the account with the MINISTER. Any amount deposited into the account with the MINISTER shall be deemed to have been pledged to the MINISTER in terms hereof. If the quantum so determined is less than the amount deposited with the MINISTER, the MINISTER will within 30 days, return such excess money to the HOLDER.

5

Notwithstanding anything to the contrary herein contained, any payment made to the MINISTER in terms of this agreement shall be irrevocable and the HOLDER or his estate shall have no claim, right or title of whatsoever nature on such monies other than for the purpose of rehabilitation. Payments from the account may only be made in terms of clauses 3 and 4 above and 10 hereunder.

6

The HOLDER hereby absolves the MINISTER and the State absolutely from, and indemnifies the MINISTER and the State against any loss, liability, damages or expense which the HOLDER may suffer as a consequence directly or indirectly, of the MINISTER or the State lawfully exercising any of its rights set out herein.

7.

If:

- 7.1 the HOLDER breaches any of the terms or conditions of this agreement or fails to perform any rehabilitation duties in terms of the Act or the regulations promulgated under the Act and fails to remedy such breach or failure within 14 days of dispatch of a notice calling upon him to do so; or
- 7.2 the HOLDER is sequestrated or surrenders his estate (in the case of an individual), is liquidated or placed under judicial management (in the case of a company or close corporation) (whether provisionally of finally); or
- 7.3 a meeting of the HOLDER's shareholders or members (in the case of a company or close corporation) is called for the purpose of a voluntary liquidation of the HOLDER or a resolution to that effect is taken; or
- 7.4 the HOLDER enters into or attempts to enter into a compromise or composition with any of its creditors; or
- 7.5 any of the HOLDER's property is attached pursuant to an order of court; or
- 7.6 a judgement is given against the HOLDER which is not satisfied within ten days of it being given, or against which an application for rescission or an appeal is not noted within that time (provided that such application or appeal is proceeded with expeditiously and with due care and diligence);

the MINISTER shall be entitled to demand immediate payment of all amounts that may be necessary to rehabilitate the surface of the land concerned and to use the money in the account to carry out the environmental management plan or programme, without prejudice to any rights which the MINISTER or the State may otherwise have as a result of that breach.

8

No variation, consensual cancellation, or novation hereof shall be of any force or effect unless reduced to writing and signed by both the HOLDER and the MINISTER.

Without detracting from the right of the HOLDER to liquidate his indebtedness to the MINISTER in full at any time hereafter, the MINISTER shall be entitled to keep any amounts deposited in terms of this agreement, until such time as the HOLDER has complied in full with all his obligations and duties in terms of this agreement. No interest will be payable by the MINISTER for any amounts deposited under this agreement.

10.

Should excess money remain in the account after a certificate has been granted to the HOLDER in terms of the Act, such excess money must be returned to the HOLDER within 30 days of such certificate being issued.

All communications with regard to this agreement shall be addressed as follows and the physical addresses chosen by the MINISTER and the HOLDER are chosen as their respective domicilium citandi et executandi:

MINISTER Postal Address

Physical Address

HOLDER Postal Address

Physical Address

Notice shall be deemed to have been duly give

- If delivered by hand, on the date of delivery.
 - If sent by post, fourteen (14) days after posting.

2	If sent by telefax, on the day follow	wing the da	ay on which	the telefax is trans	mitted.		7.5	8
	DONE and SIGNED by199		, in his cap	acity as Minister at		on the .	da	ay of
		#				1967		
		27		19 48				
212123		. 8	MINISTER	Land on the		3		
AS WI	TNESSES:	9 8 9 9						
1.	1		89	# %-		E F	E1	
10	* * *		N (1)			194 - 182		
2.				THE R. P.	40			
				*	**	*		
THUS	DONE and SIGNED by the HO	OLDER or	duly auth	norised thereto by	/			. at
			4.7	one with the second				
						14		
	Res Esperande no como esperand	\$40 X	HOLDER			0.00		
AS WI	TNESSES:		, P		•			
1.					F 51 191			
	<i>a</i>			30 35			*	

ANNEXURE 2

Maximum quantities of ores and minerals removable

Class	Mineral / Ore	Quantities that can be carried away without any payment	Maximum quantity that can be carried away by payment of royalty
11	2	3	4
Class-A	Asbestos, graphite, mica, native sulphur, auriferous rock with visible uranium mineral and uranium bearing minerals, minerals of rare earths group, beryl, tantalite, columbite-concentrates of ores of antimony, arsenic, bismuth, chromium, copper, lead, nickel, tin, titanium, tungsten, zinc.		
Class-B	Auriferous rock and gravel containing no visible gold, metalliferous ores meant for extracting cadmium, cobalt, mercury, molybdenum, silver, hellium, vanadium, barytes, bitumen, borax, corundum, emery, grossularite, felsper, fluorspar and calcite.	2 2 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	# # # # # # # # # # # # # # # # # # #
Class-C	Uraniferous rock without visible uranium minerals, metalliferous ores meant for extracting antimony, arsenic, bismuth, chromium, copper, lead, nickel, tin, titanium, tungsten, zinc and compound ores containing metals of cadmium, cobalt, mercury, molybdenum, silver, hellium and vanadium, gypsum, limestone, iron pyrites, shales, red and yellow ochre, bauxite metalliferous ores meant for extracting aluminium, iron and manganese.		
Class-D	Limestone, sillimanite, kyanite, magnesite, serpentine, steatite, vermiculite, fireclay, kaolin and other refractory materials, coal and lignite.		
Class-E	All other minerals not specified above.		W ₁ -

Dog ate your Gazette? ... read it online



www.SA GdZettes.co.za

A new information Portal keeping you up to date with news, legislation, the Parliamentary programme and which is the largest pool of SA Gazette information available on the Web.

- · Easily accessible through the www!
 - Government Gazettes from January 1994
 - Compilations of all Indexes pertaining to the past week's Government Gazettes
 - All Provincial Gazettes from September 1995
 - Parliamentary Bills as of January 1999
- Available in full-text, with keyword searching
- Sabinet Online scans, formats, edits and organize information for you. Diagrams and forms included as images.
- · No stacks of printed gazettes all on computer. Think of the storage space you save.
- · Offers Bill Tracker complementing the SA Gazettes products.

For easy electronic access to full-text gazette info, subscribe to the SA Gazettes from Sabinet Online. Please visit us at www.sagazettes.co.za



Tel: (012) 643-9500, Fax: (012) 663-3543, Toll free: 0800 11 8595, e-mail: corporate@sabinet.co.za, www: http://corporate.sabinet.co.za

Printed by and obtainable from the Government Printer, Bosman Street, Private Bag X85, Pretoria, 0001

Publications: Tel: (012) 334-4508, 334-4509, 334-4510 Advertisements: Tel: (012) 334-4673, 334-4674, 334-4504 Subscriptions: Tel: (012) 334-4735, 334-4736, 334-4737

Cape Town Branch: Tel: (021) 465-7531

Gedruk deur en verkrygbaar by die Staatsdrukker, Bosmanstraat, Privaatsak X85, Pretoria, 0001
Publikasies: Tel: (012) 334-4508, 334-4509, 334-4510
Advertensies: Tel: (012) 334-4673, 334-4674, 334-4504
Subskripsies: Tel: (012) 334-4735, 334-4736, 334-4737
Kaapstad-tak: Tel: (021) 465-7531